

# Contents

1	Introduction					
	1.1	Motivation and Results				
		1.1.1 Background about Galois Representations	4			
		1.1.2 Geometric Lifts				
		1.1.3 Relation to Previous Work	6			
	1.2	Overview of the Proof				
		1.2.1 Generalizing Ramakrishna's Method				
		1.2.2 Fontaine-Laffaille Deformation Condition				
		1.2.3 Minimally Ramified Deformation Condition				
	1.3	Acknowledgements				
2	Producing Geometric Deformations 12					
	2.1	Preliminaries about Algebraic Groups				
		2.1.1 Good and Very Good Primes				
		2.1.2 Finite Groups of Lie Type				
	2.2	Review of Galois Cohomology and Deformations				
		2.2.1 Results about Galois Cohomology				
		2.2.2 Deformations of Galois Representations	17			
	2.3 Big Representations					
		2.3.1 Big Representations	20 20			
		2.3.2 Checking the Assumptions				
	2.4	Ramakrishna's Deformation Condition				
		2.4.1 Constructing the Deformation Condition				
		2.4.2 Shrinking the Dual Selmer Group				
	2.5	Generalizing Ramakrishna's Method				
	2.0	2.5.1 Odd Galois Representations				
		2.5.2 Local Lifting Implies Global Lifting				
		2.5.3 Choosing Deformation Conditions				
		2.5.4 Lifting Representations valued in Tori				
3	Fontaine-Laffaille Deformations with Pairings 37					
	3.1	Fontaine-Laffaille Theory with Pairings				
	0.1	3.1.1 Covariant Fontaine-Laffaille Theory				
		3.1.2 Tensor Products and Freeness				
		3.1.3 Duality				
	3.2	Fontaine-Laffaille Deformation Condition				
	0.2	3.2.1 Definitions and Basic Properties				
		3.2.2 Liftability				
		3.2.3 Tangent Space				
		3.4.3 Tangent space	•)2			

4 M	inimally	y Ramified Deformations	<b>56</b>		
4.1	Repre	esentatives for Nilpotent Orbits	57		
	4.1.1	Algebraically Closed Fields	57		
	4.1.2	Representatives of Nilpotent Orbits	58		
4.2	Smoo	thness of Centralizers of Pure Nilpotents	62		
	4.2.1	Centralizers over Fields	62		
	4.2.2	Centralizers in Classical Cases	64		
	4.2.3	Checking Flatness over a Dedekind Base	66		
	4.2.4	Smooth Centralizers	67		
4.3 Deformations of Nilpotent Elements		mations of Nilpotent Elements	72		
	4.3.1	Pure Nilpotent Lifts	72		
4.4	The Minimally Ramified Deformation Condition for Tamely Ramified Representations 7				
	4.4.1	Passing between Unipotents and Nilpotents	73		
	4.4.2	Minimally Ramified Deformations	74		
	4.4.3	Deformation Conditions Based on Parabolic Subgroups	76		
4.5	Minin	nally Ramified Deformations of Symplectic and Orthogonal Groups	77		
	4.5.1	Decomposing Representations	77		
	4.5.2	Extension of Representations	80		
	4.5.3	Lifts with Pairings	83		

## Chapter 1

## Introduction

Galois representations are central in modern number theory, perhaps most famously in the proof of Fermat's Last Theorem. There are natural sources of Galois representations in algebraic geometry, and the Langlands program conjecturally connects them with automorphic forms. For example, given a classical modular eigenform there is an associated two-dimensional Galois representation which we say is modular. Another example comes from an elliptic curve defined over  $\mathbf{Q}$ : the absolute Galois group of the rationals acts on the p-adic Tate module, giving a two-dimensional Galois representation over the p-adic integers. Fermat's Last Theorem follows from the fact that the Galois representations arising from elliptic curves are modular.

More generally, Galois representations arise from algebraic geometry through the action of the Galois group on the p-adic étale cohomology of a variety defined over a number field. Such representations are ramified at finitely many primes and are potentially semistable above p. An absolutely irreducible p-adic Galois representation satisfying these latter properties is said to be geometric. The Fontaine-Mazur conjecture asserts that all geometric Galois representations arise as subsets of Tate twists of p-adic étale cohomology. A more refined version has been proven in the two-dimensional case: two-dimensional geometric Galois representations of  $Gal(\overline{\mathbb{Q}}/\mathbb{Q})$  are modular.

A key technique in answering these questions is the deformation theory of a residual Galois representation  $\bar{\rho}: \operatorname{Gal}(\overline{\mathbf{Q}}/\mathbf{Q}) \to \operatorname{GL}_n(\mathbf{F}_q)$ . Understanding the universal deformation ring (the ring  $R_{\bar{\rho}}^{\square}$  which parametrizes lifts of  $\bar{\rho}$  to complete Noetherian local rings with residue field  $\mathbf{F}_q$ ) gives results about Galois representations. Residual representations are also interesting in their own right. In the two-dimensional case, Serre's conjecture states that every odd irreducible representation is the reduction of a modular Galois representation. This was proven by Khare and Winterberger. Generalizations of Serre's conjecture to groups other than  $\operatorname{GL}_2$  have been proposed, most recently by Gee, Herzig, and Savitt [GHS15]. The combination of Serre's conjecture and the Fontaine-Mazur conjecture motivates the study of when a residual representation admits a geometric lift to characteristic zero.

In particular, we are interested in the following situation. Let K be a finite extension of  $\mathbf{Q}$  with absolute Galois group  $\Gamma_K$ . Suppose k is a finite field of characteristic p,  $\mathcal{O}$  the ring of integers in a p-adic field with residue field  $\mathcal{O}/\mathfrak{m}=k$ , and G is a connected reductive group defined over  $\mathcal{O}$ . For a continuous representation  $\overline{\rho}:\Gamma_K\to G(k)$ , in light of these conjectures it is important to study when there exists a continuous representation  $\rho:\Gamma_K\to G(\mathcal{O})$  lifting  $\overline{\rho}$  that is geometric in a suitable sense over  $\mathcal{O}[\frac{1}{p}]$  (using  $G\hookrightarrow \mathrm{GL}_p$ ).

Suppose  $K=\mathbf{Q}$ . When  $G=\mathrm{GL}_2$ , Ramakrishna developed a technique to produce geometric lifts [Ram02]. His results provided evidence for Serre's conjecture. In the symplectic and orthogonal cases, we generalize this method and prove the existence of geometric lifts in favorable conditions. Two essential hypotheses are that  $\overline{\rho}$  is odd (as defined in §2.5.1) and that  $\overline{\rho}$  restricted to the decomposition group at p looks like the reduction of a crystalline representation with distinct Hodge-Tate weights. For precise statements, see Theorem 1.1.2.2 and Theorem 2.5.3.4. This provides evidence for generalizations of Serre's conjecture. In contrast, when  $G=\mathrm{GL}_n$  with n>2, the representation  $\overline{\rho}$  cannot be odd, and the method does not apply. In such cases, there is no expectation that such lifts exist.

Ramakrishna's method works by establishing a local-to-global result for lifting Galois representations subject to local constraints. Let  $\rho$  be a lift of  $\overline{\rho}$  to  $\mathcal{O}/\mathfrak{m}^n$ . Provided a cohomological obstruction vanishes,

it is possible to lift  $\rho$  over  $\mathcal{O}/\mathfrak{m}^{n+1}$  subject to local constraints if (and only if) it possible to lift  $\rho|_{\Gamma_v}$  over  $\mathcal{O}/\mathfrak{m}^{n+1}$  for all v in a fixed set of places of  $\mathbf{Q}$  containing p and the places where  $\overline{\rho}$  is ramified. In Chapter 2, we generalize Ramakrishna's argument so it can be applied to any connected reductive group and show that the cohomological obstruction vanishes if we allow controlled ramification at additional places at which  $\overline{\rho}$  is unramified.

It remains to pick local deformation conditions at p and at places where  $\overline{\rho}$  is ramified which are liftable in the sense that it is always possible to suitably lift  $\rho|_{\Gamma_v}$ . At p, we define a Fontaine-Laffaille deformation condition in Chapter 3 by using deformations arising from Fontaine-Laffaille modules that carry extra data corresponding to a symmetric or alternating pairing.

At a prime  $\ell \neq p$  where  $\overline{\rho}$  is ramified, we generalize the minimally ramified deformation condition defined for  $\operatorname{GL}_n$  in [CHT08, §2.4.4]. In simple cases, this deformation condition controls the ramification of  $\rho$  by controlling deformations of a unipotent element  $\overline{u}$  of  $\operatorname{GL}_n(k)$ . There is a natural parabolic k-subgroup containing  $\overline{u}$ , and the deformation condition is analyzed by deforming this parabolic subgroup and then lifting  $\overline{u}$  inside this subgroup. This idea does not work for other algebraic groups. In Example 1.2.3.2 and §4.4.3, we discuss an explicit example in  $\operatorname{GSp}_4$  where the analogous deformation based on parabolics is provably not liftable. In Chapter 4, we define a minimally ramified deformation condition by instead requiring that  $\overline{u}$  deform so that "it lies in the same unipotent orbit as  $\overline{u}$ ," and explain that this is a general phenomenon. The discovery and study of this deformation condition at ramified places  $\ell \neq p$  (see §1.2.3) is the key innovation in this thesis. For  $\operatorname{GL}_n$ , our notion agrees with minimally ramified deformation of [CHT08], but for other groups it is a genuinely different (liftable) deformation condition.

### 1.1 Motivation and Results

We will now review the background and state the results in more detail.

### 1.1.1 Background about Galois Representations

Let p be a prime.

**Definition 1.1.1.1.** A Galois representation  $\rho: \Gamma_{\mathbf{Q}} \to \mathrm{GL}_2(\overline{\mathbf{Q}}_p)$  is *modular* if there exists a modular eigenform f such that the Galois representation associated to f at the prime p is isomorphic to  $\rho$ . A residual Galois representation  $\overline{\rho}: \Gamma_{\mathbf{Q}} \to \mathrm{GL}_2(\overline{\mathbf{F}}_p)$  is *modular* if it is the reduction of a modular representation  $\rho$ .

Serre's conjecture is one of the ingredients which motivated Ramakrishna's lifting result for two dimensional representations.

Fact 1.1.1.2 (Serre's Conjecture). Let  $\overline{\rho}: \Gamma_{\mathbf{Q}} \to \mathrm{GL}_2(\overline{\mathbf{F}}_p)$  be an odd irreducible representation. Then  $\overline{\rho}$  is modular.

**Remark 1.1.1.3.** More precise versions give the weight and level of an associated modular form. This conjecture was proven by Khare and Winterberger [KW09a] [KW09b].

Let K be a number field with absolute Galois group  $\Gamma_K$ . Next we define two notions necessary to state the Fontaine-Mazur conjecture.

**Definition 1.1.1.4.** An irreducible representation of  $\Gamma_K$  over  $\overline{\mathbf{Q}}_p$  comes from geometry if it is isomorphic to a sub-quotient of  $H^i_{\text{\'et}}(X_{\overline{K}}, \overline{\mathbf{Q}}_p(r))$  for some variety X over K and some  $r \in \mathbf{Z}$ .

**Definition 1.1.1.5.** A Galois representation  $\rho: \Gamma_K \to G(\overline{\mathbb{Q}}_p)$  is *geometric* if it is unramified outside of a finite set of places and is potentially semi-stable (equivalently, de Rham) at all places of K above p.

**Conjecture 1.1.1.6** (Fontaine-Mazur Conjecture). Let  $\overline{\rho}: \Gamma_K \to \operatorname{GL}_n(\overline{\mathbb{Q}}_p)$  be an irreducible representation. It is geometric if and only if it comes from geometry.

When n=2 and  $K=\mathbb{Q}$ , any irreducible geometric representation that is not the Tate-twist of an even representation factoring through a finite quotient of  $\Gamma_{\mathbb{Q}}$  is modular up to a Tate twist.

The full Fontaine-Mazur conjecture is an open problem, but the statement in the case that n=2 and  $K=\mathbf{Q}$  follows from work of Kisin and Emerton [Kis09] [Eme11].

Motivated by these conjectures, Ramakrishna developed a lifting technique that produces geometric deformations.

**Fact 1.1.1.7.** Under certain technical conditions, if  $\overline{\rho}: \Gamma_{\mathbf{Q}} \to \mathrm{GL}_2(\overline{\mathbf{F}}_p)$  is an odd irreducible representation for which inertia at p acts as a fundamental character of level two, then there exists a geometric lift that is crystalline at p.

This is the main theorem of [Ram02], which further develops the lifting technique of [Ram99]. Given this result, Serre's conjecture is a consequence of the two-dimensional version of the Fontaine-Mazur conjecture. Ramakrishna's result was seen as good evidence for Serre's conjecture before its proof by Khare and Winterberger.

### 1.1.2 Geometric Lifts

We are interested in generalizations of Ramakrishna's lifting result to reductive groups beyond  $GL_2$ , in particular symplectic and orthogonal groups. Generalizations of Serre's conjecture have been proposed in this setting, and most of the effort has been to find the correct generalization of the oddness condition and the weight (see for example the discussion in [GHS15], especially §2.1). The general flavor of these generalizations is that an odd irreducible Galois representation will be automorphic in the sense that it appears in the cohomology of an  $\overline{\mathbf{F}}_p$ -local system on a Shimura variety. For a general reductive group, there is no expectation that such representations will lift to characteristic zero. For example, as discussed in [CHT08, §1] the Taylor-Wiles method would work only if

$$[K:\mathbf{Q}](\dim G - \dim B) = \sum_{v|\infty} \dim H^0(\operatorname{Gal}(\overline{K}_v/K_v), \operatorname{ad}^0(\overline{\rho}))$$
(1.1.2.1)

where B is a Borel subgroup of G and  $\operatorname{ad}^0(\overline{\rho})$  is the adjoint representation of  $\Gamma_K$  on the Lie algebra of the derived group of G. Only under such a "numerical coincidence" do we expect to obtain automorphy lifting theorems, and hence expect geometric lifts. This coincidence cannot hold for  $\operatorname{GL}_n$  when n > 2, but can hold for  $G = \operatorname{GSp}_{2n}$  and  $G = \operatorname{GO}_m$ , and for the group  $\mathcal{G}_n$  related to  $\operatorname{GL}_n$  considered in [CHT08].

We will define the notion of an odd Galois representations in §2.5.1, for which (1.1.2.1) holds. This will imply that K is totally real and for each archimedean place v of K and any complex conjugation  $c_v$ , the invariant subspace of the action of  $\overline{\rho}(c_v)$  on  $\operatorname{ad}^0(\overline{\rho})$  has "minimal" possible dimension (equal to  $\dim G$ - $\dim B$ ). There are odd representations for symplectic and orthogonal groups, but no odd representations for  $\operatorname{GL}_n$  when n > 2. Thus we can hope to construct geometric lifts of odd representations valued in  $\operatorname{GO}_m$  or  $\operatorname{GSp}_m$ .

Remark 1.1.2.1. Our argument naturally uses connected reductive groups, so there are issues when  $GO_m$  is disconnected. In the orthogonal case we work with representations valued in  $GO_m^{\circ}$  to avoid this issue, although as discussed in Remark 2.5.3.2 it is also possible to modify the argument to apply to some disconnected reductive groups.

Consider the reductive  $\mathcal{O}$ -group scheme  $G = \mathrm{GO}_m^{\circ}$  or  $G = \mathrm{GSp}_m$  and a (continuous) residual representation  $\overline{\rho}: \Gamma_K \to G(k)$ . In favorable circumstances, we will produce geometric lifts of  $\overline{\rho}$ . The exact hypotheses needed are somewhat complicated. We will state a simple version now, and defer a more detailed statement to Theorem 2.5.3.4.

Suppose that p is unramified in K and p>16m. Let G' be the derived group of G, and assume that  $G'(k)\subset \overline{\rho}(\Gamma_K)$ . Furthermore suppose that  $\overline{\rho}$  is odd as defined in §2.5.1. At places v above p, assume that  $\overline{\rho}|_{\Gamma_{K_v}}$  is torsion crystalline with Hodge-Tate weights in an interval of length  $\frac{p-2}{2}$ , so it is Fontaine-Laffaille (these notions will be reviewed in §3.1). Furthermore, suppose that for each  $\mathbf{Z}_p$ -embedding of  $\mathcal{O}_{K_v}$  in  $\mathcal{O}_{\overline{K_v}}$ , the Fontaine-Laffaille weights for  $\overline{\rho}|_{\Gamma_{K_v}}$  with respect to that embedding are pairwise distinct.

Let  $\mu: G \to \mathbf{G}_m$  be the similitude character, and define  $\overline{\nu} = \mu \circ \overline{\rho}: \Gamma_K \to k^{\times}$ . Suppose there is a lift  $\nu: \Gamma_K \to W(k)^{\times}$  that is Fontaine-Laffaille at all places above p.

**Theorem 1.1.2.2.** Under these assumptions, there exists a geometric lift  $\rho: \Gamma_K \to G(\mathcal{O})$  of  $\overline{\rho}$  where  $\mathcal{O}$  is the ring of integers in a finite extension of  $\mathbf{Q}_p$  with residue field containing k such that  $\mu \circ \rho = \nu$ . In

particular,  $\rho$  is ramified at finitely many places of K, and for every place v of K above p the representation  $\rho|_{\Gamma_{K_n}}$  is Fontaine-Laffaille and hence crystalline.

#### 1.1.3 Relation to Previous Work

Ramakrishna developed his lifting technique when  $K = \mathbf{Q}$  and  $G = \mathrm{GL}_2$  in [Ram99] and [Ram02], and produced geometric lifts. There have been various reformulations and generalizations of this that our results build on. In particular, the formalism developed in [Tay03] (still in the case of  $\mathrm{GL}_2$ ) suggested that it should be possible to generalize the technique to algebraic groups beyonds  $\mathrm{GL}_2$ . Attempts were made in [Ham08] and [Man09] to generalize the result to  $\mathrm{GL}_n$ , but ran into the obstruction that there were no odd representations for n > 2. The results in [Ham08] simply assume the existence of liftable local deformation conditions which probably do not exist, but do provide a nice model for the arguments in Chapter 2 generalizing Ramakrishna's method. In contrast, [Man09] constructs local deformation conditions but does not aim to produce geometric lifts.

For groups beyond  $GL_n$ , [CHT08] gave a lifting result for a group  $\mathcal{G}_n$  related to  $GL_n$  but which admits odd representations. Studying the local deformation conditions for  $\mathcal{G}_n$  reduced to studying representations valued in  $GL_n$ . At primes above p, [CHT08] studied a deformation condition based on Fontaine-Laffaille theory which is generalized in Chapter 3. The idea of doing so goes back to [Ram93]. (They also discussed a deformation condition based on the notion of ordinary representations which is not used in their lifting result). At primes not above p but where  $\overline{p}$  is ramified, they defined a minimally ramified deformation condition, which we generalize in Chapter 4; this generalization is non-obvious and is our main innovation.

Building on this, Patrikis' unpublished undergraduate thesis [Pat06] explored Ramakrishna's method for symplectic groups. In particular, it generalized Ramakrishna's method to the group  $GSp_n$ , and generalized the Fontaine-Laffaille deformation condition at p. It did not generalize the minimally ramified deformation condition, so can only be applied to residual representations  $\Gamma_{\mathbf{Q}} \to GSp_n(k)$  which are unramified away from p, a stringent condition. Our results at p in Chapter 3 are a mild generalization of Patrikis' study of the Fontaine-Laffaille deformation condition.

More recently, Patrikis used Ramakrishna's method to produce geometric representations with exceptional monodromy [Pat15]. This involves generalizing Ramakrishna's method to any connected reductive group G and then modifying the technique to deform a representation valued in the principal  $SL_2 \subset G$  (coming from a modular form) to produce a geometric lift with Zariski-dense image. Chapter 2 is similar but independent of Patrikis' generalization of Ramakrishna's method. Our extensive study of local deformations conditions is not needed in [Pat15]: as the goal there is just to produce examples of geometric representations with exceptional monodromy, he could avoid generalizing the minimally ramified deformation condition.

Remark 1.1.3.1. There is also a completely different technique to produce lifts based on automorphy lifting theorems. For example, Khare and Winterberger use it in their proof of Serre's conjecture: see [KW09b, §4] especially the proof of Corollary 4.7. The finite generation needed in that argument comes from relating the Galois deformation ring to a Hecke algebra.

### 1.2 Overview of the Proof

There are three main steps in the proof Theorem 1.1.2.2. In Chapter 2 we generalize Ramakrishna's method to split connected reductive groups beyond  $GL_2$ ; this reduces the problem to defining appropriate liftable local deformation conditions. The second step is carried out in Chapter 3, where at places above p we use Fontaine-Laffaille theory to produce a liftable deformation condition. Finally in Chapter 4 we reformulate and generalize a deformation condition at places above p studied in [CHT08, §2.4.4] for  $GL_n$  to obtain a liftable deformation condition for  $GSp_{2n}$  and  $GO_m$ .

#### 1.2.1 Generalizing Ramakrishna's Method

Looking at the reformulation of Ramakrishna's method in [Tay03], it is not surprising that the method generalizes to algebraic groups beyond  $GL_2$ . It was generalized in [CHT08] to the group

$$\mathcal{G}_n = (\mathrm{GL}_n \times \mathrm{GL}_1) \times \{1, j\}$$
 where  $j(g, \mu)j^{-1} = (\mu(g^{-1})^T, \mu)$ .

We generalize the method to apply to any connected reductive group, although the method can only produce geometric lifts for groups for which there exist odd representations and for which we have nice local deformation conditions. Stefan Patrikis independently carried out this generalization [Pat15]. His results on the formalism are similar, with some minor variation in the technical hypotheses, but don't address local deformation conditions we require in Chapter 4.

Fix a prime p and finite field k of characteristic p. Let S be a finite set of places of a number field containing the places above p and the archimedean places, and define  $\Gamma_S$  to be the Galois group of the maximal extension of K unramified outside of S. Consider a continuous representation  $\overline{p}: \Gamma_S \to G(k)$  where G is a split connected reductive group scheme over the ring of integers  $\mathcal{O}$  in a p-adic field with maximal ideal  $\mathfrak{m}$  whose residue field contains k. We assume that p is very good for G (Definition 2.1.1.1), so according to Lemma 2.1.1.6 the Lie algebra of the derived group of G is a direct summand of the Lie algebra of G: we denote this summand with adjoint action of  $\Gamma_K$  by  $\operatorname{ad}^0(\overline{p})$ . The cohomology of this Galois module controls the deformation theory of  $\overline{p}$ .

The hope would be to use deformation theory of produce  $\rho_n: \Gamma_S \to G(\mathcal{O}/\mathfrak{m}^n)$  such that  $\rho_1 = \overline{\rho}$ ,  $\rho_n$  lifts  $\rho_{n-1}$  for  $n \geq 2$ , and such that  $\rho_n$  satisfies a deformation condition at places above p for which the inverse limit

$$\rho = \underline{\lim} \, \rho_n : \Gamma_S \to G(\mathcal{O})$$

restricted to the decomposition group  $\Gamma_v$  would be a lattice in a de Rham (or crystalline) representation for places v of K above p. This inverse limit would then be the desired geometric lift of  $\overline{\rho}$ . Only after a careful choice of local deformation conditions and enlarging the set S will this work. Furthermore, defining these deformation conditions may require making an extension of k, which is harmless for our applications and is why we only require that the residue field of  $\mathcal{O}$  merely contains k.

Proposition 2.5.2.1 shows a lifts exist subject to a global deformation condition  $\mathcal{D}_S$  provided the dual Selmer group  $H^1_{\mathcal{D}^\perp_S}(\Gamma_S,\operatorname{ad}^0(\overline{\rho})^*)$  vanishes. This Galois cohomology group is defined in (2.2.2.1), and encodes information about all of the local deformation conditions imposed. When it vanishes, there exists a lift of  $\rho_n$  to  $\rho_{n+1}$  satisfying local deformation conditions for  $v \in S$  provided there exists lifts of  $(\rho_n)|_{\Gamma_v}$  satisfying the deformation condition for all  $v \in S$ . This can be expressed as a local-to-global principle for lifting Galois representations with an obstruction lying in the cohomology group  $H^1_{\mathcal{D}^\perp_S}(\Gamma_S,\operatorname{ad}^0(\overline{\rho})^*)$ .

Corollary 2.4.2.6 gives a deformation condition at places where  $\overline{\rho}$  is unramified such that allowing  $\overline{\rho}$  to deform subject to this condition (i.e. enlarging S to contain such places and defining a new  $\mathcal{D}_S$ ) forces  $H^1_{\mathcal{D}_S^\perp}(\Gamma_S, \operatorname{ad}^0(\overline{\rho})^*)$  to be zero. We call this new deformation condition Ramakrishna's deformation condition, and study it in §2.4. The places of K at which we define this condition are found using the Chebotarev density theorem: each additional place where we allow ramification subject to Ramakrishna's deformation condition decreases the dimension of the dual Selmer group. For such places to exist, we need non-zero classes in certain cohomology groups, whose existence relies on the local deformation conditions satisfying the tangent space inequality (2.2.2.2)

$$\sum_{v \in S} \dim L_v \ge \sum_{v \in S} \dim H^0(\Gamma_v, \operatorname{ad}^0(\overline{\rho})),$$

where  $L_v$  is the tangent space of the local deformation condition at v. Furthermore,  $\overline{\rho}$  needs to a "big" representation in the sense of Definition 2.3.1.1 in order to define Ramakrishna's deformation condition. Being a big representation is a more precise set of technical conditions that are implied for large enough p by the condition that  $G'(k) \subset \overline{\rho}(\Gamma_K)$  appearing in Theorem 2.5.3.4. We discuss big representations in §2.3.

**Remark 1.2.1.1.** Patrikis uses a different set of technical conditions on  $\overline{\rho}$ . He does not require that  $\operatorname{ad}^0(\overline{\rho})$  be an absolutely irreducible representation of  $\Gamma_K$ , but instead imposes several weaker conditions (conditions (1), (5), and (6) of [Pat15, §5]).

For the tangent space inequality to hold, it is crucial that  $\overline{\rho}$  be an *odd* representation. The deformation conditions we will use at places v where  $\overline{\rho}$  is ramified satisfy dim  $H^0(\Gamma_v, \mathrm{ad}^0(\overline{\rho})) = \dim L_v$ . Using the Fontaine-Laffaille deformation condition at places above p, the tangent space inequality becomes

$$[K: \mathbf{Q}](\dim G - \dim B) \ge \sum_{v \mid \infty} h^0(\Gamma_v, \mathrm{ad}^0(\overline{\rho}))$$

where B is a Borel subgroup of G; this can only be satisfied if K is totally real and  $\overline{\rho}$  is odd. Thus in order to use the local-to-global principle, the residual representation must be odd and the deformation conditions we use at places above p and places where  $\overline{\rho}$  is ramified must be liftable.

Remark 1.2.1.2. In general, references for results about reductive group schemes over rings are to [Con14], which gives a self-contained development using more recent methods, even though the results are usually also found in the earlier [SGA3].

### 1.2.2 Fontaine-Laffaille Deformation Condition

Let K be a finite extension of  $\mathbf{Q}_p$ , and  $\mathcal{O}$  be the ring of integers of a p-adic field L with residue field k such that L splits K over  $\mathbf{Q}_p$ . (The latter is always possible after extending k.) Ramakrishna's method requires a deformation condition  $\mathcal{D}_{\overline{\rho}}$  for the residual representation  $\overline{\rho}: \Gamma_K \to G(k)$  such that:

- $\mathcal{D}_{\overline{\rho}}$  is liftable;
- $\mathcal{D}_{\overline{\rho}}$  is large enough, in the precise sense that its tangent space has dimension

$$[K: \mathbf{Q}_p](\dim G - \dim B) + \dim_k H^0(\Gamma_K, \mathrm{ad}^0(\overline{\rho}))$$

where B is a Borel subgroup of G;

•  $\mathcal{D}_{\overline{\rho}}(\mathcal{O})$  consists of certain lattices in crystalline representations.

We will construct such a condition using Fontaine-Laffaille theory.

Fontaine-Laffaille theory, introduced in [FL82], provides a way to describe torsion-crystalline representations in terms of semi-linear algebra when p is unramified in K. In particular, it provides an exact, fully faithful functor  $T_{\rm cris}$  from the category of filtered Dieudonné modules to the category of  $\mathcal{O}[\Gamma_K]$ -modules with continuous action (Fact 3.1.1.14), and describes the image. In [CHT08, §2.4.1], it is used to define a deformation condition for n-dimensional representations, where the allowable deformations of  $\overline{\rho}$  are exactly the deformations of the corresponding Fontaine-Laffaille module. This requires the technical assumption that the representation  $\overline{\rho}$  is torsion-crystalline with Hodge-Tate weights in an interval of length p-2 and furthermore that the Fontaine-Laffaille weights of  $\overline{\rho}$  under each embedding of K into L are distinct (see Remark 3.1.2.6).

We will adapt these ideas to symplectic and orthogonal groups under the stronger assumption that the Fontaine-Laffaille weights lie in an interval of length  $\frac{p-2}{2}$ . For symplectic groups and  $K=\mathbf{Q}_p$ , this was addressed in Patrikis's undergraduate thesis [Pat06]: Chapter 3 is a mild generalization. The key idea is to introduce a symmetric or alternating pairing into the semi-linear algebra data. To do so, it is necessary to use (at least implicitly via statements about duality) the fact that the functor  $T_{\rm cris}$  is compatible with tensor products. This requires the stronger assumption that the Fontaine-Laffaille weights lie in an interval of length  $\frac{p-2}{2}$ , which guarantees that the Fontaine-Laffaille weights of the tensor product lie in an interval of length p-2. Furthermore, it is crucial to use the covariant version of the Fontaine-Laffaille functor used in [BK90] instead of the contravariant version studied in [FL82] in order for this compatibility with tensor products to hold. For more details, see §3.1.2. Given this, it is then reasonably straightforward to check that  $T_{\rm cris}$  is compatible with duality and hence to translate the (perfect) alternating or symmetric pairing of Galois representations into a (perfect) symmetric or alternating pairing of Fontaine-Laffaille modules.

For a coefficient ring R, define  $D^{\mathrm{FL}}_{\bar{\rho}}(R)$  to be all representations  $\rho:\Gamma_K\to G(R)$  lifting  $\bar{\rho}$  and lying in the essential image of  $T_{\mathrm{cris}}$ . To study this Fontaine-Laffaille deformation condition, it suffices to study Fontaine-Laffaille modules. In particular, to show that the deformation condition is liftable (i.e. that it is always possible to lift a deformation satisfying the condition through a square-zero extension), it suffices to show that a Fontaine-Laffaille module with distinct Fontaine-Laffaille weights together with a perfect (symmetric or skew-symmetric) pairing can always be lifted through a square zero extension. This is a complicated but tractable problem in semi-linear algebra: Proposition 3.2.2.1 shows this is always possible. It is relatively simple to lift the underlying filtered module and the pairing, and requires more care to lift the semi-linear maps  $\varphi_M^i: M^i \to M$ . Likewise, to understand the tangent space of the deformation condition it suffices to study deformations of the Fontaine-Laffaille module corresponding to  $\bar{\rho}$  to the dual numbers. Again, the most involved step is understanding possible lifts of the semi-linear maps after choosing a lift of the filtration and the pairing.

**Remark 1.2.2.1.** The proof that  $D_{\overline{\rho}}^{\mathrm{FL}}$  is liftable and the computation of the dimension of its tangent space both use in an essential way the hypothesis that for each embedding of K into L the Fontaine-Laffaille weights are pairwise distinct.

Remark 1.2.2.2. An alternative deformation condition to use at primes above p is a deformation condition based on the concept of an ordinary representation. This is studied for any connected reductive group in [Pat15,  $\S4.1$ ]. It is suitable for use in Ramakrishna's method, and can give lifting results for a different class of torsion-crystalline representations.

### 1.2.3 Minimally Ramified Deformation Condition

Let  $\ell \neq p$  be primes, L be a finite extension of  $\mathbf{Q}_{\ell}$ , and k a finite field of characteristic p. For a split connected reductive group G over the valuation ring  $\mathcal{O}$  of a p-adic field K with residue field k, consider a residual representation  $\overline{p}: \Gamma_L \to G(k)$ . Ramakrishna's method requires a "nice" deformation condition for  $\overline{p}$ . If  $\overline{p}$  were unramified, the unramified deformation condition would work. The interesting case is when  $\overline{p}$  is ramified: we would like to define a deformation condition of lifts which are "ramified no worse that  $\overline{p}$ ," so the resulting deformation condition is liftable despite the fact that the unrestricted deformation condition for  $\overline{p}$  may not be liftable. To be precise, we require a deformation condition  $\mathcal{D}_{\overline{p}}$  such that  $\mathcal{D}_{\overline{p}}$  is liftable and whose tangent space has dimension (at least)  $\dim_k H^0(\Gamma_L, \operatorname{ad}^0(\overline{p}))$ .

In the case that  $G = GL_n$ , the minimally ramified deformation condition defined in [CHT08, §2.4.4] works. We will generalize this to a minimally ramified deformation condition for symplectic and orthogonal groups when p > n. Attempting to generalize the argument of [CHT08, §2.4.4] to groups besides  $GL_n$  leads to a deformation condition based on parabolics which is not liftable. Instead, inspired by the arguments of [Tay08, §3] we define a deformation condition for symplectic and orthogonal groups based on deformations of a nilpotent element of  $\mathfrak{g}_k = \text{Lie } G_k$ .

Let us first review the minimally ramified deformation condition introduced for  $GL_n$  in [CHT08, §2.4.4]. The first step is to reduce to studying certain tamely ramified representations. Recall that  $\Gamma_L^t$ , the Galois group of the maximal tamely ramified extension of L, is isomorphic to the semi-direct product

$$\widehat{\mathbf{Z}} \ltimes \prod_{p' \neq \ell} \mathbf{Z}_{p'}$$

where  $\widehat{\mathbf{Z}}$  is generated by a Frobenius  $\phi$  for L and the conjugation action by  $\phi$  on each  $\mathbf{Z}_{p'}$  is given by the p'-adic cyclotomic character. We consider tamely ramified representations which factor through the quotient  $\widehat{\mathbf{Z}} \ltimes \mathbf{Z}_p$  (recall  $p \neq \ell$ ). Picking a topological generator  $\tau$  for  $\mathbf{Z}_p$ , the action is explicitly given by

$$\phi\tau\phi^{-1}=q\tau$$

where q is the size of the residue field of L. Note q is a power of  $\ell$ , so it is relatively prime to p. Arguments in [CHT08] reduce the lifting problem to studying representations of the group  $T_q := \widehat{\mathbf{Z}} \ltimes \mathbf{Z}_p$ . This reduction generalizes without surprises to symplectic and orthogonal groups.

The second step is to specify when a lift of  $\overline{\rho}: T_q \to \operatorname{GL}_n(k)$  is "ramified no worse than  $\overline{\rho}$ ". For suitable "coefficient rings" R, a deformation  $\rho: T_q \to \operatorname{GL}_n(R)$  is minimally ramified according to [CHT08] when the natural k-linear map

$$\ker\left(\left(\rho(\tau) - 1_n\right)^i\right) \otimes_R k \to \ker\left(\left(\overline{\rho}(\tau) - 1_n\right)^i\right) \tag{1.2.3.1}$$

is an isomorphism for all i. The deformation condition is analyzed as follows:

• defining  $V_i = \ker ((\overline{\rho}(\tau) - 1_n)^i)$  gives a flag

$$0 \subset V_r \subset V_{r-1} \subset \ldots \subset V_1 \subset k^n$$
.

This flag determines a parabolic k-subgroup  $\overline{P} \subset \operatorname{GL}_n$  (points which preserve the flag) such that  $\overline{\rho}(\tau) \in (\mathcal{R}_u \overline{P})(k)$  and  $\overline{\rho}(\phi) \in \overline{P}(k)$ ;

- lift  $\overline{P}$  to a parabolic subgroup P of  $GL_n$ . The deformation functor of such lifts is formally smooth, and for any minimally ramified deformation  $\rho$  over R there is a choice of such P for which  $\rho(\tau) \in (\mathcal{R}_u P)(R)$  and  $\rho(\phi) \in P(R)$ . Conversely, any  $\rho$  with this property is minimally ramified;
- $\bullet$  Finally, for the standard block-upper-triangular choice of P, one shows the deformation functor

$$\{(T,\Phi): T \in \mathcal{R}_u P, \Phi \in P, \Phi T \Phi^{-1} = T^q, \overline{T} = \overline{\rho}(\tau), \overline{\Phi} = \overline{\rho}(\phi)\}$$

is formally smooth by building the universal lift over a power series ring: this uses explicit calculations with block-upper-triangular matrices.

To generalize beyond  $GL_n$ , we need to replace (1.2.3.1) with a more group-theoretic criterion. The naive generalization is to associate a parabolic  $\overline{P}$  to  $\overline{\rho}$  and then use the following definition.

**Definition 1.2.3.1.** For a "coefficient ring" R, say a lift  $\rho: T_q \to G(R)$  is ramified with respect to  $\overline{P}$  provided that there exists a parabolic R-subgroup  $P \subset G_R$  lifting  $\overline{P}$  such that  $\rho(\tau) \in (\mathcal{R}_u P)(R)$  and  $\rho(\phi) \in P(R)$ .

This idea does not work. Let us focus on the symplectic case to illustrate what goes wrong.

The first problem is to associate a parabolic subgroup to  $\overline{\rho}$ . Recall that parabolic subgroups of a symplectic group correspond to isotropic flags  $0 \subset V_1 \subset \ldots \subset V_r \subset V_r^{\perp} \subset \ldots \subset V_1^{\perp} \subset k^{2n}$ . There is no reason that the flag determined by (1.2.3.1) is isotropic, so we would need some other method of producing a parabolic  $\overline{P}$  such that  $\overline{\rho}(\tau) \in (\mathcal{R}_u \overline{P})(k)$ . In [BT71], Borel and Tits give a natural way to associate to the unipotent  $\overline{\rho}(\tau)$  a smooth connected unipotent k-subgroup of G. The normalizer of this subgroup is always parabolic and so gives a candidate for  $\overline{P}$ . However, working out examples in  $\mathrm{GL}_n$  for small n shows that this produces a different parabolic than the one determined by (1.2.3.1). This raises the natural question of how sensitive the smoothness of the deformation condition is to the choice of parabolic.

This leads to the second, larger problem: there are examples such that for *every* parabolic  $\overline{P}$  satisfying  $\overline{\rho}(\tau) \in (\mathcal{R}_u \overline{P})(k)$ , not all deformations ramified with respect to  $\overline{P}$  are liftable.

**Example 1.2.3.2.** Take  $L = \mathbf{Q}_{29}$  and  $k = \mathbf{F}_7$ . Consider the representation  $\overline{\rho}: T_{29} \simeq \widehat{\mathbf{Z}} \ltimes \mathbf{Z}_7 \to \mathrm{GSp}_4(\mathbf{F}_7)$  defined by

$$\overline{\rho}(\tau) = \begin{pmatrix} 1 & 0 & 1 & 0 \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 \end{pmatrix} \quad \text{and} \quad \overline{\rho}(\phi) = \begin{pmatrix} 1 & -1 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 1 & 1 \end{pmatrix}.$$

The deformation condition of lifts ramified relative to any parabolic of  $GSp_4$  whose unipotent radical contains  $\overline{\rho}(\tau)$  is not liftable: there are lifts to the dual numbers that do not lift to  $\mathbf{F}_7[\epsilon]/(\epsilon^3)$ . This is easy to check with a computer algebra system such as [SAGE], since the existence of lifts can be reduced to a problem in linear algebra.

This latter problem is a general phenomenon, which we will explain conceptually in terms of algebraic geometry in §4.4.3.

The correct approach is to define a lift  $\rho: T_q \to G(R)$  to be minimally ramified if  $\rho(\tau)$  has "the same unipotent structure" as  $\overline{\rho}(\tau)$ . It is more convenient to work with nilpotent elements, using the exponential and logarithm maps (defined for nilpotent and unipotent elements since p > n). We wish to study lifts of the nilpotent  $\overline{N} = \log(\overline{\rho}(\tau))$  to  $N \in \mathfrak{g}$  that "remain nilpotent of the same nilpotent type as  $\overline{N}$ ".

In Chapter 4, we make this notion of "same nilpotent type" rigorous for classical groups. There are combinatorial parametrization of nilpotent orbits of algebraic groups over an algebraically closed field, for example in terms of partitions or root data, which make precise the notion that the values of  $N \in \mathfrak{g}_{\mathcal{O}}$  in the special and generic fiber lie in the "same" nilpotent orbit. For each nilpotent orbit  $\sigma$  (classified by purely combinatorial data, without reference to the field), pick an element  $N_{\sigma} \in \mathfrak{g}_{\mathcal{O}}$  with this property lifting  $\overline{N} \in \mathfrak{g}_k$ . For a coefficient ring R, we define the "pure nilpotents" lifting  $\overline{N}$  to be the  $\widehat{G}(R)$ -conjugates of  $N_{\sigma}$ .

**Example 1.2.3.3.** For example, let  $G = GL_3$  and

$$\overline{N} = \begin{pmatrix} 0 & 1 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}$$

Consider the lifts

$$N_1 = \begin{pmatrix} 0 & 1 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix} \in \mathfrak{g} \quad \text{and} \quad N_2 = \begin{pmatrix} 0 & 1 & 0 \\ 0 & 0 & p \\ 0 & 0 & 0 \end{pmatrix} \in \mathfrak{g}.$$

Both are nilpotent under the embedding of  $\mathcal{O}$  into its fraction field K. The images of  $N_1$  in  $\mathfrak{g}_K$  and  $\mathfrak{g}_k$  both lie in the nilpotent orbit corresponding to the partition 2+1, so  $N_1$  is an example of the type of nilpotent lift we want to consider. On the other hand, the image of  $N_2$  in  $\mathfrak{g}_K$  lies in the nilpotent orbit corresponding to the partition 3, so we do not want to use it. The pure nilpotents lifting  $\overline{N}$  are  $\widehat{G}(R)$ -conjugates of  $N_1$ .

We then define a lift  $\rho: T_q \to G(R)$  to be minimally ramified provided  $\rho(\tau)$  is the exponential of a pure nilpotent lifting  $\log \overline{\rho}(\tau) = \overline{N}$ . In §4.4, we show that this deformation condition is liftable. The main technical fact needed to analyze this deformation condition is that the scheme-theoretic centralizer  $Z_G(N_\sigma)$  is smooth over  $\mathcal{O}$  for  $N_\sigma$  as above. The smoothness of such centralizers over algebraically closed fields is well-understood, and in §4.2 we study  $Z_G(N_\sigma)$  and show that  $Z_G(N_\sigma)$  is flat over  $\mathcal{O}$  and hence smooth: Lemma 4.2.3.1 gives a criterion for flatness that is easy to verify for classical groups which suffices for our applications; there are difficulties beyond the classical case due the structure of  $\pi_0(Z_G(N_\sigma)_{\overline{k}})$  in general.

**Remark 1.2.3.4.** It is a fortuitous coincidence (for [CHT08]) that for  $GL_n$  the lifts minimally ramified in the preceding sense are exactly the lifts ramified with respect to a parabolic subgroup of G. This rests on the fact that all nilpotent orbits of  $GL_n$  are Richardson orbits (see §4.4.3 for details).

### 1.3 Acknowledgements

I am extremely grateful for the patience, generosity, technical prowess, and support of my advisor Brian Conrad, and for his extensive and helpful comments on drafts of this thesis. I would also like to thank my committee members Aaron Pollack, Akshay Venkatesh, and Zhiwei Yun for their support and comments. I thank Brandon Levin for bringing [Pat06] to my attention, and Stefan Patrikis for his encouragement and conversations. The Stanford mathematics department was a supportive and intellectually enriching environment, and am grateful to all of the faculty members, postdocs, and graduate students who made it so. In addition to those already mentioned, I especially thank Rebecca Bellovin, Iurie Boreico, Sam Lichtenstein, Daniel Litt, Evita Nestoridi, John Pardon, Niccolo' Ronchetti, Jesse Silliman, and Ravi Vakil. Outside of Stanford, I thank Benedict Gross and Richard Taylor both for their teaching when I was an undergraduate and for subsequent conversations.

## Chapter 2

## **Producing Geometric Deformations**

In this chapter, we prove a generalization of Ramakrishna's lifting technique that applies to a wide class of reductive Chevalley groups G (Theorem 2.5.3.4). The key result is Proposition 2.5.2.1, a local-to-global principle for lifting Galois representations: under appropriate conditions, for a finite field k and a number field K with absolute Galois group  $\Gamma_K$  it is possible to lift a global representation  $\rho: \Gamma_K \to G(R/I)$  through a square zero extension  $R \to R/I$  provided it is possible to lift the local representations  $\rho|_{\Gamma_v}$  for v in a specific (finite) set of places. This reduces the problem to studying local Galois deformation rings, which we do in subsequent chapters.

This technique was first used in [Ram99] and [Ram02] to produce geometric deformations for two-dimensional representations. The method for  $GL_2$  was axiomatized by Taylor [Tay03], making it easier to generalize to other groups. We will generalize the technique to apply to any split connected reductive group, given as input local deformation conditions satisfying certain axioms; instances of such conditions are defined and studied in later chapters for symplectic and orthogonal groups. Patrikis has independently generalized this lifting technique [Pat15, Theorem 6.4]. The approach is the same, but he relies on different local deformation conditions.

Some background material about algebraic groups is reviewed in §2.1, in particular the notion of a very good prime and some facts about finite groups of Lie type. In §2.2, we review background on Galois cohomology and the deformation theory of Galois representations. Next in §2.3 we discuss a technical condition, bigness, on the residual representation necessary for the method to work, and show that if the image of  $\bar{\rho}$  contains the k-points of the derived group then these conditions hold provided the characteristic of k is large enough relative to the Coxeter number of (the root system of) G. We construct an auxiliary local condition that Ramakrishna's method needs in §2.4. Finally in §2.5 we generalize Ramakrishna's method, providing a local to global principle after allowing the representation to ramify at finitely many additional places of K subject to this auxiliary condition.

### 2.1 Preliminaries about Algebraic Groups

This section collects some results about algebraic groups. In particular, we review the notion of a very good prime and we give a few results about finite groups of Lie type.

### 2.1.1 Good and Very Good Primes

Let p be a prime,  $\Phi$  a reduced and irreducible root system, and  $P = (\mathbf{Z}\Phi^{\vee})^*$  the weight lattice for  $\Phi$ . Suppose G is a connected reductive group over a field k.

**Definition 2.1.1.1.** The prime p is good for  $\Phi$  provided that  $\mathbf{Z}\Phi/\mathbf{Z}\Phi'$  is p-torsion free for all subsets  $\Phi' \subset \Phi$ . A good prime is  $very\ good$  provided that  $P/\mathbf{Z}\Phi'$  is p-torsion free for all subsets  $\Phi' \subset \Phi$ . A prime is bad if it is not good.

Likewise, we say a prime p is good (or very good) for a general reduced root system if it is good (or very good) for each irreducible component. A prime p is good (or very good) for G provided it is good (or very good) for G provided it is good (or good) for good (or good)

good) for the root system of  $G_{\overline{k}}$ .

Let  $\{\alpha_1, \ldots, \alpha_r\}$  be a basis of positive roots for  $\Phi$ . Let  $\sum_i m_i \alpha_i$  be the highest root.

**Fact 2.1.1.2.** The prime p is bad for  $\Phi$  if and only if there is an i with  $p|m_i$ .

This is  $[SS70, \S I.4.3]$ .

**Example 2.1.1.3.** Using the characterization in terms of the highest root, we obtain:

- For type  $A_n$   $(n \ge 1)$ , all primes are good.
- For types  $B_n$   $(n \ge 2)$ ,  $C_n$   $(n \ge 2)$ , or  $D_n$   $(n \ge 4)$ , p is good if and only if  $p \ne 2$ .
- For types  $E_6$ ,  $E_7$ ,  $F_4$  or  $G_2$ , p is good if and only if  $p \neq 2, 3$ .
- For type  $E_8$ , p is good if and only if  $p \neq 2, 3, 5$ .

Furthermore, p is very good if and only if p is good and moreover  $p \nmid n+1$  when  $\Phi$  is of type  $A_n$  due to:

**Lemma 2.1.1.4.** The prime p is very good for  $\Phi$  if and only if p is good for  $\Phi$  and  $p \nmid \#\pi_1(\Phi)$ .

*Proof.* For a subset  $\Phi' \subset \Phi$ , consider the exact sequence

$$0 \to \frac{\mathbf{Z}\Phi}{\mathbf{Z}\Phi'} \to \frac{P}{\mathbf{Z}\Phi'} \to \frac{P}{\mathbf{Z}\Phi} \to 0.$$

The right term is the definition of  $\pi_1(\Phi)$ . The middle term is *p*-torsion free if the left and right terms are, proving the "only if" statement. Conversely, if *p* is very good (the middle term is *p*-torsion free) by choosing  $\Phi' = \Phi$  we get  $p \nmid \#\pi_1(\Phi)$ .

We record some consequences of p being a very good prime. Let G' be the derived group of G,  $\mathfrak{g} = \text{Lie } G$ , and  $\mathfrak{g}' = \text{Lie } G'$ . Denote the center of G by  $Z_G$ . Suppose that G is k-split.

**Lemma 2.1.1.5.** If p is very good for G, the order of the fundamental group of G' and the order of the center of G' are prime to p. Furthermore, any central isogeny  $H \to G'$  or  $G' \to H$  for a connected semisimple k-group H induces an isomorphism on Lie algebras.

*Proof.* As p is very good, the central isogeny from the simply connected cover of G' to the adjoint quotient of G' has order prime to p: the degree is the size of the fundamental group of the root system. Thus the kernel of a central isogeny between G' and H has order to prime to p, so it is étale and its Lie algebra is zero.

**Lemma 2.1.1.6.** Let p be very good for G. Then  $\text{Lie}(Z_G)$  is the center  $\mathfrak{z}_{\mathfrak{g}}$  of  $\mathfrak{g}$  and moreover  $\mathfrak{g} = \mathfrak{z}_{\mathfrak{g}} \oplus \mathfrak{g}'$ .

*Proof.* Consider the adjoint map  $\operatorname{Ad}_G: G \to \operatorname{GL}(\mathfrak{g})$ . Its scheme theoretic kernel is  $Z_G$ , so the kernel of  $\operatorname{Lie}(\operatorname{Ad}_{\mathfrak{g}}): \mathfrak{g} \to \mathfrak{gl}(\mathfrak{g})$  is  $\operatorname{Lie}Z_G$ . But  $\operatorname{Lie}(\operatorname{Ad}_G) = \operatorname{ad}_{\mathfrak{g}}$ , the kernel of which is  $\mathfrak{z}_{\mathfrak{g}}$ . As p is very good, the Lie algebra  $\mathfrak{g}'$  maps isomorphically to the Lie algebra of the adjoint quotient  $G/Z_G$  by Lemma 2.1.1.5. This gives the decomposition.

### 2.1.2 Finite Groups of Lie Type

Now suppose k is finite, with q = #k. Continue to assume the characteristic p of k is very good for G. We record a couple of classical facts from the theory of finite groups of Lie type.

**Lemma 2.1.2.1.** If G is a split, absolutely simple, connected semisimple group over k,  $\mathfrak{g} = \text{Lie } G$  is an absolutely irreducible G(k)-module provided that q > 3.

Proof. The adjoint representation  $\operatorname{Ad}_G: G_{\overline{k}} \to \operatorname{GL}(\mathfrak{g})$  is an absolutely irreducible representation of algebraic groups as p is very good [MT11, Theorem 15.20]. Its highest weight is the highest root  $\lambda = \sum_i m_i \alpha_i$ . If  $\langle \lambda, \alpha_i^\vee \rangle < q$  for all  $\alpha_i$ , then the adjoint representation will be absolutely irreducible as a G(k)-module; this was originally proven in [Ste63], with a modern proof in [Hum06, §2.12]. The value of  $\langle \lambda, \alpha_i^\vee \rangle$  can be read off from the affine Dynkin diagram of the root system, as the additional node is the lowest root. In particular, it is at most 3.

**Lemma 2.1.2.2.** Suppose that q > 3. Let G be a split connected semsimple group over k with simply connected central cover  $\pi : \widetilde{G} \to G$ . Then  $[G(k), G(k)] = \pi(\widetilde{G}(k))$ .

*Proof.* Consider the exact sequence

$$1 \to (\ker \pi)(k) \to \widetilde{G'}(k) \to G'(k) \to H^1(k, \ker \pi)$$

(using fppf  $H^1$ ). We know that  $\widetilde{G}'(k)$  is perfect:  $\widetilde{G}'$  is a product of absolutely simple semisimple simply connected groups, and the k-points of all such groups, with a few exceptions ruled out by the assumption q > 3, are perfect [MT11, Theorem 24.11]. Thus  $\pi(\widetilde{G}'(k)) \subset [G'(k), G'(k)]$ . As  $G'(k)/\pi(\widetilde{G}'(k)) \subset H^1(k, \ker \pi)$  is abelian, we conclude that  $\pi(\widetilde{G}'(k)) = [G'(k), G'(k)]$ .

Remark 2.1.2.3. While G is always perfect as an algebraic group, G(k) need not be perfect. For example, consider  $G = \operatorname{PGL}_n$ , which is perfect as an algebraic group. One can calculate that  $[\operatorname{PGL}_n(k), \operatorname{PGL}_n(k)] \simeq \operatorname{SL}_n(k)/\mu_n(k)$  which has index  $\gcd(n, q-1)$  in  $\operatorname{PGL}_n(k)$ .

Now we record and slightly generalize some results about the group cohomology from [CPS75] for G a split connected semisimple group over k.

**Proposition 2.1.2.4.** If  $p \neq 2$  is very good for G and  $q \neq 2, 3, 4, 5, 9$  then  $H^1(G(k), \mathfrak{g}) = 0$ .

Before we prove this result, we need some preparation. Let  $\Delta$  be a set of positive simple roots corresponding to a Borel  $B \subset G$  containing a split maximal torus T. Denote the unipotent radical of B by U. Consider  $V = \mathfrak{g}$  as a representation of G(k). Roots  $\alpha \in \Phi(G,T)$  give homomorphisms  $\alpha_k : T(k) \to k^{\times}$ . We say that two roots  $\alpha$  and  $\beta$  are equivalent over k if  $\ker \alpha_k = \ker \beta_k$ . We say that  $\alpha$  is equivalent over k to 0 if  $\alpha_k = 1$ .

**Remark 2.1.2.5.** This notion is called "Galois equivalence" in [CPS75]. When  $q \neq 2$  (so  $k^{\times} \neq 1$ ), such equivalences occur only for roots in a common irreducible component of  $\Phi(G,T)$ , or between a root and zero, and occur only over small fields [CPS75, Proposition 3.3]: none are possible when  $q \neq 2, 3, 4, 5, 9$ .

For the rest of §2.1.2, we assume that p and q = #k are as in Proposition 2.1.2.4.

**Lemma 2.1.2.6.** We have  $V^{B(k)} = 0$ .

*Proof.* As the characteristic is very good, it suffices to prove this for the simply connected central cover  $\pi: \widetilde{G} \to G$ : the Lie algebra is unchanged, and  $\pi^{-1}(B)$  is a Borel subgroup of  $\widetilde{G}$  containing a split maximal torus  $\pi^{-1}(T)$ . As G is now simply connected,  $\mathfrak{t} := \operatorname{Lie} T$  decomposes as

$$\mathfrak{t} = \bigoplus_{\alpha \in \Delta} \mathfrak{t}_{\alpha^\vee}$$

where  $\mathfrak{t}_{\alpha^{\vee}}$  is the coroot line associated to the simple coroot  $\alpha^{\vee}$ . As  $q \neq 2, 3$ ,  $\alpha_k : T(k) \to k^{\times}$  is non-trivial for every root  $\alpha$  by [CPS75, Proposition 3.1]. Hence  $V^{T(k)} \subset \mathfrak{t}$ , so  $V^{B(k)} \subset \mathfrak{t}$ .

Let  $t_{\alpha} = d\alpha^{\vee}(x^{-1}\partial_x|_{x=1})$  be the standard basis vector for  $\mathfrak{t}_{\alpha^{\vee}}$ . Consider an element  $v \in \mathfrak{t}$  fixed by B(k), and decompose it as

$$v = \sum_{\alpha \in \Delta} v_{\alpha} t_{\alpha}$$

with  $v_{\alpha} \in k$ . For  $\beta \in \Delta$ , fix an isomorphism  $u_{\beta} : \mathbf{G}_a \simeq U_{\beta}$  so  $e_{\beta} = du_{\beta}(\partial_x|_{x=0})$ . For  $u = u_{\beta}(1) \in U_{\beta}(k)$ , we wish to calculate

$$\operatorname{Ad}(u)(v) = \sum_{\alpha \in \Delta} v_{\alpha} \operatorname{ad}(u) t_{\alpha}.$$

We calculate that for  $z \in \mathbf{G}_m$ ,

$$u\alpha^{\vee}(z)u^{-1} = \alpha^{\vee}(z) \left(\alpha^{\vee}(z)^{-1}u_{\beta}(1)\alpha^{\vee}(z)\right) u_{\beta}(1)^{-1}$$
$$= \alpha^{\vee}(z)u_{\beta} \left(z^{-\langle \beta, \alpha^{\vee} \rangle} - 1\right).$$

Thus we obtain

$$\operatorname{Ad}(u)t_{\alpha} = t_{\alpha} - \langle \beta, \alpha^{\vee} \rangle \cdot e_{\beta}.$$

Applying  $\sum_{\alpha \in \Delta} v_{\alpha}(\cdot)$  to both sides gives

$$\operatorname{Ad}(u)v = v - \left(\sum_{\alpha \in \Delta} v_{\alpha} \langle \beta, \alpha^{\vee} \rangle v_{\alpha}\right) e_{\beta}.$$

If v is fixed by  $U_{\beta}(k)$ , hence by  $u = u_{\beta}(1)$ , it follows that

$$\sum_{\alpha \in \Lambda} v_{\alpha} \langle \beta, \alpha^{\vee} \rangle = 0$$

since  $e_{\beta} \in \mathfrak{g}$  is non-zero. If this holds for all  $\beta \in \Delta$ , then the  $\Delta$ -tuple  $(v_{\alpha}) \in k^{\Delta}$  is in the kernel of the Cartan matrix over k. But the Cartan matrix is invertible over k as the determinant is a unit in very good characteristic. (This can be verified by inspecting tables of root systems.) Hence  $V^{B(k)} = 0$ .

Now we prove Proposition 2.1.2.4. In the case that G is simply connected, we may apply [CPS75, Corollary 2.9]: for the adjoint representation  $V = \mathfrak{g}$  of G(k), letting  $V[\alpha]$  denote the weight space for  $\alpha_k$ , it states

$$\dim_k H^1(G(k), V) = \left(\sum_{\alpha \in \Delta} \dim_k V[\alpha]\right) - \dim_k V[0],$$

as there are no equivalences between distinct roots, nor between roots and 0, because of the restrictions on q. The weight space for  $\alpha \in \Delta$  (as representations of T(k)) is therefore the  $\alpha$ -root line. Since  $\#\Delta = \dim T$  and Lie T = V[0], we obtain that  $H^1(G(k), \mathfrak{g}) = 0$ .

For general G, let  $\pi: \widetilde{G} \to G$  be the simply connected central cover of G. The central k-subgroup scheme  $\ker \pi$  is finite, of order prime to p as p is very good, and  $d\pi: \operatorname{Lie} \widetilde{G} \to \mathfrak{g}$  is an isomorphism. One instance of the inflation-restriction sequence reads

$$0 \to H^1(\pi(\widetilde{G}(k)), \mathfrak{g}) \to H^1(\widetilde{G}(k), \mathfrak{g})$$

As the final term is 0, so is the middle term. Now  $\pi(\widetilde{G}(k))$  has index prime to p in G(k) as the index divides  $\#H^1(k, \ker \alpha)$ , so the composition

$$\operatorname{cor} \circ \operatorname{res} : H^1(G(k), \mathfrak{g}) \to H^1(G(k), \mathfrak{g})$$

is an isomorphism. Since  $H^1(\pi(\widetilde{G}(k)), \mathfrak{g}) = 0$ , it follows that  $H^1(G(k), \mathfrak{g}) = 0$  as desired. This completes the proof of Proposition 2.1.2.4.

### 2.2 Review of Galois Cohomology and Deformations

### 2.2.1 Results about Galois Cohomology

Let  $K_v$  be a p-adic field with normalized valuation v and absolute Galois group  $\Gamma_{K_v}$ , and let V be a finite discrete  $\Gamma_{K_v}$ -module. Let  $\mu_{\infty}$  denote the Galois module of roots of unity in  $\overline{K_v}$ . Define the (Cartier) dual of V to be

$$V^* := \operatorname{Hom}_{\mathbf{Z}}(V, \mu_{\infty}).$$

As a Galois representation it is a Tate twist of the  $\mathbf{Q}/\mathbf{Z}$  dual of V. There is a natural evaluation pairing  $V \otimes V^* \to \mu_{\infty}$  as  $\Gamma_{K_v}$ -modules. We recall some standard facts about Galois cohomology for local and global fields (found for example in [NSW08, Chapter VII, VIII]). Let  $H^i(K_v, V) := H^i(\Gamma_{K_v}, V)$  be the *i*th (continuous) Galois cohomology group.

**Fact 2.2.1.1** (Local Tate Duality). For i = 0, 1, 2 the cup product gives a non-degenerate pairing

$$H^i(K_v,V) \times H^{2-i}(K_v,V^*) \to H^2(K_v,\mu_\infty) \simeq \mathbf{Q}/\mathbf{Z}.$$

Fact 2.2.1.2 (Local Euler-Poincaré Characteristic). Let  $N = ||\#V||_v$  (the normalized absolute value, or equivalently the index of  $(\#V)\mathcal{O}_{K_v}$  in  $\mathcal{O}_{K_v}$ ). Then

$$\frac{\#H^0(K_v,V)\#H^2(K_v,V)}{\#H^1(K_v,V)} = \frac{1}{N}$$

Let  $I_{K_v}$  be the inertia subgroup of  $\Gamma_{K_v}$ .

**Definition 2.2.1.3.** The unramified ith cohomology of V is the group

$$H^i_{\mathrm{nr}}(K_v, V) := \mathrm{image}\left(H^i(\Gamma_{K_v}/I_{K_v}, V^{I_{K_v}}) \to H^i(K_v, V)\right).$$

Fact 2.2.1.4. We have 
$$H^1_{nr}(K_v, V) = \ker \left( H^1(K_v, V) \to H^1(I_{K_v}, V) \right)$$
 and  $\# H^1_{nr}(K_v, V) = \# H^0(K_v, V)$ .

Now let K be a number field with absolute Galois group  $\Gamma_K$ , and let S be a finite set of places of K containing the archimedean places. Let  $K_S$  be the maximal extension of K unramified outside of S and  $\Gamma_S = \operatorname{Gal}(K_S/K)$ . If v is a place of K, denote the decomposition group at v in  $\Gamma_K$  by  $\Gamma_v$  (well-defined up to conjugation); here we allow  $v|\infty$ . For a finite discrete  $\Gamma_S$ -module W and  $v \in S$ , there is a restriction map  $\operatorname{res}_v : H^i(\Gamma_S, W) \to H^i(K_v, W)$  where  $H^i(K_v, W) := H^i(\Gamma_v, W)$ .

**Definition 2.2.1.5.** For a place  $v \in S$  (allowing  $v|\infty$ ), a local condition N for W at v is a subgroup of  $H^1(K_v, W)$ . A global condition  $N_S$  for W is a collection of local conditions  $N_v$ , one for each  $v \in S$ . The generalized Selmer group for W with respect to  $N_S$  is

$$H^1_{N_S}(\Gamma_S,W):=\{x\in H^1(\Gamma_S,W)|\operatorname{res}_v(x)\in N_v \text{ for all }v\in S\}.$$

Given a local condition N at non-archimedean v, one can define a dual local condition  $N^{\perp} \subset H^1(K_v, W^*)$ : it is the exact annihilator of N via Tate local duality. There is also a version for achimedean places [NSW08, Theorem 7.2.17]. Then one defines a dual global condition  $N_S^{\perp}$  to be the collection  $\{N_v^{\perp}\}$  for places v of S.

Remark 2.2.1.6. Let W be a finite discrete  $\Gamma_K$ -module that is unramified outside of S. We often think of a global condition as being a collection of local conditions for *every* place of K, where for  $v \notin S$  the deformation condition is the unramified condition  $H^1_{nr}(K_v, W)$ . This allows us to work with  $\Gamma_K$  instead of  $\Gamma_S$ , which will be convenient when we want to enlarge S later. If the global condition  $N_S$  is the unramified condition outside of S, we will write

$$H^1_{N_S}(K, W) := \{ x \in H^1(K, W) | \operatorname{res}_v(x) \in N_v \text{ for all } v \}.$$

Unwinding definitions, it follows from Fact 2.2.1.4 that

$$H^1_{N_S}(\Gamma_S, W) = H^1_{N_S}(K, W).$$

This convention behaves well with respect to the notion of dual global conditions because when  $v \nmid \infty$  and #W is a v-unit (as holds for all but finitely many v), the groups  $H^1_{\rm nr}(\Gamma_v, W)$  and  $H^1_{\rm nr}(\Gamma_v, W^*)$  are exact annihilators under the Tate-local duality pairing.

The generalized Selmer groups as in Definition 2.2.1.5 fit into a long exact sequence, known as the Poitou-Tate exact sequence. For us, the relevant part is the following five term sequence.

Fact 2.2.1.7 (Modified Poitou-Tate Exact Sequence). If #W is an S-unit, there is an exact sequence

$$H^1(\Gamma_S, W) \to \bigoplus_{v \in S} H^1(K_v, W)/N_v \to H^1_{N_S^{\perp}}(K, W^*)^{\vee} \to H^2(\Gamma_S, W) \to \bigoplus_{v \in S} H^2(K_v, W).$$

This gives the following equality due to Wiles [NSW08, Theorem 8.7.9]:

$$\frac{\#H^1_{N_S}(K,W)}{\#H^1_{N_{\sigma}}(K,W^*)} = \frac{\#H^0(K,W)}{\#H^0(K,W^*)} \prod_{v \in S} \frac{\#N_v}{\#H^0(K_v,W)}.$$
 (2.2.1.1)

Note that  $\#N_v = \#H^0(K_v, W)$  for the unramified condition at  $v \in S$  by Fact 2.2.1.4, so the product is insensitive to enlarging S provided the additional local conditions are the unramified condition.

### 2.2.2 Deformations of Galois Representations

Next we recall some facts about the deformation theory for Galois representations: a basic reference is [Maz97], with the extension to algebraic groups beyond  $GL_n$  discussed in [Til96].

Let k be a finite field of characteristic p, and consider the category of complete local Noetherian rings with residue field k, where morphisms are local homomorphisms that induce the identity map on k. Objects of this category are called coefficient rings. For a coefficient ring  $\mathcal{O}$ , a coefficient  $\mathcal{O}$ -algebra is a coefficient ring which is a  $\mathcal{O}$ -algebra such that the structure morphism is a map of coefficient rings. Denote the category of coefficient  $\mathcal{O}$ -algebras by  $\widehat{\mathcal{C}}_{\mathcal{O}}$ , and the full subcategory of Artinian coefficient  $\mathcal{O}$ -algebras by  $\mathcal{C}_{\mathcal{O}}$ . Note that all coefficient rings are coefficient W(k)-algebras.

**Definition 2.2.2.1.** A small surjection of coefficient  $\mathcal{O}$ -algebras  $f: A_1 \to A_0$  is a surjection such that  $\ker(f) \cdot \mathfrak{m}_{A_1} = 0$ .

Let  $\Gamma$  be a pro-finite group satisfying the following finiteness property: for every open subgroup  $\Gamma_0 \subset \Gamma$ , there are only finitely many continuous homomorphisms from  $\Gamma_0$  to  $\mathbf{Z}/p\mathbf{Z}$ . This is true for the absolute Galois group of a local field and for the Galois group of the maximal extension of a number field unramified outside a finite set of places.

Consider a reductive group scheme G (with connected fibers) over a coefficient ring  $\mathcal{O}$  with derived group G' whose center is smooth over  $\mathcal{O}$ . Assume that p is very good for  $G_k$  (in the sense of Definition 2.1.1.1). For  $A \in \widehat{\mathcal{C}}_{\mathcal{O}}$ , define

$$\widehat{G}(A) := \ker(G(A) \to G(k))$$

We are interested in deforming a fixed  $\overline{\rho}:\Gamma\to G(k)$ . Let  $\mathfrak{g}=\mathrm{Lie}\,G$  and  $\mathfrak{g}'=\mathrm{Lie}\,G'$ .

• Let  $f: A_1 \to A_0$  be a morphism in  $\widehat{\mathcal{C}}_{\mathcal{O}}$  and  $\rho_0: \Gamma \to G(A_0)$  a continuous homomorphism. A lift of  $\rho_0$  to  $A_1$  is a continuous homomorphism  $\rho_1: \Gamma \to G(A_1)$  such that the following diagram commutes:

$$\Gamma \xrightarrow{\rho_1} G(A_1)$$

$$\downarrow^{\rho_0} \qquad \downarrow^f$$

$$G(A_0)$$

Define the functor  $D_{\overline{\rho},\mathcal{O}}^{\square}:\widehat{\mathcal{C}}_{\mathcal{O}}\to\operatorname{Sets}$  by sending a coefficient  $\mathcal{O}$ -algebra A to the set of lifts of  $\overline{\rho}$  to A.

• With the notation as before, two lifts  $\rho$  and  $\rho'$  of  $\overline{\rho}$  to  $A_1 \in \mathcal{C}_{\mathcal{O}}$  are strictly equivalent if they are conjugate by an element of  $\widehat{G}(A_1)$ . A deformation of  $\rho_0$  to  $A_1$  is a strict equivalence class of lifts. Define the functor  $D_{\overline{\rho},\mathcal{O}}:\widehat{\mathcal{C}}_{\mathcal{O}}\to \operatorname{Sets}$  by sending a coefficient  $\mathcal{O}$ -algebra A to the set of deformations of  $\overline{\rho}$  to A.

We will drop the subscript  $\mathcal{O}$  when it is clear from context.

Fact 2.2.2.2. The functor  $D_{\overline{\rho},\mathcal{O}}^{\square}$  is representable. When  $\mathfrak{g}_k^{\Gamma} = \operatorname{Lie}(Z_G)_k$ , the functor  $D_{\overline{\rho},\mathcal{O}}$  is representable.

Remark 2.2.2.3. The first part is simple, the second is a reformulation of [Til96, Theorem 3.3].

The representing objects are denoted  $R^{\square}_{\overline{\rho},\mathcal{O}}$  and (when it exists)  $R_{\overline{\rho},\mathcal{O}}$ . While we usually care about deformations, it is technically easier to work with lifts.

**Example 2.2.2.4.** For  $G = \operatorname{GL}_n$ , a homomorphism  $\rho : \Gamma \to G(A)$  is the data of an  $A[\Gamma]$ -module which is a free A-module of rank n (on which the  $\Gamma$  action is continuous) together with a basis. Deformations forget the basis. The deformation functor is representable, for example, when  $\overline{\rho}$  is absolutely irreducible, as by Schur's lemma only scalar matrices commute with  $\overline{\rho}$ .

This deformation theory is controlled by Galois cohomology. Let  $\operatorname{ad}(\overline{\rho})$  denote the representation of  $\Gamma$  on  $\mathfrak{g}_k$  via the adjoint representation. We also consider the representation  $\operatorname{ad}^0(\overline{\rho})$  of  $\Gamma$  on  $\mathfrak{g}'_k$ . By Corollary 2.1.1.6, as p is very good  $\mathfrak{g}_k = \mathfrak{g}'_k \oplus \mathfrak{z}_{\mathfrak{g}}$ , where  $\mathfrak{z}_{\mathfrak{g}}$  is the Lie algebra of  $Z_G$ . The condition in Fact 2.2.2.2 is just that

 $H^0(\Gamma, \operatorname{ad}(\overline{\rho})) = \mathfrak{z}_{\mathfrak{g}}$ , or equivalently that  $H^0(\Gamma, \operatorname{ad}^0(\overline{\rho})) = 0$ . In general, since p is very good the natural map  $H^i(\Gamma, \operatorname{ad}^0(\overline{\rho})) \to H^i(\Gamma, \operatorname{ad}(\overline{\rho}))$  is injective for all i; we often use this without comment.

The degree-1 Galois cohomology groups are related to tangent spaces. When  $D_{\overline{\rho},\mathcal{O}}$  is representable, the tangent space to  $R_{\overline{\rho},\mathcal{O}}$  can be identified with  $D_{\overline{\rho},\mathcal{O}}(k[\epsilon]/\epsilon^2)$ . The latter makes sense even when  $D_{\overline{\rho},\mathcal{O}}$  is not representable. The tangent spaces can be analyzed by the first order exponential map [Til96, §3.5]. For a smooth  $\mathcal{O}$ -group scheme G, and a small surjection  $f: A \to A/I$  of coefficient rings  $(I \cdot \mathfrak{m}_A = 0)$ , smoothness gives an isomorphism

$$\exp: \mathfrak{g} \otimes_k I \simeq \ker(G(A) \to G(A/I)) = \ker(\widehat{G}(A) \to \widehat{G}(A/I)).$$

**Definition 2.2.2.5.** For a smooth  $\mathcal{O}$ -group scheme G, and a small surjection  $f: A \to A/I$  of coefficient rings  $(I \cdot \mathfrak{m}_A = 0)$ , the *exponential map* is the induced map

$$\exp: \mathfrak{g} \otimes_k I \to \widehat{G}(A).$$

This is functorial in the  $\mathcal{O}$ -group G.

The tangent space of  $D_{\overline{\rho},\mathcal{O}}$  is identified with  $H^1(\Gamma, \operatorname{ad}(\overline{\rho}))$ . Under this isomorphism, the cohomology class of a 1-cocycle  $\tau$  corresponds to the lift  $\rho(g) = \exp(\epsilon \tau(g))\overline{\rho}(g)$ .

**Remark 2.2.2.6.** For the framed deformation ring  $R^{\square}_{\overline{\rho},\mathcal{O}}$ , the tangent space is identified with the k-vector space  $Z^1(\Gamma, \operatorname{ad}(\overline{\rho}))$  of (continuous) 1-cocycles of  $\Gamma$  valued in  $\operatorname{ad}(\overline{\rho})$ . We also observe that

$$\dim_k Z^1(\Gamma, \operatorname{ad}(\overline{\rho})) - \dim_k H^1(\Gamma, \operatorname{ad}(\overline{\rho})) = \dim_k B^1(\Gamma, \operatorname{ad}(\overline{\rho})) = \dim_k \mathfrak{g} - \dim_k H^0(\Gamma, \operatorname{ad}(\overline{\rho}))$$

since the space of coboundaries admits a surjection from  $\operatorname{ad}(\overline{\rho})$  with kernel  $\operatorname{ad}(\overline{\rho})^{\Gamma}$ . This will be useful when comparing dimensions of framed and unframed deformation rings that are smooth.

We will want to studying special classes of deformations. We work with the category  $\mathcal{C}_{\mathcal{O}}$  of Artinian coefficient rings.

**Definition 2.2.2.7.** A lifting condition is a sub-functor  $\mathcal{D}^{\square} \subset D_{\overline{\rho},\mathcal{O}}^{\square} : \mathcal{C}_{\mathcal{O}} \to \text{Sets}$  such that:

- 1. For any coefficient ring A,  $\mathcal{D}^{\square}(A)$  is closed under strict equivalence.
- 2. Given a Cartesian diagram in  $\mathcal{C}_{\mathcal{O}}$

$$\begin{array}{ccc} A_1 \times_{A_0} A_2 \stackrel{\pi_2}{\longrightarrow} A_2 \\ \downarrow^{\pi_1} & \downarrow \\ A_1 & \longrightarrow A_0 \end{array}$$

and  $\rho \in D_{\overline{\rho},\mathcal{O}}^{\square}(A_1 \times_{A_0} A_2)$ , we have  $\rho \in \mathcal{D}^{\square}(A_1 \times_{A_0} A_2)$  if and only if  $\mathcal{D}^{\square}(\pi_1) \circ \rho \in \mathcal{D}^{\square}(A_1)$  and  $\mathcal{D}^{\square}(\pi_2) \circ \rho \in \mathcal{D}^{\square}(A_2)$ .

As it is closed under strict equivalence, we naturally obtain a deformation condition, a sub-functor  $\mathcal{D} \subset D_{\overline{\rho},\mathcal{O}}$ .

By Schlessinger's criterion [Sch68, Theorem 2.11] being a lifting condition is equivalent to the functor  $\mathcal{D}^{\square}$  being pro-representable. (This is easy to check using that  $\mathcal{D}^{\square}$  is a subfunctor of a representable functor.) Likewise, the deformation condition  $\mathcal{D}$  associated to a lifting condition  $\mathcal{D}^{\square}$  is pro-representable provided that  $D_{\overline{\rho},\mathcal{O}}$  is.

Remark 2.2.2.8. To apply Schlessinger's criterion, we use the category  $\mathcal{C}_{\mathcal{O}}$  of Artinian coefficient rings. Often the functor  $\mathcal{D}$  could equally well be defined on the larger category  $\widehat{\mathcal{C}}_{\mathcal{O}}$ . Alternately, we can try to extend the functor  $\mathcal{D}$  to  $\widehat{\mathcal{C}}_{\mathcal{O}}$  by the definition  $\mathcal{D}(A) = \varprojlim \mathcal{D}(A/\mathfrak{m}_A^n)$ . It is sometimes subtle to check that this latter definition has an "expected" concrete meaning. For example, consider the case of torsion-crystalline representations: the fact that the inverse limit of torsion crystalline representations is actually a subquotient of a lattice in a crystalline representation was a conjecture of Fontaine proved by Liu [Liu07].

The tangent space of a deformation condition  $\mathcal{D}$  is a k-subspace of  $H^1(\Gamma, \operatorname{ad}(\overline{\rho}))$ , and will be denoted by  $H^1_{\mathcal{D}}(\Gamma, \operatorname{ad}(\overline{\rho}))$ . For a small surjection  $A_1 \to A_0$  and  $\rho \in \mathcal{D}(A_0)$ , the set of deformations of  $\rho$  to  $A_1$  subject to  $\mathcal{D}$  is a  $H^1_{\mathcal{D}}(\Gamma, \operatorname{ad}(\overline{\rho}))$ -torsor. This torsor-structure is compatible with the action of the unrestricted tangent space to  $D_{\overline{\rho}}$  on the space of all deformations of  $\rho$  to  $A_1$ .

**Example 2.2.2.9.** Let G' be the derived group of G. The most basic examples of deformation conditions are the conditions imposed by fixing the lift of the homomorphism  $\Gamma \to (G/G')(k)$ . To be precise, for the quotient map  $\mu: G \to G/G' =: S$ , a fixed  $\nu: \Gamma \to S(\mathcal{O})$  lifting  $\mu \circ \overline{\rho}$ , and  $A \in \widehat{\mathcal{C}}_{\mathcal{O}}$  with structure morphism  $i: \mathcal{O} \to A$ , we define a deformation condition  $\mathcal{D}_0 \subset \mathcal{D}_{\overline{\rho}}$  by

$$\mathcal{D}_{\nu}(A) = \{ \rho \in \mathcal{D}_{\overline{\rho}}(A) | \Gamma \to G(A) : \mu_A \circ \rho = \imath \circ \nu_A \}.$$

One checks this is a deformation condition. Its tangent space is  $H^1(\Gamma, \operatorname{ad}^0(\overline{\rho}))$  since p is very good. We define  $\mathcal{D}^{\square}_{\overline{\rho}}$  similarly. We will use without comment that (G/G')(A) = G(A)/G'(A) due to Lang's theorem since k is finite and G' is smooth over  $\mathcal{O}$  with  $G'_k$  connected.

Another important easy example is the *unramified* deformation condition for any non-archimedean place v where  $\rho$  is unramified: this consists of lifts that are unramified (respectively, also with a specified choice of  $\nu$ ). The tangent space is  $H^1_{\rm nr}(\Gamma_v, {\rm ad}(\overline{\rho}))$  (respectively  $H^1_{\rm nr}(\Gamma_v, {\rm ad}^0(\overline{\rho}))$ ).

For a small surjection of coefficient  $\mathcal{O}$ -algebras  $f: A_1 \to A_0$  with kernel I, by using continuous cocycles we can define an obstruction ob  $\rho_0 \in H^2(\Gamma, \operatorname{ad}(\overline{\rho})) \otimes I$  to lifting.

**Fact 2.2.2.10.** The representation  $\rho_0$  lifts to  $A_1$  if and only if ob  $\rho_0 = 0$ . When a lift exists, the set of lifts of  $\rho_0$  is naturally an  $H^1(\Gamma, \operatorname{ad}(\overline{\rho})) \otimes I$ -torsor.

**Definition 2.2.2.11.** A deformation condition  $\mathcal{D}$  is *locally liftable* (over  $\mathcal{O}$ ) if for all small surjections  $f: A_1 \to A_0$  of coefficient  $\mathcal{O}$ -algebras the natural map

$$\mathcal{D}(f): \mathcal{D}(A_1) \to \mathcal{D}(A_0)$$

is surjective.

This holds, for example, if  $H^2(\Gamma, \operatorname{ad}(\overline{\rho})) = 0$ . A geometric way to check local liftability is to show that the corresponding deformation ring (when it exists) is smooth. Obviously it suffices to check liftability for lifts instead of deformations, so we can work with the framed deformation ring and avoid representability issues for  $\mathcal{D}_{\overline{\rho}}$ .

**Example 2.2.2.12.** The unramified deformation condition is liftable: an unramified lift is completely determined by the image of Frobenius in  $G(A_0)$ , and G is smooth over  $\mathcal{O}$ .

When attempting to lift with a fixed lift  $\nu$  of  $\Gamma \to (G/G')(k)$ , the obstruction cocycle will lie in the group  $H^2(\Gamma, \mathrm{ad}^0(\overline{\rho}))$ . To see this, recall that the obstruction cocycle is defined by picking a set theoretic lift  $\rho_1$  of a given  $\rho_0 : \Gamma_K \to G(A_0)$ : the 2-cocycle records the failure of  $\rho_1$  to be a homomorphism. By choosing the continuous set-theoretic lift  $\Gamma_K \to G(A_1)$  so that  $\Gamma_K \to (G/G')(A_0)$  agrees with  $\nu$  (as we may easily do since  $\ker \rho_0$  is open in  $\Gamma_K$ ), the obstruction cocycle clearly takes values in  $\mathrm{ad}^0(\overline{\rho})$ .

We now consider global deformation conditions. Let K be a number field, S a finite set of places of K that contains all the places of K at which  $\overline{\rho}$  are ramified and all archimedean places. As before, let  $\Gamma_S$  be the Galois group of the maximal extension of K unramified outside of S and  $\Gamma_K$  be the absolute Galois group of K.

**Definition 2.2.2.13.** A global deformation condition  $\mathcal{D}_S$  for  $\overline{\rho}: \Gamma_S \to G(k)$  is a collection of local deformation conditions  $\{\mathcal{D}_v\}_{v \in S}$  for  $\overline{\rho}|_{\Gamma_v}$ . We say it is locally liftable (over  $\mathcal{O}$ ) if each  $\mathcal{D}_v$  is locally liftable (over  $\mathcal{O}$ ). A global deformation of  $\overline{\rho}: \Gamma_S \to G(k)$  subject to  $\mathcal{D}_S$  is a deformation  $\rho: \Gamma_S \to G(A)$  such that  $\rho|_{\Gamma_v} \in \mathcal{D}_v(A)$  for all  $v \in S$ .

**Remark 2.2.2.14.** Equivalently, a global deformation condition consists of a collection of local deformation conditions for every place of K such that  $\mathcal{D}_v$  is the unramified condition for  $v \notin S$ , in which case we may work in terms of deformations of  $\overline{\rho}: \Gamma_K \to G(k)$ .

For  $v \in S$ , let  $L_v$  denote the tangent space of the local deformation condition  $\mathcal{D}_v$ . A global deformation condition gives a generalized Selmer group. We will be mainly interested in the dual Selmer group

$$H^{1}_{\mathcal{D}_{S}^{\perp}}(\Gamma_{S},\operatorname{ad}(\overline{\rho})^{*}) = \{x \in H^{1}(\Gamma_{S},\operatorname{ad}(\overline{\rho})^{*}) : \operatorname{res}_{v}(x) \in L^{\perp}_{v} \text{ for all } v \in S\}.$$

$$(2.2.2.1)$$

For Ramakrishna's method to work, it is crucial that the local tangent spaces be large enough relative to the local invariants. We say that a global deformation condition satisfies the tangent space inequality if

$$\sum_{v \in S} \dim L_v \ge \sum_{v \in S} \dim H^0(\Gamma_v, \operatorname{ad}^0(\overline{\rho})). \tag{2.2.2.2}$$

**Remark 2.2.2.15.** For later use, we define  $h_v^i := \dim_k H^i(\Gamma_v, \operatorname{ad}^0(\overline{\rho}))$  and  $h_v^{*,i} := \dim_k H^i(\Gamma_v, \operatorname{ad}^0(\overline{\rho})^*)$ .

### 2.3 Big Representations

Let  $\mathcal{O}$  be the ring of integers in a p-adic field with residue field k, and let q = #k. Consider a split connected reductive group scheme G over  $\mathcal{O}$ , with derived group G', and define  $\mathfrak{g}$  and  $\mathfrak{g}'$  to be the Lie algebras of G and G' respectively. Fix a split maximal  $\mathcal{O}$ -torus T of G. Let K be a number field and  $\chi$  denote the p-adic cyclotomic character  $\chi: \Gamma_K \to \mathbf{Z}_p^{\times}$ , with reduction  $\overline{\chi}: \Gamma_K \to \mathbf{F}_p^{\times}$ .

### 2.3.1 Big Representations

The natural class of representations  $\overline{\rho}: \Gamma_K \to G(k)$  to which Ramakrishna's method will apply are those which satisfy the following conditions:

**Definition 2.3.1.1.** A big representation  $\overline{\rho}: \Gamma_K \to G(k)$  is a continuous homomorphism such that

- (i)  $\operatorname{ad}^0(\overline{\rho})$  is an absolutely irreducible representation of  $\Gamma_K$ ;
- (ii) letting  $K(\operatorname{ad}^0(\overline{\rho}))$  (respectively  $K(\operatorname{ad}^0(\overline{\rho})^*)$ ) denote the fixed field of the kernel of the action of  $\Gamma_K$  on  $\operatorname{ad}^0(\overline{\rho})$  (respectively on  $\operatorname{ad}^0(\overline{\rho})^*$ ), we have

$$H^1(\operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/K),\operatorname{ad}^0(\overline{\rho})) = 0$$
 and  $H^1(\operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho})^*)/K),\operatorname{ad}^0(\overline{\rho})^*) = 0;$ 

(iii) there exists  $\gamma \in \Gamma_K$  such that  $\overline{\rho}(\gamma)$  is regular semisimple with associated maximal torus  $Z_{G_k}(\overline{\rho}(\gamma))^{\circ}$  equal to the split maximal torus  $T_k$ , and for which there is a unique root  $\alpha \in \Phi(G,T)$  satisfying  $\alpha(\overline{\rho}(\gamma)) = \overline{\chi}(\gamma) \neq 1$ . (If dim T = 1, we furthermore require that  $\overline{\chi}(\gamma)^3 \neq 1$ . This is used only in the proof of Lemma 2.4.2.3 in cases with G of rank 1.)

**Remark 2.3.1.2.** In (iii), note that  $\alpha(\overline{\rho}(\gamma))$  makes sense because  $\overline{\rho}(\gamma) \in T(k)$ , as any semisimple element  $g \in G(k)$  satisfies  $g \in Z_{G_k}(g)^{\circ}$ .

Remark 2.3.1.3. Note that if we extend k the representation remains big: extending the field does not change the cohomological vanishing results in the second condition, and the first and third are unaffected by the extension.

Our goal is to prove a sufficient, easy-to-check condition for a representation to be big. A version of this argument goes back to [Ram99], and is similar to the independently-worked-out arguments in [Pat15, §6]. The main assumptions which will imply bigness are that the image of  $\bar{\rho}$  contains G'(k) and the representation  $\mathrm{ad}^0(\bar{\rho})$  is absolutely irreducible. For this to work, we will need the following additional assumptions (easily checkable in practice):

- (L1) p is a very good prime for G,  $q \neq 2, 3, 4, 5, 9$ , and  $\mathbf{Q}(\zeta_p) \cap K = \mathbf{Q}$  (so  $[K(\zeta_p) : K] = p 1$ );
- (L2)  $K(\operatorname{ad}^0(\overline{\rho}))$  does not contain  $\zeta_p$  (in particular, the cyclic group  $\overline{\chi}(\Gamma_{K(\operatorname{ad}^0(\overline{\rho}))}) \subset \mathbf{F}_p^{\times}$  has order d > 1);
- (L3) there exists a regular semisimple element  $g \in [G'(k), G'(k)] \subset G'(k)$  with  $Z_{G_k}(g) = T_k$ , and a non-trivial element  $x \in \overline{\chi}(\Gamma_{K(\mathrm{ad}^0(\overline{\rho}))}) \subset \mathbf{F}_p^{\times}$  such that there is a unique root  $\alpha \in \Phi(G, T)$  for which  $\alpha(g) = x$ . (If dim T = 1, we furthermore require that  $\overline{\chi}(g)^3 \neq 1$ .)

We will show these assumptions are automatic for large enough primes when the image of  $\bar{\rho}$  contains G'(k), and hold for many smaller primes as well. First, we will show these assumption imply bigness.

**Proposition 2.3.1.4.** Under assumptions (L1), (L2), and (L3),  $\overline{\rho}$  is a big representation provided the image of  $\overline{\rho}$  contains G'(k) and  $\operatorname{ad}^0(\overline{\rho})$  is an absolutely irreducible representation of  $\Gamma_K$ .

**Remark 2.3.1.5.** Patrikis instead uses the condition  $G'(k') \subset \overline{\rho}(\Gamma_K) \subset Z(k)G(k')$  for some subfield  $k' \subset k$ . His proof uses the same method.

We now assume (L1), (L2), and (L3), that  $G'(k) \subset \overline{\rho}(\Gamma_K)$ , and that  $\operatorname{ad}^0(\overline{\rho})$  is an absolutely irreducible representation of  $\Gamma_K$ .

We will prove condition (ii) of Definition 2.3.1.1 holds when  $G'(k) \subset \overline{\rho}(\Gamma_K)$ . According to Proposition 2.1.2.4,  $H^1(G'(k), \mathfrak{g}'_k) = 0$ . We will now relate this to Galois cohomology using the inflation-restriction sequence. To do so, we need several lemmas.

**Lemma 2.3.1.6.** We have naturally  $\overline{\rho}(\Gamma_K)/(Z_G(k) \cap \overline{\rho}(\Gamma_K)) \xrightarrow{\sim} \operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/K)$ .

*Proof.* Lemma 2.1.1.6 gives a decomposition  $\mathfrak{g} = \mathfrak{g}' \oplus \mathfrak{z}_{\mathfrak{g}}$ . The kernel of  $\mathrm{Ad}_G : G \to \mathrm{GL}(\mathfrak{g})$  is  $Z_G$ , and the action is trivial on  $\mathfrak{z}_{\mathfrak{g}}$ . Thus the kernel of  $\mathrm{Ad}_G \circ \overline{\rho}$  equals  $\overline{\rho}^{-1}(Z_G(k))$ , and is also the kernel of the action of  $\Gamma_K$  on  $\mathrm{ad}^0(\overline{\rho})$ . The Galois group  $\mathrm{Gal}(K(\mathrm{ad}^0(\overline{\rho}))/K)$  is the quotient.

**Lemma 2.3.1.7.** The index of G'(k) in G(k) is prime to p.

*Proof.* We have an exact sequence

$$1 \to G'(k) \to G(k) \to (G/G')(k)$$

But G/G' is a split k-torus, so the number of its k-points is relatively prime to p.

**Lemma 2.3.1.8.** The index of  $\Gamma_{K(\mathrm{ad}^0(\overline{\rho})^*)}$  in  $\Gamma_{K(\mathrm{ad}^0(\overline{\rho}))}\Gamma_{K(\mathrm{ad}^0(\overline{\rho})^*)}$  is prime to p.

*Proof.* Elementary group theory shows that the index is equal to the index of  $\Gamma_{K(\mathrm{ad}^0(\overline{\rho}))} \cap \Gamma_{K(\mathrm{ad}^0(\overline{\rho})^*)}$  in  $\Gamma_{K(\mathrm{ad}^0(\overline{\rho}))}$ . But this subgroup is the kernel of the cyclotomic character  $\overline{\chi}:\Gamma_{K(\mathrm{ad}^0(\overline{\rho}))}\to k^{\times}$ . Thus the index is relatively prime to p.

Now consider the inflation-restriction sequence

$$0 \to H^1(\overline{\rho}(\Gamma_K)/G'(k), \mathrm{ad}^0(\overline{\rho})^{G'(k)}) \to H^1(\overline{\rho}(\Gamma_K), \mathrm{ad}^0(\overline{\rho})) \to H^1(G'(k), \mathrm{ad}^0(\overline{\rho}))$$

We know the third term is 0 by Proposition 2.1.2.4. The first term is zero as the index of G'(k) in  $\overline{\rho}(\Gamma_K)$  is prime to p because of Lemma 2.3.1.7 and  $G'(k) \subset \overline{\rho}(\Gamma_K) \subset G(k)$ . This implies that  $H^1(\overline{\rho}(\Gamma_K), \operatorname{ad}^0(\overline{\rho})) = 0$ . Using Lemma 2.3.1.6, another application of the inflation-restriction sequence gives

$$0 \to H^1(\operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/K),\operatorname{ad}^0(\overline{\rho})^{Z_G(k) \cap \overline{\rho}(\Gamma_K)}) \to H^1(\overline{\rho}(\Gamma_K),\operatorname{ad}^0(\overline{\rho}))$$

But the right term is zero and  $Z_G(k)$  acts trivially on  $\operatorname{ad}^0(\overline{\rho})$ , so we conclude that

$$H^1(\operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/K), \operatorname{ad}^0(\overline{\rho})) = 0.$$

We also claim that  $H^1(\operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho})^*)/K), \operatorname{ad}^0(\overline{\rho})^*) = 0$ . To prove this, we will use the subgroup

$$\Gamma' := \Gamma_{K(\mathrm{ad}^0(\overline{\rho}))} \cdot \Gamma_{K(\mathrm{ad}^0(\overline{\rho})^*)} / \Gamma_{K(\mathrm{ad}^0(\overline{\rho})^*)} \subset \Gamma_K / \Gamma_{K(\mathrm{ad}^0(\overline{\rho})^*)} =: \Gamma.$$

The inflation-restriction sequence begins

$$0 \to H^1(\Gamma/\Gamma', (\operatorname{ad}^0(\overline{\rho})^*)^{\Gamma'}) \to H^1(\Gamma_K/\Gamma_{K(\operatorname{ad}^0(\overline{\rho})^*)}, \operatorname{ad}^0(\overline{\rho})^*) \to H^1(\Gamma', \operatorname{ad}^0(\overline{\rho})^*).$$

The action of  $\Gamma_{K(\mathrm{ad}^0(\overline{\rho}))}$  on  $\mathrm{ad}^0(\overline{\rho})^*$  is via the cyclotomic character: by (L2) this character is non-trivial, so  $\left(\mathrm{ad}^0(\overline{\rho})^*\right)^{\Gamma'}=0$ . Thus the first term is 0. The third term is zero because  $\Gamma'$  is a finite group of order prime to p (Lemma 2.3.1.8). This gives the desired vanishing, establishing (ii) in Definition 2.3.1.1.

We finally turn to producing  $\gamma \in \Gamma_K$  as in Definition 2.3.1.1(iii). Let  $L = K(\operatorname{ad}^0(\overline{\rho})) \cap K(\zeta_p)$ , and  $K(\operatorname{ad}^0(\overline{\rho}), \zeta_p)$  be the compositum of  $K(\operatorname{ad}^0(\overline{\rho}))$  and  $K(\zeta_p)$ . By Galois theory, there is a surjection

$$\Gamma_L \to \operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}), \zeta_p)/L) \simeq \operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/L) \times \operatorname{Gal}(K(\zeta_p)/L).$$
 (2.3.1.1)

Furthermore,  $\operatorname{Gal}(K(\zeta_p)/L) \simeq \operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}), \zeta_p)/K(\operatorname{ad}^0(\overline{\rho}))) = \overline{\chi}(\Gamma_{K(\operatorname{ad}^0(\overline{\rho}))}) \subset \mathbf{F}_p^{\times}$ . We also note that  $\operatorname{Gal}(L/K)$  is an abelian quotient of  $\operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/K) \simeq \overline{\rho}(\Gamma_K)/(Z_G(k) \cap \overline{\rho}(\Gamma_K))$ . This means that

$$[\overline{\rho}(\Gamma_K)/(Z_G(k)\cap\overline{\rho}(\Gamma_K)),\overline{\rho}(\Gamma_K)/(Z_G(k)\cap\overline{\rho}(\Gamma_K))]\subset \operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/L)$$

and as  $G'(k) \subset \overline{\rho}(\Gamma_K)$  we conclude

$$[G'(k)/Z_{G'}(k), G'(k)/Z_{G'}(k)] \subset \operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/L).$$

Given  $g \in [G'(k), G'(k)]$  and  $x \in \overline{\chi}(\Gamma_{K(\mathrm{ad}^0(\overline{\rho}))})$  as in (L3), using (2.3.1.1) we see there exists  $\gamma \in \Gamma_L \subset \Gamma_K$  such that  $\overline{\chi}(\gamma) = x$  (so  $\overline{\chi}(\gamma)^3 \neq 1$  when dim T = 1) and  $\overline{\rho}(\gamma) = gz$  for some  $z \in Z_G(k) \cap \overline{\rho}(\Gamma_K)$ . We are given that there is a unique root  $\alpha \in \Phi(G, T)$  such that  $\alpha(g) = x$ . We see that  $Z_{G_k}(\overline{\rho}(\gamma)) = Z_{G_k}(g) = T_k$  and  $\alpha(\overline{\rho}(\gamma)) = \alpha(g) = x$  as desired. This completes the proof of Proposition 2.3.1.4.

### 2.3.2 Checking the Assumptions

Now assume that  $G'(k) \subset \overline{\rho}(\Gamma_K)$  and (L1) holds. We wish to understand when  $\operatorname{ad}^0(\overline{\rho})$  is an absolutely irreducible, and when (L2) and (L3) hold, in terms of the root datum  $\Phi(G,T)$ . Let  $\pi: \widetilde{G'} \to G'$  be the simply connected central cover.

**Lemma 2.3.2.1.** If  $\Phi(G,T)$  is irreducible,  $\operatorname{ad}^0(\overline{\rho})$  is an absolutely irreducible  $\Gamma_K$ -representation.

*Proof.* By assumption  $\widetilde{G'}$  is absolutely simple. As q>3, Lemma 2.1.2.1 implies that the adjoint representation of  $\widetilde{G'}(k)$  on  $(\text{Lie } G')_k$  is absolutely irreducible as p is very good for G. Again using that p is very good, Lemma 2.1.1.5 implies  $(\text{Lie } \widetilde{G'})_k \simeq \mathfrak{g}'_k$ . As  $\pi(\widetilde{G'}(k)) \subset G'(k) \subset \overline{p}(\Gamma_K)$ , we conclude that  $\operatorname{ad}^0(\overline{p})$  is an absolutely irreducible  $\Gamma_K$ -representation.

Remark 2.3.2.2. Patrikis identifies some more precise conditions than absolute irreducibility that work in Ramakrishna's method [Pat15, (1), (5), (6bc) of  $\S 5$ ]. He then checks that they hold when G is simple. We have chosen to use absolute irreducibility for convenience.

The kernel of  $f: Z_G \times \widetilde{G'} \to G$ , being a subgroup of  $Z_{\widetilde{G'}}$ , has order prime to p as p is a very good prime for G.

**Lemma 2.3.2.3.** If the image of  $\overline{\rho}$  contains G'(k) then  $\#\operatorname{Gal}(K(\zeta_p) \cap K(\operatorname{ad}^0(\overline{\rho}))/K)$  divides  $\#H^1(k, \ker f)$ . In particular, if  $H^1(k, \ker f)$  does not contain any elements of order p-1 then  $\zeta_p \notin K(\operatorname{ad}^0(\overline{\rho}))$ .

*Proof.* The final assertion follows from the rest since  $[K(\zeta_p):K]=p-1$  by (L1). We will show that the abelianization of  $\operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/K)$  is a subquotient of  $G(k)/Z_G(k)\cdot\pi(\widetilde{G'}(k))$ , and that  $G(k)/Z_G(k)\cdot\pi(\widetilde{G'}(k))$  has order dividing  $\#H^1(k, \ker f)$ .

As  $G'(k) \subset \overline{\rho}(\Gamma_K)$ , observe that

$$G'(k)/(G'(k)\cap Z_G(k))\subset \overline{\rho}(\Gamma_K)/(\overline{\rho}(\Gamma_K)\cap Z_G(k))\subset G(k)/Z_G(k).$$

By Lemma 2.3.1.6,

$$\overline{\rho}(\Gamma_K)/(\overline{\rho}(\Gamma_K) \cap Z_G(k)) \simeq \operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/K).$$

Thus we see that  $\operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/K)^{\operatorname{ab}}$  is a sub-quotient of  $G(k)/Z_G(k)[G'(k),G'(k)]$ . Now consider the map

$$f: Z_G \times \widetilde{G'} \to Z_G \times G' \to G$$

By Lemma 2.1.2.2,  $[G'(k), G'(k)] = \pi(\widetilde{G'}(k))$  as q > 3. From the long exact sequence

$$Z(k)\times \widetilde{G'}(k)\to G(k)\to H^1(k,\ker f)$$

we conclude that  $\operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/K)^{\operatorname{ab}}$  is a subquotient of  $H^1(k,\ker f)$ .

In particular, the group  $H^1(k, \ker f)$  is finite and killed by  $\# \ker(f)$ , so  $p-1 > \# \ker(f)$  is sufficient for (L2). If  $G'_k$  has root system  $\Phi$ , we know that  $\# \ker(f) | \# \pi_1(\Phi)$ , so  $p > \# \pi_1(\Phi)$  suffices. Thus (L2) holds for sufficiently large p for a group G with fixed root datum. In any particular instance, this is easy to make explicit.

**Example 2.3.2.4.** Take  $G = \mathrm{GSp}_{2n}$ . Then  $G' = \mathrm{Sp}_{2n}$  is simply connected, and  $\ker f = Z_{\mathrm{Sp}_{2n}} = \mu_2$ . Thus any prime bigger than 3 is fine.

Finally, we give a sufficient condition for (L3) to hold. It is far from necessary.

Let g be a generator of  $\mathbf{F}_p^{\times}$ , d the order of  $\overline{\chi}(\Gamma_{K(\mathrm{ad}^0(\overline{\rho}))})$ , and  $e = \frac{p-1}{d} = [K(\zeta_p) \cap K(\mathrm{ad}^0(\overline{\rho})) : K]$ . Lemma 2.3.2.3 shows that for fixed root system, e divides  $\#H^1(k, \ker f)$  which is bounded independent of p. To be precise, writing  $\ker(f) = \prod \mu_{n_i}$ , we see

$$\#H^1(k, \ker f) = \prod_i \gcd(n_i, q - 1) \mid \prod n_i = \# \ker f \mid \#\pi_1(\Phi).$$

Thus for large enough p we have that d > 4(h-1), where h is the Coxeter number. Recall that upon picking a basis  $\Delta$  of the root system, the *height* ht( $\beta$ ) of a root  $\beta$  is the the sum of the coefficients when the root is written in terms of  $\Delta$ . The Coxeter number h is 1 more than the height of the highest root.

**Lemma 2.3.2.5.** Suppose dim  $T \neq 1$ . Condition (L3) holds when d > 4(h-1).

Proof. We will work with the simply connected central cover  $\pi: \widetilde{G'} \to G'$ , with  $\widetilde{T'}$  the split maximal torus preimage of  $T \cap G' \subset G'$ . Consider the root system  $\Phi = \Phi(\widetilde{G'}, \widetilde{T'})$  with chosen basis of positive roots  $\Delta$ , and the cocharacter  $\delta \in X_*(\widetilde{T'})$  which is the sum of the positive coroots of  $\Phi$ . Take  $\beta$  to be an element of order d in  $\overline{\chi}(\Gamma_{K(\mathrm{ad}^0(\overline{\rho}))}) \subset \mathbf{F}_p^{\times} \subset k^{\times}$ , and let  $s = \delta(\beta) \in \widetilde{G'}(k)$ . Recall that for  $\alpha_i \in \Delta$ ,  $\langle \delta, \alpha_i \rangle = 2$ . Thus we calculate that for any root  $\alpha$ ,

$$\alpha(s) = \beta^{2 \operatorname{ht}(\alpha)}.$$

In particular, if  $\alpha$  is the highest root then we obtain  $\beta^{2(h-1)}$ . For any other root  $\alpha'$ ,  $0 < 2 \operatorname{ht}(\alpha') < 2(h-1)$  or  $-2(h-1) \le 2 \operatorname{ht}(\alpha') < 0$  and hence as d > 4(h-1)

$$2\operatorname{ht}(\alpha) \neq 2\operatorname{ht}(\alpha') \pmod{d}$$
 and  $2\operatorname{ht}(\alpha') \neq 0 \pmod{d}$ .

Thus  $\alpha'(s) \neq \alpha(s)$  and  $\alpha'(s) \neq 1$ . Therefore we conclude that that s is regular semisimple and  $\alpha$  is the unique root on which the adjoint action on  $(\operatorname{Lie} \widetilde{G'})_k$  is given by  $\beta^{2(h-1)} \in \overline{\chi}(\Gamma_{K(\operatorname{ad}^0(\overline{\rho}))})$ .

As p is very good,  $d\pi_k : (\operatorname{Lie} \widetilde{G'})_k \to (\operatorname{Lie} G')_k$  is an isomorphism. Thus  $\pi(s) \in \pi(\widetilde{G'}(k)) = [G'(k), G'(k)]$  (using Lemma 2.1.2.2) has the required properties.

If dim T = 1, a minor modification shows that (L3) holds as long as d > 6. In particular, for fixed root datum and sufficiently large primes p, (L3) holds.

**Remark 2.3.2.6.** The above condition is far from sharp. There is no reason that the highest root must be the unique root  $\alpha$  in (L3) where the adjoint action is given by the cyclotomic character.

**Example 2.3.2.7.** Let  $G = GL_n$ . Then the Coxeter number is n, and  $Z_{G'} = \mu_n$ . Let e = (n, p - 1), so  $H^1(k, \mu_n) = \mathbf{Z}/e\mathbf{Z}$ . Assumptions (L1)-(L3) hold provided  $p > \max(5, 4(n-1)e)$ .

For  $G = \operatorname{Sp}_{2n}$ , the Coxeter number is 2n, and  $Z_{G'} = \mu_2$ . Then  $H^1(k, \mu_2) = \mathbf{Z}/2\mathbf{Z}$  and the assumptions hold when  $p > \max(5, 8(2n-1))$ . There is a similar bound for orthogonal groups.

### 2.4 Ramakrishna's Deformation Condition

Let  $\mathcal{O}$  be the ring of integers in a p-adic field with residue field k. Consider a split connected reductive group scheme G over  $\mathcal{O}$ . Let K be a number field and  $\overline{\rho}: \Gamma_K \to G(k)$  be a big representation. Recall that  $\chi: \Gamma_K \to \mathbf{Z}_p^{\times}$  denotes the cyclotomic character.

In this section, we will generalize a deformation condition that Ramakrishna introduces for  $GL_2$ , allowing controlled ramification at certain unramified places. Allowing ramification subject to this condition is a crucial tool to reduce the size of the dual Selmer group. A similar generalization is carried out in [Pat15, §4.2].

### 2.4.1 Constructing the Deformation Condition

As  $\overline{\rho}$  is big, there is a  $\gamma \in \Gamma_K$  such that  $\overline{\rho}(\gamma)$  is regular semisimple. The identity component of  $Z_{G_k}(\overline{\rho}(\gamma))$  is  $T_k$ , where T was a specified split maximal torus of G. By hypothesis, T is split and there is a unique root  $\alpha \in \Phi(G,T)$  such that  $\alpha(\overline{\rho}(\gamma)) = \overline{\chi}(\gamma)$ . We start by assuming:

(A1) there is a place v of K lying over a rational prime  $\ell$  such that  $\overline{\rho}$  is unramified at v and  $\overline{\rho}(\operatorname{Frob}_v)$  is regular semisimple element. The identity component of  $Z_{G_k}(\overline{\rho}(\operatorname{Frob}_v))$  is  $T_k$ , and there is a unique root  $\alpha \in \Phi(G,T)$  such that  $\alpha(\overline{\rho}(\operatorname{Frob}_v)) = \overline{\chi}(\operatorname{Frob}_v) = \overline{\chi}(\gamma) \neq 1$ .

Under this assumption, we will define a deformation condition consisting of certain tamely ramified lifts. Let  $K_v^{\rm t}$  be the maximal tamely ramified extension of  $K_v$ ,  $K_v^{\rm nr}$  be the maximal unramified extension, and  $\Gamma_v^{\rm t} = {\rm Gal}(K_v^{\rm t}/K_v)$ . There is a split exact sequence

$$1 \to \operatorname{Gal}(K_v^{\operatorname{t}}/K_v^{\operatorname{nr}}) \to \Gamma_v^{\operatorname{t}} \to \operatorname{Gal}(K_v^{\operatorname{nr}}/K_v) \to 1.$$

Recall that  $\operatorname{Gal}(K_v^{\operatorname{nr}}/K_v)$  is topologically generated by  $\operatorname{Frob}_v$ , while  $\operatorname{Gal}(K_v^{\operatorname{tr}}/K_v^{\operatorname{nr}}) \simeq \prod_{p' \neq \ell} \mathbf{Z}_{p'}(1)$ . The action of  $\operatorname{Gal}(K_v^{\operatorname{nr}}/K_v)$  on  $\operatorname{Gal}(K_v^{\operatorname{tr}}/K_v^{\operatorname{nr}})$  is via the prime-to- $\ell$  cyclotomic character. Concretely, the action of  $\operatorname{Frob}_v$  is given by multiplication by an integer  $q = \ell^{f(K_v/\mathbf{Q}_\ell)}$ . For a fixed splitting, we obtain a semidirect product decomposition

$$\Gamma_v^{\rm t} \simeq \operatorname{Gal}(K_v^{\rm nr}/K_v) \ltimes \operatorname{Gal}(K_v^{\rm t}/K_v^{\rm nr}).$$
 (2.4.1.1)

Fix a split maximal  $\mathcal{O}$ -torus T of G reducing to  $T_k$ : we identify  $\alpha \in \Phi(G,T) = \Phi(G_k,T_k)$  and form the usual smooth closed  $\mathcal{O}$ -subgroups  $U_{\alpha} \subset G$  (root group for  $\alpha$ ) and  $H_{\alpha} = T \ltimes U_{\alpha} \subset G$  associated to  $\alpha$  and to  $\{0,\alpha\}$  (see Theorem 5.1.13 and Proposition 5.1.16 of [Con14]).

**Definition 2.4.1.1.** We assume (A1). For a coefficient  $\mathcal{O}$ -algebra A, consider a lift  $\rho: \Gamma_v^t \to G(A)$ . The lift  $\rho$  satisfies Ramakrishna's condition relative to T provided that  $\rho(\operatorname{Frob}_v) \in T(A)$ ,  $\alpha(\rho(\operatorname{Frob}_v)) = \chi(\operatorname{Frob}_v)$ , and  $\rho(\operatorname{Gal}(K_v^t/K_v^{\operatorname{nr}})) \subset U_{\alpha}(A) \subset G(A)$ .

Define Ramakrishna's deformation condition  $\mathcal{D}_v^{\text{ram}}(A)$  to be lifts which are  $\widehat{G}(A)$ -conjugate to one which satisfies Ramakrishna's condition relative to T. Denote the tangent space of the deformation functor by  $L_v^{\text{ram}}$ .

This condition generalizes the condition for  $GL_2$  introduced by Ramakrishna [Ram99, §3]. Note that the condition can be rephrased as  $\rho \in \mathcal{D}_v^{\text{ram}}(A)$  if and only if there exists a choice of split  $\mathcal{O}$ -torus T lifting  $T_k$  such that  $\rho$  satisfies Ramakrishna's condition relative to T since  $\widehat{G}(\mathcal{O})$  acts transitively on the set of such T by [Con14, Theorem 3.2.6].

**Example 2.4.1.2.** Let  $G = GL_n$ . We may assume that the residual representation, which is unramified, sends Frob<sub>v</sub> to a diagonal element. Ramakrishna's deformation condition consists of lifts such that after conjugation  $\rho(\text{Frob}_v)$  is diagonal and the the image of  $\text{Gal}(K_v^t/K_v^{\text{nr}})$  consists of elements of the shape

$$\begin{pmatrix} 1 & * & 0 & \dots & 0 \\ 0 & 1 & 0 & \dots & 0 \\ 0 & 0 & 1 & \dots & 0 \\ & & & \dots & & \\ 0 & 0 & 0 & \dots & 1 \end{pmatrix}$$

Conjugation by Frob<sub>v</sub> is multiplication by  $q = \chi(\text{Frob}_v)$  on  $\text{Gal}(K_v^{\text{t}}/K_v^{\text{nr}})$ . On the other hand, the action of  $\rho(\text{Frob}_v)$  on  $U_{\alpha}$  is given by  $\alpha(\rho(\text{Frob}_v))$ . Thus the assumption that  $\alpha(\overline{\rho}(\text{Frob}_v)) = \overline{\chi}(\text{Frob}_v)$  is crucial to allow the component \* to be non-zero and produce lifts where  $\rho(\text{Gal}(K_v^{\text{t}}/K_v^{\text{nr}})) \neq 1$ .

**Lemma 2.4.1.3.**  $\mathcal{D}_{v}^{\text{ram}}$  is a deformation condition.

Proof.  $\mathcal{D}_v^{\mathrm{ram}}(A)$  is certainly closed under strict equivalence. Next, we will check it is subfunctor of the deformation functor at  $\overline{\rho}$ . Let  $f: A_1 \to A_0$  be a morphism of coefficient  $\mathcal{O}$ -algebras, and choose  $\rho_1 \in \mathcal{D}_v^{\mathrm{ram}}(A_1)$ . This means there is a split maximal torus T defined over  $\mathcal{O}$  such that  $\rho(\mathrm{Frob}_v) \in T(A_1)$ ,  $\alpha(\rho(\mathrm{Frob}_v)) = \chi(\mathrm{Frob}_v)$ , and  $\rho(\mathrm{Gal}(K_v^{\mathrm{t}}/K_v^{\mathrm{nr}})) \subset U_\alpha(A_1)$ . The same torus shows that  $f(\rho) \in \mathcal{D}_v^{\mathrm{ram}}(A_0)$ .

It remains to check condition (2) of Definition 2.2.2.7. Let T be a fixed split  $\mathcal{O}$ -torus lifting  $T_k$ , and let  $A_1, A_2$ , and  $A_0$  be coefficient  $\mathcal{O}$ -algebras with morphisms  $f_1:A_1\to A_0$  and  $f_2:A_2\to A_0$ . We can reduce to the case that  $f_2$  is small. It suffices to show that given deformations  $\rho_i\in\mathcal{D}^\square_{\overline{\rho}}(A_i)$  such that  $\mathcal{D}^\square_{\overline{\rho}}(f_1)(\rho_1)=\mathcal{D}^\square_{\overline{\rho}}(f_2)(\rho_2)=\rho_0$ , if  $\rho_1$  and  $\rho_2$  satisfy Ramakrishna's lifting condition then so does  $\rho_1\times_{\rho_0}\rho_2$ . This means that there are  $g_i\in\widehat{G}(A_i)$  such that  $\rho_i^{g_i}$  satisfies Ramakrishna's condition relative to T. In particular, the push-forwards of  $\rho_1^{g_1}$  and  $\rho_2^{g_2}$  to  $A_0$  both satisfy Ramakrishna's condition relative to T, and letting  $h=g_1g_2^{-1}$  both  $\rho_0^{g_2}$  and  $\rho_0^{hg_2}=(\rho_0^{g_2})^h$  satisfy Ramakrishna's condition relative to T. We will show that this implies  $h\in\widehat{H}_\alpha(A_0)$ . Using the smoothness of this group, we can lift to an element  $h'\in\widehat{H}_\alpha(A_2)$ . But then  $(g_1,hg_2)\in\widehat{G}(A_1)\times_{\widehat{G}(A_0)}\widehat{G}(A_0)=\widehat{G}(A_1\times_{A_0}A_2)$ , and this element conjugates  $\rho_1\times_{\rho_0}\rho_2$  to satisfy Ramakrishna's condition relative to T.

It remains to prove the following statement: if  $\rho: \Gamma_v \to G(R)$  satisfies Ramakrishna's condition relative to T, then

$$C_{\rho}(R) := \{ g \in \widehat{G}(R) : \rho^g(\Gamma_v) \subset H_{\alpha}(R) \} \subset \widehat{H}_{\alpha}(R).$$

If R = k, this is trivial. In general, argue by induction on the length of R. It suffices to consider a small morphism  $R \to R/I$  and assume the statement for R/I. For  $g \in C_{\rho}(R)$ , induction shows that  $\overline{g} \in \widehat{H}_{\alpha}(R/I)$  hence it suffices to check that  $\exp(x) \in \widehat{H}_{\alpha}(R)$  for all  $x \in \mathfrak{g} \otimes_k I$  for which  $\rho^{\exp(x)}(\Gamma_v) \subset H_{\alpha}(R)$ . Rewriting, we see that

$$\exp(x)\rho(\operatorname{Frob}_v)\exp(-x) \in H_\alpha(R).$$

Multiplying on the right by  $\rho(\text{Frob}_v)^{-1} \in H_\alpha(R)$ , we conclude that

$$x - \mathrm{Ad}_G(\overline{\rho}(\mathrm{Frob}_v))x \in (\mathrm{Lie}\,H_\alpha)_k \otimes_k I.$$

As  $\overline{\rho}(\operatorname{Frob}_v)$  is regular semisimple, it acts non-trivially on  $\mathfrak{g}_{\beta}$  for all  $\beta \in \Phi(G,T)$ , forcing  $x \in (\operatorname{Lie} H_{\alpha})_k \otimes_k I$  as desired.

Now we will describe the tangent space to the deformation condition. Fix a choice of  $\mathcal{O}$ -torus T lifting  $T_k$ , and let  $\mathfrak{g} := \text{Lie } G_k$  and  $\mathfrak{t} = \text{Lie } T_k$ . We have a decomposition

$$\operatorname{ad}(\overline{\rho}) = \mathfrak{t} \oplus \bigoplus_{\beta \in \Phi(G,T)} \mathfrak{g}_{\beta}. \tag{2.4.1.2}$$

Because  $\operatorname{ad}(\overline{\rho})$  is unramified at v, the Galois action is completely determined by the action of  $\operatorname{Frob}_v$ , which acts on  $\mathfrak{g}_{\beta}$  as  $\beta(\overline{\rho}(\operatorname{Frob}_v))$ . This will allow us to compute Galois cohomology at v easily:

**Lemma 2.4.1.4.** We have dim  $H^0(\Gamma_v, \operatorname{ad}(\overline{\rho})) = \dim \mathfrak{t}$  and dim  $H^1(\Gamma_v, \operatorname{ad}(\overline{\rho})) = \dim \mathfrak{t} + 1$ . Furthermore,

$$H^1(\Gamma_v, \operatorname{ad}(\overline{\rho})) = H^1(\Gamma_v, \mathfrak{t}) \oplus H^1(\Gamma_v, \mathfrak{g}_{\alpha}).$$

Proof. It is straightforward to see that  $\dim H^0(\Gamma_v, \operatorname{ad}(\overline{\rho})) = \dim H^0(\Gamma_v, \mathfrak{t}) = \dim \mathfrak{t}$  using (2.4.1.2): since  $\overline{\rho}(\operatorname{Frob}_v)$  is regular semisimple, its space of fixed vectors in  $\mathfrak{g}$  is  $\mathfrak{t}$ . We claim that  $\dim_k H^2(\Gamma_v, \operatorname{ad}(\overline{\rho})) = \dim H^2(\Gamma_v, \mathfrak{g}_{\alpha}) = 1$ . This follows from Fact 2.2.1.1 as the only piece of the decomposition on which  $\operatorname{Frob}_v$  acts as the cyclotomic character is  $\mathfrak{g}_{\alpha}$ . Then Fact 2.2.1.2 implies that  $\dim_k H^1(\Gamma_v, \operatorname{ad}(\overline{\rho})) = \dim \mathfrak{t} + 1$  and gives the decomposition of  $H^1(\Gamma_v, \operatorname{ad}(\overline{\rho}))$ .

There are also some obvious deformations to  $k[\epsilon]/\epsilon^2$  of the form  $\rho_i = \exp(\epsilon f_i)\overline{\rho}$  where  $f_i \in Z^1(\Gamma_v, \operatorname{ad}(\overline{\rho}))$  is defined as follows. Choose any non-zero homomorphism  $f_0' : \operatorname{Gal}(K_v^{\operatorname{t}}/K_v^{\operatorname{nr}}) \to \mathbf{Z}/p\mathbf{Z} \to \mathfrak{g}_{\alpha}$ , and define  $f_0 : \Gamma_v \to \mathfrak{g}_{\alpha}$  via the composition

$$\Gamma_v \to \Gamma_v^{\mathrm{t}} \stackrel{\pi}{\to} \mathrm{Gal}(K_v^{\mathrm{t}}/K_v^{\mathrm{nr}}) \stackrel{f_0'}{\to} \mathfrak{g}_{\alpha}$$

where  $\pi$  is the projection determined by the splitting (2.4.1.1). It is straightforward to check this is a cocycle using that  $\alpha(\overline{\rho}(\operatorname{Frob}_v)) = \overline{\chi}(\operatorname{Frob}_v)$ . Now choose a basis  $t_1, \ldots, t_{\dim \mathfrak{t}}$  for  $\mathfrak{t}$  such that  $t_1, \ldots, t_{\dim \mathfrak{t}-1} \in \mathfrak{t}_{\alpha}$ , where  $\mathfrak{t}_{\alpha}$  is the Lie algebra of the codimension-1 subtorus  $\ker(\alpha)_{\mathrm{red}}^0$ . Define an unramified homomorphism  $f_i: \Gamma_v \to \Gamma_v^{\mathrm{nr}} \to \mathfrak{t}$  by sending  $\operatorname{Frob}_v$  to  $t_i$ : it is clearly a cocycle since  $\overline{\rho}(\operatorname{Frob}_v) \in T(k)$ .

**Lemma 2.4.1.5.** We have dim  $L_v^{\text{ram}} = \dim \mathfrak{t} = \dim H^0(\Gamma_v, \operatorname{ad}(\overline{\rho}))$ . The cocycles  $f_0, f_1, \ldots, f_{\dim \mathfrak{t}-1}$  form a basis for  $L_v^{\text{ram}} \subset H^1(\Gamma_v, \operatorname{ad}(\overline{\rho}))$ .

*Proof.* First, we check that the cocycles  $f_0, f_1, \ldots, f_{\dim \mathfrak{t}}$  form a basis for  $H^1(\Gamma_v, \operatorname{ad}(\overline{\rho}))$ : by Lemma 2.4.1.4, it suffices to check they are linearly independent (modulo coboundaries). Note that for any 1-coboundary

$$c_x: g \mapsto \operatorname{ad}(\overline{\rho})(g)(x) - x$$

with  $x \in \mathfrak{g}$ ,  $c(\text{Frob}_v)$  has vanishing t-component. This immediately gives that  $f_1, \ldots f_{\dim \mathfrak{t}}$  are linearly independent modulo coboundaries. Furthermore, by inspection  $f_0$  is a nonzero element of  $H^1(\Gamma_v, \mathfrak{g}_\alpha)$ . In light of the decomposition from Lemma 2.4.1.4, we obtain linear independence of  $f_0, f_1, \ldots, f_{\dim \mathfrak{t}}$ .

Finally, we consider which of the lifts  $\rho_i = \exp(f_i \epsilon) \overline{\rho}$  satisfy the deformation condition  $\mathcal{D}_v^{\mathrm{ram}}$ . The requirement is that  $\alpha(\rho_i(\mathrm{Frob}_v)) = \overline{\chi}(\mathrm{Frob}_v) \in k[\epsilon]/(\epsilon^2)$ . Now  $\mathrm{ad}(\overline{\rho})(\mathrm{Frob}_v)$  acts on  $\mathfrak{g}_{\alpha}$  via multiplication by  $\overline{\chi}(\mathrm{Frob}_v)$  by (A1). Furthermore,  $\mathrm{ad}(\exp(f_i(\mathrm{Frob}_v)\epsilon)) = [f_i(\mathrm{Frob}_v), -]$  on  $\mathfrak{g}$ , and on  $\mathfrak{g}_{\alpha}$  the action is multiplication by  $\mathrm{Lie}(\alpha)(f_i(\mathrm{Frob}_v))$ . So  $\rho_i$  satisfies  $\mathcal{D}_v^{\mathrm{ram}}$  if and only if  $f_i(\mathrm{Frob}_v) \in \mathfrak{t}_{\alpha}$ . We see that  $f_0, f_1, \ldots, f_{\dim \mathfrak{t}-1}$  satisfy this requirement but  $f_{\dim \mathfrak{t}}$  does not.

**Lemma 2.4.1.6.**  $\mathcal{D}_v^{\text{ram}}$  is liftable.

*Proof.* To see it is liftable, consider trying to lift  $\rho: \Gamma_v^t \to G(A/I)$  to A, where  $A \to A/I$  is a small extension. There is a split maximal torus T of G over  $\mathcal{O}$  such that  $\rho$  factors through  $T(A/I) \ltimes U_{\alpha}(A/I)$ . Write

$$\rho(g) = t(g) \ltimes u_{\alpha}(x(g))$$

where  $t \in \operatorname{Hom}(\Gamma_v^t, T(A/I))$ ,  $u_\alpha : \mathbf{G}_a \xrightarrow{\sim} U_\alpha$  and  $x \in Z^1(\Gamma_v^t, \chi_{A/I})$  because  $\alpha(\operatorname{Frob}_v) = \chi_{A/I}(\operatorname{Frob}_v)$ . (Note that the Galois action on  $U_\alpha(R/I) = \mathbf{G}_a(R/I) = R/I$  is given by multiplication against the cyclotomic character  $\chi_{A/I}$ .) To lift  $\rho$ , it suffices to continuously lift t and x as a homomorphism and tame 1-cocycle respectively, as then the combined lift will determine a lift of  $\rho$ .

We can easily lift t: it is unramified, so just lift an image of  $\operatorname{Frob}_v$  using the smoothness of the torus. To lift x, we claim that  $H^1(\Gamma_v^t,\chi_A) \to H^1(\Gamma_v^t,\chi_{A/I})$  is surjective. To check this, it suffices to check that the next piece of the long exact sequence,  $H^2(\Gamma_v^t,\chi_I) \to H^2(\Gamma_v^t,\chi_A)$ , is injective. Since  $v \nmid p$ ,

$$H^2(\Gamma_v^t, \chi_I) = H^2(\Gamma_v, \chi_I)$$
 and  $H^2(\Gamma_v^t, \chi_A) = H^2(\Gamma_v, \chi_A)$ .

Using local duality, the claim about injectivity reduces to the evident surjectivity of  $\operatorname{Hom}(A, \mathbf{Q}/\mathbf{Z}) \to \operatorname{Hom}(I, \mathbf{Q}/\mathbf{Z})$ . Thus there is a cocycle  $x' \in Z^1(\Gamma_v^t, \chi_A)$  which reduces to x up to a (tame) 1-coboundary. But a tame coboundary valued in the unramified  $\chi_A$  obviously lifts. Therefore we can lift both t and x, hence we can produce a homomorphism  $\rho' : \Gamma_v^t \to T(A) \ltimes U_\alpha(A)$  lifting  $\rho$ .

Putting this all together, we obtain:

**Proposition 2.4.1.7.** The deformation condition  $\mathcal{D}_v^{\mathrm{ram}}$  is liftable. The dimension of its tangent space  $L_v^{\mathrm{ram}}$  equals  $\dim H^0(\Gamma_v, \mathrm{ad}(\overline{\rho}))$ . Any  $f \in L_v^{\mathrm{ram}} \subset H^1(\Gamma_v, \mathrm{ad}(\overline{\rho}))$  satisfies  $f(\mathrm{Frob}_v) \in \mathfrak{t}_{\alpha}$ .

*Proof.* Combine Lemma 2.4.1.4, 2.4.1.5, and 2.4.1.6. (The last assertion needs the additional fact, noted in the proof of Lemma 2.4.1.5, that 1-coboundaries have vanishing  $\mathfrak{t}_{\alpha}$  component.)

Denote the derived group of G by G', and let S = G/G' with quotient map  $\mu : G \to S$ .. Suppose we are given a lift  $\nu$  of  $\mu \circ \overline{\rho} : \Gamma_v \to S(k)$ . Let  $\mathcal{D}_v^{\operatorname{ram},\nu}$  denote the sub-functor of  $\mathcal{D}_v^{\operatorname{ram}}$  consisting of lifts  $\rho : \Gamma_{K_v} \to G(A)$  such that  $\nu_A = \mu \circ \rho$ . This is a deformation condition (representable), as the condition  $\mu \circ \rho = \nu$  cuts out a closed subscheme of the universal lifting ring for  $\mathcal{D}_v^{\operatorname{ram}}$ .

Corollary 2.4.1.8. Suppose that  $\nu$  is unramified. Then  $\mathcal{D}_v^{\mathrm{ram},\nu}$  is liftable, and its tangent space has dimension  $\dim H^0(\Gamma_v, \mathrm{ad}^0(\overline{\rho}))$ .

*Proof.* By construction, all the deformations in  $\mathcal{D}_{v}^{\text{ram}}$  are tame. For a cocycle

$$f \in Z^1(\Gamma_v^{\mathrm{t}}, \mathrm{ad}(\overline{\rho})) \subset Z^1(\Gamma_v, \mathrm{ad}(\overline{\rho}))$$

remember that the associated tame lift is given by  $\rho(g) = \exp_G(f(g)\epsilon)\overline{\rho}(g)$  for  $g \in \Gamma_v^t$ . By functoriality of exp, we see that

$$\mu \circ \rho(g) = \exp_S((d\mu \circ f)(g)\epsilon) (\mu \circ \overline{\rho}) (g).$$

But  $\mu \circ \overline{\rho} = \nu$ , so  $\nu = \mu \circ \rho$  provided

$$\exp(d\mu(f(g))\epsilon) = 1$$

for all  $g \in \Gamma_v^t$ . In other words, f must factor through  $\ker d\mu = \operatorname{Lie} G' = \operatorname{ad}^0(\overline{\rho})$ . As the characteristic is very good for G, the decomposition  $\operatorname{ad}(\bar{\rho}) = \operatorname{ad}^0(\bar{\rho}) \oplus \mathfrak{z}_{\mathfrak{g}}$  gives a projection  $\pi_{\mathfrak{z}_{\mathfrak{g}}} : \mathfrak{g} \to \mathfrak{z}_{\mathfrak{g}}$ . The lift  $\rho$  satisfies  $\nu = \mu \circ \rho$  if and only if  $\pi_{\mathfrak{z}_{\mathfrak{g}}} \circ f = 0$ . A basis for  $H^1(\Gamma_v, \mathfrak{t}) \cap L_v^{\mathrm{ram}}$  is given by  $f_1, \ldots f_{\dim t - 1}$ , all of which are unramified 1-cocycles. All the coboundaries are unramified, so by considering the composition of projections

$$H^1(\Gamma_v^{\mathrm{t}}, \mathrm{ad}(\overline{\rho})) \to H^1(\Gamma_v^{\mathrm{t}}, \mathfrak{t}) = H^1(\Gamma_v^{\mathrm{nr}}, \mathfrak{t}) \to H^1(\Gamma_v^{\mathrm{t}}, \mathfrak{z}_{\mathfrak{g}})$$

we see it suffices to check that  $\pi_{\mathfrak{z}_{\mathfrak{g}}} \circ f|_{\Gamma_v^{\text{nr}}} = 0$  (using the splitting (2.4.1.1)). Lemma 2.4.1.5 gives an explicit description of  $L_v^{\text{ram}}$ . The cocycle  $f_0$  certainly satisfies  $\pi_{\mathfrak{z}_{\mathfrak{g}}} \circ f = 0$  as its image lies in  $\mathfrak{g}_{\alpha}$ . The cocycles  $f_1, \ldots, f_{\dim t-1}$  give a basis for  $\operatorname{Hom}(\Gamma_v^{\operatorname{nr}}, \mathfrak{t}_{\alpha})$ . As  $\mathfrak{z}_{\mathfrak{g}}$  is a direct factor of  $\mathfrak{t}_{\alpha}$ , we see a codimension-dim  $\mathfrak{z}_{\mathfrak{g}}$  subspace of  $L_v^{\operatorname{ram}}$  satisfies the deformation condition by projecting to 0 in  $\mathfrak{z}_{\mathfrak{g}}$ . We conclude the tangent space has dimension  $h^0(\Gamma_v, \operatorname{ad}(\overline{\rho})) - \dim \mathfrak{z}_{\mathfrak{g}} = h^0(\Gamma_v, \operatorname{ad}^0(\overline{\rho})).$ 

Now consider a small surjection  $A_1 \to A_0$  with  $A_0 = A_1/I$  (so  $\mathfrak{m}_{A_1}I = 0$ ), and a lift  $\rho_0 \in \mathcal{D}_v^{\mathrm{ram},\nu,\square}(A_0)$ . To check liftability, we may factor into a sequence of small surjections where I is a 1-dimensional k-vector space, so it suffices to treat that case. By Lemma 2.4.1.6, there exists a lift  $\rho_1 \in \mathcal{D}_v^{\mathrm{ram},\square}(A_1)$ , and which may be expressed as

$$\rho_1(g) = \exp_G(h(g) \otimes i)\rho_0(g)$$

for some  $i \in I \simeq k$  and  $h \in Z^1(\Gamma_v^t, \operatorname{ad}(\overline{\rho}))$ . (The cocycle h factors through the tame quotient as all lifts in  $\mathcal{D}_v^{\mathrm{ram}}$  are tamely ramified.) We will modify h so that  $\pi_{\mathfrak{z}_{\mathfrak{g}}} \circ h = 0$ . To do so, pick a basis  $t_1, \ldots t_r$  of  $\mathfrak{z}_{\mathfrak{g}}$ , and as in the proof of Lemma 2.4.1.5 consider unramified cocycles  $c_i: \Gamma_v \to \Gamma_v^{nr} \to \mathrm{ad}(\overline{\rho})$  defined by

$$c_j(\operatorname{Frob}_v) = t_j.$$

Let  $a_j \in k$  be such that  $\pi_{\mathfrak{z}_{\mathfrak{g}}} h(\text{Frob}_v) = \sum_j a_j t_j$ . Then because we already know  $\mu \circ \rho_0 = \nu$  on  $A_0$ -points, we

$$\exp\left(-\sum_{j} a_{j} c_{j}(g) \otimes i\right) \exp(h(g) \otimes i) \rho_{0}(g) = \exp\left(\left(h(g) - \sum_{j} a_{j} c_{j}(g)\right) \otimes i\right) \rho_{0}(g) \in \mathcal{D}_{v}^{\operatorname{ram}, \nu, \square}(A_{1})$$

because we have remarked it suffices to check  $\pi_{\mathfrak{z}_{\mathfrak{g}}}\left(h(g)-\sum_{j}a_{j}c_{j}(g)\right)=0$  for  $g\in\Gamma_{v}^{\mathrm{nr}}$ . Thus we may modify  $\rho_1$  so that it is an element of  $\mathcal{D}_v^{\mathrm{ram},\nu,\square}(A_1)$ , and hence  $\mathcal{D}_v^{\mathrm{ram},\nu}$  is liftable. 

#### Shrinking the Dual Selmer Group

Finally, we will show we can find places of K where (A1) holds, as well as some additional cohomological conditions that will allow us to shrink the dual Selmer group. Let  $\overline{\rho}: \Gamma_K \to G(k)$  be a big representation, and S a finite set of places containing those above p, the archimedean places, and all places where  $\bar{p}$  is ramified. Let  $\mathcal{D}_S$  be a global deformation condition satisfying the tangent space inequality (2.2.2.2) and for which  $H^1_{\mathcal{D}^{\frac{1}{\alpha}}}(\Gamma_S, \operatorname{ad}^0(\overline{\rho})^*)$  is non-zero.

**Lemma 2.4.2.1.** Under this assumption, the group  $H^1_{\mathcal{D}_S}(\Gamma_K, \operatorname{ad}^0(\overline{\rho}))$  is non-zero..

*Proof.* Let  $L_v$  be the local deformation conditions for  $v \in S$ . The tangent space inequality (2.2.2.2) states

$$\sum_{v \in S} \dim L_v \ge \sum_{v \in S} \dim H^0(\Gamma_v, \operatorname{ad}^0(\overline{\rho})).$$

As  $\overline{\rho}$  is big,  $H^0(K, \operatorname{ad}^0(\overline{\rho}))$  and  $H^0(K, \operatorname{ad}^0(\overline{\rho})^*)$  are trivial. Then the product formula (2.2.1.1) gives that

$$\dim H^1_{\mathcal{D}_S}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})) - \dim H^1_{\mathcal{D}_S^{\perp}}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})^*) = \sum_{v \in S} \left(\dim L_v - \dim H^0(\Gamma_v, \mathrm{ad}^0(\overline{\rho})\right) \ge 0$$

As we have assumed  $H^1_{\mathcal{D}^{\perp}_{S}}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})^*)$  is non-zero, it follows that  $H^1_{\mathcal{D}_S}(\Gamma_K, \operatorname{ad}^0(\overline{\rho}))$  is non-zero as well.  $\square$ 

Pick any non-zero element  $\psi \in H^1_{\mathcal{D}_S^+}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})^*)$ , and pick a non-zero  $\xi \in H^1_{\mathcal{D}_S}(\Gamma_K, \operatorname{ad}^0(\overline{\rho}))$  using the Lemma. Now let F be the smallest field over K for which  $\Gamma_F$  is killed by  $\overline{\chi}$  and  $\operatorname{ad}^0(\overline{\rho})$ ; note that F/K is finite Galois. The absolute Galois group  $\Gamma_F$  acts trivially on  $\operatorname{ad}^0(\overline{\rho})$  and  $\operatorname{ad}^0(\overline{\rho})^*$ , so the respective restrictions of  $\xi$  and  $\psi$  to  $H^1(\Gamma_F, \operatorname{ad}^0(\overline{\rho}))$  and  $H^1(\Gamma_F, \operatorname{ad}^0(\overline{\rho})^*)$  are just homomorphisms, which we will respectively denote by  $\xi'$  and  $\psi'$ . Let  $F_{\xi'}$  and  $F_{\psi'}$  be the fixed fields of the kernel of  $\xi'$  and  $\psi'$  respectively.

**Lemma 2.4.2.2.** The elements  $\xi'$  and  $\psi'$  lie in  $H^1(\Gamma_F, \operatorname{ad}^0(\overline{\rho}))^{\operatorname{Gal}(F/K)}$  and  $H^1(\Gamma_F, \operatorname{ad}^0(\overline{\rho}))^{\operatorname{Gal}(F/K)}$  respectively, and are non-zero. Furthermore  $F_{\xi'}$  and  $F_{\psi'}$  are Galois over K.

*Proof.* By the bigness assumption,  $H^1(K(\mathrm{ad}^0(\overline{\rho}))/K,\mathrm{ad}^0(\overline{\rho})) = 0$ . The extension  $F/K(\mathrm{ad}^0(\overline{\rho}))$  is Galois of degree prime to p by Lemma 2.3.1.8, so the inflation-restriction sequence gives

$$0 \to H^1(\operatorname{Gal}(K(\operatorname{ad}^0(\overline{\rho}))/K),\operatorname{ad}^0(\overline{\rho})) \to H^1(\operatorname{Gal}(F/K),\operatorname{ad}^0(\overline{\rho})) \to H^1(\operatorname{Gal}(F/K(\operatorname{ad}^0(\overline{\rho})),\operatorname{ad}^0(\overline{\rho})) = 0$$

Therefore  $H^1(\operatorname{Gal}(F/K), \operatorname{ad}^0(\overline{\rho})) = 0$ . Another application of the inflation-restriction sequence gives

$$0 \to H^1(\operatorname{Gal}(F/K), \operatorname{ad}^0(\overline{\rho})) \to H^1(\Gamma_K, \operatorname{ad}^0(\overline{\rho})) \to H^1(\Gamma_F, \operatorname{ad}^0(\overline{\rho}))^{\operatorname{Gal}(F/K)}$$

The first term is zero, so image  $\xi' \in H^1(\Gamma_F, \operatorname{ad}^0(\overline{\rho}))^{\operatorname{Gal}(F/K)}$  is non-zero. The same argument applies to the field  $K(\operatorname{ad}^0(\overline{\rho})^*)$  and the module  $\operatorname{ad}^0(\overline{\rho})^*$ , showing that  $\psi' \in H^1(\Gamma_F, \operatorname{ad}^0(\overline{\rho}))^{\operatorname{Gal}(F/K)}$  is non-zero.

Finally, oberserve that  $F_{\xi'}$  and  $F_{\psi'}$  are Galois over K as the natural  $\Gamma_K$ -actions on  $H^1(\Gamma_F, \operatorname{ad}^0(\overline{\rho}))$  and  $H^1(\Gamma_F, \operatorname{ad}^0(\overline{\rho})^*)$  leave  $\xi'$  and  $\psi'$  invariant.

We will also need the following Lemma about the  $k[\Gamma_K]$ -modules  $\mathrm{ad}^0(\overline{\rho})$  and  $\mathrm{ad}^0(\overline{\rho})^*$ .

**Lemma 2.4.2.3.** There exist simple  $\mathbf{F}_p[\Gamma_K]$ -modules W and W' such that  $\mathrm{ad}^0(\overline{\rho}) = W^{\oplus r}$  and  $\mathrm{ad}^0(\overline{\rho})^* = (W')^{\oplus r'}$  as  $\mathbf{F}_p[\Gamma_K]$ -modules. Furthermore, W and W' are not isomorphic.

Proof. Let W and W' be simple  $\mathbf{F}_p[\Gamma_K]$ -submodules of  $\mathrm{ad}^0(\overline{\rho})$  and  $\mathrm{ad}^0(\overline{\rho})^*$ . The maps  $k \otimes_{\mathbf{F}_p} W \to \mathrm{ad}^0(\overline{\rho})$  and  $k \otimes_{\mathbf{F}_p} W' \to \mathrm{ad}^0(\overline{\rho})^*$  are surjective as  $\mathrm{ad}^0(\overline{\rho})$  and  $\mathrm{ad}^0(\overline{\rho})^*$  are irreducible over k. Furthermore the sources are direct sums of copies of W and W' respectively (as  $\mathbf{F}_p[\Gamma_K]$ -modules). Thus there exist r and r' for which  $\mathrm{ad}^0(\overline{\rho}) = W^{\oplus r}$  and  $\mathrm{ad}^0(\overline{\rho})^* = (W')^{\oplus r'}$ . We will consider the eigenspaces of the action of  $\gamma$  on W and W', and deduce that W and W' cannot be isomorphic.

The decomposition (2.4.1.2) describes how  $\gamma$  acts on  $\operatorname{ad}^0(\overline{\rho})$  and  $\operatorname{ad}^0(\overline{\rho})^*$ . It acts trivially on  $\mathfrak{t}$ , and acts as  $\alpha(\overline{\rho}(\gamma)) = \overline{\chi}(\gamma)$  on  $\mathfrak{g}_{\alpha}$ . Furthermore, there is no other root space on which  $\gamma$  acts by  $\overline{\chi}(\gamma)$ . Now consider how  $\gamma$  acts on W: it must have an eigenspace with eigenvalue 1 of dimension at least  $\frac{1}{r}\dim_{\mathbf{F}_p}\mathfrak{t}$ , and an eigenspace of eigenvalue  $\overline{\chi}(\lambda)$  of dimension  $\frac{1}{r}\dim_{\mathbf{F}_p}\mathfrak{g}_{\alpha}$ . (This uses that the eigenvalues lie in  $\mathbf{F}_p$ : in that case the eigenspace with eigenvalue  $\lambda$  on  $\mathrm{do}^0(\overline{\rho})$  is a direct sum of r copies of the eigenspace with eigenvalue  $\lambda$  on W.) On the other hand, by decomposing  $\mathrm{ad}^0(\overline{\rho})^*$  we see that  $\gamma$  acts as  $\overline{\chi}(\gamma)$  on  $\mathfrak{t}^*$  and the identity on  $\mathfrak{g}_{\alpha}^*$ , and there is no other root space on which  $\gamma$  acts as the identity. This implies  $\gamma$  acts on W' with an eigenspace of eigenvalue 1 with dimension  $\frac{1}{r'}\dim_{\mathbf{F}_p}\mathfrak{g}_{\alpha}^*$  and with an eigenspace of eigenvalue  $\overline{\chi}(\gamma)$  with dimension at least  $\frac{1}{r'}\dim_{\mathbf{F}_p}\mathfrak{t}$ . As we know  $\overline{\chi}(\gamma) \neq 1$ , as long as  $\dim_k \mathfrak{t} > 1 = \dim_k \mathfrak{g}_{\alpha}$ , this shows W and W' cannot be isomorphic.

If the dimension of the maximal torus is 1, then  $\mathfrak{g}' = \mathfrak{sl}_2$ . The eigenvalues of  $\gamma$  acting on  $\mathrm{ad}^0(\overline{\rho})$  are  $\overline{\chi}(\gamma), 1$ , and  $\overline{\chi}(\gamma)^{-1}$ , while the eigenvalues of  $\gamma$  acting on  $\mathrm{ad}^0(\overline{\rho})^*$  are  $1, \overline{\chi}(\gamma)$ , and  $\overline{\chi}(\gamma)^2$ . Since we assumed that  $\overline{\chi}(\gamma)^3 \neq 1$  when the maximal torus is one-dimensional (Definition 2.3.1.1(iii)), the eigenvalues are distinct and a similar argument shows W and W' are not isomorphic.

We can now find places v where we can define Ramakrishna's deformation problem; the key tool is the Chebotarev density theorem.

**Proposition 2.4.2.4.** There exists a place  $v \notin S$  such that:

- (i) assumption (A1) holds, so we obtain a liftable deformation condition  $\mathcal{D}_v^{\text{ram}}$  whose tangent space  $L_v^{\text{ram}}$  has dimension dim  $H^0(\Gamma_v, \text{ad}(\overline{\rho}))$ ;
- (ii) there exists  $\xi \in H^1_{\mathcal{D}_S}(\Gamma_K, \mathrm{ad}^0(\overline{\rho}))$  whose restriction to  $H^1(\Gamma_v, \mathrm{ad}^0(\overline{\rho}))$  does not lie in  $L_v^{\mathrm{ram}}$ ;

(iii) there exists  $\psi \in H^1_{\mathcal{D}_S^{\perp}}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})^*)$  whose restriction to  $H^1(\Gamma_v/I_v, \operatorname{ad}^0(\overline{\rho})^*)$  is non-zero.

*Proof.* We will always pick the place v so that (A1) holds. The first step is to find an additional condition on v so that  $\operatorname{res}_v \xi$  is not in  $L_v^{\operatorname{ram}}$ . By Proposition 2.4.1.7, it suffices to check that  $\xi(\operatorname{Frob}_v) \notin \mathfrak{t}_{\alpha}$ . We will find an element  $g \in \Gamma_F$  such that this holds whenever  $\operatorname{Frob}_v \in g\gamma\Gamma_{F_{\mathcal{E}'}}$ .

Let  $\gamma \in \Gamma_K$  be the element provided by Definition 2.3.1.1(iii). We claim there exists  $\overline{g} \in \operatorname{Gal}(F_{\xi'}/F)$  such that  $\xi(\overline{g}\gamma) \not\in \mathfrak{t}_{\alpha}$ . If  $\xi(\gamma) \not\in \mathfrak{t}_{\alpha}$ , take  $\overline{g}$  to be the identity. Otherwise, consider the image of the homomorphism  $\xi' : \Gamma_F \to \operatorname{ad}^0(\overline{\rho})$ . By Lemma 2.4.2.2  $\xi'$  is fixed by the action of  $\operatorname{Gal}(F/K)$ , so the span of the image is a representation of  $\Gamma_K$ . But  $\operatorname{ad}^0(\overline{\rho})$  is irreducible (Definition 2.3.1.1 (i)), so the image of  $\xi'$  cannot be contained in  $\mathfrak{t}_{\alpha}$  (as then its span would be a proper sub-representation). Thus, there exists  $\overline{g} \in \operatorname{Gal}(F_{\xi'}/F)$  such that  $\xi'(\overline{g}) \not\in \mathfrak{t}_{\alpha}$ . As the adjoint action of  $\overline{g}$  on  $\operatorname{ad}^0(\overline{\rho})$  is trivial, we compute that

$$\xi(\overline{g}\gamma) = \xi(\overline{g}) + \operatorname{ad}(\overline{\rho}(\overline{g}))\xi(\gamma) = \xi(\overline{g}) + \xi(\gamma) \not\in \mathfrak{t}_{\alpha}.$$

Thus if  $\operatorname{Frob}_v \in \overline{g}\gamma\Gamma_{F_{\xi'}}$ , we see  $\xi(\operatorname{Frob}_v) \not\in \mathfrak{t}_{\alpha}$  and hence  $\operatorname{res}_v \xi \not\in L_v^{\operatorname{ram}}$ .

The second step is to find a condition on v so that the image of  $\psi$  in  $H^1(\Gamma_v/I_v, \operatorname{ad}^0(\overline{\rho})^*)$  is not zero. Note that  $I_v$  acts trivially on  $\operatorname{ad}^0(\overline{\rho})^*$  since v does lie above p and  $\rho$  is unramified outside of S. By definition of  $\mathcal{D}_S^{\perp}$ , the composite map

$$H^1_{\mathcal{D}_{\Rightarrow}^\perp}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})^*) \to H^1(\Gamma_v, \mathrm{ad}^0(\overline{\rho})^*) \to H^1(I_v, \mathrm{ad}^0(\overline{\rho})^*)$$

vanishes, so by inflation-restriction we obtain a natural map

$$H^1_{\mathcal{D}_{\varsigma}^{\perp}}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})^*) \to H^1(\Gamma_v/I_v, \mathrm{ad}^0(\overline{\rho})^*).$$

Using Fact 2.2.1.4, the dimension of  $H^1(\Gamma_v/I_v, \operatorname{ad}^0(\overline{\rho})^*)$  is the dimension of  $H^0(\Gamma_v, \operatorname{ad}^0(\overline{\rho})^*)$ . But as we are assuming (A1) holds, (2.4.1.2) shows the space where  $\operatorname{Frob}_v$  acts trivially is the one-dimensional  $\mathfrak{g}_{\alpha}^*$ . We can write down a crossed homomorphism that generates it: the function sending  $\operatorname{Frob}_v$  to a non-zero  $x_{\alpha}^* \in \mathfrak{g}_{\alpha}^* \subset \operatorname{ad}^0(\overline{\rho})^*$ . It is not a 1-coboundary: as  $\operatorname{Frob}_v$  acts trivially on  $\mathfrak{g}_{\alpha}^*$ , all 1-coboundaries are zero. Any non-zero element  $f \in H^1(\Gamma_v/I_v, \operatorname{ad}^0(\overline{\rho})^*)$  must be a multiple of this function, so  $f(\operatorname{Frob}_v)(x_{\alpha})$  will be non-zero for a fixed non-zero  $x_{\alpha} \in \mathfrak{g}_{\alpha} \subset \operatorname{ad}^0(\overline{\rho})$ . So we just need to arrange that  $(\operatorname{res}_v \psi)(\operatorname{Frob}_v)(x_{\alpha}) \neq 0$ .

Now we can use the same technique as in the first step. Lemma 2.4.2.2 shows that  $\psi'$  is a non-zero homomorphism fixed by  $\Gamma_K$ . We will find  $\overline{g}' \in \operatorname{Gal}(F_{\psi'}/F)$  such that  $\psi(\overline{g}'\gamma)(x_{\alpha}) \neq 0$ . If  $\psi(\gamma)(x_{\alpha}) \neq 0$ , take  $\overline{g}'$  to be the identity in  $\operatorname{Gal}(F_{\psi'}/F)$ . Suppose instead that  $\psi(\gamma)(x_{\alpha}) = 0$ . The span of the image of  $\psi'$  is a non-zero representation of  $\Gamma_K$  contained in the irreducible  $\operatorname{ad}^0(\overline{\rho})^*$ , so there must exist  $\overline{g}' \in \operatorname{Gal}(F_{\psi'}/F)$  such that  $\psi'(\overline{g}')(x_{\alpha}) \neq 0$ . Then as  $\overline{g}'$  acts trivially under the adjoint action (by definition of F), we compute

$$\psi(\overline{g}'\gamma)(x_{\alpha}) = \psi(\overline{g}')(x_{\alpha}) + \left(\operatorname{ad}^{0}(\overline{\rho})^{*}(\overline{g}')\right)\psi(\gamma)(x_{\alpha}) = \psi(\overline{g}')(x_{\alpha}) \neq 0$$

In either case, there exists a  $\overline{g}' \in \text{Gal}(F_{\psi'}/F)$  such that  $\psi(\overline{g}'\gamma)(x_{\alpha}) \neq 0$ .

The final step is to pick the prime v so that all of these conditions hold (to be found via the Chebotarev density theorem). Let g and g' denote lifts of  $\overline{g}$  and  $\overline{g}'$  to  $\Gamma_F$ . Consider the subset  $\Sigma$  of  $\Gamma_S$  consisting of elements  $\gamma'$  such that

- 1. assumption (A1) holds with  $\gamma'$  in the role of Frob, there;
- 2.  $\gamma' \in g\gamma\Gamma_{F_{\epsilon'}};$
- 3.  $\gamma' \in g' \gamma \Gamma_{F_{ab'}}$ .

We must show  $\Sigma$  is non-empty. We claim the first condition follows from the second. As  $\Gamma_F$  acts trivially on  $\operatorname{ad}^0(\overline{\rho})$ , we know from Lemma 2.3.1.6 that  $\overline{\rho}(\Gamma_F) \subset Z_G(k) \cap \overline{\rho}(\Gamma_K)$ . In particular,  $\overline{\rho}(\gamma')$  is regular semisimple and  $Z_{G_k}(\overline{\rho}(\gamma'))^\circ = Z_{G_k}(\overline{\rho}(\gamma))^\circ$ . We chose  $\gamma$  as in Definition 2.3.1.1, so this is the torus  $T_k$ . By assumption, there is a unique root  $\alpha \in \Phi(G_k, T_k)$  for which

$$\alpha(\overline{\rho}(\gamma')) = \alpha(\overline{\rho}(\gamma)) = \overline{\chi}(\gamma) = \overline{\chi}(\gamma')$$

using that  $Z_G(k) \subset (\ker(\alpha))(k)$  and that  $\Gamma_F \subset \ker(\overline{\chi})$ . So it suffices to consider the second and third conditions.

Note that  $\Sigma$  is the preimage of its image under  $\Gamma_S \to \operatorname{Gal}(L/K)$  for any finite Galois L/K containing  $F_{\xi'}$  and  $F_{\psi'}$ . Taking L to be the compositum of  $F_{\xi'}$  and  $F_{\psi'}$ , we obtain an injection

$$\operatorname{Gal}(L/K) \hookrightarrow \operatorname{Gal}(F_{\xi'}/K) \times_{\operatorname{Gal}(F/K)} \operatorname{Gal}(F_{\psi'}/K)$$
 (2.4.2.1)

We claim that this is an equality, which is to say  $F_{\xi'} \cap F_{\psi'} = F$ . We have injections  $\xi'$  and  $\psi'$  from  $\operatorname{Gal}(F_{\xi'}/F)$  and  $\operatorname{Gal}(F_{\psi'}/F)$  into  $\operatorname{ad}^0(\overline{\rho})$  and  $\operatorname{ad}^0(\overline{\rho})^*$  respectively. For  $h \in \Gamma_K$  and  $h' \in \operatorname{Gal}(F_{\psi'}/F)$  we have  $h\xi'(hh'h^{-1}) = \xi'(h')$  as  $\xi'$  is  $\Gamma_K$ -invariant. Furthermore, for  $a, b \in \mathbb{Z}$  and  $h_1, h_2 \in \operatorname{Gal}(F_{\xi'}/F)$  we see that

$$a\xi'(h_1) + b\xi'(h_2) = \xi'(h_1^a h_2^b).$$

Thus the image of  $\xi'$  is a non-zero  $\mathbf{F}_p[\Gamma_K]$ -submodule of  $\mathrm{ad}^0(\overline{\rho})$ , and so is a direct sum of copies of W from Lemma 2.4.2.3. As  $\xi'$  is injective,  $\mathrm{Gal}(F_{\xi'}/F)$  is isomorphic to a direct sum of copies of W as  $\mathbf{F}_p[\Gamma_K]$ -modules. Likewise,  $\mathrm{Gal}(F_{\xi'}/(F_{\xi'}\cap F_{\psi'}))$  is a possibly vanishing  $\mathbf{F}_p[\Gamma_K]$ -submodule. We conclude that the  $\Gamma_K$ -invariant quotient

$$\operatorname{Gal}(F_{\xi'}/F)/\operatorname{Gal}(F_{\xi'}/(F_{\xi'}\cap F_{\psi'})) \simeq \operatorname{Gal}(F_{\xi'}\cap F_{\psi'}/F)$$

is isomorphic to a direct sum of copies of W as a  $\mathbf{F}_p[\Gamma_K]$ -module. Similarly, we see that  $\psi'(\operatorname{Gal}(F_{\psi'}/F))$  is an  $\mathbf{F}_p[\Gamma_K]$ -module that is isomorphic to a direct sum of copies of W', so the quotient  $\operatorname{Gal}(F_{\xi'} \cap F_{\psi'}/F)$  is also isomorphic to a direct sum of copies of W'. But as W and W' are not isomorphic (Lemma 2.4.2.3),  $\operatorname{Gal}(F_{\xi'} \cap F_{\psi'}/F)$  is trivial; i.e.  $F = F_{\xi'} \cap F_{\psi'}$  as desired. Thus (2.4.2.1) is an isomorphism and  $\Sigma$  is non-empty.

Now the Chebotarev density theorem implies there exists a place v outside of S (actually infinitely many) for which  $\text{Frob}_v \in \Sigma$ . The three conditions imply that

- 1. assumption (A1) holds, so we can define the condition  $\mathcal{D}_v^{\text{ram}}$ ;
- 2.  $\operatorname{res}_v \xi \not\in L_v^{\operatorname{ram}};$
- 3. The image of  $\psi$  is non-zero in  $H^1(\Gamma_v/I_v, \operatorname{ad}^0(\overline{\rho})^*)$ .

This completes the proof.

With  $\overline{\rho}$  a big representation as before and  $\mathcal{D}_S = \{\mathcal{D}_w\}_{w \in S}$  a deformation condition for  $\overline{\rho}$  satisfying the tangent space inequality (2.2.2.2), allowing  $\overline{\rho}$  to ramify at a place v as in Proposition 2.4.2.4 according to  $\mathcal{D}_v^{\text{ram}}$  will decrease the size of the dual Selmer group:

**Proposition 2.4.2.5.** For  $T = S \cup \{v\}$ , let  $\mathcal{D}_T$  be the deformation condition whose local components are  $\mathcal{D}_w$  for  $w \in S$  and  $\mathcal{D}_v^{\text{ram}}$  at v. Then

$$\dim H^1_{\mathcal{D}_T}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})) < \dim H^1_{\mathcal{D}_S}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})).$$

Furthermore,  $\mathcal{D}_T$  satisfies the tangent space inequality (2.2.2.2).

In the proof, the best perspective is that a deformation condition  $\mathcal{D}_S$  is a collection of local deformation conditions for all places of K such that it is the unramified deformation condition at those places not in S. Then  $H^1_{\mathcal{D}_S}(\Gamma_K, \operatorname{ad}^0(\overline{\rho}))$  consists of elements of  $H^1(\Gamma_K, \operatorname{ad}^0(\overline{\rho}))$  whose restrictions at each place w lie in the tangent space  $L_w$  to the local deformation condition (and likewise for T in place of S).

*Proof.* We will first analyze what happens if we weaken  $\mathcal{D}_S$  to a condition  $\mathcal{D}'_T$  where there is no restriction on deformations at v. We wish to show that  $H^1_{\mathcal{D}_S}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})) = H^1_{\mathcal{D}'_T}(\Gamma_K, \operatorname{ad}^0(\overline{\rho}))$ . The first is certainly a subset of the second.

Recall that as  $\overline{\rho}$  is big,  $H^0(\Gamma_K, \operatorname{ad}^0(\overline{\rho})) = 0$  and  $H^0(\Gamma_K, \operatorname{ad}^0(\overline{\rho})^*) = 0$ . Then applying (2.2.1.1) to  $\mathcal{D}_T'$  and using that the calculation of  $h_v^1$  and  $h_v^0$  in Lemma 2.4.1.4, we see that

$$\dim H^{1}_{\mathcal{D}'_{T}}(\Gamma_{K}, \operatorname{ad}^{0}(\overline{\rho})) - \dim H^{1}_{\mathcal{D}'_{T}^{\perp}}(\Gamma_{K}, \operatorname{ad}^{0}(\overline{\rho})^{*}) = \sum_{w \in T} \left( \dim L_{w} - \dim H^{0}(\Gamma_{w}, \operatorname{ad}^{0}(\overline{\rho})) \right)$$

$$= 1 + \sum_{w \in S} \left( \dim L_{w} - \dim H^{0}(\Gamma_{w}, \operatorname{ad}^{0}(\overline{\rho})) \right).$$

Likewise, we see that

$$\dim H^1_{\mathcal{D}_S}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})) - \dim H^1_{\mathcal{D}_S^{\perp}}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})^*) = \sum_{w \in S} \left(\dim L_w - \dim H^0(\Gamma_w, \mathrm{ad}^0(\overline{\rho}))\right).$$

To check  $H^1_{\mathcal{D}_S}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})) = H^1_{\mathcal{D}_T'}(\Gamma_K, \operatorname{ad}^0(\overline{\rho}))$ , it suffices to show that the difference

$$\dim H^1_{\mathcal{D}'_{\mathcal{T}}}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})) - \dim H^1_{\mathcal{D}_S}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})) = 1 + \dim H^1_{\mathcal{D}'_{\mathcal{T}}}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})^*) - \dim H^1_{\mathcal{D}_S^{\perp}}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})^*)$$

is non-positive. The distinction between  $H^1_{\mathcal{D}_{S}^{\perp}}(\Gamma_{K}, \operatorname{ad}^{0}(\overline{\rho})^{*})$  and  $H^1_{\mathcal{D}_{T}^{\perp}}(\Gamma_{K}, \operatorname{ad}^{0}(\overline{\rho})^{*})$  is that the first uses the condition that the restriction at v lies in  $H^1_{\operatorname{nr}}(\Gamma_{v}, \operatorname{ad}^{0}(\overline{\rho})^{*})$  while the second uses the condition that it lies in  $0 = H^1(\Gamma_{v}, \operatorname{ad}^{0}(\overline{\rho}))^{\perp}$ . The inclusion

$$\dim H^1_{\mathcal{D}'^{\frac{1}{T}}}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})^*) \subset \dim H^1_{\mathcal{D}^{\frac{1}{\sigma}}}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})^*)$$

is strict because of the element  $\psi$  in Proposition 2.4.2.4. Thus  $H^1_{\mathcal{D}_S}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})) = H^1_{\mathcal{D}_T'}(\Gamma_K, \operatorname{ad}^0(\overline{\rho}))$ . Now we study the deformation  $\mathcal{D}_T$  we actually care about. In the left exact sequence

$$0 \to H^1_{\mathcal{D}_T}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})) \to H^1_{\mathcal{D}_T'}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})) \to H^1(\Gamma_v, \mathrm{ad}^0(\overline{\rho}))/L_v^{\mathrm{ram}}$$

the middle term coincides with  $H^1_{\mathcal{D}_S}(\Gamma_K, \operatorname{ad}(\overline{\rho}))$ . The existence of the element  $\xi$  in Proposition 2.4.2.4 implies that

$$\dim H^1_{\mathcal{D}_S}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})) > \dim H^1_{\mathcal{D}_T}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})).$$

Finally, the tangent space inequality (2.2.2.2) holds for  $\mathcal{D}_T$  as dim  $L_v^{\text{ram}} = h_v^0$ .

Corollary 2.4.2.6. There is a finite set of places  $T \supset S$  such that the deformation condition  $\mathcal{D}_T$  obtained by extending  $\mathcal{D}_S$  allowing deformations according to  $\mathcal{D}_v^{\mathrm{ram}}$  for  $v \in T \backslash S$  satisfies

$$H^1_{\mathcal{D}_T^{\perp}}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})^*) = 0.$$

*Proof.* We may assume that  $H^1_{\mathcal{D}_S^{\perp}}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})^*)$  is non-zero, so Lemma 2.4.2.1 implies  $H^1_{\mathcal{D}_S}(\Gamma_K, \operatorname{ad}^0(\overline{\rho}))$  is non-zero. By Proposition 2.4.2.4 and Proposition 2.4.2.5, we can choose  $S_1 = S \cup \{v\}$  such that

$$\dim H^1_{\mathcal{D}_{S_1}}(\Gamma_K, \mathrm{ad}^0(\overline{\rho})) < \dim H^1_{\mathcal{D}_S}(\Gamma_K, \mathrm{ad}^0(\overline{\rho}))$$

and  $\mathcal{D}_{S_1}$  satisfies the tangent space inequality (2.2.2.2). Continuing in this way, we eventually reach  $T \supset S$  such that either  $H^1_{\mathcal{D}_T^{\perp}}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})^*) = 0$  (in which case we are done) or  $H^1_{\mathcal{D}_T}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})) = 0$ . By Lemma 2.4.2.1, the latter implies  $H^1_{\mathcal{D}_T^{\perp}}(\Gamma_K, \operatorname{ad}^0(\overline{\rho})^*) = 0$  as well.

### 2.5 Generalizing Ramakrishna's Method

In this section, we generalize Ramakrishna's lifting method to split connected reductive groups G defined over the ring of integers  $\mathcal{O}$  in a p-adic field with residue field k. We first recall the notion of odd Galois representations, which are the class of representations to which the method applies. Then we discuss a local to global principle for lifting Galois representations subject to a global deformation condition, and finally we choose the local deformation conditions to make this possible.

### 2.5.1 Odd Galois Representations

We first recall the notion of a split Cartan involution. Let H be a connected reductive group over an algebraically closed field k, with Lie algebra  $\mathfrak{h}$ . Denote the Lie algebra of the unipotent radical of a fixed Borel subgroup of H by  $\mathfrak{u}$ . The following result goes back to Cartan, and in this form is [Yun14, Proposition 2.2].

Fact 2.5.1.1. For any involution  $\tau \in \operatorname{Aut}(H)$ , we have  $\dim \mathfrak{h}^{\tau} \geq \dim \mathfrak{u}$ . All involutions where equality holds are conjugate under  $H^{\operatorname{ad}}(k)$ .

Such involutions  $\tau$  of H (or of  $\mathfrak{h}$ ) are called *split Cartan involutions* for H.

**Example 2.5.1.2.** Let  $H = \operatorname{GL}_2$  with  $H^{\operatorname{ad}} = \operatorname{PGL}_2$  with  $p \neq 2$ . A split Cartan involution for  $H^{\operatorname{ad}}$  is given by conjugation by  $\begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}$ : the fixed points of its action on  $\operatorname{Lie} H^{\operatorname{ad}} = \mathfrak{sl}_2$  is the standard Cartan subalgebra, and hence is one-dimensional. On the other hand,  $\dim \mathfrak{u} = 1$ , so it is indeed a split Cartan involution for  $H^{\operatorname{ad}}$ . However, conjugation by an element of H(k) is never a split Cartan involution for H. Indeed, conjugation must act trivially on the 1-dimensional  $\operatorname{Lie} Z_H$ , but conjugation would also induce an automorphism of the derived group  $H' = \operatorname{SL}_2$ . But the fixed points of that automorphism on  $\operatorname{Lie} H'$  are also at least 1-dimensional since  $\mathfrak{sl}_2 \simeq \mathfrak{psl}_2$  as  $p \neq 2$ .

There are split Cartan involutions for  $H = GL_2$ , but they are not inner automorphisms. For example, the automorphism given by transpose-inverse is a split Cartan involution for  $GL_2$ : it fixes

$$\operatorname{span}\begin{pmatrix}0&1\\-1&0\end{pmatrix}\subset\mathfrak{gl}_2$$

Letting W denote the Weyl group, we have the following useful result about the existence of split Cartan involutions [Pat15, Lemma 4.19].

Fact 2.5.1.3. If -1 belongs to the Weyl group W of H and the co-character  $\delta^{\vee}$  of  $H^{\mathrm{ad}}$  given by half the sum of the positive coroots lifts to a cocharacter of H, then H(k) contains an element c such that  $\mathrm{Ad}(c)$  is a split Cartan involution of Lie H'. Moreover, such c can be chosen to have order 2.

If  $-1 \notin W$ , then H has no inner automorphism that is a Cartan involution.

**Example 2.5.1.4.** This shows for example that no element of  $\operatorname{PGL}_n(k)$  (or  $\operatorname{GL}_n(k)$ ) can induce an inner automorphism that is a split Cartan involution when n > 2, as in those cases  $-1 \notin W \simeq S_n$  inside  $\operatorname{GL}_n(\mathbf{Z})$ .

On the other hand, for groups of type  $B_n$  or  $C_n$  with  $n \geq 2$  we have  $-1 \in W$ . If  $n \geq 4$  is even, we also have  $-1 \in W$  for type  $D_n$ . For symplectic and orthogonal *similitude* groups, the center is large enough there are no problems lifting half the sum of the positive coroots. Thus we can produce a split Cartan involution by conjugation.

For a real place v of K, let  $c_v \in \Gamma_v$  be a complex conjugation (well-defined up to conjugacy). We follow [Gro] and say that a Galois representation  $\overline{\rho}: \Gamma_K \to G(k)$  is odd at v when  $\operatorname{ad}(\overline{\rho}(c_v))$  is a split Cartan involution for  $G^{\operatorname{ad}}$  (with Lie  $G^{\operatorname{ad}} = \operatorname{ad}^0(\overline{\rho})$  since p is very good for G).

**Example 2.5.1.5.** Let  $G = \operatorname{GL}_2$  and assume  $\operatorname{char}(k) \neq 2$ . Conjugation by the matrix  $\begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}$  induces a split Cartan involution of  $\mathfrak{sl}_2$ . Any element of order 2 in  $\operatorname{GL}_2(k)$  with determinant -1 is conjugate to this matrix (think of the Jordan canonical form), so we recover the usual definition that  $\overline{\rho}: \Gamma_K \to \operatorname{GL}_2(k)$  is odd at v when  $\det \overline{\rho}(c_v) = -1$ .

**Example 2.5.1.6.** There cannot exist odd representations for  $GL_n$  with n > 2 because of Example 2.5.1.4. There can exist odd representations for  $GSp_{2n}$  when n > 1 and for  $GO_m$  when  $m \ge 5$  and  $m \not\equiv 2 \pmod{4}$ , due to Example 2.5.1.4.

### 2.5.2 Local Lifting Implies Global Lifting

Let  $\mathcal{D}_S = \{\mathcal{D}_v\}$  be a global deformation condition and  $A_1 \to A_0$  a small extension of coefficient  $\mathcal{O}$ -algebras with kernel I. Note that I is a k-vector space. Consider a lift  $\rho_0 : \Gamma_S \to G(A_0)$  of  $\overline{\rho}$  subject to  $\mathcal{D}_S$ . In favorable circumstances, we can use the following local-to-global principle to produce lifts to  $A_1$ 

Let G' be the derived group of G and  $\mu: G \to G/G'$  the quotient. We assume that the deformation condition includes the condition of fixing a lift  $\nu: \Gamma_K \to (G/G')(\mathcal{O})$  of the character  $\mu \circ \overline{\rho}: \Gamma_K \to (G/G')(k)$ . This means that all of the local deformation conditions have tangent spaces lying in  $H^1(\Gamma_v, \operatorname{ad}^0(\overline{\rho}))$ , and the obstruction cocycles automatically land in  $H^2(\Gamma_v, \operatorname{ad}^0(\overline{\rho}))$  (see Example 2.2.2.9 and Example 2.2.2.12), with similar statements for global deformation conditions.

**Proposition 2.5.2.1.** Provided  $H^1_{\mathcal{D}_S^+}(\Gamma_S, \operatorname{ad}^0(\overline{\rho})^*) = 0$ , lifting  $\rho_0$  to  $A_1$  subject to  $\mathcal{D}_S$  is equivalent to lifting  $\rho_0|_{\Gamma_v}$  to  $A_1$  subject to  $\mathcal{D}_v$  for all  $v \in S$ .

*Proof.* One direction is obvious. Conversely, suppose we have local lifts. The key input is the Poitou-Tate exact sequence (Fact 2.2.1.7):

$$H^{1}(\Gamma_{S}, \operatorname{ad}^{0}(\overline{\rho})) \to \bigoplus_{v \in S} H^{1}(\Gamma_{v}, \operatorname{ad}^{0}(\overline{\rho})) / L_{v} \to H^{1}_{\mathcal{D}_{S}^{\perp}}(\Gamma_{S}, \operatorname{ad}^{0}(\overline{\rho})^{*})^{\vee} \to H^{2}(\Gamma_{S}, \operatorname{ad}^{0}(\overline{\rho})) \to \bigoplus_{v \in S} H^{2}(\Gamma_{v}, \operatorname{ad}^{0}(\overline{\rho})).$$

The vanishing of  $H^1_{\mathcal{D}_{\overline{s}}^{\perp}}(G_S, \mathrm{ad}^0(\overline{\rho})^*)$  implies that the first map is a surjection and the last an injection.

As  $\rho_0|_{\Gamma_v}$  is liftable for all  $v \in S$ , the local obstructions to lifting vanish. The global obstruction to lifting  $\rho_0$  to  $A_1$ ,  $\operatorname{ob}(\rho_0) \in H^2(\Gamma_S, \operatorname{ad}^0(\overline{\rho})) \otimes I$ , therefore maps to 0 in  $\bigoplus_{v \in S} H^2(\Gamma_v, \operatorname{ad}^0(\overline{\rho})) \otimes I$ . As this latter restriction map is injective, there is a lift  $\rho_1$  of  $\rho_0$  to  $A_1$  on  $\Gamma_S$ . We wish to show it can be chosen subject to  $\mathcal{D}_S$ .

The set of all lifts of  $\rho_0|_{\Gamma_v}$  is an  $H^1(\Gamma_v, \operatorname{ad}^0(\overline{\rho})) \otimes I$ -torsor. The existence of local lifts means that there exist  $\phi_v \in H^1(\Gamma_v, \operatorname{ad}^0(\overline{\rho})) \otimes I$  such that  $\phi_v \cdot \rho_0|_{\Gamma_v} \in \mathcal{D}_v(A_1)$ . By the surjectivity of the first map in the sequence, there exists  $\phi \in H^1(\Gamma_S, \operatorname{ad}^0(\overline{\rho})) \otimes I$  such that  $\phi|_v$  agrees with  $\phi_v$  up to an element of  $L_v \otimes I$  for all  $v \in S$ . As the set of lifts of  $\rho_0|_{\Gamma_v}$  subject to  $\mathcal{D}_v$  is a  $L_v \otimes I$ -torsor, this implies that  $(\phi \cdot \rho_1)|_{\Gamma_v} \in \mathcal{D}_v(A_1)$ . In other words,  $\phi \cdot \rho_1$  is a lift of  $\rho_0$  to  $A_1$  satisfying  $\mathcal{D}_S$ .

### 2.5.3 Choosing Deformation Conditions

Let G' be the derived group of G and  $\mu: G \to G/G'$  be the quotient map. For a fixed lift  $\nu$  of  $\mu \circ \overline{\rho}: \Gamma_K \to (G/G')(k)$ , the heart of the matter is to choose deformations conditions so the results of §2.5.2 produce a geometric lift of  $\overline{\rho}$  with  $\mu \circ \overline{\rho} = \nu$ . We need:

- 1. Locally liftable deformation conditions at places above p whose characteristic-zero points are lattices in crystalline (or semistable representations).
- 2. Locally liftable deformation conditions at finite places away from p where  $\overline{p}$  is ramified.
- 3. The ability to choose Ramakrishna's deformation condition (§2.4) at additional finite places away from p where  $\overline{\rho}$  is unramified in order to find S so that  $H^1_{\mathcal{D}^{\perp}_{S}}(\Gamma_S, \operatorname{ad}^0(\overline{\rho})^*) = 0$ .

**Remark 2.5.3.1.** The importance of the fixed  $\nu$  is discussed in §2.5.4. The technical consequence is that deformation conditions that incorporate this fixed lift can be analyzed using the Galois cohomology of  $\operatorname{ad}^0(\overline{\rho})$ , rather than of the larger  $\operatorname{ad}(\overline{\rho})$ .

It is necessary to extend  $\mathcal{O}$  and k in order to define some of these deformation conditions: the condition that  $\overline{\rho}$  is big is unaffected (Remark 2.3.1.3), so we are free to do so. We will find such deformation conditions when  $G = \operatorname{GSp}_m$  with even  $m \geq 4$  or  $G = \operatorname{GO}_m^{\circ}$  with  $m \geq 5$ . In order to have the necessary oddness assumption on  $\overline{\rho}$ , in the latter case  $m \not\equiv 2 \pmod{4}$ .

Remark 2.5.3.2. As our analysis in this chapter applies to connected reductive groups, we need to use the connected group  $GO_m^{\circ}$ . It is possible to modify the arguments to apply to some disconnected groups (see the treatment of L-groups in [Pat15, §9]) but we do not do so here. Given  $\overline{\rho}: \Gamma_K \to GO_m^{\circ}(k) \subset GO_m(k)$ , viewing it as a representation for the group  $G = GO_m$  any deformation to a coefficient ring A will automatically factor through  $GO_m^{\circ}(A)$ , so deformation conditions for the larger group  $GO_m$  naturally give deformation conditions for  $GO_m^{\circ}$ .

At the places above p, when  $G = GO_m^{\circ}$  or  $GSp_m$  after extending k we will construct a Fontaine-Laffaille deformation condition using Fontaine-Laffaille theory in Chapter 3. This requires the assumption that  $\nu \otimes \mathcal{O}[\frac{1}{p}]$  is crystalline, p is unramified in K,  $\overline{\rho}$  is torsion-crystalline with Hodge-Tate weights in an interval of length  $\frac{p-2}{2}$ , and that the Fontaine-Laffaille weights for each  $\mathbf{Z}_p$ -embedding of  $\mathcal{O}_K$  into  $\mathcal{O}$  are pairwise distinct. The deformation condition will be proved to be liftable, and the dimension of the tangent space will be  $h^0(\Gamma_v, \operatorname{ad}^0(\overline{\rho})) + [K_v : \mathbf{Q}_p] \dim_k \mathfrak{u}$ , where  $\mathfrak{u}$  is the Lie algebra of the unipotent radical of a Borel subgroup of G. This generalizes the results for  $GL_n$  obtained in [CHT08, §2.4.2].

Remark 2.5.3.3. The restriction that p is unramified in K and that the Hodge-Tate weights of  $\overline{\rho}$  are in an interval of length  $\frac{p-2}{2}$  is required to use Fontaine-Laffaille theory. Approaches using different flavors of integral p-adic Hodge theory should be able to remove it (for example, the deformation condition based on ordinary representations worked out by Patrikis [Pat15, §4.1] does so for a special class of representations). However, most previous work on studying deformation rings using integral p-adic Hodge theory only gives results about the deformation ring with p inverted, which does not suffice for our method.

The assumption that the Hodge-Tate weights are pairwise distinct is crucial, as otherwise the expected dimensions of the local crystalline deformation rings are too small to use in Ramakrishna's method.

At the places where  $\overline{\rho}$  is ramified, in Chapter 4 we will construct a minimally ramified deformation condition by studying deformations of nilpotent (or equivalently unipotent) provided  $p \geq m$ . For each place, this will potentially require a finite extension of k. After such a further extension, this will be a liftable deformation condition at v with tangent space of dimension  $h^0(\Gamma_v, \operatorname{ad}^0(\overline{\rho}))$  (see Corollary 4.5.3.4). This generalizes the results for  $\operatorname{GL}_n$  obtained in [CHT08, §2.4.4].

We also need to specify a deformation condition at the archimedean places v: we just require lifts for which  $\mu \circ \rho|_{\Gamma_v} = \nu|_{\Gamma_v}$ . This condition is very simple to arrange, as  $\#\Gamma_v \leq 2$ . At a complex place, the dimension of the tangent space is zero and the dimension of the invariants is  $\dim_k \operatorname{ad}^0(\overline{\rho})$ . At a real place, the tangent space is zero when p > 2 and the invariants are the invariants of complex conjugation on  $\operatorname{ad}^0(\overline{\rho})$ .

In order to construct Ramakrishna's deformation condition, the tangent space inequality (2.2.2.2) must be satisfied. Let S be a set of places consisting of primes above p, places where  $\overline{p}$  is ramified, and the archimedean places. When using the local deformation conditions as above at  $s \in S$ , the inequality (2.2.2.2) says exactly that

$$[K:\mathbf{Q}] \dim_k \mathfrak{u} = \sum_{v|p} [K_v:\mathbf{Q}_p] \dim_k \mathfrak{u} \ge \sum_{v|\infty} h^0(\Gamma_v, \mathrm{ad}^0(\overline{\rho})) = \sum_{v|\infty} \mathrm{ad}^0(\overline{\rho})^{\Gamma_v}$$
 (2.5.3.1)

This is very strong:  $\dim \operatorname{ad}^0(\overline{\rho})^{\Gamma_v} \geq [K_v : \mathbf{R}] \dim_k \mathfrak{u}$  by Fact 2.5.1.1, so (2.5.3.1) holds if and only if K is totally real and  $\overline{\rho}$  is odd at all real places of K.

Assuming K is totally real and  $\bar{\rho}$  is odd and at all real places, we can allow ramification at additional places to define a deformation condition to which we can apply Proposition 2.5.2.1. In particular, using Ramakrishna's deformation condition  $\mathcal{D}_v^{\text{ram}}$  at a collection of new places as in Corollary 2.4.2.6 (again possibly extending k), we obtain a new deformation condition  $\mathcal{D}_T$  for which  $H^1_{\mathcal{D}_T^{\perp}}(\Gamma_T, \text{ad}^0(\bar{\rho})^*) = 0$ . Using Proposition 2.5.2.1, we obtain the desired lifts.

Let us collect together all of our assumptions and record the result. For  $G = \operatorname{GSp}_m$  with even  $m \geq 1$  or  $G = \operatorname{GO}_m^\circ$  with  $m \geq 5$  and a big representation  $\overline{\rho}: \Gamma_K \to G(k)$ , fix a lift  $\nu: \Gamma_K \to (G/G')(k)$  to  $\mathcal O$  of  $\mu \circ \overline{\rho}$  such that  $\nu \otimes \mathcal O[\frac{1}{p}]$  is Fontaine-Laffaille. We furthermore assume that K is totally real and that  $\overline{\rho}$  is odd at all real places (which requires  $m \not\equiv 2 \pmod{4}$ ). To use the Fontaine-Laffaille condition, we assume that p is unramified in K and that  $\overline{\rho}$  is Fontaine-Laffaille at all places above p with Fontaine-Laffaille weights in an interval of length  $\frac{p-2}{2}$ , pairwise distinct for each  $\mathbb Q_p$  embedding of K into  $\mathcal O[\frac{1}{p}]$ . In order to use the minimally ramified deformation condition of §4.5, we require that  $p \geq m$ . We extend  $\mathcal O$  (and k) so that all of the required deformation conditions may be defined.

**Theorem 2.5.3.4.** Under these conditions, there is a finite set T of places containing the archimedean places, the places above p, and the places where  $\overline{\rho}$  is ramified such that there exists a lift  $\rho: \Gamma_K \to G(\mathcal{O})$  such that

- $\mu \circ \rho = \nu$ .
- $\rho$  is ramified only at places in T
- $\rho$  is Fontaine-Laffaille at all places above p, and hence crystalline.

In particular,  $\rho$  is geometric. If we combine this with Proposition 2.3.1.4, we obtain Theorem 1.1.2.2.

**Remark 2.5.3.5.** Using the local deformation conditions for  $GL_n$  in [CHT08, §2.4.1] and [CHT08, §2.4.4], the same argument gives an identical result with  $G = GL_n$ . But for n > 2 it is impossible to satisfy the oddness hypothesis. For  $GL_2$ , this is a variant of [Ram02, Theorem 1b].

Remark 2.5.3.6. For other groups, the method will produce lifts provided appropriate local conditions exist. The deformation conditions we used are only available in full strength for symplectic and orthogonal groups. An alternative deformation condition above p is the ordinary deformation condition [Pat15, §4.1], available for any G. For ramified primes not above p, §4.4 provides a deformation condition assuming a certain nilpotent centralizer is smooth and  $\overline{\rho}|_{\Gamma_n}$  is tamely ramified of the special type considered in §4.4.

### 2.5.4 Lifting Representations valued in Tori

We will briefly explain the importance of fixing a lift of  $\overline{\nu} = \mu \circ \overline{\rho} : \Gamma_K \to (G/G')(k)$ . Pragmatically, we do so in order to use  $\mathrm{ad}^0(\overline{\rho})$  instead of the larger  $\mathrm{ad}(\overline{\rho})$  in Proposition 2.5.2.1 and connect with Corollary 2.4.2.6. This really is necessary however, as the universal deformation ring for  $\overline{\nu}$  need not be smooth. In particular, if we have no control on the homomorphism  $\Gamma_K \to G(R) \to (G/G')(R)$  induced by lifts of  $\overline{\rho}$ , we might not be able to lift at each step in Proposition 2.5.2.1 because there would be an obstruction to lifting  $\mu \circ \rho$ .

In [Til96, §4], for a split  $\mathcal{O}$ -torus T' in place of G/G' Tilouine proves that the universal deformation ring for  $\overline{\nu}$  is isomorphic to  $\mathcal{O}[[X^*(T')\otimes\Gamma_S^{\mathrm{ab},p}]]$ , where  $\Gamma_S^{\mathrm{ab},p}$  is the maximal abelian pro-p quotient of  $\Gamma_S$ . The idea is as follows. Reduce to the case of  $T' = \mathbf{G}_m$  and consider the Teichmuller lift. All other lifts to a coefficient ring R differ from this fixed lift by a continuous homomorphism  $\Gamma_S \to 1 + \mathfrak{m}_R$ , which must factor through the maximal abelian p-quotient of  $\Gamma_S$ . So a lift gives a homomorphism

$$\mathcal{O}[\Gamma_S^{\mathrm{ab},p}] \to R.$$

Taking into account the topology on  $\mathcal{O}[[\Gamma_S^{\mathrm{ab},p}]]$ , we see that it continuously extends. Conversely, a homomorphism from  $\mathcal{O}[[\Gamma_S^{\mathrm{ab},p}]]$  gives a continuous lift. Thus the universal deformation ring is  $\mathcal{O}[[\Gamma_S^{\mathrm{ab},p}]]$ .

Next, we briefly recall what class field theory tells us about the structure of  $\Gamma_S^{{\rm ab},p}$ . As before, we assume that S contains all primes above p and the archimedean places. Define

$$U_{p,1} = \{x \in \mathcal{O}_K^\times : x \equiv 1 \pmod{v} \text{ for all } v | p \}$$

and denote the closure of its image in  $\prod_{v|p} \mathcal{O}_v^{\times}$  by  $\overline{U}_{p,1}$ . The  $\mathbf{Z}_p$ -rank of  $\overline{U}_{p,1}$  is  $r_1 + r_2 - 1 - \delta$ , and Leopoldt's conjecture is that  $\delta = 0$ . Class field theory shows (as sketched in [Til96, §4.2]) that the group  $\Gamma_S^{\mathrm{ab},p}$  is the product of a finite p-group and  $\mathbf{Z}_p^{r_2+1+\delta}$ . In particular, the universal deformation ring is smooth (a power series ring) when the finite part of  $\Gamma_S^{\mathrm{ab},p}$  is trivial. In that situation, we could use deformation conditions without fixing a lift of  $\overline{\nu}$ , but in general we must fix a lift to avoid this issue.

Remark 2.5.4.1. There is always a lift of  $\overline{\nu}$ : as G/G' is split, we can reduce to the case of  $\mathbf{G}_m$  and compose with the Teichmuller character  $k^{\times} \to \mathcal{O}^{\times}$  to produce a lift. However, such a lift will not be crystalline above p as it has finite (typically ramified) image: crystalline characters are algebraic on inertia [Con14, Proposition B.4], which means such characters are either unramified or have infinite image. It is not immediately obvious that  $\overline{\nu}$  will lift to characteristic zero in such a way that it is Fontaine-Laffaille at places above p, but such a condition is an obviously necessary condition for Theorem 2.5.3.4 to hold so we do not further explore this question here.

It is worth explaining why there is no local obstruction to  $\overline{\nu}|_{\Gamma_v}$  having a Fontaine-Laffaille lift for v a place above p. Let  $\kappa$  be the residue field of  $K_v$ , and let L be a p-adic field with residue field k which splits K over  $\mathbf{Q}_p$ . Consider a crystalline character  $\Gamma_{K_v} \to \mathcal{O}_L^{\times}$ : composing with the local reciprocity homomorphism and using [Con14, Proposition B.4], on inertia this gives a map  $f: \mathcal{O}_{K_v}^{\times} \to \mathcal{O}_L^{\times}$  which is induced by a  $\mathbf{Q}_p$ -homomorphism

$$R_{K_v/\mathbf{Q}_p}(\mathbf{G}_m) \to R_{L/\mathbf{Q}_p}(\mathbf{G}_m)$$

between Weil restrictions.

By adjunction, such homomorphisms are equivalent to L-homomorphisms

$$R_{K_v/\mathbf{Q}_p}(\mathbf{G}_m) \otimes_{\mathbf{Q}_p} L \to \mathbf{G}_m.$$

As L splits  $K_v$  over  $\mathbf{Q}_p$ , this is a homomorphism

$$\prod_{ au:K_n\hookrightarrow L}(\mathbf{G}_m)_ au o \mathbf{G}_m.$$

Thus the map  $f: \mathcal{O}_{K_v}^{\times} \to \mathcal{O}_L^{\times}$  is of the form

$$x \mapsto \prod_{\tau: K_v \hookrightarrow L} \tau(x)^{n_\tau}. \tag{2.5.4.1}$$

Fix a  $\mathbf{Q}_p$ -embedding  $\tau$  of  $K_v$  into L and hence an embedding of  $\kappa$  into k: any other embedding of  $\kappa$  differs by a power of the Frobenius (and recall that  $K_v/\mathbf{Q}_p$  is unramified since we assume p is unramified in K). By appropriate choice of the exponents  $n_{\tau} \in \{0, 1, \ldots, p-1\}$ , it is clear that any map of the form

$$\kappa^{\times} \ni x \mapsto x^{a_0 + pa_1 + \dots + p^{r-1}a_{r-1}} \in k^{\times}$$

with  $r = [k : \kappa]$  and  $0 \le a_i < p$  is the reduction of something of the form (2.5.4.1).

As  $k^{\times}$  and  $\kappa^{\times}$  are cyclic, any character  $\chi_v: \mathcal{O}_{K_v}^{\times} \to k^{\times}$  is the reduction of a character  $\nu_v: \mathcal{O}_{K_v}^{\times} \to \mathcal{O}_L^{\times}$  of the form (2.5.4.1). Thus any  $\overline{\nu}_v: \Gamma_{K_v} \to k^{\times}$  agrees on inertia with the inertial restriction of the reduction of a crystalline character. The quotient is an unramified character, which may be lifted using a Teichmuller lift. Combining these lifts, we obtain a character  $\nu_v: \Gamma_{K_v} \to \mathcal{O}_L^{\times}$  lifting  $\overline{\nu}|_{\Gamma_v}$  which is algebraic on inertia, hence crystalline. The Hodge-Tate weights are the  $n_{\tau}$ , which may be taken to be in an interval of length p-1. In particular,  $\nu_v$  is Fontaine-Laffaille. (The statement of Fontaine-Laffaille theory in Chapter 3 require an interval of length p-2 to obtain a clean statement: in [FL82] an interval of length p-1 is fine as long as an additional technical condition holds: this is satisfied in our case as long as not all of the  $a_i$  are p-1.)

## Chapter 3

# Fontaine-Laffaille Deformations with Pairings

Let K be a finite extension of  $\mathbf{Q}_p$ , k a finite field of characteristic p, and G be a split connected reductive group over the valuation ring  $\mathcal{O}$  of a p-adic field L with residue field k such that L splits K over  $\mathbf{Q}_p$ . Consider a residual representation  $\overline{p}: \Gamma_K \to G(k)$ . The methods of Chapter 2 require a nice class of deformations of  $\overline{p}$ . In particular, we need a deformation condition  $\mathcal{D}_{\overline{p}}$  on the local Galois group  $\Gamma_K$  such that:

- $\mathcal{D}_{\overline{\rho}}$  is liftable;
- $\mathcal{D}_{\overline{\rho}}$  is large enough, in the sense that its tangent space has dimension

$$[K: \mathbf{Q}_p] \dim_k \mathfrak{u} + \dim_k H^0(\Gamma_K, \mathrm{ad}^0(\overline{\rho}))$$

where  $\mathfrak u$  is the Lie algebra of the unipotent radical of a Borel of G;

•  $\mathcal{D}_{\overline{\rho}}(\mathcal{O})$  consists of crystalline representations (more precisely, lattices in crystalline representations).

Fontaine-Laffaille theory provides a mechanism to study deformations when p is unramified in K (so L splits K if and only if k contains the residue field of K) and the representation  $\overline{\rho}$  is torsion-crystalline with Hodge-Tate weights in an interval of length p-2. For  $G=\operatorname{GL}_n$ , such a deformation condition was constructed in [CHT08, §2.4.1] provided that the Fontaine-Laffaille weights for each  $\mathbb{Q}_p$ -embedding  $K \hookrightarrow L$  are pairwise distinct (see Remark 3.1.2.6). We need to adapt those ideas to symplectic and orthogonal groups under the additional assumption that the Fontaine-Laffaille weights lie in an interval of length  $\frac{p-2}{2}$ . For symplectic groups and  $K = \mathbb{Q}_p$ , this was addressed in Patrikis' senior thesis [Pat06]: this chapter is a mild generalization. Keeping track of the pairing requires knowledge about how Fontaine-Laffaille modules interact with duality and tensor products, the details of which are recorded in [Pat06]. As this is not readily available, we will record proofs.

In §3.1 we will review Fontaine-Laffaille theory and how it interacts with tensor products and duality. The Fontaine-Laffaille deformation condition  $D_{\overline{\rho}}^{\mathrm{FL}}$  will be defined in §3.2: Theorem 3.2.1.2 shows it satisfies the three properties above needed in Chapter 2.

### 3.1 Fontaine-Laffaille Theory with Pairings

We begin by establishing some notation and reviewing the key results of Fontaine-Laffaille theory. It provides a mechanism to relate torsion-crystalline representations to semi-linear algebra data (Fontaine-Laffaille modules), just as p-adic Hodge theory relates crystalline representations to admissible filtered  $\varphi$ -modules. It was first studied by Fontaine and Laffaille [FL82], who introduced a contravariant functor relating torsion-crystalline representations and Fontaine-Laffaille modules. For deformation theory, in particular compatibility with tensor products, it is necessary to use a covariant version, introduced in [BK90]. The details of relating this covariant functor to the functor studied by Fontaine and Laffaille that are omitted in [BK90] are written down in [Con94]. We then studying Fontaine-Laffaille modules with the extra data of a pairing by analyzing tensor products and duals.

### 3.1.1 Covariant Fontaine-Laffaille Theory

Let  $K = W(k')[\frac{1}{p}]$  for a perfect field k' of characteristic p. Let W = W(k') and  $\sigma: W \to W$  denote the Frobenius map. Recall that there is an equivalence of categories between  $\operatorname{Rep}_{\mathbf{Q}_p}^{\operatorname{cris}}(\Gamma_K)$  (crystalline representations of  $\Gamma_K$ ) and  $\operatorname{MF}_K^{\varphi,\operatorname{ad}}$  (admissible filtered  $\varphi$ -modules). For a crystalline representation V and an admissible filtered  $\varphi$ -module M, quasi-inverse contravariant functors are given by

$$D^*_{\mathrm{cris}}(V) = \mathrm{Hom}_{\mathbf{Q}_p[\Gamma_K]}(V, B_{\mathrm{cris}}) \quad \text{and} \quad V^*_{\mathrm{cris}}(M) = \mathrm{Hom}_{\mathrm{MF}_K^{\varphi}}(M, B_{\mathrm{cris}}).$$

Fontaine and Laffaille construct a generalization of this functor that applies to  $\mathbf{Z}_p[\Gamma_K]$ -modules that arise as subquotients of lattices in crystalline representations, provided the analogue of the Hodge-Tate weights of the representation lie in an interval of length p-2. We want a generalization of the *covariant* functors

$$D_{\mathrm{cris}}(V) = (B_{\mathrm{cris}} \otimes_{\mathbf{Q}_n} V)^{\Gamma_K}$$
 and  $V_{\mathrm{cris}}(M) = \mathrm{Fil}^0 (B_{\mathrm{cris}} \otimes_K M)^{\phi = 1}$ .

Our convention will be that the Hodge-Tate weight of the cyclotomic character is -1, which will work well with these covariant functors.

**Definition 3.1.1.1.** A torsion-crystalline representation with Hodge-Tate weights in [a, b] is a  $\mathbb{Z}_p[\Gamma_K]$ -module T for which there exists a crystalline representation V with Hodge-Tate weights in [a, b] and  $\Gamma_K$ -stable lattices  $\Lambda \subset \Lambda'$  in V such that  $\Lambda'/\Lambda$  is isomorphic to T.

The analogue of torsion-crystalline representations on the semilinear algebra side are certain classes of Fontaine-Laffaille modules:

**Definition 3.1.1.2.** A Fontaine-Laffaille module is a W-module M together with a decreasing filtration  $\{M^i\}_{i\in\mathbf{Z}}$  of M by W-submodules and a family of  $\sigma$ -semilinear maps  $\{\varphi_M^i:M^i\to M\}$  such that:

- The filtration is separated and exhaustive:  $M = \bigcup_{i \in \mathbf{Z}} M^i$  and  $\bigcap_{i \in \mathbf{Z}} M^i = 0$ .
- For  $m \in M^{i+1}$ ,  $p \cdot \varphi_M^{i+1}(m) = \varphi_M^i(m)$ .

Morphisms of Fontaine-Laffaille modules  $f:M\to N$  are W-linear maps such that  $f(M^i)\subset N^i$  and  $f\circ\varphi_M^i=\varphi_N^i\circ f$  for all i. The category is denoted  $\mathrm{MF}_W$ .

Remark 3.1.1.3. Jumps in the filtration will turn out to correspond Hodge-Tate weights, so the condition  $M^a = M$  and  $M^{b+1} = 0$  with  $a \leq b$  corresponds to restricting the Hodge-Tate weights to a certain range. We call the set of jumps in the filtration the *Fontaine-Laffaille weights*. We will denote the full subcategory of Fontaine-Laffaille modules with this additional condition by  $\mathrm{MF}_W^{[a,b]}$ .

We are interested in torsion Fontaine-Laffaille modules that satisfy a version of an admissibility condition.

**Definition 3.1.1.4.** Let  $\mathrm{MF}_{W,\mathrm{tor}}^f$  denote the full subcategory of  $\mathrm{MF}_W$  consisting of M for which M is of finite length (as a W-module) and for which  $\sum_{i\in\mathbf{Z}}\varphi^i(M^i)=M$ . Let  $\mathrm{MF}_{W,\mathrm{tor}}^{f,[a,b]}$  denote the full subcategory with the additional condition that  $M^a=M$  and  $M^{b+1}=0$ .

**Fact 3.1.1.5.** The category  $MF_{W,\text{tor}}^{f,[a,b]}$  is abelian, with kernel, cokernel, and image formed as in the underlying category of W-modules. For an object  $M \in MF_{W,\text{tor}}^{f,[a,b]}$ , each  $M^i$  is a direct summand of M as a W-module.

The first statement is [FL82, Proposition 1.8], the second is [Win84, Proposition 1.4.1 (ii)]. We are also interested in a variant that allows non-torsion modules.

**Definition 3.1.1.6.** A filtered Dieudonné module M is a Fontaine-Laffaille module for which the  $M^i$  are direct summands of M as W-modules and for which

$$\sum_{i \in \mathbf{Z}} \varphi^i(M^i) = M.$$

Let  $\mathcal{D}_K$  denote the full subcategory of  $MF_W$  consisting of filtered Dieudonné modules M for which  $M^a = M$  and  $M^{b+1} = 0$  for some  $0 \le b - a \le p - 2$ .

Note that  $\mathrm{MF}_{W,\mathrm{tor}}^{f,[a,b]}$  is a full subcategory of  $\mathcal{D}_K$ . There are natural notions of tensor products and duality.

**Definition 3.1.1.7.** For Fontaine-Laffaille modules  $M_1$  and  $M_2$ , define  $M_1 \otimes M_2$  to have underlying W-module  $M_1 \otimes M_2$ , filtration given by  $(M_1 \otimes M_2)^n = \sum_{i+j=n} M_1^i \otimes M_2^j$ , and maps  $\varphi_{M_1 \otimes M_2}^n$  induced by the  $\varphi_{M_1}^i$  and  $\varphi_{M_2}^j$ .

**Definition 3.1.1.8.** For  $M \in \mathrm{MF}^f_{W,\mathrm{tor}}$ , define  $M^*$  to be  $\mathrm{Hom}_W(M,K/W)$  with the dual filtration  $M_i^* = \mathrm{Hom}_W(M/M^{1-i},K/W)$  and with  $\varphi^i_{M^*}$  characterized for  $f \in M_i^*$  and  $m \in M^j$  by  $\varphi^i_{M^*}(f)(\varphi^j(m)) = 0$  when  $j \geq 1-i$  and by  $\varphi^i_{M^*}(f)(\varphi^j(m)) = f(p^{-i-j}m)$  when j < 1-i (in which case  $-i-j \geq 0$ ).

**Lemma 3.1.1.9.** There is a unique  $(\varphi_{M^*}^i)$  satisfying these constraints. Using it,  $M^*$  is an object of  $\mathrm{MF}_{W,\mathrm{tor}}^f$ . Then  $M \mapsto M^*$  is a contravariant functor from  $\mathrm{MF}_{W,\mathrm{tor}}^f$  to itself, and  $M \simeq M^{**}$  naturally in M.

*Proof.* Uniqueness is immediate, while existence is checked in [Con94,  $\S7.5$ ]. We will use a similar argument in Lemma 3.1.3.2 and Lemma 3.1.3.3.

**Remark 3.1.1.10.** The indexing of the dual filtration is that used in [Fon82, §3.2] and elsewhere. Note that duality sends  $\mathrm{MF}_{W,\mathrm{tor}}^{f,[a,b]}$  to  $\mathrm{MF}_{W,\mathrm{tor}}^{f,[-b,-a]}$ .

To connect Fontaine-Laffaille modules and torsion-crystalline representations, we use the period ring  $A_{\rm cris}$ . For our purposes, all that is important is that  $A_{\rm cris}$  is a period ring that has an action of  $\Gamma_K$ , a ring endomorphism  $\varphi$  (coming from the pth power map) and a filtration  $\{{\rm Fil}^i\,A_{\rm cris}\}$ . In particular, it carries both an action of  $\Gamma_K$  and the structure of a Fontaine-Laffaille module. A convenient reference is [Hat, §2.2], which reviews  $A_{\rm cris}$  for the purposes of constructing the contravariant Fontaine-Laffaille functor. We use it to define an analogue of  $V_{\rm cris}$ :

**Definition 3.1.1.11.** For  $M \in \mathrm{MF}_{W \text{ tor}}^{f,[2-p,1]}$ , define

$$T_{\mathrm{cris}}(M) := \ker \left(1 - \varphi_{A_{\mathrm{cris}} \otimes M}^0 : \mathrm{Fil}^0(A_{\mathrm{cris}} \otimes M) \to A_{\mathrm{cris}} \otimes M\right).$$

Remark 3.1.1.12. A small argument (see [Hat, §2.2]) also shows that

$$A_{\mathrm{cris},\infty} := A_{\mathrm{cris}} \otimes_W K/W = \varinjlim_n A_{\mathrm{cris}}/p^n A_{\mathrm{cris}} \in \mathrm{MF}_{W,\mathrm{tor}}^{f,[0,p-1]}.$$

This allows us define a contravariant functor from  $\mathrm{MF}_{W,\mathrm{tor}}^{f,[0,p-1]}$  to  $\mathrm{Rep}_{\mathbf{Z}_p}(\Gamma_K)$  by

$$T_{\operatorname{cris}}^*(M) := \operatorname{Hom}_{\operatorname{MF}_W}(M, A_{\operatorname{cris}, \infty}).$$

This is well-defined as  $\operatorname{Hom}_{\operatorname{MF}_W}(M, A_{\operatorname{cris}, \infty})$  is  $\mathbf{Z}_p$ -finite (for example, using a contravariant version of Fact 3.1.1.14(4)). This functor agrees with the functor  $U_S$  considered by Fontaine and Laffaille [Hat, Remark 2.7].

If  $M \in \mathrm{MF}_{W \text{ tor}}^{f,[2-p,1]}$  is killed by p, then

$$T_{\mathrm{cris}}^*(M^*) = \mathrm{Hom}_{\mathrm{MF}_W}(M^*, A_{\mathrm{cris}}/pA_{\mathrm{cris}})$$

$$\simeq \ker \left(1 - \varphi_{A_{\mathrm{cris}} \otimes M}^0 : \mathrm{Fil}^0(A_{\mathrm{cris}} \otimes M) \to A_{\mathrm{cris}} \otimes M\right)$$

$$= T_{\mathrm{cris}}(M)$$

which is how Fontaine and Laffaille's results about  $T_{\text{cris}}^*$  imply results about  $T_{\text{cris}}$ .

We can extend  $T_{\text{cris}}$  to  $\mathcal{D}_K$  by defining an analogue of Tate-twisting:

**Definition 3.1.1.13.** For  $M \in \mathrm{MF}_{W,\mathrm{tor}}^{f,[a,b]}$  and an integer s, define M(s) to have the same underlying W-module with filtration  $M(s)^i = M^{i-s}$  and maps  $\varphi_{M(s)}^i = \varphi_M^{i-s}$ .

Tate-twisting allows us to shift Hodge-Tate weights and extend results in the range [2-p,1] to any interval [a,b] where  $b-a \leq p-2$ . For  $M \in \mathrm{MF}_{W,\mathrm{tor}}^{f,[a,b]}$ , we define

$$T_{\text{cris}}(M) = T_{\text{cris}}(M(-(b-1)))(b-1)$$

#### Fact 3.1.1.14. We have:

- 1. The covariant functor  $T_{\text{cris}}: \mathcal{D}_K \to \text{Rep}_{\mathbf{Z}_n}[\Gamma_K]$  is well-defined, and is exact and fully faithfully.
- 2. For  $M \in \mathcal{D}_K$ ,  $T_{cris}(M) = \underline{\lim} T_{cris}(M/p^n M)$ .
- 3. The essential image of  $T_{\text{cris}}: \mathrm{MF}_{W, \mathrm{tor}}^{f, [a,b]} \to \mathrm{Rep}_{\mathbf{Z}_p}[\Gamma_K]$  is stable under formation of sub-objects and quotients.
- 4. For  $M \in \mathrm{MF}_{W,\mathrm{tor}}^{f,[a,b]}$ , the length of M as a W-module is equal to the length of  $T_{\mathrm{cris}}(M)$  as a  $\mathbf{Z}_p$ -module.
- 5. For  $M \in \mathcal{D}_K$ , the  $\Gamma_K$ -representation  $T_{\text{cris}}(M)[\frac{1}{n}]$  is crystalline.
- 6. Any torsion-crystalline  $\mathbf{F}_p[\Gamma_K]$ -module  $\overline{V}$  whose Hodge-Tate weights lie in an interval of length p-2 is in the essential image of  $T_{\mathrm{cris}}$ .

Remark 3.1.1.15. While a torsion representation does not have Hodge-Tate weights, according to Definition 3.1.1.1 the Hodge-Tate weights of a torsion-crystalline representation lie in an interval [a, b] provided the representation is quotient of  $\Gamma_K$ -stable lattices in a crystalline representation with Hodge-Tate weights in [a, b].

This is a modified version of [BK90, Theorem 4.3]. Additional details of the proof of that theorem are recorded in [Con94,  $\S 7$ ]. The first, fourth, and fifth statements are stated explicitly in [BK90, Theorem 4.3]. The second is proven in [Con94,  $\S 7.2$ ]. The claim about the essential image follows from the results of [Con94,  $\S 8.3$ -9.6]: a formal argument shows that if  $T_{\text{cris}}^*$  takes simple objects to simple objects, the essential image is stable under formation of sub-objects and quotients. The content is that  $T_{\text{cris}}^*$  takes simple objects to simple objects. The formal argument adapts to  $T_{\text{cris}}$ , and Remark 3.1.1.12 allows us to deduce that  $T_{\text{cris}}$  takes simple objects to simple objects as all simple objects are automatically killed by p.

For the last statement, we need a fact about  $T_{\text{cris}}^*$ .

**Fact 3.1.1.16.** For  $r \in \{0, 1, ..., p-2\}$ , the functor  $T_{\text{cris}}^*$  induces an anti-equivalence between  $MF_{W,\text{tor}}^{f,[0,r]}$  and the full subcategory of  $\text{Rep}_{\mathbf{Z}_p}(\Gamma_K)$  consisting of torsion-crystalline  $\Gamma_K$  representations with Hodge-Tate weights in [-r, 0].

Dualizing  $\overline{V}$  and Tate-twisting, we may assume that the Hodge-Tate weights of  $\overline{V}^*$  lie in [0, p-2]. Then  $\overline{V}^* = T^*_{\text{cris}}(M)$  for some  $M \in \mathrm{MF}^{f,[0,p-2]}_{W,\text{tor}}$ . Remark 3.1.1.12 shows that  $T_{\text{cris}}(M^*) = \overline{V}$  using double duality (Lemma 3.1.1.9).

Remark 3.1.1.17. Our convention that the Hodge-Tate weight of the cyclotomic character is -1 makes the Fontaine-Laffaille weights and Hodge-Tate weights match under  $T_{cris}$ .

### 3.1.2 Tensor Products and Freeness

We now address two properties of  $T_{\text{cris}}$  where it is crucial to be using the covariant functor. Definition 3.1.1.7 defined a tensor product for Fontaine-Laffaille modules. If  $M_1 \in \mathrm{MF}_{W,\mathrm{tor}}^{f,[a_1,b_1]}$  and  $M_2 \in \mathrm{MF}_{W,\mathrm{tor}}^{f,[a_2,b_2]}$ , it is straightforward to verify that  $M_1 \otimes M_2 \in \mathrm{MF}_{W,\mathrm{tor}}^{f,[a_1+a_2,b_1+b_2]}$ . The functor  $T_{\mathrm{cris}}$  is compatible with tensor products in the following sense:

**Fact 3.1.2.1.** Suppose that  $M_1$ ,  $M_2$ , and  $M_1 \otimes M_2$  each has Fontaine-Laffaille weights in an interval of length at most p-2. Then  $T_{\text{cris}}(M_1) \otimes_{\mathbf{Z}_p} T_{\text{cris}}(M_2) \simeq T_{\text{cris}}(M_1 \otimes M_2)$ .

There is a natural from the left to the right coming from the multiplication of  $A_{\rm cris}$ . To check this map is an isomorphism, one first checks on simple  $M_1$  and  $M_2$  using Fontaine and Laffaille's classification of simple Fontaine-Laffaille modules when the residue field k' is algebraically closed. This is explained in [Con94, §10.6]. Then one uses a dévissage argument to reduce to the general case, as explained in [Con94, §7.11].

**Remark 3.1.2.2.** An analogue of this compatibility is stated in [FL82, Remarques 6.13(b)] for the contravariant functor  $T_{\text{cris}}^*$ . For  $T_{\text{cris}}^*$ , it only makes sense if  $M_1$  and  $M_2$  are killed by p; in that case

$$T_{\mathrm{cris}}^*(M_1) = \mathrm{Hom}_{\mathrm{MF}_W}(M_1, A_{\mathrm{cris}, \infty}) = \mathrm{Hom}_{\mathrm{MF}_W}(M_1, A_{\mathrm{cris}}/pA_{\mathrm{cris}})$$

and likewise for  $M_2$ . Then multiplication on  $A_{\text{cris}}/pA_{\text{cris}}$  gives a natural map  $T_{\text{cris}}^*(M_1) \otimes T_{\text{cris}}^*(M_2) \to T_{\text{cris}}^*(M_1 \otimes M_2)$  which can be checked to be an isomorphism by dévissage. But  $A_{\text{cris},\infty}$  is not a ring, so there is no natural map without a p-torsion hypothesis on  $M_1$  and  $M_2$ . This obstructs an analogue of dévissage to the p-torsion case for  $T_{\text{cris}}^*$ , and explains why it is crucial to work with the covariant functor  $T_{\text{cris}}$ .

For  $M \in \mathrm{MF}_{W,\mathrm{tor}}^{f,[a,b]}$ , if  $V = T_{\mathrm{cris}}(M)$  has "extra structure" then so does M. For example, if V were a deformation of a residual representation over a finite field k, V would be an  $\mathcal{O} = W(k)$ -module. As  $T_{\mathrm{cris}}$  is covariant and fully faithful, it is immediate that M is naturally an  $\mathcal{O}$ -module. The actions of  $\mathbf{Z}_p$  on M via the embeddings into  $\mathcal{O}$  and W = W(k') are obviously compatible.

Representations of  $\Gamma_K$  defined over a finite extension L of  $\mathbf{Q}_p$  can be viewed as  $\mathbf{Q}_p$ -vector spaces with the additional action of L. Assume there exists an embedding of K into L over  $\mathbf{Q}_p$ , so L splits K over  $\mathbf{Q}_p$ . These representations are modules over  $L \otimes_{\mathbf{Q}_p} K \simeq \prod_{\tau:K\hookrightarrow L} L_\tau$  via  $a\otimes b\mapsto (a\tau(b))$ . For each  $\mathbf{Q}_p$ -embedding  $\tau$ , there is a collection of Hodge-Tate weights.

We will generalize this structure to our setting. Now assume k' is finite, and more specifically that k' embeds in k, so  $\mathcal{O}\left[\frac{1}{p}\right]$  splits the finite unramified K over  $\mathbf{Q}_p$ . Hence

$$\mathcal{O} \otimes_{\mathbf{Z}_p} W \simeq \prod_{\tau: W \hookrightarrow \mathcal{O}} \mathcal{O}_{\tau}$$

as  $\mathcal{O}$ -algebras, where  $\tau$  varies over  $\mathbf{Z}_p$ -embeddings of W into  $\mathcal{O}$  and W acts on  $\mathcal{O}_{\tau}$  via  $\tau$ . We likewise obtain a decomposition of the  $\mathcal{O} \otimes_{\mathbf{Z}_p} W$ -module M as:

$$M = \bigoplus_{\tau: W \hookrightarrow \mathcal{O}} M_{\tau}.$$

Note that

$$\operatorname{Hom}_{\mathcal{O}\otimes_{\mathbf{Z}_p}W}(M,M') = \bigoplus_{\tau:W\hookrightarrow\mathcal{O}} \operatorname{Hom}_{\mathcal{O}}(M_\tau,M'_\tau).$$

**Lemma 3.1.2.3.** If V is equipped with a  $\Gamma_K$ -equivariant  $\mathcal{O}$ -module structure then for  $M^i_{\tau} := M_{\tau} \cap M^i$  we have

$$M^i = \bigoplus_{\tau: W \hookrightarrow \mathcal{O}} M^i_\tau$$

and furthermore the  $\sigma$ -semilinear map  $\varphi_M^i|_{M^i_{\underline{\tau}}}:M^i_{\underline{\tau}}\to M$  factors through  $M_{\sigma\tau}$ .

Note that  $\sigma \tau = \tau \sigma$ .

*Proof.* It is immediate that  $M^i = \bigoplus_{\tau} M^i_{\tau}$ , as the induced action of  $\mathcal{O}$  on M preserves the filtration. Let  $j_i: M^i_{\tau} \to M^i$  be the inclusion, so  $j_i(wm) = \tau(w)j_i(m)$  for  $w \in W$  and  $m \in M^i_{\tau}$ . The fact that  $\varphi^i_M$  is  $\sigma$ -semilinear implies that for  $m \in M^i_{\tau}$  and  $w \in W$ ,

$$\sigma\tau(w)\varphi_M^i(j_i(m)) = \varphi_M^i(\tau(w)j_i(m)) = \varphi_M^i(j_i(wm)).$$

**Lemma 3.1.2.4.** The length of M as an  $\mathcal{O}$ -module equals the length of V as an  $\mathcal{O}$ -module multiplied by  $[K: \mathbf{Q}_p]$ .

*Proof.* The length of M as a W-module equals the length of V as a  $\mathbf{Z}_p$ -module by Fact 3.1.1.14(4). The rest is bookkeeping. The  $\mathbf{F}_p$ -dimension of  $\mathcal{O}/p\mathcal{O}$  is  $t = [\mathcal{O}[\frac{1}{p}] : \mathbf{Q}_p]$ , and the k'-dimension of  $\mathcal{O}/p\mathcal{O}$  is  $s = [\mathcal{O}[\frac{1}{p}] : K]$ . It follows that

$$s \lg_{\mathcal{O}}(M) = \lg_{W}(M) = \lg_{\mathbf{Z}_{p}}(V) = t \lg_{\mathcal{O}}(V).$$

Hence  $\lg_{\mathcal{O}}(M) = [K : \mathbf{Q}_p] \lg_{\mathcal{O}}(V)$ .

We can prove a result about freeness when V and M are R-modules for an artinian coefficient  $\mathcal{O}$ -algebra R with residue field k:

**Lemma 3.1.2.5.** We have V is a free R-module if and only if M is a free R-module. When M is a free R-module, all of the  $M_{\tau}^{i}$  are free R-direct summands. All of the  $M_{\tau}$  has the same rank.

*Proof.* A finitely generated R-module N is free if and only if  $\lg_{\mathcal{O}}(N) = n \lg_{\mathcal{O}}(R)$  for  $n = \dim_k N/\mathfrak{m}_R N$ , as we see via Nakayama's lemma applied to a map  $R^n \to N$  inducing an isomorphism modulo  $\mathfrak{m}_R$ . From the exact sequence of Fontaine-Laffaille modules

$$0 \to \mathfrak{m}_R M \to M \to M/\mathfrak{m}_R M \to 0$$

and the fact that  $T_{\text{cris}}$  is covariant and exact, we see that  $T_{\text{cris}}(M/\mathfrak{m}_R M) = V/\mathfrak{m}_R V$ . By Lemma 3.1.2.4, if  $\dim_k V/\mathfrak{m}_R V = n$  then  $M/\mathfrak{m}_R M$  is a k-vector space of dimension  $[K: \mathbf{Q}_p] n$ . Thus to relate R-freeness of M and V we just need to show that  $\lg_{\mathcal{O}}(M) = [K: \mathbf{Q}_p] \lg_{\mathcal{O}}(V)$ , which is Lemma 3.1.2.4.

Now suppose M is a free R-module. By functoriality, the  $\mathbf{Z}_p$ -module direct summands  $M_\tau$  of M are each R-submodules, so each  $M_\tau$  is an R-module direct summand of M. Hence each  $M_\tau$  is R-free when M is free. To deduce the same for each  $M_\tau^i$ , we just need that each  $M_\tau^i$  is an R-module summand. By R-freeness of M, it suffices to show that each  $M_\tau^i/\mathfrak{m}_R M_\tau^i \to M/\mathfrak{m}_R M$  is injective. Since  $M_\tau^i$  is the " $\tau$ -component" of  $M^i$  by Lemma 3.1.2.3 it is an R-module summand of  $M^i$ . Thus it suffices to show that

$$M^i/\mathfrak{m}_R M^i \to M/\mathfrak{m}_R M$$

is injective for all i. But this follows from the abelianness in Fact 3.1.1.5.

To check that all of the  $M_{\tau}$  have the same rank, by freeness it suffices to check that  $\dim_k \overline{M}_{\tau}$  is independent of  $\tau$ . As all  $\mathbf{Z}_p$ -embeddings of the *unramified* W into  $\mathcal{O}$  are of the form  $\sigma^i \tau$  for some fixed  $\mathbf{Z}_p$ -embedding  $\tau$  and  $\sigma$  has finite order, it suffices to show that

$$\dim_k \overline{M}_{\tau} \ge \dim_k \overline{M}_{\sigma\tau}.$$

As each  $\overline{M}_{\tau}^{i}$  is a k-module direct summand of  $\overline{M}_{\tau}$ ,  $\overline{M}_{\tau}$  is isomorphic to  $\operatorname{gr}^{\bullet} \overline{M}_{\tau}$ . Because  $\varphi_{\overline{M}}^{i}(\overline{M}^{i+1}) = 0$ , we obtain a map

$$\sum_{i} \varphi_{M_{\tau}}^{i} : \operatorname{gr}^{\bullet} \overline{M}_{\tau} \to \overline{M}_{\sigma\tau}.$$

As Fontaine-Laffaille modules satisfy

$$\overline{M} = \sum_{i} \varphi^{i}(\overline{M}^{i})$$

the map  $\sum_{i} \varphi_{M_{\tau}}^{i}$  is surjective. This completes the proof.

Remark 3.1.2.6. We get a set of Fontaine-Laffaille weights for each  $\mathbf{Z}_p$ -embedding  $\tau: W \hookrightarrow \mathcal{O}$ . We can also define the multiplicity of a weight  $w_{\tau}$  to be the rank of the R-module  $M_{\tau}^{w_{\tau}}/M_{\tau}^{w_{\tau}+1}$ . The number of Fontaine-Laffaille weights, counted with multiplicity, for each embedding is the same. We say the Fontaine-Laffaille weights with respect to an embedding are distinct if each has multiplicity 1. This is analogous to the way a Hodge-Tate representation of  $\Gamma_K$  over a p-adic field splitting K over  $\mathbf{Q}_p$  has a set of Hodge-Tate weights for each  $\mathbf{Q}_p$ -embedding of K into that field.

We can now define a notion of a tensor product for Fontaine-Laffaille modules that are also R-modules for a coefficient ring R over  $\mathcal{O}$ .

**Definition 3.1.2.7.** Define  $M_1 \otimes_R M_2$  to be the module  $M_1 \otimes_R M_2$  together with filtration defined by  $(M_1 \otimes_R M_2)^n = \sum_{i+j=n} M_1^i \otimes_R M_2^j$  and with  $\varphi^n_{M_1 \otimes_R M_2}$  defined in the obvious way on the pieces.

**Lemma 3.1.2.8.** Suppose that  $M_1$ ,  $M_2$ , and  $M_1 \otimes_R M_2$  are all in  $\mathrm{MF}_{W,\mathrm{tor}}^{f,[a,b]}$  with  $0 \leq b-a \leq p-2$ . The natural map  $T_{\mathrm{cris}}(M_1) \otimes_R T_{\mathrm{cris}}(M_2) \to T_{\mathrm{cris}}(M_1 \otimes_R M_2)$  is an isomorphism.

*Proof.* We have an exact sequence

$$0 \to J \to M_1 \otimes M_2 \to M_1 \otimes_R M_2 \to 0$$

where J is generated by the extra relations imposed by R-bilinearity (beyond W-bilinearity). For  $r \in R$ , define  $\mu_r : M_1 \otimes M_2 \to M_1 \otimes M_2$  by

$$\mu_r(m_1 \otimes m_2) = (rm_1) \otimes m_2 - m_1 \otimes (rm_2).$$

Then  $J = \sum_{r \in R} \operatorname{Im}(\mu_r)$ ; this is an object of  $\operatorname{MF}_{W,\operatorname{tor}}^{f,[a,b]}$  by Fact 3.1.1.5. We will show that  $T_{\operatorname{cris}}(J)$  is the kernel of  $T_{\operatorname{cris}}(M_1 \otimes M_2) \to T_{\operatorname{cris}}(M_1) \otimes_R T_{\operatorname{cris}}(M_2)$ .

It suffices to show that if  $N_1$  and  $N_2$  are sub-objects of  $M_1 \otimes M_2$  then  $T_{\text{cris}}(N_1 + N_2) = T_{\text{cris}}(N_1) + T_{\text{cris}}(N_2)$ . Indeed, granting this we would know that

$$T_{\text{cris}}(J) = \sum_{r \in R} T_{\text{cris}}(\mu_r).$$

But by functoriality  $T_{\text{cris}}(\mu_r)$  is the map  $T_{\text{cris}}(M_1) \otimes T_{\text{cris}}(M_2) \to T_{\text{cris}}(M_1) \otimes T_{\text{cris}}(M_2)$  given by  $v_1 \otimes v_2 \mapsto rv_1 \otimes v_2 - v_1 \otimes rv_2$ , so  $T_{\text{cris}}(J)$  is the kernel of  $T_{\text{cris}}(M_1 \otimes M_2) \to T_{\text{cris}}(M_1) \otimes_R T_{\text{cris}}(M_2)$  as desired.

To prove that  $T_{\text{cris}}(N_1 + N_2) = T_{\text{cris}}(N_1) + T_{\text{cris}}(N_2)$ , consider the exact sequence

$$0 \to N_1 \cap N_2 \to N_1 \oplus N_2 \to N_1 + N_2 \to 0.$$

As  $T_{\rm cris}$  preserves direct sums, it suffices to show that

$$T_{\text{cris}}(N_1) \cap T_{\text{cris}}(N_2) = T_{\text{cris}}(N_1 \cap N_2).$$

But this follows from the exactness of  $T_{\rm cris}$  and the left exact sequence

$$0 \to N_1 \cap N_2 \to N_1 \oplus N_2 \to M_1 \otimes M_2$$

where the second map is  $(n_1, n_2) \mapsto n_1 - n_2$ .

### 3.1.3 Duality

Let R be a coefficient ring over  $\mathcal{O}$  and  $M \in \mathrm{MF}_{W,\mathrm{tor}}^f$  have the structure of a free R-module. Fix  $L \in \mathrm{MF}_{W,\mathrm{tor}}^f$  with an R-structure so that for each  $\tau$ ,  $L_{\tau}$  is a free R-module of rank 1 with  $L_{\tau}^{s_{\tau}} = L_{\tau}$  and  $L_{\tau}^{s_{\tau}+1} = 0$  for some  $s_{\tau}$  (the analogue of a character taking values in  $R^{\times}$ ). We will define a dual relative to L akin to Cartier duality. This will be useful for studying pairings.

**Definition 3.1.3.1.** For an M as above, define  $M^{\vee} = \operatorname{Hom}_{R \otimes_{\mathbf{Z}_n} W}(M, L)$  with a filtration given by

$$\operatorname{Fil}^{i} M^{\vee} = \{ \psi \in \operatorname{Hom}_{R \otimes_{\mathbf{Z}_{n}} W}(M, L) : \psi(M^{j}) \subset L^{i+j} \text{ for all } j \in \mathbf{Z} \}.$$

For  $\psi \in \operatorname{Fil}^i M^{\vee}$ , define  $\varphi_{M^{\vee}}^i(\psi)$  to be the unique function in  $\operatorname{Hom}_{R \otimes_{\mathbf{Z}_p} W}(M, L)$  such that

$$\varphi_{M^{\vee}}^{i}(\psi)(\varphi_{M}^{j}(m))=\varphi_{L}^{i+j}(\psi(m)).$$

for all  $m \in M^j$  and j.

If  $\varphi^i_{M^\vee}$  exists, it is unique since the images of the  $\varphi^j_{M}$ 's span M additively. Likewise, if  $\varphi^i_{M^\vee}$  exists for all i they are automatically  $\sigma$ -semilinear and satisfy  $p\varphi^{i+1}_{M^\vee} = \varphi^i_{M^\vee}|_{\mathrm{Fil}^{i+1}M^\vee}$ . We check  $\varphi^i_{M^\vee}(\psi)$  is well-defined in the following lemma. The fact that all of the  $M^i_{\tau}$  are free R-module direct summands of  $M_{\tau}$  (by Lemma 3.1.2.5) is crucial.

**Lemma 3.1.3.2.** The function  $\varphi_{M^{\vee}}^i(\psi)$  is well-defined, and the filtration can equivalently be described as

$$\operatorname{Fil}^{i} M^{\vee} = \bigoplus_{\tau: W \hookrightarrow \mathcal{O}} \operatorname{Hom}_{R}(M_{\tau}/M_{\tau}^{1+s_{\tau}-i}, L_{\tau}).$$

*Proof.* We first establish the alternate description of  $\mathrm{Fil}^i M^{\vee}$ . Because

$$\operatorname{Hom}_{R\otimes_{\mathbf{Z}_p}W}(M,L) = \bigoplus_{\tau: W \hookrightarrow \mathcal{O}} \operatorname{Hom}_R(M_\tau, L_\tau),$$

and  $L_{\tau}^{s_{\tau}} = L_{\tau}$  while  $L_{\tau}^{s_{\tau}+1} = 0$ , an element  $\psi_{\tau} \in \operatorname{Hom}_{R}(M_{\tau}, L_{\tau})$  satisfies  $\psi_{\tau}(M_{\tau}^{j}) \subset L_{\tau}^{i+j}$  if and only if  $\psi_{\tau}(M_{\tau}^{j}) = 0$  whenever  $i+j > s_{\tau}$ . This says exactly that  $\psi_{i}$  factors through  $M_{\tau}/M_{\tau}^{1+s_{\tau}-i}$ . Because  $M_{\tau}^{1+s_{\tau}-i}$  is an R-module direct summand, hence free with free complement, a morphism  $M_{\tau}/M_{\tau}^{1+s_{\tau}-i} \to L_{\tau}$  is equivalent to a  $\psi_{\tau} : M_{\tau} \to L_{\tau}$  such that  $\psi_{\tau}(M^{1+s_{\tau}-i}) = 0$ . Thus  $\operatorname{Fil}^{i} M_{\tau}^{\vee} = \operatorname{Hom}_{R}(M_{\tau}/M_{\tau}^{1+s_{\tau}-i}, L_{\tau})$  as desired.

We will construct  $\varphi^i_{M^\vee}: \mathrm{Fil}^i M^\vee \to M^\vee$  using the exact sequence

$$0 \to \bigoplus_{r=a+1}^{b} M^r \to \bigoplus_{r=a}^{b} M^r \to M \to 0$$
 (3.1.3.1)

of [FL82, Lemme 1.7]. The first map sends  $(m_r)_{r=a+1}^{r=b}$  to  $(pm_r-m_{r+1})_{r=a}^{r=b}$  (with the convention that  $m_a=0$  and  $m_{b+1}=0$ ), and the second map is  $\sum_{r=a}^b \varphi_M^i$ . For  $\psi\in\operatorname{Fil}^iM^\vee$ , consider the map

$$\phi: \bigoplus_{r=a}^b M^r \to L_\sigma$$

induced by the  $\varphi_L^{i+r} \circ \psi : M^r \to L_\sigma$ . For  $(m_r)_{r=a+1}^{r=b}$  in  $\bigoplus_{r=a+1}^b M^r$ , we compute that

$$\phi((m_r)_{r=a+1}^{r=b}) = \sum_{j=a}^{b} \varphi_L^{i+j} (\psi((pm_j - m_{j+1})))$$

$$= \sum_{j=a}^{b} p \varphi_L^{i+j} (\psi(m_j)) - \sum_{j=a}^{b} \varphi_L^{i+j} (\psi(m_{j+1})).$$

But  $\varphi_L^{i+j}|_{L^{i+j+1}} = p\varphi_L^{i+j+1}$ , so this difference is

$$\sum_{i=a}^{b} p \varphi_L^{i+j}(\psi(m_j)) - \sum_{i=a+1}^{b+1} p \varphi_L^{i+j}(\psi(m_j))$$

which vanishes as  $m_{b+1}=0$  and  $m_a=0$ . Hence  $\phi$  factors through the quotient M of (3.1.3.1), giving the desired well-defined map  $\varphi_{M^{\vee}}^i$ .

**Lemma 3.1.3.3.** The Fontaine-Laffaille module  $M^{\vee}$  is an object of  $MF_{W,tor}^f$ .

*Proof.* We need to show that  $\sum_i \varphi_{M^\vee}^i(\operatorname{Fil}^i M^\vee) = M^\vee$ . It suffices to show that the inclusion

$$\sum_{i} \varphi^{i}_{M^{\vee}}(\operatorname{Fil}^{i} M^{\vee}) \hookrightarrow M^{\vee}$$

is an equality. By Nakayama's lemma, it suffices to show that the reduction modulo  $\mathfrak{m}_R$  is surjective. For an R-module N, let  $\overline{N}$  denote the reduction modulo  $\mathfrak{m}_R$ . We may pick free R-modules  $N^i_{\tau}$  such that  $M^i_{\tau} = N^i_{\tau} \oplus M^{i+1}_{\tau}$  as each  $M^i_{\tau}$  is a (free) direct summand of the R-free  $M_{\tau}$  that is an R-free direct summand of M. Because  $p \cdot \varphi^{i+1}_M = \varphi^i_M|_{M^{i+1}}$ , we see  $\varphi^i_M(\overline{M}^i_{\tau}) = \varphi^i_M(\overline{N}^i_{\tau})$ , so

$$\overline{M}_{\sigma\tau} = \sum_{i} \varphi_{M}^{i}(\overline{N}_{\tau}^{i}).$$

By dimension reasons  $\varphi_M^i|_{N_\tau^i}$  is injective and the sum is direct. We also know that  $\varphi_L^i|_{\overline{L_\tau}}=0$  for  $i< s_\tau$  because  $p\cdot \varphi_L^{j+1}=\varphi_L^j|_{L^{j+1}}$ .

As  $M_{\tau}$  and  $L_{\tau}$  are free R-module summands of M and L for all  $\tau$ ,  $\overline{M^{\vee}} = \overline{M^{\vee}}$  by Lemma 3.1.3.2. We can describe an element  $\psi \in \operatorname{Fil}^i \overline{M^{\vee}}$  as a collection of  $\psi_{\tau,j} \in \bigoplus_{\tau,j} \operatorname{Hom}_R(\overline{N}_{\tau}^j, \overline{L}_{\tau}^{i+j})$ . But  $\overline{L}_{\tau}^{i+j}$  is one-dimensional over k if  $i+j \leq s_{\tau}$ , and is zero otherwise. Then for  $f = \varphi_{M^{\vee}}^i(\psi)$  and  $m = \sum_{\tau,j} \varphi_M^j(n_{\tau,j})$  with  $n_{\tau,j} \in \overline{N}_{\tau}^j$ , by construction we have

$$f(m) = \sum_{\tau,j} \varphi_L^{i+j}(\psi(n_{\tau,j})).$$

But  $\varphi_L^{i+j}(\psi(n_{\tau,j}))$  is forced to be zero unless  $i+j=s_\tau$ , in which case it can take on any non-zero value in  $\overline{L}_\tau$  (depending on the choice of  $\psi$ ). Thus

$$\varphi_{M^{\vee}}^{i}(\operatorname{Fil}^{i}\overline{M}^{\vee}) = \bigoplus_{\tau} \operatorname{Hom}\left(\varphi_{M}^{s_{\tau}-i}(\overline{N}_{\tau}^{s_{\tau}-i}), \overline{L}_{\sigma\tau}\right).$$

Summing over i, and using the sum decomposition  $\overline{M} = \sum_{\tau,i} \varphi^i_M(\overline{N}^i_\tau)$  gives that

$$\sum_{i} \varphi_{\overline{M}^{\vee}}^{i}(\operatorname{Fil}^{i} \overline{M}^{\vee}) = \operatorname{Hom}(\overline{M}, \overline{L}).$$

This shows the desired surjectivity.

Remark 3.1.3.4. For fixed  $\mathbb{Z}_p$ -embedding  $\tau: W \hookrightarrow \mathcal{O}$ , if the Fontaine-Laffaille weights (Remark 3.1.2.6) of M with respect to  $\tau$  are  $\{w_{\tau,i}\}_i$  then the Fontaine-Laffaille weights of  $M^{\vee}$  with respect to  $\tau$  are  $\{s_{\tau} - w_{\tau,i}\}_i$ . This is just bookkeeping.

Letting  $\nu$  be the Galois representation on the free rank-1 R-module corresponding to L, we define the dual  $V^{\vee} = \operatorname{Hom}_{R[\Gamma_K]}(V, R(\nu))$  for a discrete  $\Gamma_K$ -representation on a finite free R-module V.

**Lemma 3.1.3.5.** For a morphism  $f: M \to N$  in  $\operatorname{MF}_{W,\operatorname{tor}}^{f,[a,b]}$  with  $b-a \leq \frac{p-2}{2}$ , there is a natural isomorphism  $T_{\operatorname{cris}}(M^{\vee}) \simeq T_{\operatorname{cris}}(M)^{\vee}$  and  $T_{\operatorname{cris}}(f^{\vee}) = T_{\operatorname{cris}}(f)^{\vee}$ .

*Proof.* We prove this by studying the evaluation pairing  $M \otimes_R M^{\vee} \to L$ . It is straightforward to verify that this pairing is a morphism of Fontaine-Laffaille modules. Because  $b - a \leq \frac{p-2}{2}$ , we obtain a pairing of Galois-modules

$$T_{\text{cris}}(M) \otimes_R T_{\text{cris}}(M^{\vee}) = T_{\text{cris}}(M \otimes_R M^{\vee}) \to T_{\text{cris}}(L) = \nu_R.$$
 (3.1.3.2)

We will now prove that this pairing is perfect when R = k. We will do so by inducting on the dimension of the k-vector space M. The case of dimension 0 is clear. Also, if  $M \neq 0$  the pairing of Fontaine-Laffaille modules is non-zero (look at the pairing of vector spaces  $M_{\tau} \times \text{Hom}(M_{\tau}, L_{\tau}) \to L_{\tau}$ ). Thus the pairing of Galois-modules is non-zero if  $M \neq 0$  as  $T_{\text{cris}}$  is faithful.

Now we use induction, so we can assume  $M \neq 0$ . The annihilator of  $T_{\text{cris}}(M^{\vee})$  is  $T_{\text{cris}}(M_1)$  for some  $f: M_1 \hookrightarrow M$  because the essential image of  $T_{\text{cris}}$  is closed under taking sub-objects. We know  $M_1$  is a proper sub-object as the pairing is non-zero. Observe that we may define the dual  $f^{\vee}: M^{\vee} \to M_1^{\vee}$  by precomposition: it is obviously surjective since we are over a field. For  $v_1 \in T_{\text{cris}}(M_1)$  and  $v_2 \in T_{\text{cris}}(M^{\vee})$ , we must have

$$0 = \langle v_1, f^{\vee} v_2 \rangle = \langle f(v_1), v_2 \rangle.$$

But the pairing  $T_{\text{cris}}(M_1) \otimes T_{\text{cris}}(M_1^{\vee}) \to T_{\text{cris}}(L)$  is non-degenerate by induction, and  $f^{\vee}$  is surjective, so this means that  $v_1 = 0$ . Thus  $T(M_1)$  and hence  $M_1$  are trivial. Over the field k, this ensures the pairing is perfect.

For the general case, we shall use the basic fact that for a coefficient ring R, if  $N_1$  and  $N_2$  are free R-modules of the same rank with an R-bilinear pairing  $N_1 \times N_2 \to R$ , the pairing is perfect if the reduction (modulo  $\mathfrak{m}_R$ )  $\overline{N_1} \times \overline{N_2} \to k$  is perfect. Apply this to  $T_{\mathrm{cris}}(M) \times T_{\mathrm{cris}}(M^{\vee}) \to T_{\mathrm{cris}}(L)$ .

The statement 
$$T_{\text{cris}}(f^{\vee}) = T_{\text{cris}}(f)^{\vee}$$
 is just functoriality.

### 3.2 Fontaine-Laffaille Deformation Condition

Let  $G = \operatorname{GSp}_r$  or  $\operatorname{GO}_r$ , and consider a representation  $\overline{\rho}: \Gamma_K \to G(k)$  with similitude character  $\overline{\nu}$ , where  $K = W[\frac{1}{p}]$  for W = W(k') with finite k'. Let  $\overline{V}$  be the underlying vector space for  $\overline{\rho}$  using the standard representation of G. Take  $\mathcal{O}$  to be the Witt vectors of k, and assume  $\mathcal{O}[\frac{1}{p}]$  splits K over  $\mathbf{Q}_p$ . Fix a lift  $\nu: \Gamma_K \to \mathcal{O}^{\times}$  of  $\overline{\nu}$  that is crystalline with Hodge-Tate weights  $\{s_{\tau}\}_{\tau}$  in an interval of length p-2, corresponding to a filtered Dieudonné module L.

We suppose that  $\overline{\rho}$  is torsion-crystalline with Hodge-Tate weights in an interval [a,b] with  $0 \le b-a \le \frac{p-2}{2}$  so we can use Fontaine-Laffaille theory. Let  $\overline{M}$  be the corresponding Fontaine-Laffaille module (using Fact 3.1.1.14(6)), with Fontaine-Laffaille weights  $\{w_{\tau,i}\}_{\tau,i}$ . In this section we define and study the Fontaine-Laffaille deformation condition assuming that for each  $\mathbb{Z}_p$ -embedding  $\tau: W \hookrightarrow \mathcal{O}$  the Fontaine-Laffaille weights are multiplicity-free as in Remark 3.1.2.6 (rank 1 jumps in the filtration).

### 3.2.1 Definitions and Basic Properties

As  $\overline{V}$  is an k-linear representation of  $\Gamma_K$ ,  $\overline{M}$  becomes a  $k' \otimes_{\mathbf{F}_p} k$ -module and in particular a k-vector space.

**Definition 3.2.1.1.** For a coefficient ring R over  $\mathcal{O} = W(k)$ , define  $D_{\overline{\rho}}^{\operatorname{FL}}(R)$  to be the collection of lifts  $\rho: \Gamma_K \to G(R)$  of  $\overline{\rho}$  with similitude character  $\nu_R$  that lie in the essential image of  $T_{\operatorname{cris}}$  restricted to the full subcategory  $\operatorname{MF}_{W,\operatorname{tor}}^{f,[a,b]}$ . Such a deformation is called a *Fontaine-Laffaille deformation*.

We will analyze this deformation condition when the Fontaine-Laffaille weights of  $\overline{\rho}$  are multiplicity-free for each fixed embedding  $\tau: W \hookrightarrow \mathcal{O}$  (when the jumps in the filtration of each  $\overline{M}_{\tau}$  are 1-dimensional over k).

**Theorem 3.2.1.2.** Under the multiplicity-free assumption on Fontaine-Laffaille weights,  $D_{\overline{\rho}}^{\mathrm{FL}}$  is liftable. If  $\mathfrak{u}$  is the Lie algebra of the unipotent radical of a Borel subgroup of G, the dimension of the tangent space of  $D_{\overline{\rho}}^{\mathrm{FL}}$  is

$$[K: \mathbf{Q}_p] \dim_k \mathfrak{u} + H^0(\Gamma_K, \mathrm{ad}^0(\overline{\rho})).$$

If  $\rho: \Gamma_K \to G(\mathcal{O})$  is an inverse limit of Fontaine-Laffaille deformations of  $\overline{\rho}$  to  $\mathcal{O}/p^n\mathcal{O}$  for all  $n \geq 1$ , it is a lattice in a crystalline representation with the same Fontaine-Laffaille weights as  $\overline{\rho}$ .

The proof of this theorem will occur over the remainder of §3.2. The key pieces are Proposition 3.2.1.7, Proposition 3.2.2.1, and Proposition 3.2.3.1.

To understand  $D_{\overline{\rho}}^{\mathrm{FL}}$ , we must express the orthogonal or symplectic pairing in the language of Fontaine-Laffaille modules. For a Galois module V which is a free R-module, recall we defined  $V^{\vee} = \mathrm{Hom}_{R[\Gamma_K]}(V, \nu_R)$ . For a deformation of  $\overline{\rho}$  to a coefficient ring R, we obtain an  $R[\Gamma_K]$ -module V together with an isomorphism  $\eta: V \simeq V^{\vee}$  coming from the pairing. Let  $\epsilon = 1$  for  $G = \mathrm{GO}_r$  and  $\epsilon = -1$  for  $G = \mathrm{GSp}_r$ . The fact that  $\langle v, w \rangle = \epsilon \langle w, v \rangle$  is equivalent to  $\eta^* = \epsilon \eta$ , where  $\eta^*$  is the map  $V \simeq V^{\vee\vee} \to V^{\vee}$  induced by double duality.

There is a corresponding Fontaine-Laffaille module M such that  $T_{\text{cris}}(M) = V$ , and M is a free R-module by Lemma 3.1.2.5.

**Lemma 3.2.1.3.** For a coefficient ring R, suppose V is a lift of  $\overline{V}$  as an  $R[\Gamma_K]$ -module that is finite free over R, corresponding to a Fontaine-Laffaille module M. An isomorphism of  $R[\Gamma_K]$ -modules

$$n: V \simeq V^{\vee}$$

such that  $\eta(v)(w) = \epsilon \eta(w)(v)$  is equivalent to an R-linear isomorphism of Fontaine-Laffaille modules

$$\gamma: M \simeq M^{\vee}$$

such that  $\gamma(m)(n) = \epsilon \gamma(n)(m)$ .

*Proof.* As the Hodge-Tate weights of  $\overline{\rho}$  lie in an interval of length  $\frac{p-2}{2}$ , Lemma 3.1.2.8 and Lemma 3.1.3.5 hold. In particular,  $T_{\text{cris}}(M^{\vee}) = T_{\text{cris}}(M)^{\vee}$ . As  $T_{\text{cris}}$  is fully faithful in this range, we see that a map  $\eta$  is

equivalent to a map  $\gamma$ , and one is an isomorphism if and only if the other one is. It remains to check that  $\gamma$  is symmetric or alternating if and only if  $\eta$  is. Let  $\eta^*$  and  $\gamma^*$  denote the isomorphisms respectively given by

$$V \simeq V^{\vee\vee} \xrightarrow{\eta^{\vee}} V^{\vee}$$
 and  $M \simeq M^{\vee\vee} \xrightarrow{\gamma^{\vee}} M^{\vee}$ .

A straightforward check shows that  $T_{\text{cris}}$  carries  $\eta^*$  to  $\gamma^*$ , and hence  $\eta = \epsilon \eta^*$  if and only if  $\gamma = \epsilon \gamma^*$ .

**Lemma 3.2.1.4.** An R-linear isomorphism of Fontaine-Laffaille modules  $\gamma: M \simeq M^{\vee}$  for which  $\gamma(m)(n) = \epsilon \gamma(n)(m)$  is equivalent to a perfect  $\epsilon$ -symmetric R-bilinear pairing  $\langle \cdot, \cdot \rangle: M \times M \to L_R$  satisfying

- $\langle M^i, M^j \rangle \subset L^{i+j}$ ;
- $\langle \varphi_M^i(m), \varphi_M^j(n) \rangle = \varphi_L^{i+j} \langle m, n \rangle.$

*Proof.* This is just writing out what  $\gamma: M \to M^{\vee}$  being a morphism of Fontaine-Laffaille modules means for the pairing  $\langle m, n \rangle = \gamma(m)(n)$ .

For  $\gamma$  to preserve the filtration says exactly that

$$\gamma(M^i) \subset \operatorname{Fil}^i M^{\vee} = \{ \psi \in \operatorname{Hom}_R(M, L) : \psi(M^j) \subset L^{i+j} \}.$$

This is equivalent to  $\langle M^i, M^j \rangle \subset L^{i+j}$  for all i, j. The compatibility of  $\gamma$  with the  $\varphi$ 's says exactly that for  $m \in M^i$ 

$$\varphi^i_{M^\vee}(\gamma(m)) = \gamma(\varphi^i_M(m)).$$

Evaluating on any  $\varphi_M^j(n) \in M$  and using the definition of  $M^{\vee}$  we see

$$\varphi_{M^{\vee}}^{i}(\gamma(m))(\varphi_{M}^{j}(n)) = \varphi_{L}^{i+j}(\gamma(m)(n)) = \varphi_{L}^{i+j}(\langle m, n \rangle).$$

Evaluating  $\gamma(\varphi_M^i(m))$ , we see that

$$\gamma(\varphi_M^i(m))(\varphi_M^j(n)) = \langle \varphi_M^i(m), \varphi_M^j(n) \rangle.$$

Thus,  $\gamma$  being compatible with  $\varphi$ 's is equivalent to  $\langle \varphi_M^i(m), \varphi_M^j(n) \rangle = \varphi_L^{i+j}(\langle m, n \rangle)$ .

In particular, the pairing  $\overline{V} \times \overline{V} \to \overline{\nu}$  gives a perfect pairing  $\langle \cdot, \cdot \rangle_{\overline{M}} : \overline{M} \times \overline{M} \to \overline{L}$ .

Corollary 3.2.1.5. For a coefficient ring R, a lift  $\rho \in D^{\mathrm{FL}}_{\overline{\rho}}(R)$  is equivalent to a Fontaine-Laffaille module  $M \in \mathrm{MF}^{f,[a,b]}_{W,\mathrm{tor}}$  that is free as an R-module for which there exists a perfect  $\epsilon$ -symmetric R-bilinear pairing  $\langle \cdot, \cdot \rangle : M \times M \to L_R$  satisfying

- $\langle M^i, M^j \rangle \subset L^{i+j}$ :
- $\bullet \ \langle \varphi_M^i(m), \varphi_M^j(n) \rangle = \varphi_L^{i+j} \langle m, n \rangle.$

such that  $(M, \langle \cdot, \cdot \rangle)$  reduces to  $(\overline{M}, \langle \cdot, \cdot \rangle_{\overline{M}})$ .

*Proof.* This follows by combining the two previous lemmas. Note that the pairing  $\langle \cdot, \cdot \rangle$  is automatically perfect as it lifts the perfect pairing  $\langle \cdot, \cdot \rangle_{\overline{M}}$ .

Corollary 3.2.1.6.  $D_{\overline{\rho}}^{\mathrm{FL}}$  is a deformation condition.

Proof. This follows from the fact that for a morphism of coefficient rings  $R \to R'$ ,  $R' \otimes_R T_{\text{cris}}(M) = T_{\text{cris}}(R' \otimes_R M)$ , exactness properties of  $T_{\text{cris}}$  on  $\text{MF}_{W,\text{tor}}^f$ , and Corollary 3.2.1.5. For example, to check that  $D_{\overline{\rho}}^{\text{FL}}$  is a sub-functor of  $\mathcal{D}_{\overline{\rho}}$ , let R be a coefficient ring and M be the Fontaine-Laffaille module corresponding to  $\rho \in D_{\overline{\rho}}^{\text{FL}}(R)$ . Then  $R' \otimes_R T_{\text{cris}}(M)$ ) lies in the essential image of  $T_{\text{cris}}$ , and  $R' \otimes_R M$  admits a perfect  $\epsilon$ -symmetric R'-bilinear pairing as in Corollary 3.2.1.5 given by extending the pairing on M. This shows that  $\rho_{R'} \in \mathcal{D}_{\overline{\rho}}^{\text{FL}}(R')$ . A similar argument checks Definition 2.2.2.7(2).

It is simple to understand characteristic-zero points of the deformation functor.

**Proposition 3.2.1.7.** Suppose we are given a compatible collection of Fontaine-Laffaille deformations  $\rho_i$ :  $\Gamma_K \to G(R_i)$ , where  $\{R_i\}$  is a co-final system of artinian quotients of the valuation ring R of a finite extension of  $\mathcal{O}[\frac{1}{p}]$  with the same residue field as  $\mathcal{O}$ . Then  $\rho = \varprojlim \rho_i$  is crystalline (more precisely, a lattice in a crystalline representation) with indexed tuple of Hodge-Tate weights equal to the corresponding indexedtuple of Fontaine-Laffaille weights of  $\bar{\rho}$ .

*Proof.* It is straightforward to verify that the inverse limit of the Fontaine-Laffaille modules corresponding to  $\rho_i$  is in  $\mathcal{D}_K$ . Then the Proposition follows from combining Fact 3.1.1.14(2) and (5). Our convention that the cyclotomic character has Hodge-Tate weight -1 makes the Hodge-Tate weights and Fontaine-Laffaille weights match (Remark 3.1.1.17).

### 3.2.2 Liftability

In this section, we analyze liftability by constructing lifts of Fontaine-Laffaille modules. Lifting the underlying module, filtration, and pairing will be relatively easy. Constructing lifts of the  $\varphi_M^i$  compatible with these choices requires substantial work. Let  $\mathcal{W}_{\mathrm{FL},\tau}$  denote the Fontaine-Laffaille weights of  $\overline{\rho}$  with respect to a  $\mathbf{Z}_v$ -embedding  $\tau: W \hookrightarrow \mathcal{O}$ , corresponding to the jumps in the filtration of  $\overline{M}_{\tau}$ .

**Proposition 3.2.2.1.** Under the assumption that the Fontaine-Laffaille weights lie in an interval of length  $\frac{p-2}{2}$  and are multiplicity-free for each  $\tau: W \hookrightarrow \mathcal{O}$ , the deformation condition  $D_{\overline{\rho}}^{\mathrm{FL}}$  is liftable.

Let  $\rho: \Gamma_K \to G(R)$  be a Fontaine-Laffaille deformation of  $\overline{\rho}$ . Let M and  $\overline{M}$  be the corresponding Fontaine-Laffaille modules for  $\rho$  and  $\overline{\rho}$ , which decompose as

$$M = \bigoplus_{\tau} M_{\tau}$$
 and  $\overline{M} = \bigoplus_{\tau} \overline{M}_{\tau}$ .

Each  $M_{\tau}$  is a free R-module by Lemma 3.1.2.5. Recall that there is a filtration  $\{M_{\tau}^{i}\}$  on  $M_{\tau}$  given by R-module direct summands, and that  $\varphi_{M}^{i}(M_{\tau}^{i}) \subset M_{\sigma\tau}$ . In particular, there exist free rank-1 R-modules  $N_{\tau}^{i} \subset M_{\tau}^{i}$  such that  $M_{\tau}^{i} = N_{\tau}^{i} \oplus M_{\tau}^{i+1}$ . As the pairing is  $\mathcal{O}$ -bilinear, the pairings  $M_{\tau} \times M_{\tau} \to L_{\tau}$  are collectively equivalent to the pairing  $M \times M \to L$ , so to lift the pairing and check compatibility it suffices to do so on  $M_{\tau}$ . To analyze liftability, we may work with each  $M_{\tau}$  separately (since  $R \otimes_{\mathbf{Z}_{p}} W = \prod_{\tau} R_{\tau}$  with  $\tau$  varying through  $\mathbf{Z}_{p}$ -embeddings  $W \hookrightarrow \mathcal{O} \to R$ ).

By a basis for  $M_{\tau}$ , we mean a basis for it as an R-module. By Lemma 3.1.2.5, the rank of  $M_{\tau}$  is r. For  $G = \mathrm{GSp}_r$  with r even, the standard alternating pairing with respect to a chosen basis is the one given by the block matrix

$$\begin{pmatrix} 0 & I'_{r/2} \\ -I'_{r/2} & 0 \end{pmatrix}$$

where  $I'_m$  denotes the anti-diagonal matrix with 1's on the diagonal. For  $G = GO_r$ , the standard symmetric pairing with respect to the basis is the one given by the matrix  $I'_r$ .

**Example 3.2.2.2.** Take R = k and fix an embedding  $\tau : W \hookrightarrow \mathcal{O}$ . Let  $w_1, \ldots, w_r$  be the Fontaine-Laffaille weights of  $M_{\tau}$ , and recall that  $w_i + w_{r+1-i} = s_{\tau}$  because  $M \simeq M^{\vee}$ . Pick  $v_i \in M_{\tau}^{w_i} - M_{\tau}^{w_i+1}$ . Since  $\varphi_M^i|_{M_{i+1}} = p\varphi_M^{i+1} = 0$ ,

$$M_{\sigma\tau} = \sum_{i} \varphi^{i}(M_{\tau}^{i}) = \operatorname{span}_{k} \varphi_{M}^{w_{i}}(v_{i}).$$

Note that  $\{\varphi_M^{w_i}(v_i)\}$  is a k-basis for  $M_{\sigma\tau}$ , as the left side has k-dimension r and there are r Fontaine-Laffaille weights for  $\sigma\tau$ . Furthermore, compatibility with the pairing means that

$$\langle \varphi_M^{w_i}(v_i), \varphi_M^{w_j}(v_i) \rangle = \varphi_L^{w_i + w_j}(\langle v_i, v_j \rangle).$$

But  $\varphi_L^h|_{L_\tau} = 0$  unless  $h = s_\tau$ : for  $h > s_\tau$  this is because  $L_\tau^h = 0$ , while for  $h < s_\tau$  this is because  $L_\tau^h = L_\tau^{h+1} = L_\tau$  and  $\varphi_L^h|_{L_\tau^{h+1}} = p\varphi_L^{h+1} = 0$ . Thus  $\langle \varphi_M^{w_i}(v_i), \varphi_M^{w_j}(v_j) \rangle = 0$  unless  $w_i + w_j = s_\tau$ , in which case the pairing must be non-zero as it is perfect. If  $i \neq j$ , by rescaling  $v_i$  we may arrange for  $\langle \varphi_M^{w_i}(v_i), \varphi_M^{w_j}(v_j) \rangle$  to be an arbitrary unit. For  $G = \text{GSp}_\tau$  or  $G = \text{GO}_\tau$  with r even this means after rescaling the pairing

may be taken to be standard with respect to the basis  $n_i = \varphi^{w_i}(v_i)$  of  $M_{\sigma\tau}$ . For  $G = \mathrm{GO}_r$  with r odd and i = [r/2] + 1, defining  $c_\tau := \langle \varphi^{w_i}(v_i), \varphi^{w_i}(v_i) \rangle \in k^\times$  and rescaling  $v_1, \ldots, v_{i-1}$  then brings us to the case that the pairing is  $c_\tau$  times the standard pairing with respect to the basis  $n_i = \varphi^{w_i}(v_i)$  of  $M_{\sigma\tau}$ .

**Remark 3.2.2.3.** The constant  $c_{\tau}$  depends on the choice of basis  $\{v_i\}$  for  $M_{\tau}$ , so in particular is not independent of  $\tau$ . This will not cause problems in later arguments.

Remark 3.2.2.4. There is a lot of notation in the following arguments. With  $\tau$  fixed, we will use  $v_i$  to denote elements of  $M_{\tau}^{w_i}$ , and  $m_i$  to denote elements of  $M_{\sigma\tau}$ . Usually we will have  $\varphi_M^{w_i}(v_i) = m_i$ . If we want to index by Fontaine-Laffaille weights instead of the integers  $\{1, 2, \ldots, r\}$ , we will use  $v'_{w_i} := v_i$  and  $m'_{w_i} := m_i$ .

**Lemma 3.2.2.5.** Let  $w_1 < w_2 < \ldots < w_r$  denote the Fontaine-Laffaille weights of M with respect to  $\tau$ . There exists an R-basis  $m_1 \ldots, m_r$  of  $M_{\sigma\tau}$  such that  $m_i = \varphi_M^{w_i}(v_i)$  where  $v_i$  is an R-basis for a complement to  $M_{\tau}^{w_i+1}$  in  $M_{\tau}^{w_i}$  and such that the pairing  $\langle \cdot, \cdot \rangle$  on  $M_{\sigma\tau}$  is an  $R^{\times}$ -multiple of the standard pairing with respect to the basis  $\{m_i\}$ .

*Proof.* Example 3.2.2.2 shows that such a basis  $\overline{v}_i$  exists over  $R/\mathfrak{m}_R$ : pick a lift  $v_i \in N_{\tau}^i$  of  $\overline{v}_i$ , and define  $m_i = \varphi_M^{w_i}(v_i)$ . We know that

$$\langle \varphi_M^{w_i} v_i, \varphi_M^{w_j} v_j \rangle = \varphi_L^{w_i + w_j} (\langle v_i, v_j \rangle).$$

If  $w_i + w_j > s_\tau$ , this is zero because  $L_\tau^{s_\tau + 1} = 0$ . If  $w_i + w_j < s_\tau$ , this is not a unit as  $\varphi_L^{w_i + w_j}|_{L_\tau^{s_\tau}} = p^{s_\tau - w_i - w_j} \varphi_L^{s_\tau}$ . If  $w_i + w_j = s_\tau$  (equivalently, i + j = r + 1), it is a unit of R as the pairing is perfect.

We will modify the lifts  $v_i$  and then  $m_i = \varphi_M^{w_i}(v_i)$  accordingly. For  $0 \le j \le r/2$  (so j < r+1-j), we will inductively arrange that:

- 1. for  $i \leq j$ ,  $\langle m_i, m_h \rangle = 0$  for  $h \neq r + 1 i$ ;
- 2.  $v_i$  is an R-basis for a complement to  $M_{\tau}^{w_i+1}$  in  $M_{\tau}^{w_i}$ ;
- 3.  $\langle m_i, m_{r+1-i} \rangle$  is a unit for all  $1 \leq i \leq r$ .

For j=0, the first condition is vacuous and the other two conditions hold by our choice of lift. Given that these conditions hold for j-1 with  $1 \le j \le r$ , we will show how to modify the  $v_i$  so that these conditions hold for j. Let  $c = \langle m_j, m_{r+1-j} \rangle \in R^{\times}$ . For j < h < r+1-j, define

$$\widetilde{v}_h := v_h - \langle m_j, m_h \rangle c^{-1} v_{r+1-j}.$$

As  $j \neq r+1-h$ ,  $\langle m_i, m_h \rangle \in \mathfrak{m}_R$ . We compute that

$$\langle m_j, \varphi_M^{w_h} \widetilde{v}_h \rangle = \langle m_j, m_h \rangle - \langle m_j, m_h \rangle c^{-1} \langle m_j, m_{r+1-j} \rangle = 0.$$

For i < j, as  $r+1-i \ne h, r+1-h$  we know  $m_i$  is orthogonal to both  $m_h$  and  $m_{r+1-h}$  by the inductive hypothesis and hence  $\langle m_i, \varphi_M^{w_h} \widetilde{v}_h \rangle = 0$ . Thus (1) holds for the R-basis  $v_1, \ldots, v_j, \widetilde{v}_{j+1}, \ldots, \widetilde{v}_{r-j}, v_{r-j+1}, \ldots, v_r$ .

As  $\widetilde{v}_h - v_h \in M_{\tau}^{w_{r+1-j}}, \widetilde{v}_h$  is still an R-basis for a complement to  $M_{\tau}^{w_h+1}$  in  $M_{\tau}^{w_h}$  (since  $w_{r+1-j} > w_h$  as h < r+1-j), so (2) holds for this new R-basis of  $M_{\tau}$ . Furthermore, we see that

$$\langle \varphi_M^{w_h} \widetilde{v}_h, \varphi_M^{w_{r+1-h}} \widetilde{v}_{r+1-h} \rangle - \langle m_h, m_{r+1-h} \rangle \in \mathfrak{m}_R.$$

As  $\langle m_h, m_{r+1-h} \rangle$  is a unit,  $\langle \varphi_M^{w_h} \widetilde{v}_h, \varphi_M^{w_{r+1-h}} \widetilde{v}_{r+1-h} \rangle$  is a unit and (3) holds. Thus we may modify the lifts  $v_i$  and then accordingly  $m_i$  to satisfy the inductive hypothesis.

Take such a basis for j = [r/2]. By (1),

$$\langle m_i, m_{i'} \rangle = 0$$

if  $i+i'\neq r+1$  and one of i or i' is at most r/2. Otherwise i'>r+1-i so  $w_i+w_{i'}>s_\tau$  and hence the pairing is zero automatically. If r is even, rescale  $v_1,\ldots,v_{r/2}$  so that  $\langle m_i,m_{r+1-i}\rangle=1$  for  $i\leq r/2$  using (3). If r is odd (so  $G=\mathrm{GO}_r$ ), let  $c_\tau=\langle v_{[r/2]+1},v_{[r/2]+1}\rangle\in R^\times$  and rescale  $v_1,\ldots,v_{[r/2]}$  so that  $\langle m_i,m_{r+1-i}\rangle=c_\tau$  for  $1\leq i\leq [r/2]$ . In these cases, the pairing with respect to the basis  $v_1,\ldots,v_r$  is  $c_\tau$  times the standard pairing.

**Remark 3.2.2.6.** When r is odd (so  $G = GO_r$ ), to choose a basis where the pairing is standard we would need to rescale  $v_{[r/2]+1}$  by a square root of the unit  $\langle m_{[r/2]+1}, m_{[r/2]+1} \rangle$ . This might not exist in R. Also note that the orthogonal similitude group  $GO_r$  is unaffected by a unit scaling of the quadratic form.

Now we begin the proof of Proposition 3.2.2.1. Let R' oup R be a small surjection with kernel I. To lift  $\rho$  to  $\rho': \Gamma_K \to G(R')$ , we can reduce to the case when I is killed by  $\mathfrak{m}_R$  and  $\dim_k I = 1$ . Lift the R-module  $M_\tau$  together with its pairing  $\langle \cdot, \cdot \rangle$  over R' as follows. Choose the basis  $\{m_i\}$  provided by Lemma 3.2.2.5, with respect to which  $\langle \cdot, \cdot \rangle$  is  $\omega_\tau$  times the standard pairing for some  $\omega_\tau \in R^\times$ . We take  $M'_{\sigma\tau}$  to be a free R'-module with basis  $\{n_i\}$  reducing to the basis  $\{m_i\}$  of  $M_{\sigma\tau}$ . Lift  $\omega_\tau$  to some  $\omega'_\tau \in (R')^\times$  and define a pairing on  $M'_\tau$  to be  $\omega'_\tau$  times the standard pairing on  $M'_\tau$  with respect to  $\{n_i\}$ . Pick a lift  $u_i \in M'_\tau$  of  $v_i$ , and define a filtration on  $M'_\tau$  by

$$(M'_{\tau})^j = \operatorname{span}_{R'}(u_i : w_i \ge j).$$

We define the module  $M' = \bigoplus_{\tau:W \hookrightarrow \mathcal{O}} M'_{\tau}$  over  $W \otimes_{\mathbf{Z}_p} R$  with filtration  $(M')^i = \bigoplus_{\tau:W \hookrightarrow \mathcal{O}} (M'_{\tau})^i$ . It is clear the filtration reduces to the filtration on  $M_{\tau}$ . Furthermore, the pairing  $M'_{\tau} \times M'_{\tau} \to L_{\tau}$  with respect to  $\{n_i\}$  is the standard one.

It remains to produce  $\varphi^i_{M'}$  lifting  $\varphi^i_{M}$ . As always, it suffices to lift all of the  $\varphi^i_{M_\tau}: M^i_\tau \to M_{\sigma\tau}$  separately. We note that the  $\varphi^j_{M'_\tau}: M'^{jj}_\tau \to M'_{\sigma\tau}$  are determined by the function  $\varphi^{j+1}_{M'_\tau}$  on  $M'^{j+1}_\tau$  (via the relation  $p\varphi^{j+1}_{M'_\tau} = \varphi^j_{M'_\tau}|_{M'^{w_j+1}_\tau}$ ) together with the values  $\varphi^{w_i}_{M'_\tau}(u_i)$  for  $w_i \in \mathcal{W}_{\mathrm{FL},\tau}$ . We will define  $\varphi^{w_i}_{M'_\tau}(u_i)$  for each  $w_i \in \mathcal{W}_{\mathrm{FL},\tau}$  to obtain the desired set of maps  $\varphi^j_{M'}: M'^j \to M'$ .

It will be more convenient to index via weights, so let  $n'_{w_i} = n_i$  and  $u'_{w_i} = u_i$ . Let us consider defining

$$\varphi_{M'_{\tau}}^{w}(u'_{w}) = \sum_{i \in \mathcal{W}_{\text{FL}, \sigma\tau}} c_{iw} n'_{i} := x_{w}$$

for  $c_{iw}$  to be determined with the obvious restriction that  $c_{iw}$  must lift the corresponding coefficient for  $\varphi_M^w(v_w')$ . We will study for which choices of  $\{c_{iw}\}$  these maps are compatible with the pairing.

**Lemma 3.2.2.7.** For any choice of  $\{c_{iw}\}$ , the elements  $x_w$  form a basis for  $M'_{\sigma\tau}$ .

*Proof.* Note that the Fontaine-Laffaille weights of  $\overline{M}$ , M, and M' are the same. Consider the map

$$\sum_{i \in \mathcal{W}_{\mathrm{FL},\tau}} \varphi^i_{M'_{\tau}} : M'^i_{\tau} \to M'_{\sigma\tau}.$$

Quotienting by the maximal ideal of R', as  $\varphi_M'^w$  is a lift of  $\varphi_M^w$  we obtain a surjection

$$\sum_{i \in \mathcal{W}_{\mathrm{FL},\tau}} \varphi_{\overline{M}}^i \colon \overline{M}_{\tau}^i \twoheadrightarrow \overline{M}_{\sigma\tau}$$

as  $\overline{M}_{\sigma\tau} = \sum_{i} \varphi^{i}_{\overline{M}_{\tau}}(\overline{M}_{\tau}^{i})$ . By Nakayama's lemma, the original map is also a surjection. Thus  $\{x_{w}\}$  spans the free R-module  $M'_{\sigma\tau}$ . But  $\#\{x_{w}\} = \operatorname{rk}_{R'}(M'_{\sigma\tau}) = r$ , so  $\{x_{w}\}$  is a basis for  $M'_{\sigma\tau}$ .

The compatibility condition with the pairing is that

$$\langle \varphi^{i}_{M'_{\tau}}(x), \varphi^{j}_{M'_{\tau}}(y) \rangle = \varphi^{i+j}_{L_{\tau}}\left(\langle x, y \rangle\right).$$

Let  $\epsilon=1$  for  $\mathrm{GO}_r$  and  $\epsilon=-1$  for  $\mathrm{GSp}_r$  with even r. Recall that for a Fontaine-Laffaille weight  $i\in\mathcal{W}_{\mathrm{FL}}\tau$ , we define  $i^*$  to satisfy  $i+i^*=s_\tau$ , so  $n_i'$  and  $n_{i^*}'$  pair non-trivially. By linearity and the relations  $p\varphi_{M'}^{w+1}=\varphi_{M'}^w|_{(M')^{w+1}}$  and  $\langle x,y\rangle=\epsilon\langle y,x\rangle$ , it suffices to check compatibility with the pairing only when  $i,j\in\mathcal{W}_{\mathrm{FL}}$ ,  $x=n_i'$  and  $y=n_j'$  and i< j or  $i=j=i^*$  (provided we have arranged that  $p\varphi_{M'}^{w+1}=\varphi_{M'}^w|_{M'^{w+1}}$ ).

**Remark 3.2.2.8.** The case  $i = j = i^*$  only occurs when the pairing is orthogonal and r is odd, for the weight of the unique basis vector which pairs with itself giving a unit.

Of course, there is no reason to expect our initial arbitrary choice of  $\{c_{iw}\}$  to work. Any other choice is of the form  $\{c_{iw} + \delta_{iw}\}$  where  $\delta_{i,w} \in I$ . The compatibility condition on  $M'_{\tau}$  becomes

$$\sum_{w,w'\in\mathcal{W}_{\mathrm{FL},\tau}} (c_{iw} + \delta_{iw})(c_{jw'} + \delta_{jw'}) \langle n'_w, n'_{w'} \rangle = \varphi_{L_{\tau}}^{i+j} \left( \langle n'_i, n'_j \rangle \right).$$

Expanding and using the fact that  $I^2 = 0$ , we see that we wish to choose  $\{\delta_{iw}\}$  so that

$$\sum_{w,w' \in \mathcal{W}_{\text{FI},\tau}} \left( c_{iw} \delta_{jw'} + c_{jw'} \delta_{iw} \right) \left\langle n'_w, n'_{w'} \right\rangle = \omega'_{\tau} C_{ij}$$

where the constant  $C_{ij} := (\omega_{\tau}')^{-1} \left( \varphi_{L_{\tau}}^{i+j}(n_i', n_j') - \sum_{w, w' \in \mathcal{W}_{\mathrm{FL}, \tau}} c_{iw} c_{jw'} \langle n_w', n_{w'}' \rangle \right)$  lies in I as  $\varphi_M^i$  is compatible with the pairing.

Now we can simplify based on the explicit form of the pairing with respect to the basis  $\{n'_w\}$ . As  $n'_w$  pairs non-trivially with  $n'_{w^*}$ , for i < j or  $i = j = i^*$  we obtain the relation

$$\sum_{w \le w^*} \left( c_{iw} \delta_{jw^*} + c_{jw^*} \delta_{iw} \right) + \epsilon \sum_{w > w^*} \left( c_{iw} \delta_{jw^*} + c_{jw^*} \delta_{iw} \right) = C_{ij}. \tag{3.2.2.1}$$

To show that this system of linear equations has a solution, we shall interpret it as a linear transformation. It is now convenient to index using  $\{1,2,3\ldots,r\}$ . Recall that the Fontaine-Laffaille weights of  $M_{\tau}$  are denoted  $w_1 < w_2 < \ldots < w_r$ . Let  $U = I^{\oplus r^2}$ , and decompose U as  $\bigoplus_{i=1}^r U_i$ , where the coordinates of  $U_i = I^{\oplus r}$  are denoted  $\{\delta_{w_i,w_j}\}_{j=1}^r$ . Let  $U' = I^{\oplus \frac{r(r-1)}{2} + \sigma_r}$ , where  $\sigma_r = 1$  if there is a  $w \in \mathcal{W}_{\mathrm{FL},\tau}$  for which  $w = w^*$  and 0 otherwise. (So  $\sigma_r$  is zero unless  $G = \mathrm{GO}_r$  and r is odd.) We may write  $U' = \bigoplus_{i=1}^{r-r} U_i'$ , where the coordinates of  $U_i' = I^{\oplus r-i}$  are denoted  $\{C_{w_iw_j}\}_{j=i+1}^r$ , except if  $\sigma_r = 1$  and  $w_i = w_i^*$ . In that case, instead take  $U_i' = I^{\oplus r-i+1}$  with coordinates denoted  $\{C_{w_iw_j}\}_{j=i}^r$ .

Consider the function  $T: U \to U'$  given by

$$T\left((\delta_{w_{i},w_{h}})_{ih}\right) = \left(\sum_{w_{h} \leq w_{h}^{*}} \left(c_{w_{i}w_{h}}\delta_{w_{j}w_{h}^{*}} + c_{w_{j}w_{h}^{*}}\delta_{w_{i}w_{h}}\right) + \epsilon \sum_{w_{h} > w_{h}^{*}} \left(c_{w_{i}w_{h}}\delta_{w_{j}w_{h}^{*}} + c_{w_{j}w_{h}^{*}}\delta_{w_{i}w_{h}}\right)\right)_{ij}$$

where the  $c_{ww'} \in R'$  matter only through their images in k since  $\mathfrak{m}_R I = 0$ . It suffices to show that T is surjective. As we arranged for I to be 1-dimensional over  $R/\mathfrak{m}_R = k$ , this is question of linear algebra over k upon fixing a k-basis of I.

We will studying particular k-linear maps  $U_i \to U_i'$ . To simplify notation, let  $\epsilon_i = 1$  if  $w_i \leq w_i^*$  and  $\epsilon_i = -1$  otherwise.

**Lemma 3.2.2.9.** Suppose  $w_i \neq w_i^*$ . The linear transformation  $T_i: U_i \to U_i'$  defined on

$$(\delta_{w_i w_h})_h \mapsto (C_{w_i w_j} = \sum_{h=1}^r \epsilon_h c_{w_j w_h^*} \delta_{w_i w_h})_j$$

is surjective. It is the composition  $U_i \to U \xrightarrow{T} U' \to U'_i$ .

*Proof.* As I is one-dimensional over  $R/\mathfrak{m}_R = k$ , it suffices to study the matrix for this linear transformation with respect to a fixed k-basis of I. Fix  $w_{h'} \in \mathcal{W}_{\mathrm{FL},\tau}$ . If we take  $\delta_{w_iw_h} = 0$  for  $w_h \neq w_{h'}$  and  $\delta_{w_iw_{h'}} = 1$ , the image of  $\{\delta_{w_iw_h}\}_h \in U_i$  under  $T_i$  has coordinates  $C_{w_iw_j} = \epsilon_{w_{h'}} c_{w_jw_{h'}}$ . Thus the matrix for  $T_i$  is

$$\begin{pmatrix} \epsilon_1 c_{w_{i+1}w_1^*} & \epsilon_2 c_{w_{i+1}w_2^*} & \dots & \epsilon_r c_{w_{i+1}w_r^*} \\ \epsilon_1 c_{w_{i+2}w_1^*} & \epsilon_2 c_{w_{i+2}w_2^*} & \dots & \epsilon c_{w_{i+1}w_r^*} \\ \dots & \dots & \dots \\ \epsilon_1 c_{w_rw_1^*} & \epsilon_2 c_{w_rw_2^*} & \dots & \epsilon_r c_{w_rw_r^*} \end{pmatrix}.$$

Multiplying the columns where  $w_i > w_{i^*}$  by -1, the columns of this matrix are exactly the coordinates of  $x_{w_j}$  with respect to the basis  $\{n'_w\}_{w \in \mathcal{W}_{\mathrm{FL},\sigma\tau}}$  as in Lemma 3.2.2.7 except that the first i rows are removed. As the  $\{x_w\}$  form a basis, the columns of this matrix span  $U'_i$ .

The last statement follows from the definition.

**Remark 3.2.2.10.** The statement for  $w_i = w_i^*$  is similar. In that case, we must have  $\epsilon = 1$ , and we have

$$C_{w_i w_i} = 2 \sum_j c_{w_i w_j^*} \delta_{w_i w_j}.$$

Extending the definition of  $T_i$  in Lemma 3.2.2.9, we again see that the columns of the matrix representing this transformation are truncated versions of the coordinates of  $x_{w_j}$  with some signs changed and one coordinate multiplied by 2. The image of a basis under the transformation multiplying one coordinate by 2 is still a basis, so again  $T_i$  is surjective.

**Lemma 3.2.2.11.** The composition  $T_{ij}: U_i \to U \xrightarrow{T} U' \to U'_i$  is zero whenever i < j.

Informally, this is saying that T is block lower-triangular with diagonal blocks that are surjective.

*Proof.* The coordinates of  $U_i$  are  $\delta_{w_i w_h}$ . The coordinates of  $U'_j$  are  $C_{w_j w_h}$  for j < h (or  $j \le h$  if  $w_j = w_j^*$ ). The formulas for the  $C_{w_j w_h}$  in the definition of T depend only on  $w_j$  and  $w_h$ . In particular, as  $i < j \le h$ , these coordinates are always zero on the image of the inclusion  $U_i \to U$ .

### Corollary 3.2.2.12. T is surjective.

*Proof.* The composition of  $U_i \to U \to U' \to U'_i$  is exactly  $T_i$ , hence surjective. For  $v \in U'$ , by descending induction on i, we will construct  $u_i \in U_i$  so that

$$T(u_i + \ldots + u_r) - v \in U'_1 \oplus \ldots \oplus U'_{i-1}$$

(meaning  $T(u_1 + \ldots + u_r) = v$  when i = 1). For i = r, take  $u_r$  be a preimage under  $T_r$  of the component of v in  $U'_r$ . Now suppose we have selected  $u_{i+1}, \ldots u_r$ . Pick a preimage  $u_i \in U_i$  of the projection of  $T(u_{i+1} + \ldots u_r) - v$  to  $U'_i$  using the surjectivity of  $T_i$ . We know that  $T_{ij}(u_i) = 0$  for j > i, so

$$T(u_i + \ldots + u_r) - v \in U_1 \oplus \ldots \oplus U_{i-1}.$$

For i = 1, we have  $T(u_1 + \ldots + u_r) = v$  as desired.

By Corollary 3.2.2.12, we can choose the  $\{\delta_{ih}\}$  so that the compatibility relations (3.2.2.1) are satisfied. This defines  $\varphi_{M'_{\tau}}^{w}(n'_{w})$ , and hence we can extend to a map  $\varphi_{M'}^{i}: M' \to M'$  compatible with the pairing. We can finish the proof of Proposition 3.2.2.1 as follows.

Given the deformation  $\rho$  to a coefficient ring R with associated Fontaine-Laffaille module

$$M = \bigoplus_{\tau: W \hookrightarrow \mathcal{O}} M_{\tau},$$

and a small surjection  $R' \to R$  whose kernel I is 1-dimensional over the field  $R/\mathfrak{m}_R$ , we have constructed a free R'-module M' together with a filtration  $\{(M')^i\}$  and maps  $\varphi^i_{M'}$  by lifting the  $M_\tau$ . The filtration and  $\{\varphi^i_{M'}\}$  make M' into a Fontaine-Laffaille module. There is an obvious  $R' \otimes_{\mathbf{Z}_p} W$ -module structure. The condition  $M' = \sum_i \varphi^i_{M'}((M')^i)$  follows from Lemma 3.2.2.7. We also constructed a lift of the pairing  $M' \times M' \to N$ , and the filtration and  $\varphi^i_{M'}$  are compatible with it (in the sense of Corollary 3.2.1.5) by our choice of  $(\delta_{ih})_{ih}$ . By Corollary 3.2.1.5 and Lemma 3.1.2.5,  $T_{\text{cris}}(M')$  gives a representation  $\rho' : \Gamma_K \to G(R')$  lifting  $\rho$ .

### 3.2.3 Tangent Space

The final step in the proof of Theorem 3.2.1.2 is to analyze the tangent space of  $D_{\overline{\rho}}^{\mathrm{FL}}$ . It is a subspace  $L_{\overline{\rho}}^{\mathrm{FL}}$  of the tangent space  $H^1(\Gamma_K, \mathrm{ad}^0(\overline{\rho}))$  of deformations with fixed similitude character  $\nu$ . We are mainly interested in its dimension as a vector space over k, and will analyze it by considering deformations  $\rho$  of  $\overline{\rho}$  to the dual numbers  $k[t]/(t^2)$ . Recall that  $G = \mathrm{GSp}_r$  (with even r) or  $G = \mathrm{GO}_r$ ; let  $\mathfrak{u}$  be the Lie algebra of the unipotent radical of a Borel subgroup of G.

**Proposition 3.2.3.1.** Under the standing assumption that  $\overline{\rho}$  is torsion-crystalline with pairwise distinct Fontaine-Laffaille weights for each  $\tau: W \hookrightarrow \mathcal{O}$  contained in an interval of length  $\frac{p-2}{2}$ ,

$$\dim_k L_{\overline{\rho}}^{\mathrm{FL}} - \dim_k H^0(\Gamma_K, \mathrm{ad}^0(\overline{\rho})) = [K : \mathbf{Q}_p] \dim_k \mathfrak{u}.$$

Let  $\overline{V}$  be the Galois module given by  $\overline{\rho}$ , and for a lift  $\rho$  of  $\overline{\rho}$  over  $k[t]/(t^2)$  let V be the corresponding Galois module. The submodule tV is naturally isomorphic to  $\overline{V}$ , and we have an exact sequence

$$0 \to tV \to V \to \overline{V} \to 0.$$

Let  $\overline{M}$  be the Fontaine-Laffaille module corresponding to  $\overline{\rho}$ , with pairing  $\langle \cdot, \cdot \rangle : \overline{M} \times \overline{M} \to L_k$ . We know  $\overline{M}$  is a k-vector space of dimension  $r[K: \mathbf{Q}_p]$ . Let M be the Fontaine-Laffaille module corresponding to  $\rho$ . It is a free  $k[t]/(t^2) \otimes W$ -module, and fits in an exact sequence

$$0 \to tM \to M \to \overline{M} \to 0$$

of Fontaine-Laffaille modules. The map  $\overline{M} \to tM$  induced by  $t: M \to M$  is an isomorphism of Fontaine-Laffaille modules since it is so on underlying k-vector spaces by  $k[t]/(t^2)$ -freeness of M. As before, we have decompositions

$$M = \bigoplus_{\tau: W \hookrightarrow \mathcal{O}} M_{\tau}$$
 and  $\overline{M} = \bigoplus_{\tau: W \hookrightarrow \mathcal{O}} \overline{M}_{\tau}$ 

from Lemma 3.1.2.3.

Using Lemma 3.2.2.5, pick a basis  $\{v_{\tau,i}\}_{i=1}^r$  of the  $k[t]/(t^2)$ -module  $M_\tau$  such that  $v_{\tau,i}$  is a basis for a  $k[t]/(t^2)$ -complement to  $M_\tau^{w_i+1}$  in  $M_\tau^{w_i}$  and such that the pairing  $M_{\sigma\tau} \times M_{\sigma\tau} \to L_{\sigma\tau}$  with respect to the  $m_{\tau,i} := \varphi_M^{w_i}(v_{\tau,i})$  is  $\omega_\tau$ -times of the standard pairing. As 1-units admit square roots, we may assume that  $\omega_\tau \in k^\times$ . Note that  $\{m_{\tau,i}\} \cup \{tm_{\tau,i}\}$  is a basis for  $M_{\sigma\tau}$  as a k-vector space, and  $\{m_{\tau,i}\}_{\tau,i}$  is a basis for M as a  $k[t]/(t^2)$  module.

Let  $M_0$  be the submodule of M spanned by the  $\{v_{\tau,i}\}_{\tau,i}$  as a k-vector space. We have that  $tM_0 = tM \simeq \overline{M}$  as vector spaces, and have an obvious decomposition

$$M_0 = \bigoplus_{\tau: W \hookrightarrow \mathcal{O}} M_{\tau,0}.$$

We obtain a filtration on  $M_0$  by intersection:  $M_{\tau,0}^i = M^i \cap M_{\tau,0}$ . We also obtain a pairing  $M_0 \times M_0 \to L$  by restriction.

**Lemma 3.2.3.2.** We have that  $M_{\tau}^{i} = M_{\tau,0}^{i} \otimes k[t]/(t^{2})$ , and hence  $M^{i} = M_{0}^{i} \otimes k[t]/(t^{2})$ .

*Proof.* We know that the  $k[t]/(t^2)$ -span of  $v_i$  is a  $k[t]/(t^2)$ -complement to  $M_{\tau}^{w_i+1}$  in  $M_{\tau}^{w_i}$ . Hence  $M_{\tau}^{w_i}/M_{\tau}^{w_i+1}$  is isomorphic to the k-span of  $v_i$  and  $tv_i$ . As the filtration is automatically split (i.e.  $M_{\tau}^i$  is a direct summand of  $M_{\tau}$ , and hence  $M_{\tau}^i$  is a direct summand of  $M_{\tau}^{i-1}$ ), this suffices.

Observe that the surjection of Fontaine-Laffaille modules  $M \to \overline{M}$  carries  $M_0$  isomorphically onto  $\overline{M}$ . Under the isomorphism of k-vector spaces  $M_0 \to \overline{M}$ , the pairing on  $M_0$  and the pairing on  $\overline{M}$  are identified because by choice of basis the pairing on  $M_0$  is a  $k^\times$ -multiple of the standard pairing. Furthermore, extending the pairing  $M_0 \times M_0 \to L$  by  $k[t]/(t^2)$ -bilinearity recovers the pairing on M. Using  $M_0 \simeq \overline{M}$ , we can also define  $\varphi^i_{M_0}: M^i_0 \to M_0$  to be the lift of  $\varphi^i_{\overline{M}}$  to  $M^i_0$ . It is compatible with the pairing on  $M_0$ . Note that it is not the same as  $\varphi^i_{M_0}|_{M^i_0}$ .

Our goal is to describe the set of strict equivalence classes of deformations M of  $\overline{M}$ , so by making these identifications it remains to study ways to lift  $\varphi^i_{\overline{M}}$  to a map  $\varphi^i_{M_0 \otimes k[t]/(t^2)} : M^i_0 \otimes k[t]/(t^2) \to M_0 \otimes k[t]/(t^2)$ . For  $n, n' \in M^i_0$  we may write

$$\varphi_M^i(n+tn') = \varphi_{M_0}^i(n) + t(\varphi_{M_0}^i(n') + \delta_i(n))$$

for some  $\sigma$ -semilinear  $\delta_i: M_0^i \to M_0$  which completely determines  $\varphi_M^i$ . It is clear that for  $n \in M_0^{i+1}$  we have  $\delta_i(n) = 0$  due to the relation  $\varphi_{M_0}^i(n) = p\varphi_{M_0}^{i+1}(n) = 0$ . Thus,  $\delta_i$  factors through  $M_0^i/M_0^{i+1}$ , and together the  $\delta_i$  define a  $\sigma$ -semilinear

$$\delta: \operatorname{gr}^{\bullet}(M_0) \to M_0.$$

Compatibility with the pairing says exactly

$$\langle \varphi_M^i(n+tn'), \varphi_M^j(m+tm') \rangle = \varphi_L^{i+j}(\langle n+tn', m+tm' \rangle)$$

for  $n, n' \in M_0^i$  and  $m, m' \in M_0^j$  and all i and j. Expanding and using the compatibility of the  $\varphi_{M_0}^i$  with the pairing, we see that it is necessary and sufficient that

$$\langle \delta_i(n), \varphi_{M_0}^j(m) \rangle + \langle \varphi_{M_0}^i(n), \delta_j(m) \rangle = 0 \tag{3.2.3.1}$$

for  $n \in M_0^i$  and  $m \in M_0^j$  and all i and j. As  $\overline{M} = \sum_i \varphi_{\overline{M}}^i(\overline{M}^i)$  and we defined  $\varphi_{M_0}^i$  to lift  $\varphi_{\overline{M}}^i$ , it follows that  $M_0 = \sum_i \varphi_{M_0}^i(M_0^i)$ . Furthermore, we have an isomorphism  $\varphi : \operatorname{gr}^{\bullet} M_0 \to M_0$ . This allows us to rewrite (3.2.3.1) as

$$\langle \delta' \varphi(n), \varphi(m) \rangle + \langle \varphi(n), \delta' \varphi(m) \rangle = 0$$

where  $\delta'$  is the k-linear composition of  $\varphi^{-1}$  with  $\delta$ . In other words,

$$\langle \delta' x, y \rangle + \langle x, \delta' y \rangle = 0$$

for all  $x, y \in M_0$ . Note that  $\delta'$  is compatible with the filtration, the pairing, and the  $k \otimes W$ -module structure. Denote the collection of all such  $\delta'$  by  $\operatorname{End}_{k \otimes W}(M_0, \langle \cdot, \cdot \rangle)$ : it is isomorphic to  $\mathfrak{sp}_r(k \otimes W)$  or  $\mathfrak{so}_r(k \otimes W)$ , which have dimension  $[K: \mathbb{Q}_p](\dim G - 1)$  over k.

**Lemma 3.2.3.3.** For such a choice of  $\delta'$ , we obtain a Fontaine-Laffaille module  $M \in \mathrm{MF}_{W,\mathrm{tor}}^f$  together with a pairing  $M \times M \to L$  as in Corollary 3.2.1.5.

*Proof.* This is just bookkeeping. First, observe that  $\sum_i \varphi_M^i(M^i)$  is a  $k[t]/(t^2)$ -module containing  $\varphi_{M_0}^i(M_0^i) = M_0$ . Thus it is M. It is immediate that the pairing is compatible with the filtration. We chose  $\delta'$  so that the pairing is compatible with the  $\varphi_M^i$ .

Of course, different  $\delta'$  may give isomorphic deformations of  $\overline{M}$ . Suppose that we are given  $\delta$  and  $\gamma$  such that the Fontaine-Laffaille modules they create are strictly equivalent as deformations of  $\overline{M}$ . We have shown that the underlying module, pairing, and filtration can be identified with the fixed data  $M = M_0 \otimes k[t]/(t^2)$ ,  $\langle \cdot, \cdot \rangle \otimes k[t]/(t^2)$ , and  $M_0^i \otimes k[t]/(t^2)$  so that the isomorphism reduces to the identity modulo t (by strictness). This means there exists an isomorphism  $\alpha: M_0 \to M_0$  compatible with the pairing, filtration, and module structure such that

$$(1+t\alpha)\left(\varphi_{M_0}^i(n) + t(\varphi_{M_0}^i(n') + \delta_i(n))\right) = \varphi_{M_0}^i(n) + t(\varphi_{M_0}^i(\alpha(n) + n') + \gamma_i(n)).$$

Simplifying, this is the condition that

$$\gamma_i(n) - \delta_i(n) = \alpha(\varphi_{M_0}^i(n)) - \varphi_{M_0}^i(\alpha(n)).$$

In other words,  $\delta, \gamma \in \operatorname{End}(M_0, \langle \cdot, \cdot \rangle)$  define the same deformation if and only if  $\gamma_i - \delta_i$  is of the form  $\alpha \circ \varphi_{M_0}^i - \varphi_{M_0}^i \circ \alpha$  for all i and some  $k \otimes W$ -linear  $\alpha : M_0 \to M_0$  that is compatible with the filtration, pairing, and module structures. Under the identification of  $\operatorname{End}_{k \otimes W}(M_0, \langle \cdot, \cdot \rangle)$  with the Lie algebra of a symplectic or orthogonal group valued in  $k \otimes W$ , these are the elements in the Lie algebra of the Borel subgroup corresponding to the filtration. (The assumption that the Fontaine-Laffaille weights for each  $\tau$  are pairwise distinct is what makes it a Borel subgroup.) This has dimension  $[K: \mathbb{Q}_p](\dim B - 1)$  as a k-vector space, where B is a Borel subgroup of G.

Finally, we must understand when  $\alpha$  and  $\beta$  satisfy

$$\alpha \circ \varphi_{M_0}^i - \varphi_{M_0}^i \circ \alpha = \beta \circ \varphi_{M_0}^i - \varphi_{M_0}^i \circ \beta.$$

This happens exactly when  $\alpha - \beta$  commutes with the  $\varphi^i_{M_0}$  (as well as being compatible with the filtration, pairing, and module structure). In other words,  $\alpha - \beta \in \operatorname{End}_{\operatorname{MF}_W}(M_0, \langle \cdot, \cdot \rangle)$ . But under  $T_{\operatorname{cris}}$ , this is identified with endomorphisms of  $\overline{\rho}$  preserving the pairing (not just up to a similitude factor), and in particular has dimension  $\dim_k H^0(\Gamma_K, \operatorname{ad}^0(\overline{\rho}))$ .

We can express this analysis as the exact sequence

$$0 \to \operatorname{End}_{\operatorname{MF}_W}(M_0, \langle \cdot, \cdot \rangle) \to \operatorname{Fil}^0\left(\operatorname{End}_{k \otimes W}(M_0, \langle \cdot, \cdot \rangle)\right) \to \operatorname{End}_{k \otimes W}(M_0, \langle \cdot, \cdot \rangle) \to D^{\operatorname{FL}}_{\overline{\rho}}(k[t]/(t^2)) \to 0.$$

In particular, we see that the space of deformations of  $\overline{\rho}$  to  $k[t]/(t^2)$  has dimension

$$[K:\mathbf{Q}_p](\dim G-1)-[K:\mathbf{Q}_p[(\dim B-1)+\dim_k H^0(\Gamma_K,\mathrm{ad}^0(\overline{\rho}))=[K:\mathbf{Q}_p]\dim_k \mathfrak{u}+\dim_k H^0(\Gamma_K,\mathrm{ad}^0(\overline{\rho}))$$

(since the Borel subgroup of the derived group of G has codimension-1 in a Borel subgroup of G). This completes the proof of Proposition 3.2.3.1.

# Chapter 4

# Minimally Ramified Deformations

Let  $\ell \neq p$  be primes and L be a finite extension of  $\mathbf{Q}_{\ell}$ . For a split connected reductive group G over the valuation ring  $\mathcal{O}$  of a p-adic field K with residue field k, consider a residual representation  $\overline{\rho}: \Gamma_L \to G(k)$ . The methods of Chapter 2 require a nice class of deformations of  $\overline{\rho}$ . If  $\overline{\rho}$  were unramified, the unramified deformation condition (Example 2.2.2.9) would often work for our purposes. The interesting case is when  $\overline{\rho}$  is ramified: we would like to define a deformation condition that "restricts the ramification" of lifts of  $\overline{\rho}$  so the resulting deformation condition is liftable, despite the fact that the unrestricted deformation condition for  $\overline{\rho}$  may not be liftable. To be precise, we require a deformation condition  $\mathcal{D}_{\overline{\rho}}$  such that  $\mathcal{D}_{\overline{\rho}}$  is liftable and has tangent space with dimension (at least) dim $_k H^0(\Gamma_L, \mathrm{ad}^0(\overline{\rho}))$ . We will obtain such a deformation condition after making an étale local extension of  $\mathcal{O}$ , which is harmless for our application.

In the case that  $G = \operatorname{GL}_m$ , the minimally ramified deformation condition defined in [CHT08, §2.4.4] works. We will generalize this to a minimally ramified deformation condition for symplectic and orthogonal similitude groups on m-dimensional spaces when  $p \geq m$ . The key inspiration comes from the arguments of [Tay08, §3], where a similar deformation condition for  $\operatorname{GL}_m$  is analyzed by studying deformations of a nilpotent element of  $\mathfrak{g}_k = \operatorname{Lie} G_k$ . In §4.1, we review facts about nilpotent orbits over algebraically closed fields of very good characteristic and find "nice" integral representatives N for such orbits. In §4.2, we study the centralizer scheme  $Z_G(N)$  over  $\mathcal{O}$  and show that it is smooth. We can reduce checking  $\mathcal{O}$ -smoothness to the problem of finding elements  $g \in Z_G(N)(\mathcal{O})$  such that  $g_k$  lies in any specified component of  $Z_{G_k}(N_k)/Z_{G_k}(N_k)^{\circ}$ . Then in §4.3 we define the notion of a "pure nilpotent" deforming  $N_k$  and study the space of such deformations. Finally in §4.4 and §4.5 we define the minimally ramified deformation condition using our study of nilpotents, first for a special class of tamely ramified representations and then in general.

The "minimally ramified" deformation condition is defined only for symplectic and orthogonal similitude groups, essentially because of an argument used in §4.5 to reduce the study of general representations to a special class of tamely ramified representations. We have adapted the argument of [CHT08, §2.4.4] to keep track of a symmetric or alternating pairing, so this is genuinely specific to symplectic or orthogonal groups. The arguments in earlier sections are phrased in such a way as to easily generalize to any split connected reductive  $\mathcal{O}$ -group G, given certain general facts about nilpotent orbits and centralizers of nilpotent elements. In particular, we use the following:

- (N1) for every nilpotent orbit of  $G(\overline{k})$  in  $\mathfrak{g}_{\overline{k}}$ , there exists  $N \in \mathfrak{g}$  such that both  $N_{\overline{K}} \in \mathfrak{g}_{\overline{K}}$  and  $N_{\overline{k}} \in \mathfrak{g}_{\overline{k}}$  "lie in that nilpotent orbit" (using the combinatorial parametrization of nilpotent orbits discussed in §4.1 to relate nilpotent orbits in  $\mathfrak{g}_{\overline{k}}$  and  $\mathfrak{g}_{\overline{K}}$ ) and such that  $Z_{G_k}(N_k)$  and  $Z_{G_K}(N_K)$  are smooth;
- (N2) for a suitable such  $N \in \mathfrak{g}$  and every component of  $Z_{G_{\overline{k}}}(N_{\overline{k}})$  there exists  $g \in Z_G(N)(\mathcal{O})$  such that  $g_k$  lies in that component (which will imply the flatness and hence smoothness of the scheme-theoretic centralizer  $Z_G(N)$  over  $\mathcal{O}$ );
- (N3) for tame  $\overline{\rho}$  and a tame inertial generator  $\tau$ , there is a way to pass between a nice class of deformations of  $N_k = \overline{\rho}(\tau)$  and deformations of  $\overline{\rho}$  as made precise in §4.4. In particular, this uses the exponential map (4.4.1.1) to convert between nilpotents and unipotents, as well as elements  $\Phi \in G(\mathcal{O})$  such that  $\mathrm{ad}_G(\Phi)N = qN$  for some specified  $q \in \mathbf{Z}$ .

We will check all of these assumptions for  $GL_m$ ,  $GSp_m$ , and  $GO_m$  when  $p \geq m$ , which suffices for our applications. It would be of interest to give uniform proofs, or at least to verify them for the exceptional groups (where there are just finitely many nilpotent orbits to check).

### 4.1 Representatives for Nilpotent Orbits

### 4.1.1 Algebraically Closed Fields

In this subsection, we let k be algebraically closed of characteristic  $p \geq 0$  and take G to be a connected reductive group over k with p good for G. (By convention, characteristic zero is good for all G.) Let  $\mathfrak{g} = \operatorname{Lie} G$ . For a nilpotent  $N \in \mathfrak{g}$ , the orbit  $O_N$  of N under G can be defined as the locally closed (reduced) image of the orbit map through N. It is a smooth locally closed subscheme of  $\mathfrak{g}$  by the closed orbit lemma, and is called a *nilpotent orbit* (for G). As G is connected, this orbit is irreducible. Let  $Z_G(N)$  be the scheme-theoretic centralizer of a nilpotent N, representing the functor

$$R \mapsto \{g \in G(R) : \operatorname{Ad}_G(g)N_R = N_R\}$$

for k-algebras R.

In good characteristic, the finite number of nilpotent orbits can be described by the Bala-Carter method as we review below. (More information can be found in [Jan04,  $\S4$ ], and a uniform proof without case-checking in small characteristic is due to Premet [Pre03].) To state this, we need to define some terminology. Let H be a connected reductive k-group with p good for H, and  $\mathfrak{h} = \text{Lie } H$ .

- A nilpotent  $N \in \mathfrak{h}$  is a distinguished nilpotent if each torus contained in  $Z_H(N)$  is contained in the center of H.
- For a parabolic  $P \subset H$  with unipotent radical U, the Richardson orbit associated to P is the unique nilpotent orbit of H with dense intersection with Lie U. Its intersection with Lie P is a single orbit under P.
- A parabolic subgroup  $P \subset H$  with unipotent radical U is a distinguished parabolic if  $\dim P/U = \dim U/\mathscr{D}(U)$ .

Bala and Carter classified nilpotent orbits when the characteristic is good. One can check that if p is good for G, it will also be good for any Levi factor of a parabolic subgroup of G.

**Fact 4.1.1.1.** If p is a good prime for G, the nilpotent orbits for G are in bijection with G(k)-conjugacy classes of pairs (L, P) where L is a Levi factor of a parabolic subgroup of G and P is a distinguished parabolic of L. The nilpotent orbit for G associated to (L, P) is the unique one meeting Lie(P) in its Richardson orbit for L.

**Example 4.1.1.2.** For  $G = GL_n$ , conjugacy classes of parabolic subgroups of  $GL_n$  are indexed by partitions  $n = n_1 + \ldots + n_r$ , with a Levi subgroup given by the product  $GL_{n_1} \times \ldots \times GL_{n_r}$  for the associated "standard" parabolic subgroup. The only distinguished parabolic subgroups of  $GL_{n_i}$  are the Borel subgroups. Taking into account conjugation, we conclude that conjugacy classes of nilpotents are in bijection with partitions of n. A representative for each orbit is given by a block matrix in Jordan canonical form with eigenvalues zero and blocks of size  $n_1, \ldots, n_r$ . This is worked out in detail in [Jan04, §4.1,4.4,4.8]. The nilpotent orbits are the same for  $SL_n$  [Jan04, §1.2].

The Bala-Carter data C for G is the set of G(k)-conjugacy classes of pairs (L, P) as above. It turns out C is independent of k in the sense that it can be described completely in terms of the root datum of G as follows. All Levi subgroups L of a parabolic k-subgroup Q of G are a single  $\mathcal{R}_{u,k}(Q)$ -orbit, so in Fact 4.1.1.1 we may restrict to one Q per G(k)-conjugacy class and one L per Q. We may pick L so that it contains a (split) maximal torus T. After conjugation by L(k), the distinguished parabolic subgroup  $P \subset L$  may be assumed to contain T as well. But we know that parabolic subgroups Q of G containing T are in bijection with parabolic subsets of  $\Phi(G,T)$  via  $Q \mapsto \Phi(Q,T)$ , so the possible Levi factors L of Q containing T are

described just in terms of the root datum. Likewise, parabolic subgroups P of L containing T are in bijection with parabolic subsets of  $\Phi(L,T)$ . If we can characterize the condition that P is distinguished just in terms of the root data, this would mean that the Bala-Carter data can be described solely in terms of the root data and so is completely combinatorial.

We do so by constructing a grading on the Lie algebra of a parabolic P. Pick a Borel subgroup  $B \subset G$  satisfying  $T \subset B \subset P$ . Let  $\mathfrak{t} = \operatorname{Lie}(T)$  and  $\Delta \subset \Phi = \Phi(L,T)$  be the set of positive simple roots determined by B. There is a unique subset  $I \subset \Delta$  such that  $P = BW_IB$  where  $W_I$  is the subset of the Weyl group generated by reflections with respect to roots in I. Define a group homomorphism  $f: \mathbf{Z}\Phi \subset \mathbf{Z}^{\Delta} \to \mathbf{Z}$  by specifying that on the basis  $\Delta$  we have

$$f(\alpha) = \begin{cases} 2 & \alpha \in \Delta - I, \\ 0 & \alpha \in I. \end{cases}$$

This function gives a grading on l = Lie(L):

$$\mathfrak{l}(i) = \bigoplus_{f(\alpha)=i} \mathfrak{l}_{\alpha} \quad \text{and} \quad \mathfrak{l}(0) = \left(\bigoplus_{f(\alpha)=0} \mathfrak{l}_{\alpha}\right) \oplus \mathfrak{t}$$

(sums indexed by  $\alpha \in \Phi$ ). With respect to this grading,

$$\operatorname{Lie} P = \bigoplus_{i \geq 0} \mathfrak{l}(i) \quad \text{and} \quad \operatorname{Lie} U = \bigoplus_{i > 0} \mathfrak{l}(i).$$

The condition that P is distinguished is equivalent to the condition that

$$\dim \mathfrak{l}(0) = \dim \mathfrak{l}(2) + \dim Z_L$$

by [Car85, Corollary 5.8.3] as p is good for L. But this condition depends only on the root datum. Thus the Bala-Carter data for G can be described in a manner independent of the choice of algebraically closed field.

**Definition 4.1.1.3.** For  $\sigma \in \mathcal{C}$ , let  $O_{\sigma} \subset \mathfrak{g}$  (or  $O_{k,\sigma}$  if it is necessary to specify the field) be the corresponding nilpotent orbit.

From the classification of nilpotent orbits over algebraically closed fields, it is known that the corresponding nilpotent orbits in characteristic zero and characteristic p have the same dimension.

**Example 4.1.1.4.** Let  $G = GL_n$  over an algebraically closed field k. For a partition  $n = n_1 + n_2 + \ldots + n_r$ , define  $d_i(\sigma)$  inductively by  $d_0(\sigma) = 0$  and  $d_i(\sigma) = d_{i-1}(\sigma) + \#\{j : n_j \geq i\}$ . The orbit closure  $\overline{O}_{\sigma}$  consists of nilpotents  $N \in \mathfrak{g}$  such that the  $(n+1-d_i(\sigma)) \times (n+1-d_i(\sigma))$  minors of  $N^i$  vanish for all  $i=1,\ldots,n$ .

**Remark 4.1.1.5.** We consider connected reductive groups, but natural groups like  $O_n$  are disconnected. The nilpotent orbits of  $O_n$  and  $SO_n$  are different but closely related; as explained in [Jan04, §1.12], the only change is that certain pairs of orbits for  $SO_n$  may become a single  $O_n$ -orbit.

#### 4.1.2 Representatives of Nilpotent Orbits

Now return the case when G is a split reductive group scheme with connected fibers over a complete discrete valuation ring  $\mathcal{O}$  with  $\mathfrak{g} = \text{Lie } G$ . Let k be the residue field of characteristic p > 0. Assume p is very good for  $G_k$ . For  $\sigma \in \mathcal{C}$ , we seek elements

$$N_{\sigma} \in \mathfrak{g} \text{ such that } (N_{\sigma})_k \in O_{\overline{k},\sigma} \text{ and } (N_{\sigma})_K \in O_{\overline{K},\sigma}.$$
 (4.1.2.1)

In the case  $G = GL_n$ , we will see that the standard representatives in  $G(\mathcal{O})$  for nilpotent orbits in Jordan canonical form satisfy this condition. We will also explicitly describe such  $N_{\sigma}$  for symplectic and orthogonal groups, as we will need this concrete description to analyze the centralizer  $Z_G(N)$  as an  $\mathcal{O}$ -scheme.

**Remark 4.1.2.1.** In general, one can build such an  $N_{\sigma}$  using the root system as follows. Let T be a split maximal torus of G. Pick a pair (L, P) consisting of a Levi factor L of a parabolic  $\mathcal{O}$ -subgroup of G and a fiber-wise distinguished parabolic  $\mathcal{O}$ -subgroup P of L corresponding to  $\sigma$  such that  $T \subset P$ . For any subset  $J \subset \Phi(L, T)$ , consider the element

$$N_J := \sum_{\alpha \in J} N_\alpha \in \mathfrak{g}$$

where  $N_{\alpha}$  is an  $\mathcal{O}$ -basis for the root line  $\mathfrak{g}_{\alpha} \subset \mathfrak{g}$ . For any  $\sigma = (L, P)$ , according to [SS70, III.4.29] there is some subset J of a positive system of roots in  $\Phi(G, T)$  with  $|J| < \dim(T)$  for which  $N_J$  lies in  $O_{\overline{k}, \sigma}$  and  $O_{\overline{K}, \sigma}$ . (The method used to prove [SS70, III.4.29] requires a stronger condition on p, as the Jacobson-Morozov theorem is used in the analysis of nilpotent orbits presented in [SS70]. There should be no difficulty in removing this assumption, but we don't treat this as our final results only apply to symplectic and orthogonal groups, for which we make direct constructions below.)

For  $GL_n$ , it is easy to describe an  $N_\sigma$  as in (4.1.2.1) in terms of the root data via Jordan canonical form:

Example 4.1.2.2. Let  $G = \operatorname{GL}_n$  with nilpotent orbit corresponding to the partition  $n_1 + n_2 + \ldots + n_r = n$ . For the diagonal torus T and standard upper triangular Borel subgroup B, let  $e_i$  be the character extracting the ith diagonal entry of the torus. The positive simple roots in  $\Phi(G,T)$  with respect to B are  $\{\alpha_i = e_i - e_{i+1} : i = 1, \ldots, n-1\}$ . For a partition  $n = n_1 + \ldots + n_r$ , the corresponding nilpotent orbit as in Example 4.1.1.2 corresponds to the standard upper triangular Borel subgroup  $B_L$  of  $L = \operatorname{GL}_{n_1} \times \operatorname{GL}_{n_2} \times \ldots \times \operatorname{GL}_{n_r}$ . Now  $\Phi(L,T)$  contains as its set  $\Delta_L$  of simple roots with respect to  $B_L$  those  $e_i - e_{i+1}$  with  $i \neq n_1, n_1 + n_2, \ldots n_r$  (in other words, i and i+1 lie in the "same block"). We consider

$$N = \sum_{\alpha \in \Delta_L} N_{\alpha}$$

where  $N_{\alpha}$  is a basis element for the root line  $\mathfrak{g}_{\alpha}$ . Identifying  $\mathfrak{g}_k$  with  $\operatorname{End}(k^n)$ ,  $N_k$  is the nilpotent matrix in Jordan canonical form whose blocks (in order) are of sizes  $n_1, n_2, \ldots, n_r$ . Thus, N satisfies (4.1.2.1).

For symplectic and orthogonal groups, explicitly describing the nilpotents constructed from Bala-Carter data is more complicated. We will sketch the approach, but not carry out the details because it turns out to be more convenient to construct and analyze representatives in these cases using a partition-based classification rather than via Bala-Carter data; this approach gives additional information about the centralizers that will be needed in §4.2.2.

**Example 4.1.2.3.** Consider the groups  $\operatorname{Sp}_m$  with m=2n, or  $\operatorname{SO}_m$  with m=2n or m=2n+1 (so  $\left[\frac{m}{2}\right]=n$ ) and assume  $p\neq 2$  and  $n\geq 2$ . We will first describe the Levi subgroups of parabolic subgroups [Jan04, §4.5]. Choose an integer  $0< n_0\leq m$  with  $n_0\equiv m\pmod 2$ . In the orthogonal case, assume that  $n_0\neq 2$ . Choose a partition  $2d_1+2d_2+\ldots+2d_r\geq 0$  of  $m-n_0$  into even parts. Choose a partition

$$\{1,\ldots,n\} = \overline{S}_0 \coprod \overline{S}_1 \coprod \ldots \coprod \overline{S}_r$$

where  $|\overline{S}_i| = d_i$  for  $1 \le i \le r$  and hence  $|\overline{S}_0| = [n_0/2]$ . For  $1 \le i \le r$ , let  $S_i = \overline{S}_i \cup (n + \overline{S}_i)$ . If m is even, let  $S_0 = \overline{S}_0 \cup (n + \overline{S}_i)$ , while if m is odd let  $S_0 = \overline{S}_0 \cup (n + \overline{S}_i) \cup \{0\}$ . Clearly  $\{S_i\}$  is a partition of  $\{1, 2, \ldots, 2n\}$  if m = 2n and of  $\{0, 1, 2, \ldots, 2n\}$  if m = 2n + 1.

Let  $\{v_i\}$  be the standard basis  $k^m$ . Define  $W_i = \operatorname{span}_{j \in \overline{S}_i} v_j$ ,  $\overline{W}_i = \operatorname{span}_{j \in \overline{S}_i} v_{n+j}$ , and  $V_i = \operatorname{span}_{j \in S_i} v_j$ . This gives an orthogonal direct sum decomposition

$$V_0 \oplus V_1 \oplus V_2 \ldots \oplus V_r = k^m$$

with respect to the standard pairings. For  $1 \leq i \leq r$ , we have  $V_i = W_i \oplus \overline{W}_i$  with subspaces  $W_i$  and  $\overline{W}_i$  in perfect duality via the chosen standard pairing, and an automorphism of  $W_i$  covariantly induces an automorphism of  $\overline{W}_i$  via "dual-inverse", giving an inclusion of  $GL(W_i)$  inside  $SO(V_i)$  or  $Sp(V_i)$ . The subgroup

$$L = \operatorname{Sp}(V_0) \times \operatorname{GL}(W_1) \times \ldots \times \operatorname{GL}(W_r)$$

is a Levi subgroup of a parabolic subgroup in  $Sp_m$ . The subgroup

$$L = SO(V_0) \times GL(W_1) \times \ldots \times GL(W_r)$$

is a Levi subgroup of a parabolic subgroup in  $SO_m$ . Up to conjugation, all Levi subgroups of parabolic subgroups arise in this way [Jan04, §4.5].

A distinguished parabolic subgroup of L is a direct product of distinguished parabolic subgroups of  $Sp(V_0)$  or  $SO(V_0)$  and each of the  $GL(W_i)$ . There is a unique conjugacy class of distinguished parabolic subgroups in  $GL(W_i)$ , namely the Borel subgroups. The conjugacy classes of distinguished parabolic subgroups in  $G_0 = Sp(V_0)$  or  $SO(V_0)$  are described in the tables of [Car85, §5.9]. There are several classes, and it will be much more convenient to describe them using an approach based on partitions.

Now we provide partition-based classifications for symplectic and orthogonal groups, similar to the classification for  $GL_m$  we have already seen in Example 4.1.1.2. We will use them to construct elements as in (4.1.2.1). Let  $G = \operatorname{Sp}_m$  with m = 2n, or  $G = \operatorname{O}_m$  with m = 2n or m = 2n + 1. As usual, we assume  $n \geq 2$ . Recall that  $\operatorname{Sp}_m$  and  $\operatorname{O}_m$  are defined using standard pairings on a free  $\mathcal{O}$ -module M of rank m. For m = 2n, the standard alternating pairing  $\varphi_{\text{std}}$  on  $\mathcal{O}^m$  is the one given by the block matrix

$$\begin{pmatrix} 0 & I_n' \\ -I_n' & 0 \end{pmatrix},$$

where  $I'_n$  denotes the anti-diagonal matrix with 1's on the diagonal. The standard symmetric pairing  $\varphi_{\text{std}}$  on  $\mathcal{O}^m$  is the one given by the matrix  $I'_m$ .

Remark 4.1.2.4. We chose to work with  $O_m$  instead of  $SO_m$ , as the classification is cleaner for  $O_m$ . The nilpotent orbits are almost the same for  $SO_m$ , except that certain nilpotent orbits of  $O_m$  (the ones where the partition contains only even parts) split into two  $SO_m$ -orbits [Jan04, Proposition 1.12] (conjugation by an element of  $O_m$  with determinant -1 carries one such orbit into the other).

**Definition 4.1.2.5.** Let  $\sigma$  denote a partition  $m = m_1 + m_2 + \ldots + m_r$  of m. It is admissible if

- every even  $m_i$  appears an even number of times when  $G = \mathcal{O}_m$ ;
- every odd  $m_i$  appears an even number of times when  $G = \operatorname{Sp}_m$ .

The admissible partitions of m are in bijection with nilpotent orbits of  $\operatorname{Sp}_m$  or  $\operatorname{O}_m$  over an algebraically closed field [Jan04, Theorem 1.6]. The corresponding orbit is the intersection of  $\mathfrak{g} \subset \mathfrak{gl}_m$  with the  $\operatorname{GL}_m$ -orbit corresponding to that partition of m. Note that  $\operatorname{GL}_m$ -orbit representatives in Jordan canonical form need not lie in  $\mathfrak{g}$ .

We will construct nilpotents together with a pairing, and then show how to relate the constructed pairing to the standard pairings used to define G. Let  $\epsilon=1$  in the case of  $\mathcal{O}_m$ , and  $\epsilon=-1$  in the case of  $\mathcal{Sp}_m$ .

**Definition 4.1.2.6.** Let  $d \geq 2$  be an integer. Define  $M(d) = \mathcal{O}^d$ , with basis  $v_1, \dots v_d$  and a perfect symmetric or alternating pairing  $\varphi_d$  such that

$$\varphi_d(v_i, v_j) = \begin{cases} (-1)^i, & i+j = d+1\\ 0, & \text{otherwise} \end{cases}$$

(alternating for even d, symmetric for odd d). Define a nilpotent  $X_d \in \text{End}(M(d))$  by  $X_dv_i = v_{i-1}$  for  $1 < i \le d$  and  $X_dv_1 = 0$ .

Similarly, define  $M(d,d) = \mathcal{O}^{2d}$  with basis  $v_1, \ldots, v_d, v_1', \ldots, v_d'$  and a perfect symmetric or alternating pairing  $\varphi_{d,d}$  by extending

$$\varphi_{d,d}(v_i,v_j) = \varphi_{d,d}(v_i',v_j') = 0 \quad \text{and} \quad \varphi_{d,d}(v_i,v_j') = \begin{cases} (-1)^i, & i+j=d+1\\ 0, & \text{otherwise} \end{cases}$$

Define a nilpotent  $X_{d,d} \in \text{End}(M(d,d))$  by  $X_{d,d}v_i = v_{i-1}$  and  $X_{d,d}v_i' = v_{i-1}'$  for  $1 < i \le d$ , and  $X_{d,d}v_1 = X_{d,d}v_1' = 0$ .

It is straightforward to verify the pairings are perfect and that  $X_d$  and  $X_{d,d}$  are skew with respect to the corresponding pairing. The pairing  $\varphi_{d,d}$  can be symmetric or alternating. Given an admissible partition  $\sigma: m = m_1 + m_2 + \ldots + m_r$ , we will construct a free  $\mathcal{O}$ -module of rank m with a symmetric or alternating pairing respectively and a nilpotent endomorphism respecting that pairing such that the Jordan block structure of nilpotent endomorphism on the geometric special fiber over Spec  $\mathcal{O}$  is given by  $\sigma$ . Let  $n_i(\sigma) = \#\{j: m_j = i\}$ .

• If  $G = \mathcal{O}_m$  then  $n_i(\sigma)$  is even for even i, so we can define

$$M_{\sigma} = \bigoplus_{i \text{ odd}} M(i)^{\oplus n_i(\sigma)} \oplus \bigoplus_{i \text{ even}} M(i,i)^{\oplus n_i(\sigma)/2}.$$

• If  $G = \operatorname{Sp}_m$  then  $n_i(\sigma)$  is even for odd i, so we can define

$$M_{\sigma} = \bigoplus_{i \text{ odd}} M(i,i)^{\oplus n_i(\sigma)/2} \oplus \bigoplus_{i \text{ even}} M(i)^{\oplus n_i(\sigma)}.$$

Let  $\varphi_{\sigma}$  and  $X_{\sigma}$  denote the pairing and nilpotent endomorphism defined by the pairing and nilpotent endomorphism on each piece using Definition 4.1.2.6. In all cases,  $M_{\sigma}$  is a free  $\mathcal{O}$ -module of rank m. For each  $\sigma$ , let  $G_{\sigma}$  be the automorphism scheme  $\operatorname{Aut}(M_{\sigma}, \varphi_{\sigma})$ , so for an algebraically closed field F over  $\mathcal{O}$  we have an isomorphism  $(G_{\sigma})_F \simeq G_F$  well-defined up to G(F)-conjugation by using F-linear isomorphisms  $(M_{\sigma}, \varphi_{\sigma}) \simeq (F^m, \varphi_{\text{std}})$ . Thus, the element  $X_{\sigma} \in \operatorname{Lie} G_{\sigma}$  goes over to a G(F)-orbit in  $\mathfrak{g}_F$ .

Let  $e_1, e_2, \dots e_m$  be the standard basis for  $\mathcal{O}^m$ . The elements  $e_i$  and  $e_{m+1-i}$  pair non-trivially under the standard pairing. When m = 2n + 1,  $e_{n+1}$  pairs non-trivially with itself under the standard pairing. We now relate the standard pairings to the pairings  $\varphi_{\sigma}$ .

**Lemma 4.1.2.7.** For all admissible partitions of m, the specializations of the  $X_{\sigma}$  at geometric points  $\xi$  of Spec  $\mathcal{O}$  constitute a set of representatives for the nilpotent orbits of  $G_{\xi}$ .

*Proof.* The set of admissible partitions of m is in bijection with the set of nilpotent orbits over any algebraically closed field [Jan04, Theorem 1.6]. The  $X_{\sigma}$  we constructed are integral versions of the representatives constructed in [Jan04, §1.7].

**Proposition 4.1.2.8.** Suppose that  $\sqrt{-1}$ ,  $\sqrt{2} \in \mathcal{O}^{\times}$ . Then  $\varphi_{\sigma}$  is equivalent to the standard pairing over  $\mathcal{O}$ . There exists an  $\mathcal{O}$ -basis  $\{v_i\}$  of  $M_{\sigma}$  with respect to which the pairing is given by  $\varphi_{\sigma}$  and  $X_{\sigma}$  satisfies the condition in (4.1.2.1).

Proof. The standard pairings are very similar to  $\varphi_{\sigma}$ . In the case of  $\operatorname{Sp}_m$ , each basis vector pairs trivially against all but one other basis vector, with which it pairs as  $\pm 1$ . So after reordering the basis,  $\varphi_{\sigma}$  is the standard pairing. The case of  $\operatorname{O}_m$  is slightly more complicated. Let  $\sigma: m = m_1 + m_2 + \ldots + m_r$  be an admissible partition. The construction of  $M_{\sigma}$  and  $\varphi_{\sigma}$  gives a basis  $\{v_{i,j}\}$  where  $1 \leq i \leq r$  and  $1 \leq j \leq m_i$ . From the construction of  $\varphi_{\sigma}$ , we see that  $v_{i,j}$  pairs trivially against all basis vectors except for  $v_{i,m_i+1-j}$ . So as long as  $2j \neq m_i + 1$ , we obtain a pair of basis vectors which are orthogonal to all others and which pair to  $\pm 1$ . For each odd  $m_i$ , the vector  $v_{i,(m_i+1)/2}$  pairs non-trivially with itself. The standard pairing with respect to the basis  $e_i$  has such a vector only when m = 2n + 1 and then only for one  $e_i$ .

We must change the basis over  $\mathcal{O}$  so that  $\varphi_{\sigma}$  becomes the standard symmetric pairing. Let  $v = v_{i,(m_i+1)/2}$  and  $v' = v_{j,(m_j+1)/2}$  be two distinct vectors which pair non-trivially with themselves. In particular,  $\varphi_{\sigma}(v,v) = (-1)^{(m_i+1)/2} := \eta$  and  $\varphi_{\sigma}(v',v') = (-1)^{(m_j+1)/2} := \eta'$ . Define

$$w = \frac{\sqrt{\eta}v - \sqrt{-\eta'}v'}{\sqrt{2}}$$
 and  $w' = \frac{\sqrt{\eta}v + \sqrt{-\eta'}v'}{\sqrt{2}}$ .

Then we see that  $\varphi_{\sigma}(w, w) = 0 = \varphi_{\sigma}(w', w')$  and  $\varphi_{\sigma}(w, w') = 1$ . Making this change of variable over  $\mathcal{O}$  (which requires  $\sqrt{-1}, \sqrt{2} \in \mathcal{O}^{\times}$ ), we have reduced the number of basis vectors which pair non-trivially with themselves by two, and produced a new pair of basis vectors orthogonal to the others and which pair to 1. By induction, we may therefore pick a basis  $v'_1, \ldots, v'_m$  for which at most one basis vector pairs non-trivially with itself under  $\varphi_{\sigma}$ . After re-ordering, we may further assume that  $\varphi_{\sigma}(v'_i, v'_j) = 0$  unless i + j = m + 1, in which case  $\varphi_{\sigma}(v'_i, v'_j) = \pm 1$ . Suppose j = m + 1 - i. If  $i \neq j$ , by scaling  $v'_i$  we may assume that  $\varphi_{\sigma}(v'_i, v'_j) = 1$ . If i = j, we already know that  $\varphi_{\sigma}(v'_i, v'_j) = 1$ . With respect to this basis,  $\varphi_{\sigma}$  is the standard pairing.  $\square$ 

### 4.2 Smoothness of Centralizers of Pure Nilpotents

Keeping the notation of §4.1.2, we next study the centralizer  $Z_G(N_\sigma)$  in more detail where  $N_\sigma \in \mathfrak{g}$  is an element satisfying (4.1.2.1). In particular, this centralizer will be shown to be smooth when G is symplectic or orthogonal. We first review the known theory over fields, and then develop and apply a technique to deduce smoothness over  $\mathcal{O}$  (i.e.  $\mathcal{O}$ -flatness) from the known smoothness in the field case.

### 4.2.1 Centralizers over Fields

In this section, let k be an algebraically closed field, G be a connected reductive group over k, and N a nilpotent element of  $\mathfrak{g} = \text{Lie } G$ . As the formation of the scheme-theoretic centralizer commutes with base change, smoothness results for  $Z_G(N)$  over k will imply such results over general fields (not necessarily algebraically closed).

The group scheme  $Z_G(N)$  is the fiber over  $0 \in \mathfrak{g}$  of the composition

$$G \stackrel{\mathrm{Ad}_G}{\longrightarrow} \mathrm{GL}(\mathfrak{g}) \stackrel{T \mapsto TN - N}{\longrightarrow} \mathfrak{g}.$$

Hence Lie  $Z_G(N)$  is the kernel of

$$\mathfrak{g} \stackrel{\mathrm{ad}_{\mathfrak{g}}}{\longrightarrow} \mathrm{End}(\mathfrak{g}) \stackrel{T \mapsto TN}{\longrightarrow} \mathfrak{g}$$

which is the Lie algebra centralizer  $\mathfrak{z}_{\mathfrak{g}}(N)$ . Recall that  $O_N$  denotes the smooth locally closed orbit  $G.N \subset \mathfrak{g}$ .

**Lemma 4.2.1.1.** The following are equivalent:

- 1. The centralizer  $Z_G(N)$  is smooth.
- 2. The map  $ad(N): T_e(G) \to T_N(O_N)$  is surjective.
- 3. The orbit map  $\mu_N: G \to O_N$  is smooth.

Proof. The tangent space of  $Z_G(N)$  at e is the kernel of  $d\mu_N : T_e(G) \to T_N(O_N)$ , so  $Z_G(N)$  is smooth if and only if this kernel has the same dimension as  $Z_G(N)$ . As G and  $O_N$  are smooth and we have  $\dim G - \dim Z_G(N) = \dim O_N$ , dimension considerations force  $d\mu_N$  to be surjective when  $Z_G(N)$  is smooth. Thus (1) implies (2). To check that a morphism between smooth schemes is smooth, it suffices to check the map is surjective on tangent spaces. We need only check at the identity because of translations, so (2) implies (3). As  $Z_G(N)$  is the fiber over  $N \in O_N$ , (3) implies (1).

**Remark 4.2.1.2.** In references using the language of varieties rather than schemes (such as [Jan04]),  $Z_G(N)$  is usually *defined* to be reduced (via its geometric points) and hence smooth, so the content of the analogue of Lemma 4.2.1.1 in such references is that the variety  $Z_G(N)$  has Lie algebra  $\mathfrak{z}_{\mathfrak{g}}(N)$ ; saying exactly that this variety agrees with the scheme-theoretic centralizer.

In a wide range of situations, all nilpotent centralizers are smooth. A direct calculation shows that this holds for  $G = GL_n$  (see [Jan04, §2.3]), and a criterion of Richardson leverages this to many other cases:

**Proposition 4.2.1.3.** Suppose G is a smooth closed subgroup of GL(V) and there exist a subspace  $W \subset \mathfrak{gl}(V)$  such that  $\mathfrak{gl}(V) = \mathfrak{g} \oplus W$  and  $[\mathfrak{g}, W] \subset W$ . Then  $Z_G(N)$  is smooth for any nilpotent  $N \in \mathfrak{g}$ .

*Proof.* Let  $\widetilde{\mathcal{O}}_N$  be the orbit of N under  $\mathrm{GL}(V)$  in  $\mathfrak{gl}(V)$ , and  $O_N$  the orbit of N under G. We know that  $T_N(\widetilde{\mathcal{O}}_N) = [\mathfrak{gl}(V), N]$  by Lemma 4.2.1.1(2) and the known smoothness of the centralizer for  $\mathrm{GL}(V)$ . The decomposition  $\mathfrak{gl}(V) = \mathfrak{g} \oplus W$  then implies that

$$T_N(\widetilde{\mathcal{O}}_N) = [\mathfrak{g}, N] + [W, N].$$

But  $[W,N] \subset W$ , so  $T_N(\widetilde{\mathcal{O}}_N) \cap \mathfrak{g} \subset [\mathfrak{g},N] \subset T_N(O_N)$ . On the other hand,  $T_N(O_N)$  is obviously inside  $T_N(\widetilde{\mathcal{O}}_N)$  and  $\mathfrak{g}$ . This implies that

$$[\mathfrak{g}, N] = T_N(O_N).$$

Then Lemma 4.2.1.1(2) gives smoothness.

The criterion in Proposition 4.2.1.3 automatically holds in characteristic 0 if G is reductive, because  $\mathfrak{g}$  has a complement as a  $\mathfrak{g}$ -representation in  $\mathfrak{gl}(V)$ ; of course, this is not interesting because the smoothness conclusion holds for all group schemes of finite type over a field of characteristic zero. In characteristic p > 0, the criterion in Proposition 4.2.1.3 can checked in many cases [Jan04, §2.6], such as:

- for  $SL_n$  when  $p \nmid n$  using the embedding  $SL(V) \hookrightarrow GL(V)$  and taking  $W = k \cdot id_V$ ;
- for a simple adjoint exceptional groups G when p is good for G;
- for  $SO(V, \varphi)$  or  $Sp(V, \varphi)$  when  $p \neq 2$  using the natural embeddings into GL(V) and taking  $W = \{f \in \mathfrak{gl}(V) : f^* = f\}$ .

(Here the non-degenerate pairing  $\varphi$  defines an adjoint  $f^*$  for any  $f \in \mathfrak{gl}(V) = \operatorname{End}(V)$  via the requirement  $\varphi(fv, w) = \varphi(v, f^*w)$ .)

The same techniques work to prove the same result  $GSp_{2n}$  and  $GO_n$ ; there does not appear to be a reference in the literature, so we now give a proof. Let V be a vector space over k with non-degenerate bilinear form  $\varphi: V \times V \to k$  that is symmetric or alternating. Suppose  $p \neq 2$ .

**Proposition 4.2.1.4.** If  $p \nmid \dim(V)$ , the group  $G = GO(V, \varphi)$  (respectively  $G = GSp(V, \varphi)$ ) with the natural inclusion into GL(V) admits a decomposition

$$\mathfrak{gl}(V) = \mathfrak{g} \oplus M$$

with  $[\mathfrak{g}, M] \subset M$ , so the G-centralizer of any nilpotent  $N \in \mathfrak{g}$  is smooth.

*Proof.* Let's recall why  $\operatorname{tr}(f^*) = \operatorname{tr}(f)$  for any  $f \in \mathfrak{gl}(V) = \operatorname{End}(V)$ . Fix a basis of V Representing the pairing  $\varphi$  as an (invertible) matrix J and letting A and  $A^*$  be the matrices for f and  $f^*$ , the identity  $\varphi(fv,w) = \varphi(v,f^*w)$  says

$$v^T A^T J w = v^T J A^* w$$

for all  $v, w \in V$ . This implies that  $A^* = J^{-1}A^TJ$ , which has the same trace as A.

We can express the Lie algebra as  $\mathfrak{g} = \{ f \in \mathfrak{gl}(V) : f + f^* \in k \cdot \mathrm{Id} \}$ . As  $p \nmid \dim(V)$ , we can define a complement to the scalar matrices by requiring the trace to be 0: define

$$M = \{ f \in \mathfrak{gl}(V) : \operatorname{tr}(f) = 0 \text{ and } f = f^* \}.$$

For  $f \in \mathfrak{g}$ , define  $\lambda_f \in k$  by  $f + f^* = \lambda_f$  Id. We compute that

$$\mathfrak{g} \cap M = \{ f \in \mathfrak{gl}(V) : 2f = \lambda_f \text{ Id} \text{ and } \operatorname{tr}(f) = 0 \} = \{ 0 \}.$$

On the other hand,  $\mathfrak{g} + M = V$ : the scalar matrices are in  $\mathfrak{g}$  (G is a similitude group) and if  $\operatorname{tr}(f) = 0$  then  $\frac{f - f^*}{2} \in \mathfrak{g}$  and  $\frac{f + f^*}{2} \in M$ . Thus we have a decomposition

$$\mathfrak{gl}(V) = \mathfrak{g} \oplus M.$$

For  $f \in \mathfrak{g}$  and  $g \in M$ , we compute that  $\operatorname{tr}([f,g]) = 0$  and

$$(fg - gf)^* = g^*f^* - f^*g^* = g(-f + \lambda_f \operatorname{Id}) - (-f + \lambda_f \operatorname{Id})g = fg - gf.$$

Therefore  $[\mathfrak{g}, M] \subset M$  as desired. We now apply Proposition 4.2.1.3.

**Remark 4.2.1.5.** When  $Z_G(N)$  is smooth over k and p is good for G, the dimension of the centralizer is independent of k (given the combinatorial classification of the G-orbit of N). This is apparent as the classification of nilpotent orbits and their dimensions are independent of k.

**Remark 4.2.1.6.** Additional analysis gives a broader criterion:  $Z_G(N)$  is smooth for a connected reductive group G provided the derived group G' is simply connected, p is good for G, and there exists a G-invariant non-degenerate bilinear form on  $\mathfrak{g}$  [Jan04, §2.9].

### 4.2.2 Centralizers in Classical Cases

For later use, we now explicitly describe the centralizer  $Z_G(N)$  for symplectic and orthogonal groups over an algebraically closed field k of characteristic  $p \neq 2$  and for suitable N. Let  $G = \mathcal{O}_m$  or  $G = \operatorname{Sp}_m$  with  $m \geq 4$ , acting on  $M = k^m$  preserving the standard pairing  $\varphi$ . Choose  $\epsilon = 1$  in the case of  $\mathcal{O}_m$  and  $\epsilon = -1$  in the case of  $\operatorname{Sp}_m$ . Let  $\sigma$  be an admissible partition  $m = m_1 + \ldots + m_r$ . We will work with  $(M_{\sigma} = \mathcal{O}^m, \varphi_{\sigma}, X_{\sigma})$  as defined above Lemma 4.1.2.7, so  $\operatorname{Aut}(M_{\sigma}, \varphi_{\sigma})_k \simeq G$ . Write X to denote the nilpotent  $(X_{\sigma})_k$  with  $\sigma$  fixed, and M to denote  $(M_{\sigma})_k = k^m$ . We know there exist vectors  $v_1, \ldots, v_r \in M$  such that

$$v_1, Xv_1, \dots, X^{m_1-1}v_1, v_2, Xv_2, \dots, X^{m_r-1}v_r$$

is a basis for M. Furthermore,  $X^{d_i}v_i=0$  for  $i=1,\ldots r$ , and the pairing between basis elements is given as near the start of the proof of Proposition 4.1.2.8. In particular, each  $v_i$  pairs non-trivially with only one other basis vector  $X^{d_i-1}v_{i^*}$ , for some  $i^* \in \{1,\ldots,r\}$ . (This is completely spelled out in [Jan04, §1.11].)

To understand the G-centralizer of X, we construct an associated grading of V as in [Jan04, §3.3,3.4]. This is motivated by the Jacobson-Morosov theory of  $\mathfrak{sl}_2$ -triples over a field of sufficiently large characteristic, but for symplectic and orthogonal groups it is constructed by hand in characteristic  $p \neq 2$  below.

**Remark 4.2.2.1.** Every nilpotent X gives a filtration of V defined by  $\mathrm{Fil}^i = \ker(X^i)$ . For  $\mathrm{GL}_n$ , this is a good filtration and is used in [CHT08] to define the minimally ramified deformation condition for  $\mathrm{GL}_n$ . However, this filtration need not be isotropic with respect to the pairing, so we will construct a nicer grading associated to X.

**Remark 4.2.2.2.** The motivation for the grading to be associated to X comes from the Jacobson-Morosov theorem. Let  $\mathfrak g$  be the Lie algebra of a reductive group G over an algebraically closed field of characteristic p, where either p=0 or p>3(h-1) where h is the Coxeter number of G. An  $\mathfrak{sl}_2$ -triple is triple of non-zero elements  $H,X,Y\in\mathfrak g$  such that

$$[H, X] = 2X, \quad [H, Y] = -2Y, \quad [X, Y] = H.$$

(They span a subalgebra isomorphic to  $\mathfrak{sl}_2$ ). The Jacobson-Morosov theorem says every non-zero nilpotent X can be extended to a  $\mathfrak{sl}_2$ -triple. The advantage is that we understand the representation theory of  $\mathfrak{sl}_2$  in characteristic zero: the irreducible representations are the representation on  $L_d = \operatorname{Sym}^d k^2$ . In positive characteristic there are differences from the characteristic zero case in dimension greater than p, but the assumption that p > 3(h-1) ensures that these problems are not relevant for our purposes. Viewing the adjoint representation as a representation of  $\mathfrak{sl}_2$ , it decomposes as a direct sum. On a summand  $L_d$ , X and Y act as raising and lowering operators, and H has eigenvalues  $d, d-2, \ldots, -d$ . The one-dimensional eigenspaces of H give a grading on the Lie algebra: the ith graded piece is

$$\{g\in \mathfrak{g}: H.g=i\cdot g\}.$$

**Definition 4.2.2.3.** Let M(s) be the span of  $X^j v_i$  for all i and j such that  $s = 2j + 1 - d_i$ . We set  $M^{(s)} = \bigoplus_{t > s} M(t)$ , and also define N(s) to be the span of  $\{v_i : v_i \in M(s)\}$ .

**Example 4.2.2.4.** Take  $G = O_8$  and  $\sigma$  to be the admissible partition 1 + 2 + 2 + 3. There are vectors  $v_1, v_2, v_3, v_4 \in k^8 = M$  such that

$$v_1, v_2, Xv_2, v_3, Xv_3, v_4, Xv_4, X^2v_4$$

form a basis for M. Then  $M(-2) = \operatorname{Span}_k(v_4)$ ,  $M(-1) = \operatorname{Span}_k(v_2, v_3)$ ,  $M(0) = \operatorname{Span}_k(v_1, Xv_4)$ ,  $M(1) = \operatorname{Span}_k(Xv_2, Xv_3)$ , and  $M(2) = \operatorname{Span}_k(X^2v_4)$ . Furthermore,  $N(0) = \operatorname{Span}_k(v_1)$ ,  $N(-1) = \operatorname{Span}_k(v_2, v_3)$ , and  $N(-2) = \operatorname{Span}_k(v_4)$ . The nilpotent X raises the degree by 2, and N(s) plays the role of a space of "lowest-weight vectors" inside M(s) with respect to the operator X.

We now record some elementary properties of the preceding construction; all are routine to check, and may be found in [Jan04, §3.4]. We have that  $M = \bigoplus_{s} M(s)$ , and

$$v_i \in M(-(d_i-1)), Xv_i \in M(-(d_i-1)+2), \dots, X^{d_i-1}v_r \in M(d_r-1).$$

Furthermore, we know  $XM(s) \subset M(s+2)$  and  $M(s) = XM(s-2) \oplus N(s)$  for  $s \leq 0$ .

The dimension of M(s) is  $m_s(\sigma) := \{j : d_j - 1 \ge |s|\}$ . The dimension of N(s) equals  $n_s(\sigma) := m_{s+1}(\sigma) - m_s(\sigma)$ . Furthermore, the pairing  $\varphi$  interacts well with the grading: a computation with basis elements gives that

$$\varphi(M(s), M(t)) \neq 0 \implies s + t = 0.$$

In particular,  $(M^{(r)})^{\perp} = M^{(-r)}$ .

The above grading on M corresponds to the one-parameter subgroup  $\lambda: \mathbf{G}_m \to G$  for which the action of  $t \in \mathbf{G}_m$  on M(s) is given by scaling by  $t^s$ . The dynamic method (see [CGP15, §2.1, Proposition 2.2.9]) associates to  $\lambda$  a parabolic subgroup  $P_G(\lambda)$  with Levi  $Z_G(\lambda)$ . Define  $C_X$  and  $U_X$  to be the scheme-theoretic intersections

$$C_X = Z_G(X) \cap Z_G(\lambda) = \{g \in Z_G(X) : gM(i) = M(i) \text{ for all } i\}$$
  
 $U_X = Z_G(X) \cap U_G(\lambda) = \{g \in Z_G(X) : (g-1)M^{(i)} \subset M^{(i+1)} \text{ for all } i\}.$ 

Fact 4.2.2.5. The group-scheme  $Z_G(X)$  is a semi-direct product of  $C_X$  and the smooth connected unipotent subgroup  $U_X$ .

This is [Jan04, Proposition 3.12]. The existence of  $\lambda$  and this decomposition is not specific to symplectic and orthogonal groups [Jan04, Proposition 5.10].

We finally give a concrete description of  $C_X$ . We first define a pairing on N(s). Recall that the space N(s) of "lowest weight vectors" in M(s) has basis  $\{v_i : 1 - d_i = s\}$ . We define a pairing on N(s) by

$$\psi_s(v, w) = \varphi(v, X^{1-s}w).$$

A direct calculation shows that  $\psi_s$  is non-degenerate and that  $\psi_s$  is symmetric if  $(-1)^{1-s} = -\epsilon$  and is alternating if  $(-1)^{1-s} = \epsilon$  [Jan04, §3.7].

A point of  $C_X$  preserves the grading on M, and since it commutes with "raising operator" X its action on M is determined by its action on the space N(s) of "lowest weight vectors" in M(s), the following fact is no surprise.

Fact 4.2.2.6. There is an isomorphism of algebraic groups

$$C_X \simeq \prod_{s \le -1} \underline{\operatorname{Aut}}(N(s), \psi_s)$$

This is [Jan04, §3.8 Proposition 2, 3].

**Example 4.2.2.7.** Let  $G = \operatorname{Sp}_m$ . Unraveling when  $\psi_s$  is symmetric or alternating, we see that

$$C_X \simeq \prod_{s < -1; s \text{ even}} O(N(s), \psi_s) \times \prod_{s < -1; s \text{ odd}} \operatorname{Sp}(N(s), \psi_s).$$

The symplectic factors are connected, while the orthogonal factors have two connected components. The connected components of  $Z_G(X)$  are the same as those for  $C_X$  by Fact 4.2.2.5. There are  $2^t$  of them, where t is the number of even s for which  $N(s) \neq 0$ .

**Example 4.2.2.8.** Let  $G = \mathcal{O}_m$ . We likewise see that

$$C_X \simeq \prod_{s \le -1; s \text{ odd}} O(N(s), \psi_s) \times \prod_{s \le -1; s \text{ even}} \operatorname{Sp}(N(s), \psi_s).$$

The connected components of  $Z_G(X)$  are the same as those for  $C_X$  by Fact 4.2.2.5. There are  $2^t$  of them, where t is the number of odd s for which  $N(s) \neq 0$ .

Now suppose that  $G = SO_m$ . The elements X we considered in this section are representatives for some of the nilpotent orbits of  $SO_m$ . The group  $C_X$  has the same structure as for  $G = O_m$ , except we require that the overall determinant be 1; this has  $2^{t-1}$  connected components. Though  $SO_m$  has more nilpotent orbits than  $O_m$ , according to Remark 4.1.2.4 their representatives are conjugate by an element of  $O_m(k) - SO_m(k)$  to the representatives constructed in Proposition 4.1.2.8. But such a transformation is just a change of basis preserving the pairing, so our analysis in this section applies. Thus, we have an explicit description of the component group of  $Z_G(X)$  in the  $SO_m$  case as well.

Remark 4.2.2.9. Assume  $\sqrt{-1}$ ,  $\sqrt{2} \in \mathcal{O}^{\times}$ . By using the element  $N_{\sigma} \in \mathfrak{g}$  constructed in Proposition 4.1.2.8 (which came with the data of a basis  $\{v'_i\}$  for a free  $\mathcal{O}$ -module M of rank m), much of the discussion in this section continues to hold if we work over  $\mathcal{O}$ : in particular, there is no problem defining the grading M(s) or cocharacter  $\lambda : \mathbf{G}_m \to G$  over  $\mathcal{O}$ . Furthermore, the dynamic construction in [CGP15, §2.2] is carried out over rings.

Suppose q is a square in  $\mathcal{O}^{\times}$ . For use in the proof of Proposition 4.4.2.3, we need the existence of an element  $\Phi \in G(\mathcal{O})$  such that  $\mathrm{ad}_G(\Phi)N_{\sigma} = qN_{\sigma}$ . If  $\alpha^2 = q$ , taking  $\Phi = \lambda(\alpha)$  would work:  $\Phi$  would scale  $N_{\sigma}^{j}v_{i} \in M(s)$  by  $\alpha^{s}$ , and  $N_{\sigma}$  increases the degree by 2.

This  $\Phi$  is a version for symplectic and orthogonal groups of the diagonal matrix denoted  $\Phi(\sigma, a, q)$  whose diagonal entries are increasing powers of q used in [Tay08, §2.3]. There it is checked that  $\mathrm{ad}_G(\Phi(\sigma, a, q))N_\sigma = qN_\sigma$  where  $N_\sigma$  is the nilpotent representative in Jordan canonical form considered in Example 4.1.2.2 for the partition  $\sigma$  of m.

### 4.2.3 Checking Flatness over a Dedekind Base

We want to analyze smoothness of centralizers in the relative setting (especially over Spec  $\mathcal{O}$ ). If  $Z_G(N_\sigma) \to \operatorname{Spec} \mathcal{O}$  is flat and the special and generic fibers are smooth then  $Z_G(N)$  is smooth over  $\mathcal{O}$ . The following lemma gives a way to check that a morphism to a Dedekind scheme is flat.

**Lemma 4.2.3.1.** Let  $f: X \to S$  be finite type for a connected Dedekind scheme S. Then f is flat provided the following all hold:

- 1. for each  $s \in S$ ,  $X_s$  is reduced and non-empty;
- 2. for each  $s \in S$ ,  $X_s$  is equidimensional with dimension independent of s;
- 3. there are sections  $\{\sigma_i \in X(S)\}$  to f such that for every irreducible component of a fiber above a closed point, there is a section  $\sigma_i$  which meets the fiber only in that component.

**Remark 4.2.3.2.** This lemma is a modification of [GY03, Proposition 6.1] to allow several connected components in the fibers.

*Proof.* It suffices to prove the result when  $S = \operatorname{Spec}(A)$  for A a discrete valuation ring with uniformizer  $\pi$ . Let  $X_{\eta}$  be the generic fiber and  $X_s$  the special fiber. Consider the schematic closure  $i: X' \hookrightarrow X$  of the generic fiber. The scheme X' is flat over  $\operatorname{Spec}(A)$  since flatness is equivalent to being torsion-free over a discrete valuation ring, and there is an exact sequence

$$0 \to J \to \mathcal{O}_X \to \iota_* \mathcal{O}_{X'} \to 0 \tag{4.2.3.1}$$

where J is a coherent sheaf killed by a power of  $\pi$ . We will show that i is an isomorphism by analyzing the special fiber.

First, we claim that the dimension of each irreducible component on the special fiber of X' is the same as the dimension of the equidimensional  $X'_{\eta} = X_{\eta}$ . We will get this from flatness of X'. The generic fiber of X' is  $X_{\eta}$ , which is equidimensional and non-empty by hypothesis. Furthermore, X' is the union of the closures  $Z_i$  of the reduced irreducible components  $X_{\eta,i}$  of  $X_{\eta}$ , and each  $Z_i$  is A-flat with integral  $\eta$ -fiber, hence integral. We just need to analyze the dimension of irreducible components of  $(Z_i)_s$  when  $(Z_i)_s \neq \emptyset$ . Since  $Z_i$  is integral, we can apply [Mat89, Theorem 15.1, 15.5] to such  $Z_i$  to conclude that the dimension of each irreducible component of the special fiber of X' is the same as the dimension of the generic fiber.

Observe that the sections  $\sigma_i$  factor through the closed subscheme  $X' \subset X$ , as we can check this on the generic fiber since X' is A-flat. Thus, X' meets every irreducible component of  $X_s$  away from the other irreducible components of  $X_s$ , and so we would have that  $|X_s'| = |X_s|$  if  $X_s'$  is equidimensional of the same dimension as the equidimensional  $X_s$ . We have shown the dimension of any irreducible component in  $X_s'$  is the same dimension as the common dimension of irreducible components of the generic fiber  $X_\eta$  of X'. By hypothesis, the dimension of any irreducible component of the generic fiber of X is the same as the dimension of any irreducible component of  $X_s$  is the same as the dimension of each irreducible component of  $X_s$ , giving that  $|X_s'| = |X_s|$ . As  $X_s$  is reduced, this forces  $i_s: X_s' \hookrightarrow X_s$  to be an isomorphism.

Now tensoring (4.2.3.1) with the residue field of A gives an exact sequence

$$0 \to J/\pi J \to \mathcal{O}_{X,s} \to \imath_* \mathcal{O}_{X',s} \to 0$$

because  $\mathcal{O}_{X'}$  is A-flat. Therefore  $J/\pi J=0$ . Hence  $J=\pi J=\pi^2 J=\ldots=\pi^n J=0$  for n large, so X=X' is flat over A.

Corollary 4.2.3.3. In the situation of the lemma, if the fibers are also smooth then X is smooth.

*Proof.* For a flat morphism of finite type between Noetherian schemes, smoothness of all fibers is equivalent to smoothness of the morphism.  $\Box$ 

#### 4.2.4 Smooth Centralizers

We now return the case when G is a split reductive group scheme with connected fibers over a discrete valuation ring  $\mathcal{O}$  with uniformizer  $\pi$  and residue field k of very good characteristic p > 0. Let  $T \subset G$  be a fiber-wise maximal split  $\mathcal{O}$ -torus, and G' denote the derived group of G over  $\mathcal{O}$ . Denote the field of fractions of  $\mathcal{O}$  by K, and set  $\mathfrak{g} := \text{Lie } G$ . Suppose we are given  $N = N_{\sigma} \in \mathfrak{g}$ , an integral representative for the nilpotent orbit on geometric fibers corresponding to  $\sigma \in \mathcal{C}$  as in (4.1.2.1). Proposition 4.1.2.8 provides such N in symplectic and orthogonal cases when  $\sqrt{-1}$ ,  $\sqrt{2} \in \mathcal{O}^{\times}$ . We wish to show that the  $Z_G(N)$  is smooth over  $\mathcal{O}$  in many situations. An obviously necessary condition is that

$$Z_{G_K}(N_K)$$
 and  $Z_{G_k}(N_k)$  are smooth. (4.2.4.1)

Remark 4.2.4.1. Some assumption on N is essential. Otherwise  $N_{\overline{K}}$  and  $N_{\overline{k}}$  can lie in "different" nilpotent orbits (in terms of the combinatorial characteristic-free classification of geometric orbits), and so  $Z_{G_K}(N_K)$  and  $Z_{G_k}(N_k)$  could have different dimensions, in which case  $Z_G(N)$  cannot be  $\mathcal{O}$ -flat. An example of this is the element  $N_2$  in Example 1.2.3.3.

Corollary 4.2.3.3 gives an approach to proving smoothness. It suffices to produce a finite set of elements of  $Z_G(N)(\mathcal{O})$  collectively meeting each connected component of each geometric fiber of  $Z_G(N)$  over Spec  $\mathcal{O}$  provided that (4.2.4.1) holds. Note that we may first make a local flat extension of  $\mathcal{O}$ , as it suffices to check flatness after such an extension. In particular, we may reduce to the case that k is algebraically closed.

In this section, we will establish the following result:

**Proposition 4.2.4.2.** Suppose that all the irreducible factors of the root system  $\Phi(G,T)$  are of classical type, and  $\sqrt{-1}, \sqrt{2} \in \mathcal{O}^{\times}$ . Construct  $N \in \mathfrak{g}' = \text{Lie } G'$  by decomposing  $\mathfrak{g}'$  according to the irreducible components of  $\Phi(G,T)$  and using the integral representatives provided by Proposition 4.1.2.8 for symplectic and orthogonal factors (types B, C, and D) and Example 4.1.2.2 for  $\mathfrak{sl}_n$  (type A). Suppose that the fibers of  $Z_G(N)$  over  $\text{Spec } \mathcal{O}$  are smooth. Then  $Z_G(N)$  is smooth over  $\text{Spec } \mathcal{O}$ .

**Remark 4.2.4.3.** In particular, this gives smoothness for G equal to  $GSp_{2n}$   $(n \ge 2)$  and  $GO_m$   $(m \ge 4)$  using Proposition 4.2.1.4.

The main step of the proof of Proposition 4.2.4.2 is to establish that the scheme-theoretic centralizer  $Z_G(N)$  is  $\mathcal{O}$ -flat. First, some preliminaries. We may and do assume that  $\mathcal{O}$  is Henselian (or even complete) by scalar extension. For  $N \in \mathfrak{g}$ , we define

$$A(N) = (Z_{G_k}(N_k)/Z_{G_k}(N_k)^{\circ})(k) = Z_{G_k}(N_k)(k)/Z_{G_k}(N_k)^{\circ}(k),$$

and study when the following holds:

each element of 
$$A(N)$$
 arises from some  $s \in Z_G(N)(\mathcal{O})$ . (4.2.4.2)

This is very easy for  $GL_n$ :

**Example 4.2.4.4.** Consider the nilpotent orbits of  $GL_n$  as in Example 4.1.1.2, with representatives given in Example 4.1.2.2. But  $Z_{G_k}(N_k)$  is connected for every nilpotent orbit [Jan04, Proposition 3.10], so the identity section shows (4.2.4.2) holds.

Let  $\pi: \widetilde{G'} \to G'$  be the simply connected central cover of the derived group G' over  $\mathcal{O}$ . As p is very good,  $\widetilde{G'}$  and G' have isomorphic Lie algebras via  $\pi$  and Lie G' is a direct factor of Lie G with complement Lie  $(Z_G)$ , so we may abuse notation and view N as an element of all of these Lie algebras over  $\mathcal{O}$ .

Let  $S \subset T$  be the (split) maximal central torus in G. Consider the isogeny  $S \times \widetilde{G'} \to G$ . As S acts trivially on N, we see that  $S \times Z_{\widetilde{G'}}(N)$  is the preimage of  $Z_{G'}(N)$  under this isogeny. As p is very good for G, we obtain finite étale surjections

$$Z_{\widetilde{G}'}(N) \to Z_{G'}(N)$$
 and  $S \times Z_{\widetilde{G}'}(N) \to Z_G(N)$ 

over  $\mathcal{O}$ .

**Lemma 4.2.4.5.** The condition (4.2.4.2) holds for  $\widetilde{G}'$  if and only if (4.2.4.2) holds for G.

Proof. Assume  $\widetilde{G'}$  satisfies (4.2.4.2). Pick a connected component of C of  $Z_{G_k}(N_k)$ . The preimage of C under  $S \times Z_{\widetilde{G'}}(N) \to Z_G(N)$  is a union of k-fiber components of the form  $S_k \times C'$  where C' is a connected component of  $Z_{\widetilde{G'}_k}(N_k)$ . By assumption, there exists  $s \in Z_{\widetilde{G'}}(N)(\mathcal{O})$  meeting any such C'. The image of (1,s) is a point of  $Z_G(N)(\mathcal{O})$  meeting C.

Conversely, assume G satisfies (4.2.4.2). Pick a connected component C' of  $Z_{\widetilde{G'}_k}(N_k)$ . Under  $S \times Z_{\widetilde{G'}}(N) \to Z_G(N)$ ,  $S_k \times C'$  maps onto a connected component C of  $Z_{G_k}(N_k)$ . By assumption, there exists  $s \in Z_G(N)(\mathcal{O})$  such  $s_k \in C$ . As k is algebraically closed, there is  $s'_k \in (S \times Z_{\widetilde{G'}}(N))(k)$  lifting  $s_k$  and lying in C'. As  $S \times Z_{\widetilde{G'}}(N) \to Z_G(N)$  is a finite étale cover and  $\mathcal{O}$  is Henselian, there exists  $s' \in (S \times Z_{\widetilde{G'}}(N))(\mathcal{O})$  lifting  $s_k$  and reducing to  $s'_k$ .

Lemma 4.2.4.5 allows us to reduce to the case when G is semisimple and simply connected. In such a situation,  $G = \prod_j G_j$  with  $G_j$  having irreducible root system (and p very good for each  $G_j$ ) and likewise  $T = \prod_j T_j$  for split maximal  $\mathcal{O}$ -tori  $T_j \subset G_j$ . The element N was defined by decomposing the Lie algebra, so we obtain a decomposition  $N = \sum_j N_j$  such that  $N_j \in \mathfrak{g}_j$  and  $Z_G(N) = \prod_j Z_{G_j}(N_j)$ .

**Lemma 4.2.4.6.** If  $G = \prod_i G_i$  and (4.2.4.2) holds for each  $G_i$ , it holds for G.

*Proof.* The component group of a product is the product of the component groups over an algebraically closed field.  $\Box$ 

Lemma 4.2.4.6 allows us to reduce to a semisimple, simply connected G with irreducible root system. We can now check types  $A_n$ ,  $B_n$ ,  $C_n$ , and  $D_n$  using previous work, and use Lemma 4.2.4.5 to reduce the cases that are traditionally studied:

- 1. For type  $A_n$ , take  $G = GL_n$  and use Example 4.2.4.4.
- 2. For type  $C_n$  with  $n \geq 2$ , use Example 4.2.2.7 and Proposition 4.1.2.8 for  $\operatorname{Sp}_{2n}$ . Note that the representatives constructed are obviously integral.
- 3. For type  $B_n$  with  $n \ge 2$  or  $D_n$  with  $n \ge 4$ , use Example 4.2.2.8 and Proposition 4.1.2.8 for  $SO_{2n+1}$  and  $SO_{2n}$  respectively. (Recall that  $D_3 = A_3$  and  $D_2 = A_1 \times A_1$ .)

We now prove Proposition 4.2.4.2.

*Proof.* We are given that  $Z_{G_k}(N_k)$  and  $Z_{G_K}(N_K)$  are smooth. By the classification of nilpotent orbits over algebraically closed fields, the dimension of the orbit only depends on the combinatorial classification for the orbit in very good characteristic and in characteristic 0, so these fibers are equidimensional of the same dimension. By Corollary 4.2.3.3, it suffices to find  $s \in Z_G(N)(\mathcal{O})$  meeting each desired connected component of  $Z_{G_k}(N_k)$ .

Using Lemmas 4.2.4.5 and 4.2.4.6, it suffices to do so for the irreducible root systems  $A_n$ ,  $B_n$   $(n \ge 2)$ ,  $C_n$   $(n \ge 2)$ , and  $D_n$   $(n \ge 4)$ . We have done so above.

The above argument worked for classical groups, which is our main application. However, we will briefly discuss some possible approaches to producing a finite set of  $\mathcal{O}$ -points collectively meeting all components of the geometric special fiber of a centralizer in general. Let G be a split reductive  $\mathcal{O}$ -group with split maximal  $\mathcal{O}$ -torus T, and let  $N = N_{\sigma} \in \mathfrak{g}_{\mathcal{O}}$  be the element constructed in terms of root data as in Remark 4.1.2.1. Lemmas 4.2.4.5 and 4.2.4.6 show that to make "enough" points in  $Z_G(N)(\mathcal{O})$  we can reduce to the simple and simply connected case if we so choose.

As all split reductive groups descend as such to  $\mathbf{Z}$ , and the description of N in terms of the root datum shows it is compatibly defined over  $\mathbf{Z}$ , we may assume that the residue field k is  $\overline{\mathbf{F}}_p$  for the purposes of checking (4.2.4.2).

**Proposition 4.2.4.7.** Suppose G is semisimple and simply connected and  $\mathcal{O}$  is Henselian. For any  $s_k \in Z_{G_k}(N)(k) \cap T(k)$ , there exists  $s \in Z_G(N)(\mathcal{O})$  lifting  $s_k$ .

Remark 4.2.4.8. Before proving Proposition 4.2.4.7, we record a general observation. A representative  $s_k \in Z_{G_k}(N_k)$  for a coset of A(N) may always be chosen to be semisimple. A proof for the analogous assertion for centralizers of unipotent elements in G is given in [MS03, Corollary 13]; the proof easily adapts to nilpotents in the Lie algebra, or we can invoke the Springer isomorphism to translate between unipotents and nilpotents in very good characteristic. However, it is not always the case that such a component group representative can be chosen in a *single* maximal torus T as  $s_k$  varies. Indeed, if it could then the component group A(N) would be commutative since T is commutative, but some nilpotent orbits in exceptional groups have centralizer with a non-abelian component group. When all representatives can be chosen in a single T, then by Proposition 4.2.4.7 we could verify (4.2.4.2).

*Proof.* Let  $\Phi = \Phi(G, T)$  and  $\Delta$  be a basis for a set  $\Phi^+$  of positive roots. By hypothesis, we may represent N as  $N = \sum_{\beta \in J} N_{\beta}$  where  $N_{\beta}$  is an  $\mathcal{O}$ -basis for the root line  $\mathfrak{g}_{\beta}$  and  $J \subset \Phi^+$ . As G is simply connected, the simple coroots span the cocharacter lattice, so we may write

$$s_k = \prod_{\alpha \in \Delta} \alpha^{\vee}(t_\alpha)$$

for some  $t_{\alpha} \in \mathbf{G}_m(k) = k^{\times}$ . As  $s_k \in Z_{G_k}(N_k)$ , the action of  $s_k$  on  $N_{\beta}$  is trivial for each  $\beta \in J$ , so for all  $\beta \in J$ 

$$\prod_{\alpha} t_{\alpha}^{\langle \beta, \alpha^{\vee} \rangle} = 1.$$

As we have used the existence of Chevalley groups over  $\mathbf{Z}$  to reduce to the case  $k = \overline{\mathbf{F}}_p$ , all of the  $t_{\alpha}$  are of finite order. Thus we pick a root of unity  $t_k \in k^{\times}$  such that each  $t_{\alpha}$  is a power of  $t_k$ . Let the order of  $t_k$  be m, which we know is prime to p. Choose  $n'_{\alpha}$  so that  $t_k^{n'_{\alpha}} = t_{\alpha}$ . Then we know for all  $\beta \in J$ 

$$t_k^{\sum_\alpha n_\alpha'\langle\beta,\alpha^\vee\rangle}=1.$$

In other words,  $\sum_{\alpha} n'_{\alpha} \langle \beta, \alpha^{\vee} \rangle \equiv 0 \pmod{m}$ .

Now pick a lift  $t \in \mathcal{O}^{\times}$  of  $t_k$  that is an mth root of unity (as we may do since  $p \nmid m$  and  $\mathcal{O}$  is Henselian). The element

$$s:=\prod_{\alpha\in\Delta}\alpha^\vee(t)^{n_\alpha'}\in T(\mathcal{O})$$

also lies in  $Z_G(N)(\mathcal{O})$  because for any  $\beta \in J$ 

$$t^{\sum_{\alpha} n'_{\alpha} \langle \beta, \alpha^{\vee} \rangle} = 1$$

(as this mth root of unity in  $\mathcal{O}^{\times}$  has trivial reduction in  $k^{\times}$ ). This says that s acts trivially on  $N_{\beta}$  for any  $\beta \in J$ .

We now review a description of the component groups  $A(N) = Z_{G_k}(N_k)/Z_{G_k}(N_k)^{\circ}$  over an algebraically closed field k and sketch an approach that reduces the question of whether (4.2.4.2) holds for the exceptional groups to extensive case-checking.

For a connected reductive group H of over an algebraically closed field F of very good characteristic, there is a classification of the possibilities of the component group of a centralizer  $Z_H(X)$  for nilpotent  $X \in \text{Lie}(H)(F)$ . The classification uses the notion of a pseudo-Levi subgroup, which is a subgroup of the form  $Z_H(s)^{\circ}$  for a semisimple  $s \in H(F)$ . This classification will be in terms of triples  $(L, sZ^{\circ}(F), X)$  where L is a pseudo-Levi subgroup of H, Z is the center of L,  $sZ^{\circ}(F)$  is a coset of  $Z^{\circ}(F)$  such that  $L = Z_H(sZ^{\circ}(F))^{\circ}$ , and X is a distinguished unipotent in Lie(L) (recall that being distinguished means that every maximal torus contained in  $Z_H(X)$  is contained in the center of H).

**Fact 4.2.4.9.** There is a bijection between H(F)-conjugacy classes of triples  $(L, sZ^{\circ}(F), X)$  as above and H(F)-conjugacy classes of pairs (X, C) where X is a distinguished nilpotent in Lie(H) and C is a conjugacy class of  $(Z_H(X)/Z_H(X)^{\circ})(F)$ , taking the class of  $(L, sZ^{\circ}(F), X)$  to the class of (X, [s]).

This is the main theorem of [MS03]. There it is stated for unipotent elements of H(F), but in very good characteristic the Springer isomorphism identifies the unipotent and nilpotent varieties. For simple groups of adjoint type, an alternate proof is given in [Pre03, Theorem 2].

**Example 4.2.4.10.** Let us give an explicit construction of pseudo-Levi subgroups when the root system of H is irreducible. Pick a maximal torus  $T \subset H$ , and let  $\Phi = \Phi(H,T)$  be the root system. Let  $\Delta = \{\alpha_1, \ldots, \alpha_r\}$  be a basis of a system of positive roots, and  $I = \{1, 2, \ldots, r\}$ . Let  $\widetilde{\alpha} = \sum_{i \in I} n_i \alpha_i$  be the highest root. Define  $\alpha_0 = -\widetilde{\alpha}$ ,  $n_0 = 1$ , and  $I_0 = I \cup \{0\}$ . For a proper subset  $J \subset I_0$ , let  $\Phi_J$  the set of all roots of the form  $\sum_{i \in J} a_i \alpha_i$ . The subgroup  $L_J$  of H generated by T and the root groups  $U_{\alpha}$  for  $\alpha \in \Phi_J$  is a pseudo-Levi subgroup, and all pseudo-Levi subgroups are conjugate to one of this form [MS03, Propositions 30, 32].

Consider the case  $H = \operatorname{GL}_4$  with diagonal torus T and upper triangular Borel subgroup B. The pair (B,T) determines a basis of positive simple roots  $\{\alpha_1,\alpha_2,\alpha_3\}$ . The other positive roots are  $\alpha_1 + \alpha_2, \alpha_1 + \alpha_3$ , and the highest root  $\widetilde{\alpha} = \alpha_1 + \alpha_2 + \alpha_3$ . Suppose  $J \subset \{1,2,3\}$ . Then  $L_J$  is a Levi factor of a parabolic subgroup of H. For example, if  $J = \{1\}$  then  $L_J$  is generated by T,  $U_{\alpha_1}$  and  $U_{-\alpha_1}$ ; this  $L_J$  is the identity component of the centralizer of a diagonal element of the form

$$s = \begin{pmatrix} \lambda_1 & 1 & 0 & 0 \\ 0 & \lambda_1 & 0 & 0 \\ 0 & 0 & \lambda_2 & 0 \\ 0 & 0 & 0 & \lambda_3 \end{pmatrix}$$

with pairwise distinct  $\lambda_1, \lambda_2, \lambda_3 \in k^{\times}$ , and is also a Levi factor of the parabolic subgroup of block upper triangular matrices with blocks of size 2, 1 and 1. A distinguished nilpotent element in the Lie algebra of  $L_J$  is

and the centralizer of X is connected (as is always the case for  $GL_n$ ). The pair  $(L_J, X)$  corresponds to the pair (X, [s]), where s lies in the identity component of the centralizer of X

If  $0 \in J$ , the pseudo-Levi  $L_J$  may be less familiar. Consider  $J' = \{0,1\}$ . Then  $L_{J'}$  is generated by T,  $U_{\alpha_1}$ ,  $U_{\alpha_1+\alpha_2+\alpha_3}$ ,  $U_{\alpha_2+\alpha_3}$ ,  $U_{-\alpha_1-\alpha_2-\alpha_3}$ ,  $U_{-\alpha_1}$ , and  $U_{-\alpha_2-\alpha_3}$ , and it is also the identity component of the centralizer of a diagonal element s' of the form

$$s' = \begin{pmatrix} \lambda_1 & 0 & 0 & 0 \\ 0 & \lambda_1 & 0 & 0 \\ 0 & 0 & \lambda_2 & 0 \\ 0 & 0 & 0 & \lambda_1 \end{pmatrix}$$

with pairwise distinct  $\lambda_1, \lambda_2 \in k^{\times}$ . Alternately,  $L_{J'}$  is a Levi factor of the parabolic subgroup determined by the flag  $0 \subset \text{span}(e_1, e_2, e_4) \subset F^4$  (where  $\{e_1, e_2, e_3, e_4\}$  is the standard basis of  $F^4$ ).

Remark 4.2.4.11. In our example, when  $0 \in J$  we saw that  $L_J$  is a Levi factor of a parabolic subgroup of H. This is not true in general. In fact, a pseudo-Levi subgroup is a Levi factor of a parabolic subgroup if and only if the pair (L, X) corresponds to (X, [1]) [Pre03, Theorem 3.7]. But  $Z_H(X)$  has more than one connected component in general.

We now return to the relative situation over  $\mathcal{O}$ , and remember that we have extended  $(\mathcal{O}, k)$  so that k is algebraically closed.

**Lemma 4.2.4.12.** Suppose that for every conjugacy class of A(N), there exists an element  $s \in Z_G(N)(\mathcal{O})$  such that  $sZ_{G_k}(N_k)^{\circ}$  lies in that conjugacy class. Then for every coset in A(N) there exists an element  $s \in Z_G(N)(\mathcal{O})$  representing that coset.

*Proof.* Consider the subset

$$\Sigma = \{ s Z_{G_k}(N_k)^{\circ} : s \in Z_G(N)(\mathcal{O}) \} \subset A(N)$$

It is certainly a subgroup, and it meets every conjugacy class of the finite group A(N). Consider the decomposition  $A(N) = \bigcup_{g \in A(N)/\Sigma} g\Sigma g^{-1}$ . Each  $g\Sigma g^{-1}$  contains the identity, so by comparing sizes we see that  $A(N) = \Sigma$ .

Thus it would suffice to show that for each triple  $(L_k, s_k Z_k^{\circ}, N_k)$  produced in Fact 4.2.4.9 (applied to  $G_k$  over k in the role of H and F),  $s_k$  is the reduction of an element  $s \in G(\mathcal{O})$  which acts trivially on N. We may conjugate so that  $s_k \in T(k)$ , in which case  $T_k \subset L_k$ . Doing so changes  $N_k$ , so we no longer have an explicit description of N in terms of root data. Without such a description, it is not clear how to lift  $s_k$  using cocharacters as in Proposition 4.2.4.7.

We may conjugate so that  $L_k$  is of the form  $L_J$  which contains  $T_k$ . The k-group  $L_J$  is reductive [MS03, Lemma 14], and p is good for  $L_J$  [MS03, Proposition 16]. The nilpotent  $N_k$  lies in a nilpotent orbit for  $L_J$ , so as in Remark 4.1.2.1 it is  $L_J(k)$ -conjugate to some nilpotent  $\overline{N}'$  which can be expressed in terms of the roots of  $\Phi(L_J, T_k)$ . In particular,  $\overline{N}'$  has a natural lift N' to  $G(\mathcal{O})$ . However, it is not clear that N' and the original N are  $G(\mathcal{O})$ -conjugate. Both are described in terms of roots of  $\Phi(G, T)$ , so for the *finite* number of nilpotent orbits in exceptional groups this assertion could be attacked by (unpleasant) case-checking. As  $s_k$  belongs to  $T_k$  and N' is described in terms of  $\Phi(G, T)$ , Proposition 4.2.4.7 would then lift  $s_k$ . Of course, a uniform approach would be preferable.

Remark 4.2.4.13. McNinch analyzes the centralizer of an "equidimensional nilpotent" in [McN08]. An equidimensional nilpotent is an element  $N \in \mathfrak{g}$  such that  $N_K$  is nilpotent and the dimension of the special and generic fibers of  $Z_G(N)$  are the same. [McN08, §5.2] claims that such  $Z_G(N)$  are  $\mathcal{O}$ -smooth because the fibers are smooth of the same dimension. This deduction is incorrect: it relies on [McN08, 2.3.2] which uses the wrong definition of an equidimensional morphism and thereby incorrectly applies [SGA1, Exp. II, Prop 2.3].

According to [SGA1, Exp. II, Prop 2.3] (or see [EGAIV<sub>3</sub>, §13.3, 14.4.6, 15.2.3]), for a Noetherian scheme Y, a morphism  $f: X \to Y$  locally of finite type, and points  $x \in X$  and y = f(x) with  $\mathcal{O}_y$  normal, f is smooth at x if and only if f is equidimensional at x and  $f^{-1}(y)$  is smooth over k(y) at x. But by definition in [EGAIV<sub>3</sub>, 13.3.2], an equidimensional morphism is more than just a morphism all of whose fibers are of the same dimension (the condition checked in [McN08, 2.3.2]): a locally finite type morphism f is called equidimensional of dimension d at  $x \in X$  when there exists an open neighborhood U of x such that for every irreducible component Z of U through x, f(Z) is dense in some irreducible component of Y containing y and for all  $x' \in U$  the fiber  $f^{-1}(f(x')) \cap U$  has all irreducible components of dimension d.

This is much stronger than the fibers simply being of the same dimension. To see the importance of the extra conditions, consider a discrete valuation ring  $\mathcal{O}$  with field of fractions K and residue field k, and the morphism from X, the disjoint union of Spec K and Spec k, to  $Y = \operatorname{Spec} \mathcal{O}$ . The fibers are of the same dimension (zero) and smooth but the morphism is not flat. This morphism is also not equidimensional at Spec k: the only irreducible component of K containing Spec K is the point itself, with image the closed point of Spec K. This is not dense in Spec K, the only irreducible component of the only open set containing the closed point of Spec K.

The smoothness of centralizers of an equidimensional pure nilpotent is important to proving the main results of [McN08]. In particular, the results in §6 and §7 in [McN08] crucially rely on the smoothness of the centralizers of such nilpotents, leaving a gap in the proof of Theorem B in [McN08] concerning the component group of centralizers. The method we have discussed here reverses this, understanding the geometric component group well enough to *produce* sufficiently many  $\mathcal{O}$ -valued points in order to deduce smoothness of the centralizer in classical cases in very good characteristic via Lemma 4.2.3.1.

### 4.3 Deformations of Nilpotent Elements

As before, let  $\mathcal{O}$  be a discrete valuation ring with residue field k of characteristic p > 0, and let G be a split reductive group scheme over  $\mathcal{O}$  (with connected fibers) such that p is very good for G. Let  $\mathfrak{g} = \text{Lie } G$ . For a nilpotent element  $\overline{N} \in \mathfrak{g}_k$ , we define the notion of a *pure nilpotent* lift of  $\overline{N}$  in  $\mathfrak{g}$  and study the space of such lifts.

#### 4.3.1 Pure Nilpotent Lifts

Let  $\overline{N} \in \mathfrak{g}_k$  be a nilpotent with Bala-Carter data  $\sigma \in C$ . Suppose there exists  $N_{\sigma} \in \mathfrak{g}$  lifting  $\overline{N}$  such that  $(N_{\sigma})_{\overline{K}} \in O_{\overline{K},\sigma}$  and  $Z_G(N_{\sigma})$  is smooth over  $\mathcal{O}$ . We will define the notion of a "pure nilpotent"  $N \in \mathfrak{g}$  lifting  $\overline{N}$  with  $N_K$  nilpotent.

Remark 4.3.1.1. For classical groups, Proposition 4.2.4.2 shows that for any nilpotent  $\overline{N} \in \mathfrak{g}_k$ , there exists  $N'_{\sigma} \in \mathfrak{g}$  such that  $(N'_{\sigma})_{\overline{k}} \in O_{\overline{k},\sigma}$ ,  $Z_G(N'_{\sigma})$  is  $\mathcal{O}$ -smooth, and such that  $(N'_{\sigma})_k$  and  $\overline{N}$  are  $G(\overline{k})$ -conjugate. Thus  $(N'_{\sigma})_k$  and  $\overline{N}$  are conjugate by  $\overline{g} \in G(k')$  for some finite extension k'/k. Lift  $\overline{g}$  to an element  $g \in G(\mathcal{O}')$  for a Henselian discrete valuation ring local over  $\mathcal{O}$  and having residue field k'. The element  $gN'_{\sigma}g^{-1} \in \mathfrak{g}_{\mathcal{O}'}$  reduces to  $\overline{N}_{k'}$  and has the required properties. So the above hypothesis is satisfied for classical groups after a finite flat local extension of  $\mathcal{O}$ .

**Definition 4.3.1.2.** Define the functor  $Nil_{\overline{N}}: \widehat{\mathcal{C}}_{\mathcal{O}} \to Sets$  by

$$\operatorname{Nil}_{\overline{N}}(R) = \{ N \in \mathfrak{g}_R | \operatorname{Ad}_G(g)(N_{\sigma}) = N \text{ for some } g \in \widehat{G}(R), N_k = \overline{N} \}.$$

Call these  $N \in \text{Nil}_{\overline{N}}(R)$  the pure nilpotents lifting  $\overline{N}$ .

This is obviously a subfunctor of the formal neighborhood of  $\overline{N}$  in the affine space  $\underline{\mathfrak{g}}$  over  $\mathcal{O}$  attached to  $\mathfrak{g}$ . The key to analyzing  $\operatorname{Nil}_{\overline{N}}$  is that  $Z_{G_R}(N)$  is smooth over R since  $Z_G(N_\sigma)$  is  $\mathcal{O}$ -smooth and  $(N_\sigma)_R$  is in the G-orbit of N. To ease notation below, we shall write  $gNg^{-1}$  rather than  $\operatorname{Ad}_G(g)(N)$  for  $g \in \widehat{G}(R)$ .

**Lemma 4.3.1.3.** The functor  $Nil_{\overline{N}}$  is pro-representable.

*Proof.* We will use Schlessinger's criterion to check pro-representability. As  $\operatorname{Nil}_{\overline{N}}$  is a subfunctor of the formal neighborhood of the scheme  $\underline{\mathfrak{g}}$  at  $\overline{N}$ , the only condition to check is the analogue of Definition 2.2.2.7(2): given a Cartesian diagram in  $\mathcal{C}_{\mathcal{O}}$ 

$$\begin{array}{ccc} R_1 \times_{R_0} R_2 \xrightarrow{\pi_2} & R_2 \\ & \downarrow^{\pi_1} & & \downarrow \\ & R_1 & \longrightarrow & R_0 \end{array}$$

and  $N_i \in \operatorname{Nil}_{\overline{N}}(R_i)$  such that  $N_1$  and  $N_2$  reduce to  $N_0$ , we want to check that  $N_1 \times N_2 \in \operatorname{Nil}_{\overline{N}}(R_1 \times_{R_0} R_2)$ . By definition, there exists  $g_1 \in \widehat{G}(R_1)$  and  $g_2 \in \widehat{G}(R_2)$  such that  $N_1 = g_1 N_{\sigma} g_1^{-1}$  and  $N_2 = g_2 N_{\sigma} g_2^{-1}$ . Consider the element  $g_1 g_2^{-1} \in \widehat{G}(R_0)$ . Observe that

$$g_1g_2^{-1}N_{\sigma}g_2g_1^{-1} = g_1N_0g_1^{-1} = N_{\sigma} \in \mathfrak{g}_{R_0}.$$

In particular,  $g_1g_2^{-1} \in Z_G(N_\sigma)(R_0)$ . The extension  $R_2 \to R_0$  has nilpotent kernel, so as  $Z_G(N_\sigma)$  is smooth over  $\mathcal{O}$  there exists  $h \in Z_G(N_\sigma)(R_2)$  lifting  $g_1g_2^{-1}$ . The element

$$(g_1, hg_2) \in R_1 \times_{R_0} R_2$$

conjugates  $N_1 \times N_2$  to  $N_{\sigma}$ . Hence  $N_1 \times N_2 \in \text{Nil}_{\overline{N}}(R_1 \times_{R_0} R_2)$ .

**Lemma 4.3.1.4.** The functor  $\operatorname{Nil}_{\overline{N}}$  is formally smooth, in the sense that for a small surjection  $R_2 \to R_1$  of coefficient  $\mathcal{O}$ -algebras the map

$$Nil_{\overline{N}}(R_2) \to Nil_{\overline{N}}(R_1)$$

is surjective. Moreover,  $\operatorname{Nil}_{\overline{N}}$  has relative dimension  $\dim G_k - \dim Z_{G_k}(N_k)$  over  $\mathcal{O}$ .

Proof. Given  $N \in \operatorname{Nil}_{\overline{N}}(R_1)$ , there exists  $g \in \widehat{G}(R_1)$  such that  $gNg^{-1} = N_{\sigma}$ . As G is smooth over  $\mathcal{O}$ , we may find  $g' \in \widehat{G}(R_2)$  lifting g. Then  $(g')^{-1}N_{\sigma}g'$  is a lift of N to  $R_2$ . From its definition, the tangent space to  $\operatorname{Nil}_{\overline{N}}$  is  $\mathfrak{g}_k/\mathfrak{z}_{\mathfrak{g}}(N_k)$ , so the formally smooth  $\operatorname{Nil}_{\overline{N}}$  has relative dimension  $\dim G_k - \dim Z_{G_k}(N_k)$  since  $Z_G(N)$  is  $\mathcal{O}$ -smooth.

Now suppose that A is a complete local Noetherian  $\mathcal{O}$ -algebra with residue field k.

**Lemma 4.3.1.5.** The inverse limit  $\varprojlim \operatorname{Nil}_{\overline{N}}(A/\mathfrak{m}_A^n)$  equals  $\{N \in \mathfrak{g}_A : N = gN_{\sigma}g^{-1} \text{ for some } g \in G(A)\}.$ 

*Proof.* It is immediate that the second is a subset of the first. On the other hand, suppose we had compatible elements  $N_i \in \operatorname{Nil}_{\overline{N}}(A/\mathfrak{m}_A^i)$  such that  $N_i$  is  $\widehat{G}(A/\mathfrak{m}_A^i)$ -conjugate to  $N_{\sigma}$ .

By induction, we will show there exists  $g_i \in \widehat{G}(A/\mathfrak{m}_A^i)$  such that  $N_i = g_i N_\sigma g_i^{-1}$  and  $g_i$  reduces to  $g_{i-1}$ . The base case i=1 is just the assertion that  $(N_\sigma)_k$  equals  $N_1$ . Given  $g_i \in \widehat{G}(A/\mathfrak{m}_A^i)$ , we know there is some element  $g'_{i+1} \in \widehat{G}(A/\mathfrak{m}_A^{i+1})$  such that  $N_i = g'_{i+1} N_\sigma (g'_{i+1})^{-1}$ . The element  $(g'_{i+1})^{-1}g_i$  lies in  $Z_G(N_\sigma)(A/\mathfrak{m}_A^i)$ . As  $Z_G(N_\sigma)$  is smooth over  $\mathcal{O}$ , we may lift to produce an element  $\widetilde{g} \in Z_G(N_\sigma)(A/\mathfrak{m}_A^{i+1})$  for which  $N_i = g'_{i+1}\widetilde{g}N_\sigma(g'_{i+1}\widetilde{g})^{-1}$  and such that  $g'_{i+1}\widetilde{g}$  reduces to  $g_i \in \widehat{G}(A/\mathfrak{m}_A^i)$ . This completes the induction. Finally let  $g \in \widehat{G}(A)$  be the limit of the  $g_i$  and observe  $gN_\sigma g^{-1}$  is the limit of the  $N_i$ .

**Remark 4.3.1.6.** If we had defined  $Nil_{\overline{N}}$  on the larger category  $\widehat{\mathcal{C}}_{\mathcal{O}}$  in the obvious way, Lemma 4.3.1.5 would say that  $Nil_{\overline{N}}$  is continuous.

Remark 4.3.1.7. One could define a scheme-theoretic "nilpotent cone" over  $\mathcal{O}$  as the vanishing locus of the ideal of non-constant homogeneous G-invariant polynomials on  $\mathfrak{g}$ . The arguments in this section could be rephrased as constructing a formal scheme of pure nilpotents inside the formal neighborhood of  $\overline{N}$  in  $\underline{\mathfrak{g}}$ . A natural question is whether there is a broader notion of pure nilpotents that gives a locally closed subscheme of the scheme-theoretic nilpotent cone. For instance, for  $N, N' \in \mathfrak{g}$ , if their images in  $\mathfrak{g}_K$  and  $\mathfrak{g}_k$  are nilpotent in orbits with the same combinatorial parameters, are N and N' conjugate under G over a discrete valuation ring local over  $\mathcal{O}$ ?

When  $G = GL_n$ , this has been explored by Taylor in the course of constructing local deformation conditions [Tay08, Lemma 2.5]. The method uses the explicit description of the orbit closures in Example 4.1.1.4 to define an analogue over  $\mathcal{O}$ . It would be interesting to find a way to do the same for a general split connected reductive group.

# 4.4 The Minimally Ramified Deformation Condition for Tamely Ramified Representations

In this section, we will generalize the minimally ramified deformation condition introduced in [CHT08, §2.4.4] for  $GL_n$  to symplectic and orthogonal groups. We also explain why another more immediate notion based on parabolic subgroups, giving the same deformation condition for  $GL_n$ , is *not liftable* in general (even for  $GSp_4$ ).

Let G be either  $\mathrm{GSp}_m$  or  $\mathrm{GO}_m$  (or  $\mathrm{GL}_m$  to recover the results of [CHT08, §2.4.4]) over the ring of integers  $\mathcal{O}$  in a p-adic field with uniformizer  $\pi$  and residue field k of characteristic p > 0 with  $m \ge 4$ . As always, we assume that p is very good for  $G_k$  (i.e.  $p \ne 2$ ). Let  $\mathfrak{g} = \mathrm{Lie}(G)$ .

#### 4.4.1 Passing between Unipotents and Nilpotents

As in §4.3.1, we work with a pure nilpotent  $N_{\sigma} \in \mathfrak{g}$  for which  $Z_G(N_{\sigma})$  is  $\mathcal{O}$ -smooth,  $(N_{\sigma})_{\overline{K}} \in O_{\overline{K},\sigma}$ , and  $(N_{\sigma})_{\overline{k}} \in O_{\overline{k},\sigma}$ . Define  $\overline{N} := (N_{\sigma})_k$ . We studied deformations of  $\overline{N}$  in §4.3.1, but will ultimately want to analyze deformations of Galois representations which take on unipotent values at certain elements of a local Galois group. Thus, we need a way to pass between unipotent and nilpotent elements. For classical groups, we can use a truncated version of the exponential and logarithm maps:

**Fact 4.4.1.1.** Suppose that  $p \ge m$  and that R is an  $\mathcal{O}$ -algebra. If  $A \in \operatorname{Mat}_m(R)$  has characteristic polynomial  $x^m$  then

$$\exp(A) := 1 + A + A^2/2 + \dots + A^{m-1}/(m-1)!$$

has characteristic polynomial  $(x-1)^m$ . If  $B \in \operatorname{Mat}_m(R)$  has characteristic polynomial  $(x-1)^m$  then

$$\log(B) := (B-1) - (B-1)^2/2 + \ldots + (-1)^m (B-1)^{m-1}/(m-1)$$

has characteristic polynomial  $x^m$ . Furthermore for  $C \in GL_m(R)$  we have  $\exp(CAC^{-1}) = C \exp(A)C^{-1}$ ,  $\log(CBC^{-1}) = C \log(B)C^{-1}$ ,  $\log(\exp(A)) = A$ ,  $\exp(\log(B)) = B$ ,  $\exp(qA) = \exp(A)^q$ , and  $\log(B^q) = q \log(B)$  for any integer q.

This is [Tay08, Lemma 2.4]. The key idea is that because all the higher powers of A and B-1 vanish and all of the denominators appearing are invertible as  $p \ge m$ , we can deduce these facts from results about the exponential and logarithm in characteristic zero.

Suppose J is the matrix for a perfect symmetric or alternating pairing over R.

Corollary 4.4.1.2. For A and B as in Fact 4.4.1.1 with  $\exp(A) = B$ ,  $A^T J + J A = 0$  if and only if  $B^T J B = J$ .

*Proof.* Directly from the definitions we see that 
$$\exp(A^T) = \exp(A)^T$$
. Observe that  $\exp(JAJ^{-1}) = JBJ^{-1}$  and  $\exp(-A^T) = (B^T)^{-1}$ . Thus  $JAJ^{-1} = -A^T$  if and only if  $(B^T)^{-1} = JBJ^{-1}$ .

We shall use this exponential map to convert pure nilpotents into unipotent elements. Let R be a coefficient ring over  $\mathcal{O}$ . By Definition 4.3.1.2, any pure nilpotent  $N \in \operatorname{Nil}_{\overline{N}}(R)$  is G(R)-conjugate to  $N_{\sigma}$ , so it has characteristic polynomial  $x^m$ . Denoting the derived group of G by G', any element of  $\mathfrak{g}$  nilpotent in  $\mathfrak{g}_k$  lies in  $\mathfrak{g}' = \operatorname{Lie} G'$  (if  $G = \operatorname{GSp}_m$  for example, this means that NJ + JN = 0, not just  $NJ + JN = \lambda J$  for some  $\lambda \in \mathcal{O}$ ). Thus, Corollary 4.4.1.2 shows that  $\exp(N) \in G(R)$ . This gives an exponential map

$$\exp: \mathrm{Nil}_{\overline{N}} \to G \tag{4.4.1.1}$$

such that for  $g \in \widehat{G}(R)$ ,  $N \in \text{Nil}_{\overline{N}}(R)$ , and  $q \in \mathbf{Z}$  we have  $\exp(qN) = \exp(N)^q$  and  $g \exp(N)g^{-1} = \exp(\text{Ad}_G(g)N)$ .

**Remark 4.4.1.3.** This is a realization over  $\mathcal{O}$  of a special case of the Springer isomorphism identifying the nilpotent and unipotent varieties in very good characteristic. For later purposes, we will use that the identification to be compatible with the multiplication in the sense that  $\exp(qA) = \exp(A)^q$ . In the case of  $\mathrm{GL}_m$ , a Springer isomorphism that works in any characteristic is given by  $X \to 1 + X$  for nilpotent X, but this is not compatible with multiplication.

**Remark 4.4.1.4.** Let G be a split reductive  $\mathcal{O}$ -group with U the unipotent radical of a parabolic  $\mathcal{O}$ -subgroup P of G. Let r be the nilpotence class of  $U_K$  (the smallest integer for which  $x^r = 1$  for all  $x \in U_K$ ) and suppose that p > r. According to [Sei00, §5] (following [Ser94]), there is an exponential isomorphism  $\exp_U : \text{Lie } U \to U$  defined over  $\mathcal{O}$ . Making Lie U into an  $\mathcal{O}$ -group using the Hausdorff formula, this exponential is a map of  $\mathcal{O}$ -groups, so  $\exp_U(qN) = \exp_U(N)^q$  for  $N \in \text{Lie } U$  and  $q \in \mathbb{Z}$ .

When p is larger than the Coxeter number of G, this provides an approach to converting pure nilpotents  $N \in \operatorname{Nil}_{\overline{N}}(R)$  into elements of G(R). Having to select a U in order to define the exponential map adds complexity; as we only need results for the classical groups, we do not purse this here.

### 4.4.2 Minimally Ramified Deformations

Assume the residue field k of  $\mathcal{O}$  is finite of characteristic p, with  $\mathcal{O}$  complete. Now suppose L is a finite extension of  $\mathbf{Q}_{\ell}$  (with  $\ell \neq p$ ), and denote its absolute Galois group by  $\Gamma_L$ . Consider a representation  $\overline{p}:\Gamma_L\to G(k)$ . We wish to define a (large) smooth deformation condition for  $\overline{p}$  generalizing the minimally ramified deformation condition for  $\mathrm{GL}_n$  defined in [CHT08, §2.4.4]. In this section we do so for a special class of tamely ramified representations. This requires making an étale local extension of  $\mathcal{O}$ , which will be harmless for our purposes.

Recall that  $\Gamma_L^t$ , the Galois group of the maximal tamely ramified extension of L, is isomorphic to the semi-direct product

$$\widehat{\mathbf{Z}} \ltimes \prod_{p' 
eq \ell} \mathbf{Z}_{p'}$$

where  $\widehat{\mathbf{Z}}$  is generated by a Frobenius  $\phi$  and the conjugation action by  $\phi$  on the direct product is given by the cyclotomic character. We consider representations of  $\Gamma_L^t$  which factor through the quotient  $\widehat{\mathbf{Z}} \ltimes \mathbf{Z}_p$ . Picking a topological generator  $\tau$  for  $\mathbf{Z}_p$ , the action is explicitly given by

$$\phi \tau \phi^{-1} = q \tau$$

where q is the size of the residue field of L. Note q is a power of  $\ell$ , so it is relatively prime to p. This leads us to study representations of the group  $T_q = \widehat{\mathbf{Z}} \times \mathbf{Z}_p$ .

Let  $\overline{\rho}: T_q \to G(k)$  be a representation. Informally, the deformation condition will be lifts  $\rho: T_q \to G(R)$  such that  $\rho(\tau)$  lies in the "same" unipotent orbit as  $\overline{\rho}(\tau)$ . To make this meaningful over an infinitesimal thickening of k, we shall use the notion of pure nilpotents as in Definition 4.3.1.2 since unipotence and unipotent orbits are not good notions when not over a field.

First, let us check that  $\overline{\rho}(\tau) \in G(k)$  actually is unipotent. This element decomposes as a commuting product of semi-simple and unipotent elements of G(k). The order of a semi-simple element in G(k) is prime to p, while by continuity there is an  $r \geq 0$  such that  $\tau^{p^r} \in \ker(\overline{\rho})$ . Thus  $\overline{\rho}(\tau)$  is unipotent. In particular,  $\log(\overline{\rho}(\tau)) = \overline{N}$  is nilpotent, and by Remark 4.3.1.1 after making an étale local extension of  $\mathcal{O}$  we may assume that there exists a pure nilpotent  $N_{\sigma} \in \mathfrak{g}$  lifting  $\overline{N}$  for which  $Z_G(N_{\sigma})$  is smooth. Making a further extension if necessary, we may also assume that the unit  $q \in \mathcal{O}^{\times}$  is a square. We obtain an exponential map  $\exp: \operatorname{Nil}_{\overline{N}} \to G$  as in (4.4.1.1).

**Definition 4.4.2.1.** For a coefficient ring R over  $\mathcal{O}$ , a continuous lift  $\rho: T_q \to G(R)$  of  $\overline{\rho}$  is minimally ramified if  $\rho(\tau) = \exp(N)$  for some  $N \in \operatorname{Nil}_{\overline{N}}(R)$ .

**Example 4.4.2.2.** Take  $G = \operatorname{GL}_n$ . Then  $X \mapsto 1_n + X$  gives an identification of nilpotents and unipotents. Up to conjugacy, over algebraically closed fields parabolic subgroups correspond to partitions of n and every nilpotent orbit is the Richardson orbit of such a parabolic. Let  $\overline{\rho}(\tau) - 1_n =: \overline{N}$  correspond to the partition  $\sigma = n_1 + n_2 + \ldots + n_r$ . By Example 4.1.2.2, the lift  $N_{\sigma}$  of  $\overline{N}$  is conjugate to a block nilpotent matrix with blocks of size  $n_1, n_2, \ldots, n_r$ . The points  $N \in \operatorname{Nil}_{\overline{N}}(R)$  are the  $\widehat{G}(R)$ -conjugates of  $N_{\sigma}$ . It is clear (since p > n) that if  $\rho(\tau) \in \operatorname{Nil}_{\overline{N}}(R)$  then

$$\ker(\rho(\tau) - 1_n)^i \otimes_R k \to \ker(\overline{\rho}(\tau) - 1_n)^i \tag{4.4.2.1}$$

is an isomorphism for all i. Conversely, repeated applications of [CHT08, Lemma 2.4.15] show that any  $\rho(\tau)$  satisfying this collection of isomorphism conditions is  $\widehat{G}(R)$ -conjugate to  $N_{\sigma}$ . Thus the minimally ramified deformation condition for  $GL_n$  defined in [CHT08] agrees with our definition. Note that the identification  $X \mapsto 1_n + X$  does not satisfy  $qX \to (1+X)^q$ , so it will not work in our argument. The proof of [CHT08, Lemma 2.4.19] uses a different method for which this non-homomorphic identification suffices.

Our goal is to show that the functor of minimally ramified lifts is pro-representable over  $\mathcal{O}$  and that the representing object  $R_{\overline{\rho}}^{\text{m.r.}\square}$  is formally smooth over  $\mathcal{O}$ .

**Proposition 4.4.2.3.** Under our assumptions on G, the local deformation ring  $R_{\overline{\rho}}^{\text{m.r.} \square}$  is formally smooth over  $\mathcal{O}$  of relative dimension dim  $G_k$ .

*Proof.* Let  $\overline{\Phi} = \overline{\rho}(\phi) \in G(k)$  and let  $\widehat{G}_{\overline{\Phi}}$  be the formal completion of G at  $\overline{\Phi}$ . Using the relation

$$\overline{\rho}(\phi)\overline{\rho}(\tau)\overline{\rho}(\phi)^{-1} = \overline{\rho}(\tau)^q,$$

we deduce that  $\overline{\Phi} \, \overline{N} \, \overline{\Phi}^{-1} = q \overline{N}$ . Therefore we study the functor  $M_{\overline{N}}$  on  $\widehat{\mathcal{C}}_{\mathcal{O}}$  defined by

$$M_{\overline{N}}(R) = \{(\Phi,N): N \in \operatorname{Nil}_{\overline{N}}(R), \, \Phi \in \widehat{G}_{\overline{\Phi}}(R), \, \Phi N \Phi^{-1} = qN\} \subset \operatorname{Nil}_{\overline{N}}(R) \times \widehat{G}_{\overline{\Phi}}(R).$$

Any such lift  $(\Phi, N)$  to a coefficient ring R determines a homomorphism  $T_q \to G(R)$  lifting  $\overline{\rho}$  via  $\phi \mapsto \Phi$  and  $\tau \mapsto \exp(N)$ , and it is continuous because  $\exp(\overline{N})$  is unipotent. We will analyze  $M_{\overline{N}}$  through the composition

$$M_{\overline{N}} \to \operatorname{Nil}_{\overline{N}} \to \operatorname{Spf} \mathcal{O}.$$

First, observe that  $M_{\overline{N}} \to \operatorname{Nil}_{\overline{N}}$  is relatively representable as " $\Phi N = qN\Phi$ " is a formal closed condition on points  $\Phi$  of  $(\widehat{G}_{\overline{\Phi}})_R$  for each  $N \in \operatorname{Nil}_{\overline{N}}(R)$ .

From Lemma 4.3.1.4, we know that  $\operatorname{Nil}_{\overline{N}}$  is formally smooth over  $\mathcal{O}$ , and the universal nilpotent is  $gN_{\sigma}g^{-1}$  for some  $g \in \widehat{G}(\operatorname{Nil}_{\overline{N}})$ . To check formal smoothness of the map  $M_{\overline{N}} \to \operatorname{Nil}_{\overline{N}}$ , it therefore suffices to check the formal smoothness of the fiber of  $M_{\overline{N}}$  over the  $\mathcal{O}$ -point  $N_{\sigma}$  of  $\operatorname{Nil}_{\overline{N}}$ .

We have written down  $\Phi_{\sigma} \in G(\mathcal{O})$  satisfying  $\Phi_{\sigma} N_{\sigma} \Phi_{\sigma}^{-1} = q N_{\sigma}$  in Remark 4.2.2.9. Observe that  $\overline{\Phi} \overline{\Phi}_{\sigma}^{-1} \in Z_G(N_{\sigma})(k)$ . By smoothness, we may lift  $\overline{\Phi} \overline{\Phi}_{\sigma}^{-1}$  to an element  $s \in Z_G(N_{\sigma})(\mathcal{O})$ . Then  $s\Phi_{\sigma}$  reduces to  $\overline{\Phi}$  and satisfies  $(s\Phi_{\sigma})N_{\sigma}(s\Phi_{\sigma})^{-1} = qN_{\sigma}$ , so the fiber of  $M_{\overline{N}}$  over  $N_{\sigma}$  has an  $\mathcal{O}$ -point. The relative dimension of the formally smooth  $\operatorname{Nil}_{\overline{N}}$  is  $\dim G_k - \dim Z_{G_k}(\overline{N})$  by Lemma 4.3.1.4, and  $M_{\overline{N}} \to \operatorname{Nil}_{\overline{N}}$  is a  $\widehat{Z_G(N_{\sigma})}$ -torsor since it has an  $\mathcal{O}$ -point over  $N_{\sigma}$ . As  $Z_G(N_{\sigma})$  is smooth it follows that  $M_{\overline{N}}$  is formally smooth over  $\operatorname{Spf} \mathcal{O}$  of relative dimension  $\dim G_k$ .

**Example 4.4.2.4.** This recovers [CHT08, Lemma 2.4.19] in the case  $G = GL_n$ .

Let S be the (torus) quotient of G by its derived group G', and  $\mu: G \to S$  the quotient map. For use elsewhere, we now study a variant where we fix a lift  $\nu: T_q \to S(\mathcal{O})$  of  $\mu \circ \overline{\rho}: T_q \to S(k)$ :

Corollary 4.4.2.5. The deformation condition of minimally ramified lifts  $\rho: T_q \to G(R)$  satisfying  $\mu \circ \rho = \nu$  is a liftable deformation condition. The framed deformation ring  $R_{\overline{\rho}}^{\text{m.r.},\nu,\square}$  is of relative dimension dim  $G_k-1$ .

*Proof.* The quotient torus S = G/G' is split of rank 1, so the subscheme  $R_{\overline{\rho}}^{\text{m.r.},\nu,\square} \subset R_{\overline{\rho}}^{\text{m.r.},\square}$  is the vanishing of locus of a single function. As  $R_{\overline{\rho}}^{\text{m.r.},\square}$  is formally smooth over  $\mathcal{O}$  with relative dimension dim  $G_k$ , it suffices to check that the "reduced" tangent space of  $R_{\overline{\rho}}^{\text{m.r.},\nu,\square}$  over k is a proper subspace of the "reduced" tangent space of  $R_{\overline{\rho}}^{\text{m.r.},\square}$ .

Let Z be the maximal central torus of G. On the level of Lie algebras, we know that Lie G splits over  $\mathcal{O}$  as a direct sum of Lie G' and Lie  $S \simeq \text{Lie } Z$  as p is very good for G. We can modify a lift  $\rho_0$  over  $R = k[\epsilon]/(\epsilon^2)$  by multiplying against an unramified non-trivial character  $T_q \to Z(R)$  with trivial reduction, changing  $\mu \circ \rho_0$ . Thus the "reduced" tangent space of  $R_{\overline{\rho}}^{\text{m.r.},\nu,\square}$  is a proper subspace of that of  $R_{\overline{\rho}}^{\text{m.r.}}$ 

#### 4.4.3 Deformation Conditions Based on Parabolic Subgroups

The use of nilpotent orbits is not the only approach to defining a minimally ramified deformation condition. As discussed in §1.2.3, the method used to prove [CHT08, Lemma 2.4.19] suggests a generalization from  $GL_n$  to other groups G based on deformations lying in certain parabolic subgroups of G. This deformation condition is not smooth for algebraic groups beyond  $GL_n$ , so it does not work in Ramakrishna's method. In this section we give a conceptual explanation for this phenomenon.

Let  $P \subset G$  be a parabolic  $\mathcal{O}$ -subgroup. The Richardson orbit for  $P_k$  intersects (Lie P) $_k$  in a dense open set which is a single geometric orbit under  $P_k$ . Suppose that  $\overline{\rho}(\tau)$  is the exponential of a k-point  $\overline{N}$  in the Richardson orbit, and consider deformations  $\rho: T_q \to G(\mathcal{O})$  of  $\overline{\rho}$  ramified with respect to P in the sense that  $\rho(\tau) \in P$  (compare with Definition 1.2.3.1). This requires specifying a lift of  $\overline{N}$  that lies in Lie P. One could hope that such lifts would automatically be  $G(\mathcal{O})$ -conjugate to the fixed lift  $N_{\sigma}$  defined in Proposition 4.1.2.8, reminiscent of the definition we gave for  $\operatorname{Nil}_{\overline{N}}$ , a situation in which the associated (framed) deformation ring is smooth.

We now show that often smoothness fails if  $\overline{N}$  does not lie in the Richardson orbit of  $P_k$ . Lifts of  $\overline{N}$  can "change nilpotent type" yet still lie in a parabolic lifting  $P_k$ , such as the example of the standard Borel subgroup in  $GL_3$  with

$$N = \begin{pmatrix} 0 & 1 & 0 \\ 0 & 0 & p \\ 0 & 0 & 0 \end{pmatrix} \quad \text{lifting} \quad \overline{N} = \begin{pmatrix} 0 & 1 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}$$

In particular, we easily obtain non-pure nilpotents. This is very bad: the nilpotent orbits over a field are smooth but the nilpotent cone is not smooth, so the deformation problem of deforming with respect to P should not be smooth because "it sees multiple orbits". Furthermore, by working with lifts to  $\mathcal{O}$  even if we could lift  $\overline{\rho}(\tau)$  appropriately, there would still be problems lifting the image of  $\phi$  because the centralizer of a non-pure nilpotent is not smooth over  $\mathcal{O}$  (the special and generic fiber typically have different dimensions). So it is crucial to choose a parabolic such that  $\overline{N}$  lies in the Richardson orbit of  $P_k$ .

For  $GL_n$ , all nilpotent orbits are Richardson orbits. This is not true in general. In particular, we should not expect the deformation condition of being ramified with respect to a parabolic to be liftable. Example 1.2.3.2 illustrates this phenomenon for  $GSp_4$ , which we now revisit in a more conceptual manner.

**Example 4.4.3.1.** Take  $G = \mathrm{GSp}_4$ . Parabolic subgroups correspond to isotropic flags. Up to conjugacy, these subgroups are G (the trivial parabolic) and stabilizers of the flags

$$0 \subset \operatorname{Span}(v_1) \subset \operatorname{Span}(v_1, v_2) \subset \operatorname{Span}(v_1, v_2, v_3) \subset k^4,$$
  
$$0 \subset \operatorname{Span}(v_1) \subset \operatorname{Span}(v_1, v_2, v_3) \subset k^4, \quad 0 \subset \operatorname{Span}(v_1, v_2) \subset k^4$$

where  $\{v_1, v_2, v_4\}$  is the standard basis of  $k^4$ . Their Richardson orbits correspond to the respective nilpotent orbits indexed by the partitions 1+1+1+1, 4, 4, and 2+2. In particular, the same Richardson orbit is associated to two of these. There is one more partition of 4 with odd numbers appearing an even number of times: 2+1+1. This is a nilpotent orbit that is not a Richardson orbit; for the representation in Example 1.2.3.2,  $\log(\bar{\rho}(\tau))$  is in this nilpotent orbit.

# 4.5 Minimally Ramified Deformations of Symplectic and Orthogonal Groups

We continue the notation of the previous section. We have defined the minimally ramified deformation condition for representations factoring through the quotient  $T_q = \hat{\mathbf{Z}} \ltimes \mathbf{Z}_p$  of the tame Galois group  $\Gamma_L^t$  at a place away from p. In this section, we will adapt the matrix-theoretic methods in [CHT08, §2.4.4], making use of more conceptual module-theoretic arguments, to define the minimally ramified deformation condition for any representation when  $G = \mathrm{GSp}_m$  or  $G = \mathrm{GO}_m$ . (Minor variants of this method work for  $\mathrm{Sp}_m$  and  $\mathrm{SO}_m$ , and the original method of [CHT08, §2.4.4] works for  $\mathrm{GL}_m$ .) We naturally embed G into  $\mathrm{GL}(M)$  for a free  $\mathcal{O}$ -module M of rank m, and let V denote the reduction of M, a vector space over the residue field k.

We consider a representation  $\overline{\rho}: \Gamma_L \to G(k) \subset \operatorname{GL}(V)(k)$  which may be wildly ramified (with L an  $\ell$ -adic field for  $\ell \neq p$ ). We will define a deformation condition for  $\overline{\rho}$  in terms of the minimally ramified deformation condition for certain associated tamely ramified representations, after possibly extending  $\mathcal{O}$ . In §4.5.1, we analyze  $\overline{\rho}$  as being built out of two pieces of data: representations of a closed normal subgroup  $\Lambda_L$  of  $\Gamma_L$  whose pro-order is prime to p, and tamely ramified representations of  $\Gamma_L/\Lambda_L$ . The representation theory of  $\Lambda_L$  is manageable since its pro-order is prime to p, and representations of  $\Gamma_L/\Lambda_L$  can be understood using the results of the previous section.

#### 4.5.1 Decomposing Representations

We begin with a few preliminaries about representations over rings. Let  $\Lambda'$  be a profinite group and R be an Artinian coefficient ring with residue field k. If  $\Lambda'$  has pro-order prime to p, the representation theory over k is nice: every finite-dimensional continuous representation is a direct sum of irreducibles, and every such representation is projective over  $k[\Lambda]$  for any finite discrete quotient  $\Lambda$  of  $\Lambda'$  through which the representation factors. We are also interested in corresponding statements over an Artinian coefficient ring R.

**Fact 4.5.1.1.** Suppose the pro-order of  $\Lambda'$  is prime to p. Let P and P' be  $R[\Lambda']$ -modules that are finitely generated over R with continuous action of  $\Lambda'$ , and F be a  $k[\Lambda']$ -module that is finite dimensional over k with continuous action of  $\Lambda'$ . Let  $\Lambda$  be a finite discrete quotient of  $\Lambda'$  through which the  $\Lambda'$ -actions on P, P', and F factor.

1. If P is free as an R-module, it is projective as a  $R[\Lambda]$ -module.

- 2. If P and P' are projective over  $R[\Lambda]$ , they are isomorphic if and only if  $\overline{P}$  and  $\overline{P'}$  are isomorphic.
- 3. There exists a projective  $R[\Lambda]$ -module (unique up to isomorphism) whose reduction is F.

These statements are special cases of results in [Ser77, §14.4]. We now record two lemmas which do not need the assumption that the pro-order of  $\Lambda'$  is prime to p.

**Lemma 4.5.1.2.** Let P and P' be  $R[\Lambda']$ -modules, finitely generated over R with continuous action of  $\Lambda'$  factoring through a finite discrete quotient  $\Lambda$  of  $\Lambda'$ . Assume P and P' are  $R[\Lambda]$ -projective. The natural map gives an isomorphism

$$\operatorname{Hom}_{\Lambda'}(P, P') \otimes_R k \to \operatorname{Hom}_{\Lambda'}(\overline{P}, \overline{P'}).$$

*Proof.* We may replace  $\operatorname{Hom}_{\Lambda'}$  with  $\operatorname{Hom}_{\Lambda}$ . Note that  $\mathfrak{m}P' = \mathfrak{m} \otimes_R P'$ , so  $\operatorname{Hom}_{\Lambda}(P, \mathfrak{m}P') = \operatorname{Hom}_{\Lambda}(P, P') \otimes_R \mathfrak{m}$  as P and P' are  $R[\Lambda]$ -projective. Then apply  $\operatorname{Hom}_{\Lambda}(P, -)$  to the exact sequence  $0 \to \mathfrak{m}P' \to P' \to P'/\mathfrak{m}P' \to 0$ .

**Lemma 4.5.1.3.** Let  $\Lambda$  be a finite group and let M and M' be finite  $R[\Lambda]$ -modules whose reductions  $\overline{M}$  and  $\overline{M}'$  are non-isomorphic irreducible  $k[\Lambda]$ -modules. Then  $\operatorname{Hom}_{R[\Lambda]}(M,M')=0$ .

*Proof.* Filter M' by the composition series  $\{\mathfrak{m}^i M'\}$ , and consider the surjection

$$\mathfrak{m}^i/\mathfrak{m}^{i+1}\otimes \overline{M}' \twoheadrightarrow \mathfrak{m}^i M'/\mathfrak{m}^{i+1} M'.$$

The action of  $\Lambda$  on  $\mathfrak{m}^i/\mathfrak{m}^{i+1} \otimes M'$  is solely on the irreducible  $\overline{M}'$ , so as a  $k[\Lambda]$ -module  $\mathfrak{m}^i M'/\mathfrak{m}^{i+1} M'$  is isomorphic to a direct sum of copies of  $\overline{M}'$ . Thus

$$\operatorname{Hom}_{R[\Lambda]}(M,\mathfrak{m}^{i}M'/\mathfrak{m}^{i+1}M') = \operatorname{Hom}_{k[\Lambda]}(\overline{M},\mathfrak{m}^{i}M'/\mathfrak{m}^{i+1}M') = 0$$

as  $\overline{M}$  and  $\overline{M}'$  are non-isomorphic  $k[\Lambda]$ -modules.

By descending induction on i, we shall show that

$$\operatorname{Hom}_{R[\Lambda]}(M,\mathfrak{m}^iM')=0.$$

For large i,  $\mathfrak{m}^i M' = 0$ . Consider the exact sequence

$$0 \to \mathfrak{m}^{i+1}M' \to \mathfrak{m}^iM' \to \mathfrak{m}^iM'/\mathfrak{m}^{i+1}M' \to 0.$$

Applying  $\operatorname{Hom}_{R[\Lambda]}(M,-)$ , we obtain a left exact sequence

$$0 \to \operatorname{Hom}_{R[\Lambda]}(M,\mathfrak{m}^{i+1}M') \to \operatorname{Hom}_{R[\Lambda]}(M,\mathfrak{m}^{i}M') \to \operatorname{Hom}_{R[\Lambda]}(M,\mathfrak{m}^{i}M'/\mathfrak{m}^{i+1}M')$$

The left term is 0 by induction, and the right term is 0 by the above calculation. This completes the induction.  $\Box$ 

Given  $\overline{\rho}: \Gamma_L \to G(k) \subset \operatorname{GL}(V)(k)$  and a lift  $\rho: \Gamma_L \to G(R) \subset \operatorname{GL}(M)(R)$  for some  $R \in \mathcal{C}_{\mathcal{O}}$ , we now turn to decomposing the  $R[\Gamma_L]$ -module M. Let  $I_L \subset \Gamma_L$  be the inertia group, and pick a surjection  $I_L \to \mathbf{Z}_p$ . Define  $\Lambda_L$  to be the kernel of this surjection (normal in  $\Gamma_L$ ). This is a pro-finite group with pro-order prime to p, and is independent of the choice of surjection. Define the quotient

$$T_L := \Gamma_L / \Lambda_L$$

which is a quotient of the tamely ramified Galois group  $\Gamma_L^t$  and of the form  $T_q = \hat{\mathbf{Z}} \ltimes \mathbf{Z}_p$  as in §4.4. We wish to compatibly decompose V and M as  $\Lambda_L$ -modules and then understand the action of  $\Gamma_L$  on the decomposition.

We first make a finite extension of k (and hence of  $\mathcal{O}$ ) so that all of the (finitely many) irreducible representations of  $\Lambda_L$  over k occurring in V are absolutely irreducible over k.

Because  $\Lambda_L$  has order prime to p,  $\operatorname{Res}_{\Lambda_L}^{\Gamma_L}(V)$  is completely reducible and we can write

$$\operatorname{Res}_{\Lambda_L}^{\Gamma_L}(V) = \bigoplus_{\tau} V_{\tau}$$

where  $\tau$  runs through the set  $\Sigma$  of isomorphism classes of irreducible representations of  $\Lambda_L$  over k occurring in V, and each  $V_{\tau}$  is the  $\tau$ -isotypic component. We will obtain an analogous decomposition for M.

Let  $\Gamma$  be a finite discrete quotient of  $\Gamma_L$  through which  $\rho$  factors, and let  $\Lambda$  be the image of  $\Lambda_L$  in  $\Gamma$ . Using Fact 4.5.1.1(3) we can lift  $\tau$  to a projective  $R[\Lambda]$ -module  $\widetilde{\tau}$  unique up to isomorphism. We will eventually want this lift to have additional properties (see §4.5.2), but this is not yet necessary. We set  $W_{\tau} := \operatorname{Hom}_{\Lambda_L}(\widetilde{\tau}, M)$  and consider the natural morphism

$$\bigoplus_{\tau} \widetilde{\tau} \otimes_R W_{\tau} \to M.$$

Note that M is  $R[\Lambda]$ -projective by Fact 4.5.1.1(1).

**Lemma 4.5.1.4.** This map is an isomorphism of  $R[\Lambda_L]$ -modules.

*Proof.* It suffices to check it is an isomorphism of  $R[\Lambda]$ -modules. When R = k, we know  $\operatorname{End}_{\Lambda}(\tau) = k$  as we extended k so that all of the irreducible representations of  $\Lambda$  over k occurring inside V are absolutely irreducible. Splitting up V as a direct sum of irreducibles, we obtain the desired isomorphism.

In the general case, the map is an isomorphism after reducing modulo  $\mathfrak{m}$  (use Lemma 4.5.1.2). Thus by Nakayama's lemma it is surjective. Since M is R-projective, the formation of the kernel commutes with reduction modulo  $\mathfrak{m}$ . Thus, again using Nakayama's lemma the kernel is zero.

We define  $M_{\tau}$  to be the image of  $\widetilde{\tau} \otimes_R \operatorname{Hom}_{\Lambda_L}(\widetilde{\tau}, M)$  in M. It is the largest  $R[\Lambda_L]$ -direct summand whose reduction is a direct sum of copies of  $\tau$ .

We next seek to understand the action of  $\Gamma_L$  on this canonical decomposition of M. For  $g \in \Gamma_L$ , consider the  $R[\Lambda_L]$ -module  $gM_{\tau}$ : it is a direct summand of M over R whose reduction is a direct sum of copies of the representation  $\tau^g$  defined by  $\tau^g(h) = \tau(g^{-1}hg)$  for  $h \in \Lambda_L$ . Thus we see that  $gM_{\tau} = M_{\tau^g}$  inside M, and  $\Gamma_L$  permutes the  $M_{\tau}$ 's. The orbits corresponds to sets of conjugate representations.

Consider the stabilizer of  $V_{\tau}$ :

$$\Gamma_{L,\tau} = \{g \in \Gamma_L : gV_\tau = V_\tau \text{ inside } V\} = \{g \in \Gamma_L : \tau^g \simeq \tau\} \subset \Gamma_L$$

with corresponding image

$$\Gamma_{\tau} = \{g \in \Gamma : gV_{\tau} = V_{\tau} \text{ inside } V\} = \{g \in \Gamma : \tau^g \simeq \tau\} \subset \Gamma.$$

Then the k-span of the members of the  $\Gamma_L$ -orbit of  $V_{\tau}$  is exactly the representation  $\operatorname{Ind}_{\Gamma_L,\tau}^{\Gamma_L} V_{\tau} = \operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma} V_{\tau}$ . Letting  $[\tau]$  denote the set of  $R[\Lambda_L]$ -isomorphism classes of  $\Lambda$ -representations  $\Gamma_L$ -conjugate to  $\tau$ , by taking into account the action of  $\Gamma_{\tau}$  the isomorphism in Lemma 4.5.1.4 becomes an isomorphism of  $R[\Gamma_L]$ -modules

$$M = \bigoplus_{[\tau]} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} M_{\tau} \tag{4.5.1.1}$$

using one representative  $\tau$  per  $\Gamma_L$ -conjugacy class  $[\tau]$ .

For orthogonal or symplectic representations, this decomposition interacts with duality. Denote the similitude character by  $\mu$ , and let  $\overline{\nu} := \mu \circ \overline{\rho} : \Gamma_L \to k^{\times}$ . Let N be a free  $\mathcal{O}$ -module of rank 1 on which  $\Gamma_L$  acts by a specified continuous  $\mathcal{O}^{\times}$ -valued lift  $\nu$  of  $\overline{\nu}$ , and let  $\overline{N}$  be its reduction modulo  $\mathfrak{m}$ . For an  $\mathcal{O}$ -module M, define  $M^{\vee} = \operatorname{Hom}_{\mathcal{O}}(M, N)$  with the evident  $\Gamma_L$ -action. The perfect pairing gives an isomorphism of  $R[\Gamma_L]$ -modules  $M \simeq M^{\vee}$ . In particular,

$$M_{\tau}^{\vee} \simeq M_{\tau^*}$$

for some irreducible representation  $\tau^*$  of  $\Lambda_L$  occurring in V. Note that  $\tau^* \simeq \tau^{\vee}$  as  $k[\Lambda_L]$ -modules. There are three cases:

- Case 1:  $\tau$  is not conjugate to  $\tau^*$ ;
- Case 2:  $\tau$  is isomorphic to  $\tau^*$ ;
- Case 3:  $\tau$  is conjugate to  $\tau^*$  but not isomorphic.

In the second case, we claim that the isomorphism of  $k[\Lambda_L]$ -modules  $\iota : \tau \simeq \tau^{\vee}$  gives a sign-symmetric perfect pairing on  $\tau$ . Note that  $W_{\tau} = \operatorname{Hom}_{\Lambda}(\tau, V) \simeq \operatorname{Hom}_{\Lambda}(\tau^{\vee}, V^{\vee}) \simeq W_{\tau}^{\vee}$  as  $V_{\tau} \simeq V_{\tau}^{\vee}$ . This isomorphism  $\varphi_{\tau}$  defines a pairing  $\langle , \rangle_{W_{\tau}}$  on  $W_{\tau}$  via

$$\langle w_1, w_2 \rangle_{W_\tau} := \varphi_\tau(w_1)(w_2).$$

Let  $\psi: V \to V^{\vee}$  be the isomorphism given by  $m \mapsto \langle m, - \rangle_V$ , and define  $\langle v_1, v_2 \rangle_{\tau} := i(v_1)(v_2)$  for  $v_1, v_2 \in \tau$ . We have a commutative diagram

The commutativity says that for elementary tensors  $m_i = v_i \otimes w_i \in V_\tau = \tau \otimes W_\tau$  we have

$$\langle m_1, m_2 \rangle_M = \psi(m_1)(m_2) = (i(v_1) \otimes \varphi_{\tau}(w_1)) (v_2 \otimes w_2) = i(v_1)(v_2) \cdot \varphi_{\tau}(w_1)(w_2) = \langle v_1, v_2 \rangle_{\tau} \langle w_1, w_2 \rangle_{W_{\tau}}.$$
(4.5.1.2)

**Lemma 4.5.1.5.** The pairing  $\langle , \rangle_{\tau}$  is a sign-symmetric.

*Proof.* Suppose there exists  $v \in \tau$  such that  $\iota(v)(v) \neq 0$ . For  $w_1, w_2 \in W_\tau$ , (4.5.1.2) gives

$$i(v)(v)\varphi_{\tau}(w_1)(w_2) = \langle v \otimes w_1, v \otimes w_2 \rangle_V = \epsilon \langle v \otimes w_2, v \otimes w_1 \rangle_V = \epsilon i(v)(v)\varphi_{\tau}(w_2)(w_1).$$

Canceling i(v)(v), we conclude that  $\langle w_1, w_2 \rangle_{W_{\pi}} = \epsilon \langle w_2, w_1 \rangle_{W_{\pi}}$ . Using (4.5.1.2), we conclude that

$$\epsilon i(v_2)(v_1) \cdot \varphi_{\tau}(w_2)(w_1) = \epsilon \langle m_2, m_1 \rangle_V = \langle m_1, m_2 \rangle_V = i(v_1)(v_2) \cdot \varphi_{\tau}(w_1)(w_2) = \epsilon i(v_1)(v_2) \cdot \varphi_{\tau}(w_2)(w_1).$$

Choosing  $w_1$  and  $w_2$  with  $\langle w_2, w_1 \rangle_{W_{\tau}} \neq 0$  (possible as  $\langle , \rangle_V$  is perfect), we conclude that  $\langle v_1, v_2 \rangle_{\tilde{\tau}} = \langle v_2, v_1 \rangle_{\tilde{\tau}}$ . Otherwise  $\iota(v)(v) = 0$  for all  $v \in \tau$ , in which case  $\langle , \rangle_{\tilde{\tau}}$  is alternating.

In §4.5.2 we will see that the action of  $\Lambda_L$  on the module underlying  $\tilde{\tau}$  can be extended to an action of  $\Gamma_{L,\tau}$  factoring through  $\Gamma_{\tau}$ . Therefore,  $W_{\tau} = \operatorname{Hom}_{\Lambda_L}(\tilde{\tau}, M)$  is naturally a representation of  $T_{L,\tau} := \Gamma_{L,\tau}/\Lambda_L$ , and of  $T_{\tau} := \Gamma_{\tau}/\Lambda$  (a finite quotient of  $T_{L,\tau}$ ). In §4.5.3, we will use the minimally ramified deformation condition of §4.4 to specify which deformations  $W_{\tau}$  are allowed. Together with the decomposition (4.5.1.1)

$$\bigoplus_{[\tau]} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_{L}} \left( \widetilde{\tau} \otimes W_{\tau} \right) \to M$$

this defines a deformation condition for  $\overline{\rho}$ . Some care is needed to ensure compatibility with the pairing on M, which will require breaking into cases in the next sections.

#### 4.5.2 Extension of Representations

We continue the notation of the previous section, where  $\tau$  is an absolutely irreducible representation of  $\Lambda_L$  over k. We need to lift this to a representation over  $\mathcal{O}$  and extend it to a representation of  $\Gamma_{L,\tau}$ . We will have to do something extra for the representation to be compatible with a pairing, depending on how  $\tau$  and  $\tau^*$  are related.

In Case 1, we ignore the pairing. The results of [CHT08, §2.4.4] let us pick a  $\mathcal{O}[\Gamma_{L,\tau}]$ -module  $\widetilde{\tau}$  that is a free  $\mathcal{O}$ -module and reduces to  $\tau$ . In this case,  $\widetilde{\tau}^{\vee}$  is a free  $\mathcal{O}$ -module reducing to  $\tau^*$ .

In Case 2, from Lemma 4.5.1.5 it follows that  $\tau$  is a symplectic or orthogonal representation. We will adapt the  $GL_n$ -technique of [CHT08] to produce a symplectic or orthogonal extension  $\tilde{\tau}$ . Letting  $n = \dim \tau$ , the representation  $\tau$  gives a homomorphism  $\tau : \Lambda_L \to G(k)$  where G is  $GSp_n$  or  $GO_n$ .

First, we claim that there is a continuous lift  $\tilde{\tau}: \Lambda_L \to G(W(k))$ : without the pairing, this would be Fact 4.5.1.1(3). To also take into account the pairing, consider deformation theory for the residual representation  $\tau$ . This is a smooth deformation condition as  $H^2(\Lambda_L, \operatorname{ad} \tau) = 0$ :  $\Lambda_L$  has pro-order prime

to p and  $\operatorname{ad} \tau$  has order a power of p. Therefore the desired lift exists. It is unique (up to conjugations which preserve  $\tau$ ) because the tangent space is zero dimensional as  $H^1(\Lambda_L, \operatorname{ad} \tau) = 0$ . By considering representations of the group  $\Lambda_L/\ker(\tau)$ , we may and do assume arrange that  $\ker(\tilde{\tau}) = \ker(\tau)$  as subgroups of  $\Lambda_L$ .

**Remark 4.5.2.1.** For  $g \in \Gamma_{L,\tau}$ , the isomorphism of  $k[\Lambda_L]$ -modules  $\tau^g \simeq \tau$  lifts to an isomorphism  $\widetilde{\tau}^g \simeq \widetilde{\tau}$  of  $\mathcal{O}[\Lambda_L]$ -modules by uniqueness. Thus  $\Gamma_{L,\tau} = \{g \in \Gamma_L : \widetilde{\tau}^g \simeq \widetilde{\tau}\}.$ 

We will now show how to continuously extend  $\tilde{\tau}$  to  $\Gamma_{L,\tau}$ . The first step in constructing the extension is to understand the structure of  $\Gamma_{L,\tau}$  and  $I_L \cap \Gamma_{L,\tau}$ , where  $I_L$  is the inertia group.

Recall that  $T_L = \Gamma_L/\Lambda_L$  is the semi-direct product of  $\hat{\mathbf{Z}}$  and  $\mathbf{Z}_p$ , where  $\hat{\mathbf{Z}}$  is generated by a lift of Frobenius  $\phi$  and  $\mathbf{Z}_p$  is generated by an element  $\sigma$ , with  $\phi\sigma\phi^{-1} = \sigma^q$  where  $q = \ell^a$  is the size of the residue field of L.

#### Lemma 4.5.2.2. The exact sequence

$$1 \to \Lambda_L \to \Gamma_L \to T_L \to 1$$

is topologically split, so  $\Gamma_L$  is a semi-direct product.

*Proof.* Let S be a Sylow pro-p subgroup of  $I_L$ , which must be isomorphic to  $\mathbb{Z}_p$ . Let  $\phi$  be a lift of Frobenius to  $\Gamma_L$ . Then  $\phi S \phi^{-1}$  is another Sylow pro-p subgroup of  $I_L$ , and hence is conjugate to S by an element of  $I_L$ . By choosing the lift  $\phi$ , we may thereby assume that  $\phi$  normalizes S. But then it is clear that S and  $\phi$  together topologically generate  $T_L$ , giving the desired splitting.

For  $T_{L,\tau} := \Gamma_{L,\tau}/\Gamma_L$ , this gives a topological splitting of

$$1 \to \Lambda_L \to \Gamma_{L,\tau} \to T_{L,\tau} \to 1.$$

As  $\Gamma_{L,\tau}$  is an open subgroup of  $\Gamma_L$ , we observe that  $T_{L,\tau}$  is an open subgroup of  $T_L$ . Note that  $T_{L,\tau}$  is an open normal subgroup of  $T_L$  topologically generated by some powers of  $\phi$  and  $\sigma$  which will be denoted by  $\phi_{\tau}$  and  $\sigma_{\tau}$  (since any open subgroup of a semidirect product  $C \ltimes C'$  for pro-cyclic C and C' is of the form  $C_0 \ltimes C'_0$  for open subgroups  $C_0 \subset C$  and  $C'_0 \subset C'$ ). In particular,  $T_{L,\tau}$  is itself isomorphic to  $T_{q'}$  for some q'. The element  $\sigma_{\tau}$  and  $\Lambda_L$  together topologically generate  $\Gamma_{L,\tau} \cap I_L$ .

Before extending  $\tilde{\tau}$ , we need several technical lemmas.

#### **Lemma 4.5.2.3.** The centralizer of the image of $\tilde{\tau}$ is $\mathcal{O}$ .

*Proof.* As  $\tau$  is absolutely irreducible,  $\operatorname{End}_{\Lambda_L}(\tau) = k$ . By Lemma 4.5.1.2, we see that the reduction of  $\operatorname{End}_{\Lambda_L}(\widetilde{\tau})$  modulo the maximal ideal of  $\mathcal{O}$  is k, so the map  $\mathcal{O} \hookrightarrow \operatorname{End}_{\Lambda_L}(\widetilde{\tau})$  is surjective by Nakayama's lemma.

#### **Lemma 4.5.2.4.** The dimension of $\tau$ is not divisible by p.

*Proof.* As  $\tau$  is continuous and  $\Lambda_L$  has pro-order prime to p, the representation  $\tau$  factors through a finite discrete quotient  $\Lambda$  of  $\Lambda_L$  whose order is prime to p. Such a representation is the reduction of a projective  $\mathcal{O}[\Lambda]$ -module by Fact 4.5.1.1(3). Inverting p, we obtain a representation of  $\Lambda$  in characteristic zero that is absolutely irreducible since the "reduction"  $\tau$  is absolutely irreducible over k. By [Ser77, §6.5 Corollary 2], the dimension of this representation (equal to the dimension of  $\tau$ ) divides the order of  $\Lambda$ .

We will now extend  $\tilde{\tau}$  from  $\Lambda_L \subset I_L$  to  $\Gamma_{L,\tau}$  by defining it on the topological generators  $\sigma_{\tau}$  and  $\phi_{\tau}$ . We say that such an extension has tame determinant if  $\det(\tilde{\tau}(\sigma_{\tau}))$  has finite order which is prime to p.

**Lemma 4.5.2.5.** There is a unique continuous extension  $\widetilde{\tau}: \Gamma_{L,\tau} \cap I_L \to G(\mathcal{O})$  with tame determinant.

*Proof.* A continuous extension of  $\widetilde{\tau}$  to  $\Gamma_{L,\tau} \cap I_L$  is determined by its value on  $\sigma_{\tau}$ . As  $\sigma_{\tau} \in \Gamma_{L,\tau}$ , in light of Remark 4.5.2.1 there is an  $A \in G(\mathcal{O})$  such that for  $g \in \Lambda_L$  we have

$$\widetilde{\tau}(\sigma_{\tau}g\sigma_{\tau}^{-1}) = A\widetilde{\tau}(g)A^{-1}.$$

We would like to send  $\sigma_{\tau}$  to the element A. However, this might not produce a continuous extension, and even if it does it might not have tame determinant unless we pick A correctly. As  $\sigma_{\tau}$  is a topological generator for a group isomorphic to  $\mathbf{Z}_p$ , the existence of a continuous extension satisfying  $\sigma_{\tau} \mapsto A$  is equivalent to some p-power of A having trivial reduction. We wish to show that there is a unique choice of such A that also makes the extension have tame determinant.

We will first show that some power  $A^{p^b}$  lies in the centralizer of the image  $\tilde{\tau}(\Lambda_L)$ . Consider the conjugation action of  $\langle \sigma_{\tau} \rangle$  on  $\Lambda_L$ . As  $\ker \tilde{\tau} = \ker \tau$  is a normal subgroup of  $\Gamma_{L,\tau}$  (if  $g \in \Gamma_{L,\tau}$  and  $\tau(h) = 1$ , then  $\tau^g(h)$  is conjugate to  $\tau(h) = 1$  by Remark 4.5.2.1) we get an action of  $\langle \sigma_{\tau} \rangle$  on  $\Lambda_L / \ker \tau \simeq \tau(\Lambda_L)$ . The action is continuous, so there is a power  $p^b$  such that for all  $g \in \Lambda_L$  we have

$$\tau(\sigma_{\tau}^{p^b}g\sigma_{\tau}^{-p^b}) = \tau(g).$$

As  $\ker \tilde{\tau} = \ker \tau$ , we see that

$$A^{p^b}\widetilde{\tau}(g)A^{-p^b} = \widetilde{\tau}(\sigma_{\tau}^{p^b}g\sigma_{\tau}^{-p^b}) = \widetilde{\tau}(g).$$

Therefore  $A^{p^b}$  lies in the centralizer of  $\widetilde{\tau}(\Lambda_L)$ .

By Lemma 4.5.2.3, this centralizer is  $\mathcal{O}$ . We claim that by multiplying A by some unit in  $\mathcal{O}$ , we can arrange for the continuous extension  $\tilde{\tau}$  to exist and have tame determinant. We will use the fact that an element of  $\mathcal{O}^{\times}$  is the product of a 1-unit and a Teichmuller lift of an element of  $k^{\times}$ . As  $A^{p^b} \in \mathcal{O}^{\times}$  and the pth power map is an automorphism of  $k^{\times}$ , we may multiply A by a unit scalar so that  $A^{p^b}$  reduces to the identity matrix. By Lemma 4.5.2.4, the dimension n of  $\tau$  is prime to p so we may multiply A by a 1-unit so that det(A) has finite order prime to p. Sending  $\sigma_{\tau}$  to this particular A gives a continuous extension with tame determinant.

Let's show this extension is unique. Any extension must send  $\sigma_{\tau}$  to an element of the form wA for  $w \in \mathcal{O}^{\times}$  (the centralizer of the image  $\tilde{\tau}(\Lambda_L)$ ). By continuity, there is a power  $p^b$  such that  $(wA)^{p^b}$  reduces to the identity. This means that  $w^{p^b}$  reduces to the identity, and hence that w reduces to the identity. On the other hand,  $\det(wA)\det(A)^{-1}=w^n$ . The left side has finite order that is relatively prime to p, so  $w^n$  does too. This forces  $w^n=1$  since its reduction is 1. But as n is prime to p (Lemma 4.5.2.4), the only nth roots of unity in  $\mathcal{O}^{\times}$  are Teichmuller lifts. Therefore w=1.

**Lemma 4.5.2.6.** There is a continuous extension  $\tilde{\tau}: \Gamma_{L,\tau} \to G(\mathcal{O})$ .

*Proof.* We extend  $\widetilde{\tau}$  in Lemma 4.5.2.5 continuously to  $\Gamma_{L,\tau}$  by defining it on  $\phi_{\tau}$ . As  $\phi_{\tau} \in \Gamma_{L,\tau}$ , there is an element  $A \in G(\mathcal{O})$  conjugating  $\widetilde{\tau} : \Lambda_L \to G(\mathcal{O})$  to  $\widetilde{\tau}^{\phi_{\tau}} : \Lambda_L \to G(\mathcal{O})$ . Each has a unique extension to a continuous morphism from  $I_L \cap \Gamma_{L,\tau}$  to  $G(\mathcal{O})$  with tame determinant. Therefore for  $g \in I_L \cap \Gamma_{L,\tau}$  we have

$$\widetilde{\tau}(\phi_{\tau}g\phi_{\tau}^{-1}) = A\widetilde{\tau}(g)A^{-1}$$

since the right side has the same (tame) determinant as  $\tilde{\tau}$  on  $I_L \cap T_\tau$ . We can continuously extend  $\tilde{\tau}$ :  $I_L \cap \Gamma_{L,\tau} \to G(\mathcal{O})$  by sending  $\phi_\tau$  to A since A has reduction with finite order.

This gives the desired lift and extension of  $\tau$  in the case that  $\tau \simeq \tau^*$ .

In Case 3,  $\tau$  is conjugate to  $\tau^*$  but not isomorphic. The argument follows the same structure as the previous case, but we make a few modifications to treat  $\tau \oplus \tau^*$  together. In particular, we can pick a copy of the  $k[\Lambda_L]$ -module  $\tau$  inside  $V_{\tau}$  and a copy of  $\tau^* \simeq \tau^{\vee}$  inside V such that the pairing restricted to  $\tau \oplus \tau^*$  is perfect.

Define  $\Gamma_{L,\tau\oplus\tau^*} = \{g \in \Gamma_L : g(\tau\oplus\tau^*) = \tau\oplus\tau^*\}$ . It contains  $\Gamma_{L,\tau}$  with index 2, as any automorphism either preserves  $\tau$  and  $\tau^*$  or swaps them. Arguing as in the paragraph after Lemma 4.5.2.2, we obtain a split exact sequence

$$0 \to \Lambda_L \to \Gamma_{L,\tau \oplus \tau^*} \to T_{L,\tau \oplus \tau^*} \to 1$$

where  $T_{L,\tau\oplus\tau^*}$  is an open normal subgroup of  $T_L$  topologically generated by some powers of  $\phi$  and  $\sigma$  which we denote by  $\phi_{\tau\oplus\tau^*}$  and  $\sigma_{\tau\oplus\tau^*}$ . We may arrange that either

- Case 3a:  $\phi_{\tau \oplus \tau^*}^2 = \phi_{\tau}$  and  $\sigma_{\tau \oplus \tau^*} = \sigma_{\tau}$  or
- Case 3b:  $\phi_{\tau \oplus \tau^*} = \phi_{\tau}$  and  $\sigma^2_{\tau \oplus \tau^*} = \sigma_{\tau}$ .

In Case 3a, we begin by lifting  $\tau$  to  $\mathcal{O}$  as a representation of  $\Lambda_L$ : as before, we do this using the fact that the pro-order of  $\Lambda_L$  is prime to p, and obtain a lift  $\widetilde{\tau}$  unique up to isomorphism. We extend  $\widetilde{\tau}$  to be a representation of  $\Gamma_{L,\tau} \cap I_L$  by defining it on  $\sigma_{\tau}$  using the  $\mathrm{GL}_n$ -version of Lemma 4.5.2.5, [CHT08, Lemma 2.4.11]. There it is shown all such extensions are unique up to equivalence. In particular,  $\widetilde{\tau}$  and  $(\widetilde{\tau}^{\phi_{\tau}\oplus\tau^*})^{\vee}$  are isomorphic  $\mathcal{O}[\Gamma_{L,\tau}\cap I_K]$ -modules. We can use this to define a sign-symmetric perfect pairing on  $\widetilde{\tau} \oplus \widetilde{\tau}^{\phi_{\tau}\oplus\tau^*}$  that is compatible with the action of  $\Gamma_{L,\tau} \cap I_K$  and  $\phi_{\tau\oplus\tau^*}$ , hence of  $\Gamma_{L,\tau\oplus\tau^*}$ .

In Case 3b, as  $\tau^{\vee}$  and  $\tau^{\sigma_{\tau \oplus \tau^*}}$  are isomorphic  $k[\Lambda_L]$ -modules it follows that  $\widetilde{\tau}^{\vee}$  and  $\widetilde{\tau}^{\sigma_{\tau \oplus \tau^*}}$  are isomorphic  $\mathcal{O}[\Lambda_L]$ -modules. In particular, this isomorphism gives a natural way to define a sign-symmetric perfect pairing on  $M = \widetilde{\tau} \oplus \widetilde{\tau}^{\sigma_{\tau \oplus \tau^*}}$  lifting the residual one. This pairing is compatible with the action of  $\Gamma_{L,\tau \oplus \tau^*} \cap I_L$  (which is generated by  $\Lambda_L$  and  $\sigma_{\tau \oplus \tau^*}$ ). Finally, we claim that M and  $M^{\phi_{\tau}}$  are isomorphic. As  $\phi_{\tau} \in \Gamma_{L,\tau}$  preserves  $\tau$ , the reductions of M and  $M^{\phi_{\tau}}$  are isomorphic by an isomorphism which identifies  $\tau$  and  $\tau^{\phi_{\tau}}$ . By uniqueness of the lift of  $\tau$  as a  $\mathcal{O}[\Lambda_L]$ -module, we obtain an isomorphism of  $\widetilde{\tau}$  and  $\widetilde{\tau}^{\phi_{\tau}}$  and hence of M and  $M^{\phi_{\tau}}$  preserving the pairing. Then we proceed as in the proof of Lemma 4.5.2.6, defining an image of  $\phi_{\tau}$  using this isomorphism.

In conclusion, we have shown:

**Lemma 4.5.2.7.** In case 3, there exists an  $\mathcal{O}[\Gamma_{L,\tau\oplus\tau^*}]$ -module  $\widetilde{\tau\oplus\tau^*}$  with pairing lifting  $\tau\oplus\tau^*$  together with its pairing.

#### 4.5.3 Lifts with Pairings

We continue the notation of §4.5.1, and analyze how the duality pairing interacts with the decomposition (4.5.1.1). Recall that we obtained an isomorphism  $M \simeq M^{\vee}$  of  $R[\Gamma_L]$ -modules which gave isomorphisms  $M_{\tau} \simeq M_{\tau^*}^{\vee}$  of  $R[\Gamma_{L,\tau}]$ -modules. The key point is that for any lift and extension  $\tau'$  of  $\tau$  to an  $\mathcal{O}[\Gamma_{L,\tau}]$ -module, the isomorphism of  $R[\Lambda_L]$ -modules

$$\tau' \otimes \operatorname{Hom}_{\Lambda_L}(\tau', M) \to M_{\tau}$$

is compatible with the  $\Gamma_{L,\tau}$ -action.

To do this, it is convenient to break into the cases introduced at the end of §4.5.1. For an irreducible  $k[\Lambda]$ -module  $\tau$  occurring in V, note that  $(\tau^g)^{\vee} = (\tau^{\vee})^g$  for any  $g \in \Gamma_L$ , so if  $\tau \simeq \tau^*$  then  $\tau^g \simeq (\tau^g)^*$ . We let

- $\Sigma_n$  denote the set of  $\Gamma_L$ -conjugacy classes of such  $\tau$  for which  $\tau$  is not conjugate to  $\tau^*$ ;
- $\Sigma_e$  denote the set of  $\Gamma_L$ -conjugacy classes of such  $\tau$  for which  $\tau \simeq \tau^*$ ;
- $\Sigma_c$  denote the set of  $\Gamma_L$ -conjugacy classes of such  $\tau$  for which  $\tau^*$  is conjugate to  $\tau$  but  $\tau \not\simeq \tau^*$ .

From (4.5.1.1), we obtain a decomposition

$$M = \bigoplus_{\tau \in \Sigma_n} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} (\tau' \otimes W_{\tau}) \oplus \bigoplus_{\tau \in \Sigma_e} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} (\tau' \otimes W_{\tau}) \oplus \bigoplus_{\tau \in \Sigma_c} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} (\tau' \otimes W_{\tau})$$

$$(4.5.3.1)$$

where  $\tau'$  is any lift and extension of  $\tau$  to  $\Gamma_{\tau}$  and  $W_{\tau} = \operatorname{Hom}_{\Lambda}(\tau', M)$  is a representation of  $T_{L,\tau}$ . Note that  $W_{\tau}$  is free as an R-module (since M and  $\tau'$  are, with  $\tau' \neq 0$  and R local), and hence that  $W_{\tau}$  is tamely ramified of the type considered in §4.4.

We may rewrite this to make use of the special extensions constructed in §4.5.2. In particular, for  $\tau \in \Sigma_c$  we rewrite

$$\operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_{L}}(\tau' \otimes W_{\tau}) = \operatorname{Ind}_{\Gamma_{L,\tau \oplus \tau^{*}}}^{\Gamma_{L}} \left( \widetilde{\tau \oplus \tau^{*}} \otimes W_{\tau \oplus \tau^{*}} \right)$$

where  $W_{\tau \oplus \tau^*} := \operatorname{Hom}_{\Lambda'_L}(\widetilde{\tau \oplus \tau^*}, M)$ . This uses the notation and results from Case 3 in §4.5.2, in particular the fact that  $\tau \oplus \tau^*$  is an irreducible representation of the group  $\Lambda'_L$  generated by  $\Lambda_L$  and any  $g \in \Gamma_L$  with  $\tau^* \simeq \tau^g$ . Note that  $W_{\tau \oplus \tau^*}$  is a representation of  $T_{L,\tau \oplus \tau^*}$ , which is a subgroup of  $T_L = \Gamma_L/\Lambda_L$ , hence of the form  $T_q$  as considered in §4.4. Using the extensions  $\widetilde{\tau}$  and  $\widetilde{\tau} \oplus \tau^*$  from §4.5.2, we obtain a decomposition

$$M = \bigoplus_{\tau \in \Sigma_n} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} \left( \widetilde{\tau} \otimes W_{\tau} \right) \oplus \bigoplus_{\tau \in \Sigma_e} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} \left( \widetilde{\tau} \otimes W_{\tau} \right) \oplus \bigoplus_{\tau \in \Sigma_c} \operatorname{Ind}_{\Gamma_{L,\tau \oplus \tau^*}}^{\Gamma_L} \left( \widetilde{\tau \oplus \tau^*} \otimes W_{\tau \oplus \tau^*} \right). \tag{4.5.3.2}$$

Now let M' be another  $R[\Gamma_L]$ -module that is finite free over R such that the irreducible representations of  $\Lambda_L$  occurring in  $V' := M'/\mathfrak{m}M'$  are among the same  $\tau$ 's, so

$$M' = \bigoplus_{\tau \in \Sigma_n} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} \left( \widetilde{\tau} \otimes W'_{\tau} \right) \oplus \bigoplus_{\tau \in \Sigma_e} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} \left( \widetilde{\tau} \otimes W'_{\tau} \right) \oplus \bigoplus_{\tau \in \Sigma_c} \operatorname{Ind}_{\Gamma_{L,\tau \oplus \tau^*}}^{\Gamma_L} \left( \widetilde{\tau \oplus \tau^*} \otimes W'_{\tau \oplus \tau^*} \right).$$

Lemma 4.5.3.1. The natural map

$$\bigoplus_{\tau \in \Sigma_n} \operatorname{Hom}_{T_{L,\tau}}(W_\tau, W_\tau') \oplus \bigoplus_{\tau \in \Sigma_e} \operatorname{Hom}_{T_{L,\tau}}(W_\tau, W_\tau') \oplus \bigoplus_{\tau \in \Sigma_c} \operatorname{Hom}_{T_{L,\tau \oplus \tau^*}}(W_{\tau \oplus \tau^*}, W_{\tau \oplus \tau^*}') \to \operatorname{Hom}_{\Gamma_L}(M, M')$$

is an isomorphism.

*Proof.* We may immediately pass to working with representations of the finite discrete groups  $\Gamma$  and  $\Lambda$ . Notice that

$$\operatorname{Hom}_{\Gamma}(\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma}(M_{\tau}),\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma}(M_{\tau}')) \simeq \operatorname{Hom}_{\Gamma_{\tau}}(M_{\tau},\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma}(M_{\tau}')) \simeq \operatorname{Hom}_{\Gamma_{\tau}}(M_{\tau},M_{\tau}')$$

where the second isomorphism uses that  $\operatorname{Hom}_{\Gamma_{\tau}}(M_{\tau}, M'_{\tau^g}) = 0$  by Lemma 4.5.1.3 when  $\tau$  and  $\tau^g$  are non-isomorphic. Furthermore, if  $\tau_1$  and  $\tau_2$  are not  $\Gamma$ -conjugate then

$$\operatorname{Hom}_{\Gamma}(\operatorname{Ind}_{\Gamma_{\tau_1}}^{\Gamma}(M_{\tau_1}),\operatorname{Ind}_{\Gamma_{\tau_2}}^{\Gamma}(M'_{\tau_2}))=0$$

again using Lemma 4.5.1.3. Then using (4.5.1.1) we see that

$$\operatorname{Hom}_{\Gamma}(M,M') = \bigoplus_{[\tau_1], [\tau_2]} \operatorname{Hom}_{\Gamma}(\operatorname{Ind}_{\Gamma_{\tau_1}}^{\Gamma}(M_{\tau_1}), \operatorname{Ind}_{\Gamma_{\tau_2}}^{\Gamma}(M'_{\tau_2})) = \bigoplus_{[\tau]} \operatorname{Hom}_{\Gamma_{\tau}}(M_{\tau}, M'_{\tau}).$$

All the irreducible finite-dimensional representations of  $\Lambda$  occurring in V and V' are absolutely irreducible over k by design. For  $\tau \in \Sigma_n \cup \Sigma_e$ , consider the natural inclusion

$$\operatorname{Hom}_{R}(W_{\tau}, W_{\tau}') \hookrightarrow \operatorname{Hom}_{\Lambda}(\widetilde{\tau} \otimes W_{\tau}, \widetilde{\tau} \otimes W_{\tau}') = \operatorname{Hom}_{\Lambda}(\widetilde{\tau}, \widetilde{\tau}) \otimes_{R} \operatorname{Hom}_{R}(W_{\tau}, W_{\tau}'), \tag{4.5.3.3}$$

using that  $W_{\tau}$  and  $W'_{\tau}$  are R-free of finite rank and  $\Lambda$  has no effect on them. But  $R \hookrightarrow \operatorname{Hom}_{\Lambda}(\widetilde{\tau}, \widetilde{\tau})$  is an isomorphism because  $\operatorname{End}_{\Lambda}(\tau) = k$  and because surjectivity can be checked modulo  $\mathfrak{m}_R$  using Lemma 4.5.1.2. As  $M_{\tau} \simeq \widetilde{\tau} \otimes W_{\tau}$ , this implies that

$$\operatorname{Hom}_{\Gamma_{\tau}}(M_{\tau},M_{\tau}') = \operatorname{Hom}_{\Lambda}(M_{\tau},M_{\tau}')^{T_{\tau}} = \operatorname{Hom}_{\Lambda}(\widetilde{\tau} \otimes W_{\tau},\widetilde{\tau} \otimes W_{\tau}')^{T_{\tau}} = \operatorname{Hom}_{R}(W_{\tau},W_{\tau})^{T_{\tau}} = \operatorname{Hom}_{T_{\tau}}(W_{\tau},W_{\tau}')^{T_{\tau}} = \operatorname{Hom}_{T_{\tau}}(W_{\tau},W_$$

where  $T_{\tau}$  is the image of  $T_{L,\tau}$  in  $\Gamma_{\tau}$ . An analogous computation in the case  $\tau \in \Sigma_c$  completes the proof.  $\square$ 

We can now consider the duality isomorphism  $M \simeq M^{\vee}$ . By Lemma 4.5.3.1, this is equivalent to a collection of isomorphisms of  $R[T_{L,\tau}]$ -modules  $\varphi_{\tau}: W_{\tau} \simeq W_{\tau^*}^{\vee}$  for  $\tau \in \Sigma_e \cup \Sigma_n$  and an isomorphism of  $R[T_{L,\tau \oplus \tau^*}]$ -modules  $\varphi_{\tau}: W_{\tau \oplus \tau^*} \simeq W_{\tau \oplus \tau^*}^{\vee}$  for  $\tau \in \Sigma_c$ . We analyze the cases separately.

 $R[T_{L,\tau\oplus\tau^*}]$ -modules  $\varphi_{\tau}:W_{\tau\oplus\tau^*}\simeq W_{\tau\oplus\tau^*}^{\vee}\simeq W_{\tau\oplus\tau^*}^{\vee}$  for  $\tau\in\Sigma_c$ . We analyze the cases separately. In **Case 1**  $(\tau\in\Sigma_n)$ , note that  $\operatorname{Ind}_{\Gamma_L,\tau}^{\Gamma_L}M_{\tau}$  is an isotropic subspace of M as  $\tau\not\simeq\tau^*$ . In particular, the perfect sign-symmetric pairing on  $\operatorname{Ind}^{\Gamma_L}\Gamma_{L,\tau}M_{\tau}\oplus\operatorname{Ind}^{\Gamma_L}\Gamma_{L,\tau^*}M_{\tau^*}$  is equivalent to an isomorphism of  $R[\Gamma_L]$ -modules

$$\operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_{L}} M_{\tau} \simeq \left(\operatorname{Ind}_{\Gamma_{L,\tau^{*}}}^{\Gamma_{L}} M_{\tau^{*}}\right)^{\vee},$$

which is equivalent to the isomorphism of  $R[T_{L,\tau}]$ -modules  $\varphi_{\tau}:W_{\tau}\simeq W_{\tau^*}^{\vee}$ . (Note that the similitude character  $\nu$  is present in the use of the dual.)

In Case 2  $(\tau \in \Sigma_e)$ , the perfect sign-symmetric pairing on  $\operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} M_{\tau}$  is equivalent to an isomorphism  $W_{\tau} \simeq W_{\tau}^{\vee}$  of  $R[T_{L,\tau}]$ -modules. Thus it gives a pairing  $\langle , \rangle_{W_{\tau}}$  on  $W_{\tau}$  via

$$\langle w_1, w_2 \rangle_{W_\tau} := \varphi_\tau(w_1)(w_2).$$

We claim this pairing is sign-symmetric.

From §4.5.2 we have an isomorphism  $i: \widetilde{\tau} \simeq \widetilde{\tau}^{\vee}$  of  $R[\Gamma_{L,\tau}]$ -modules. Let  $\psi: M \to M^{\vee}$  be the isomorphism of  $R[\Gamma_L]$ -modules given by  $m \mapsto \langle m, - \rangle_M$ , and define  $\langle v_1, v_2 \rangle_{\widetilde{\tau}} := i(v_1)(v_2)$ . We have a commutative diagram

The commutativity says that for an elementary tensor  $m_i = v_i \otimes w_i \in M_\tau = \widetilde{\tau} \otimes W_\tau$  we have

$$\langle m_1, m_2 \rangle_M = \psi(m_1)(m_2) = (i(v_1) \otimes \varphi_{\tau}(w_1)) (v_2 \otimes w_2) = i(v_1)(v_2) \cdot \varphi_{\tau}(w_1)(w_2) = \langle v_1, v_2 \rangle_{\widetilde{\tau}} \langle w_1, w_2 \rangle_{W_{\tau}}.$$
(4.5.3.4)

The pairings are perfect and  $\langle \cdot, \cdot \rangle_{\tau}$  is sign-symmetric, so the pairing on  $W_{\tau}$  is sign-symmetric if and only if the pairing on  $M_{\tau}$  is sign-symmetric.

In Case 3  $(\tau \in \Sigma_c)$ , an analogous argument using the isomorphism  $\widetilde{\tau \oplus \tau^*} \simeq \widetilde{\tau \oplus \tau^*}$  of  $R[\Gamma_{L,\tau \oplus \tau^*}]$ modules shows that the pairing induced by  $\varphi_\tau : W_{\tau \oplus \tau^*} \simeq W_{\tau \oplus \tau^*}^{\vee}$  is sign-symmetric if and only if the pairing on

$$\operatorname{Ind}_{\Gamma_{L,\tau\oplus\tau^*}}^{\Gamma_L}\left(\widetilde{\tau\oplus\tau^*}\otimes W'_{\tau\oplus\tau^*}\right)$$

induced from the pairing on M is sign-symmetric.

We can now define the minimally ramified deformation condition for  $\overline{\rho}: \Gamma_L \to G(k)$ , under the continuing assumption that we have extended k so all irreducible representations of  $\Lambda_L$  occurring in V are absolutely irreducible over k. From (4.5.3.2), we obtain a decomposition

$$V = \bigoplus_{\tau \in \Sigma_n} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} \left( \widetilde{\tau} \otimes \overline{W}_{\tau} \right) \oplus \bigoplus_{\tau \in \Sigma_e} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} \left( \widetilde{\tau} \otimes \overline{W}_{\tau} \right) \oplus \bigoplus_{\tau \in \Sigma_c} \operatorname{Ind}_{\Gamma_{L,\tau \oplus \tau^*}}^{\Gamma_L} \left( \widetilde{\tau} \oplus \overline{\tau^*} \otimes \overline{W}_{\tau \oplus \tau^*} \right). \tag{4.5.3.5}$$

where  $\overline{W}_{\tau}$  is a representation of  $T_{L,\tau}$  over k. If  $\tau \in \Sigma_n$ , define  $\overline{G}_{\tau} := \underline{\operatorname{Aut}}(\overline{W}_{\tau})$ . If  $\tau \in \Sigma_e$ , there is a sign-symmetric perfect pairing  $\langle \cdot, \cdot \rangle_{\overline{W}_{\tau}}$  on  $\overline{W}_{\tau}$ : in that case define  $\overline{G}_{\tau} := \underline{\operatorname{GAut}}(\overline{W}_{\tau}, \langle \cdot, \cdot \rangle_{\overline{W}_{\tau}})$ . (The notation  $\underline{\operatorname{GAut}}$  means automorphisms preserving the pairing up to scalar.) If  $\tau \in \Sigma_c$ , there is a sign-symmetric perfect pairing on  $\overline{W}_{\tau \oplus \tau^*}$ : in that case define  $\overline{G}_{\tau} := \underline{\operatorname{GAut}}(\overline{W}_{\tau \oplus \tau^*}, \langle \cdot, \cdot \rangle_{\overline{W}_{\tau \oplus \tau^*}})$ . Make a finite extension of k so that all the pairings are split. Lift  $\overline{G}_{\tau}$  to a split reductive group  $G_{\tau}$  over  $\mathcal O$  by lifting the split linear algebra data.

**Definition 4.5.3.2.** Let  $\rho: \Gamma_L \to G(R)$  be a continuous Galois representation lifting  $\overline{\rho}$  as above, with associated  $R[\Gamma]$ -module

$$M = \bigoplus_{\tau \in \Sigma_n} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} \left( \widetilde{\tau} \otimes W_{\tau} \right) \oplus \bigoplus_{\tau \in \Sigma_e} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} \left( \widetilde{\tau} \otimes W_{\tau} \right) \oplus \bigoplus_{\tau \in \Sigma_e} \operatorname{Ind}_{\Gamma_{L,\tau \oplus \tau^*}}^{\Gamma_L} \left( \widetilde{\tau \oplus \tau^*} \otimes W_{\tau \oplus \tau^*} \right).$$

We say that  $\rho$  is minimally ramified with similitude character  $\nu$  if each  $W_{\tau}$  and  $W_{\tau \oplus \tau^*}$  is minimally ramified in the sense of Definition 4.4.2.1 as a representation of  $T_{L,\tau}$  or  $T_{L,\tau \oplus \tau^*}$  valued in the group  $G_{\tau}$ . (Note that defining the minimally ramified deformation condition as in §4.4 may require an additional étale local extension of  $\mathcal{O}$ , which as always is harmless for applications.)

Let  $D^{\mathrm{m.r.},\nu}_{\overline{\rho}}$  denote the deformation functor for  $\overline{\rho}$  with specified similitude character  $\nu$ , and  $\mathcal{D}^{\mathrm{m.r.},\nu}_{G_{\tau}}$  (respectively  $D^{\mathrm{m.r.},\nu}_{G_{\tau}}$ ) denote the deformation functor for  $\overline{W}_{\tau}$  or  $\overline{W}_{\tau \oplus \tau^*}$  viewed as a representation valued in  $G_{\tau}$  (respectively with specified similitude character  $\nu$ ). In particular, letting  $r = \dim \overline{W}_{\tau}$  (or  $\dim \overline{W}_{\tau \oplus \tau^*}$  when  $\tau \in \Sigma_c$ ), we have that the adjoint representation ad  $\overline{W}_{\tau}$  is the Lie algebra of  $\overline{G}_{\tau}$ , which is the Lie algebra of  $\mathrm{GSp}_r$  or  $\mathrm{GO}_r$  when  $\tau \in \Sigma_e$  or  $\Sigma_c$ , and the Lie algebra of  $\mathrm{GL}_r$  when  $\tau \in \Sigma_n$ . Let  $\Sigma'_n$  consist of one representative for each pair of representations  $\tau, \tau^* \in \Sigma_n$ .

Proposition 4.5.3.3. The natural map

$$D^{\text{m.r. }\nu}_{\overline{\rho}}(R) \to \prod_{\tau \in \Sigma_n'} D^{\text{m.r. }}_{G_\tau}(R) \times \prod_{\tau \in \Sigma_e} D^{\text{m.r. }\nu}_{G_\tau}(R) \times \prod_{\tau \in \Sigma_c} D^{\text{m.r. }\nu}_{G_\tau}(R)$$

is an isomorphism.

Proof. This expresses the decomposition obtained in this section: given a lift  $\rho$  of  $\overline{\rho}$ , we obtain a decomposition of M as in Definition 4.5.3.2. Our analysis with pairings shows that when  $\tau \in \Sigma_e$ ,  $W_{\tau}$  is a deformation of  $\overline{W}_{\tau}$  together with its sign-symmetric perfect pairing. Likewise, when  $\tau \in \Sigma_c$  we know that  $W_{\tau \oplus \tau^*}$  is a deformation of  $\overline{W}_{\tau \oplus \tau^*}$  together with its pairing. When  $\tau \in \Sigma_n$ , we know  $W_{\tau} \simeq W_{\tau^*}^{\vee}$ . This gives the natural map: to  $\rho \in D_{\overline{\rho}}^{\mathrm{m.r.} \ \nu}(R)$  associate the collection of the  $W_{\tau}$  for  $\tau \in \Sigma_e \cup \Sigma_c \cup \Sigma'_n$ .

Conversely, given  $W_{\tau}$  for  $\tau \in \Sigma_e \cup \Sigma_c \cup \Sigma'_n$ , and defining  $W_{\tau^*} := W_{\tau}^{\vee}$  for  $\tau \in \Sigma'_n$  we can define a lift

$$M := \bigoplus_{\tau \in \Sigma_r} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} \left( \widetilde{\tau} \otimes W_{\tau} \right) \oplus \bigoplus_{\tau \in \Sigma_r} \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L} \left( \widetilde{\tau} \otimes W_{\tau} \right) \oplus \bigoplus_{\tau \in \Sigma_r} \operatorname{Ind}_{\Gamma_{L,\tau \oplus \tau^*}}^{\Gamma_L} \left( \widetilde{\tau \oplus \tau^*} \otimes W_{\tau \oplus \tau^*} \right).$$

as in (4.5.1.1). (Note that the groups  $\Gamma_{L,\tau}$  depend only on the fixed residual representation V.) For  $\tau \in \Sigma_e$ , the sign-symmetric perfect pairing on the lift  $W_{\tau}$  gives an isomorphism  $\varphi_{\tau}: W_{\tau} \simeq W_{\tau}^{\vee}$  of  $R[T_{L,\tau}]$ -modules, which gives a sign-symmetric pairing on  $\operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L}(\tilde{\tau} \otimes W_{\tau})$  (using (4.5.3.4)). Likewise,  $\tau \in \Sigma_c$  the sign-symmetric pairing on  $W_{\tau \oplus \tau^*}$  gives one on  $\operatorname{Ind}_{\Gamma_{L,\tau} \oplus \tau^*}^{\Gamma_L}(\tilde{\tau} \otimes W_{\tau})$ . For  $\tau \in \Sigma_n$ , we obtain an isomorphism  $\varphi_{\tau}: W_{\tau} \simeq W_{\tau^*}^{\vee}$  of  $R[T_{L,\tau}]$ -modules and hence a sign-symmetric perfect pairing on  $(\tilde{\tau} \otimes W_{\tau}) \oplus (\tilde{\tau}^{\vee} \otimes W_{\tau^*})$  which gives one on  $\operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L}(\tilde{\tau} \otimes W_{\tau}) \oplus \operatorname{Ind}_{\Gamma_{L,\tau}}^{\Gamma_L}(\tilde{\tau}^{\vee} \otimes W_{\tau^*})$ . Putting these together, we obtain a sign-symmetric pairing on M; the action of  $\Gamma_L$  preserves it up to scalar, giving a continuous homomorphism  $\rho: \Gamma_L \to G(R)$ .

Finally, we claim that these constructions are compatible with strict equivalence of lifts, giving an identification of the deformation functors. For  $g \in \widehat{G}(R)$ , decompose the g-conjugate  $\Gamma_L$ -representation  $M^g$  according to (4.5.3.2). As g reduces to the identity, it must respect the decomposition into  $\tau$ -isotypic pieces, so gives automorphisms  $g_{\tau} \in \operatorname{Aut}(W_{\tau})$  and  $g_{\tau} \in \operatorname{Aut}(W_{\tau \oplus \tau^*})$ . If  $\tau \in \Sigma_e$  or  $\Sigma_c$ , as g is compatible with the pairing on M we see  $g_{\tau}$  is compatible with the pairing as well. For  $\tau \in \Sigma_e$ , the  $g_{\tau}$ -conjugate  $T_{L,\tau}$ -representation  $W_{\tau}^{g_{\tau}}$  is minimally ramified as minimally ramified lifts of  $\overline{W}_{\tau}$  for the group  $T_{L,\tau}$  are a deformation condition, and likewise for  $\tau \in \Sigma_c$  and  $\tau \in \Sigma'_n$ .

Conversely, given  $g_{\tau} \in \operatorname{Aut}(W_{\tau})$  reducing to the identity (compatible with the pairing on  $W_{\tau}$  or  $W_{\tau \oplus \tau^*}$  if there is one), using (4.5.1.1) and acting on each piece we obtain a lift of the form  $M^g$  for  $g \in \widehat{G}(R)$ . Thus the identification is compatible with strict equivalence.

**Corollary 4.5.3.4.** The minimally ramified deformation condition with fixed similitude character is liftable. The dimension of the tangent space is  $h^0(\Gamma_L, \operatorname{ad}^0(\overline{\rho}))$ .

Proof. Liftability is a consequence of Proposition 4.5.3.3 and the smoothness of the framed minimally ramified deformation ring for representations of  $T_{L,\tau}$  (Proposition 4.4.2.3 and Corollary 4.4.2.5). Recall that the dimension of the tangent space of a deformation condition of a representation  $\theta: T_{L,\tau} \to G_{\tau}(k)$  is the dimension of the tangent space of the framed deformation ring minus the relative dimension of  $G_{\tau}$  plus the dimension of  $H^0(T_{L,\tau}, \operatorname{ad} \theta)$  (see Remark 2.2.2.6). By Corollary 4.4.2.5, for  $\tau \in \Sigma_e$  the dimension of the tangent space of  $D_{G_{\tau}}^{\operatorname{m.r.},\nu}$  is  $h^0(T_{L,\tau}, \operatorname{ad} \overline{W}_{\tau}) - 1 = h^0(T_{L,\tau}, \operatorname{ad}^0 \overline{W}_{\tau})$ , and for  $\tau \in \Sigma_c$  the dimension is  $h^0(T_{L,\tau\oplus\tau^*}, \operatorname{ad} \overline{W}_{\tau\oplus\tau^*}) - 1 = h^0(T_{L,\tau\oplus\tau^*}, \operatorname{ad}^0 \overline{W}_{\tau\oplus\tau^*})$ . For  $\tau \in \Sigma'_n$ , by Proposition 4.4.2.3 the dimension of the tangent space of  $D_{G_{\tau}}^{\operatorname{m.r.}}$  is  $h^0(T_{L,\tau}, \operatorname{ad} \overline{W}_{\tau})$ . Using Proposition 4.5.3.3, we see that the dimension of the tangent space of the minimally ramified deformation condition is

$$\sum_{\tau \in \Sigma_e} h^0(T_{L,\tau}, \operatorname{ad}^0 \overline{W}_{\tau}) + \sum_{\tau \in \Sigma_c} h^0(T_{L,\tau \oplus \tau^*}, \operatorname{ad}^0 \overline{W}_{\tau \oplus \tau^*}) + \sum_{\tau \in \Sigma_n'} h^0(T_{L,\tau}, \operatorname{ad} \overline{W}_{\tau}).$$

It remains to identify this quantity with  $h^0(\Gamma_L, ad^0(\overline{\rho}))$ . Using Lemma 4.5.3.1

$$\begin{split} H^0(\Gamma_L, \operatorname{End}(V)) &= \operatorname{Hom}_{k[\Gamma_L]}(V, V) = \bigoplus_{\tau \in \Sigma_c \cup \Sigma_n} \operatorname{Hom}_{T_{L,\tau}}(\overline{W}_\tau, \overline{W}_\tau) \oplus \bigoplus_{\tau \in \Sigma_c} \operatorname{Hom}_{T_{L,\tau \oplus \tau^*}}(\overline{W}_{\tau \oplus \tau^*}, \overline{W}_{\tau \oplus \tau^*}) \\ &= \bigoplus_{\tau \in \Sigma_c \cup \Sigma_n} H^0(T_{L,\tau}, \operatorname{End}(\overline{W}_\tau)) \oplus \bigoplus_{\tau \in \Sigma_c} H^0(T_{L,\tau \oplus \tau^*}, \operatorname{End}(\overline{W}_{\tau \oplus \tau^*})). \end{split}$$

We are interested in  $H^0(\Gamma_L, \operatorname{ad}^0(\overline{\rho}))$ : the elements  $\psi \in H^0(\Gamma_L, \operatorname{End}(V))$  compatible with the pairing on V in the sense that for  $v, v' \in V$ 

$$\langle \psi v, \psi v' \rangle = \langle v, v' \rangle.$$

The pairing on  $V_{\tau} = \tau \otimes \overline{W}_{\tau}$  is induced by the pairings on  $\overline{W}_{\tau}$  and  $\tau$  when  $\tau \in \Sigma_e$ , and is induced by the pairings on  $\overline{W}_{\tau \oplus \tau^*}$  and  $\tau \oplus \tau^*$  when  $\tau \in \Sigma_c$ . When  $\tau \in \Sigma'_n$ , the pairing on  $V_{\tau} \oplus V_{\tau^*}$  comes from the  $\Gamma_{L,\tau}$ -isomorphism  $\overline{W}_{\tau} \simeq \overline{W}_{\tau^*}^{\vee}$ . So  $\psi$  is compatible with the pairing if and only if

- when  $\tau \in \Sigma_e$ , the associated  $\psi_{\tau} \in H^0(T_{L,\tau}, \operatorname{End}(\overline{W}_{\tau}))$  is compatible with the pairing on  $\overline{W}_{\tau}$ ;
- when  $\tau \in \Sigma_c$ , the associated  $\psi_{\tau} \in H^0(T_{L,\tau \oplus \tau^*}, \operatorname{End}(\overline{W}_{\tau \oplus \tau^*}))$  is compatible with the pairing on  $\overline{W}_{\tau \oplus \tau^*}$ ;
- when  $\tau \in \Sigma'_n$ , the associated  $\psi_{\tau}$  and  $\psi_{\tau^*}$  are identified by duality and the isomorphism  $\overline{W}_{\tau} \simeq \overline{W}_{\tau^*}^{\vee}$ .

In the first two cases,  $\operatorname{ad}^0 \overline{W}_{\tau}$  and  $\operatorname{ad}^0 \overline{W}_{\tau \oplus \tau^*}$  are the symplectic or orthogonal Lie algebra, consisting exactly of endomorphisms compatible with the pairing on  $\overline{W}_{\tau}$ . In the third, we just choose one of  $\psi_{\tau}$  and  $\psi_{\tau^*}$  without restriction, which determines the other. Thus we see

$$H^{0}(\Gamma_{L}, \operatorname{ad}^{0}(\overline{\rho})) = \bigoplus_{\tau \in \Sigma_{e}} H^{0}(T_{L,\tau}, \operatorname{ad}^{0} \overline{W}_{\tau}) \oplus \bigoplus_{\tau \in \Sigma_{c}} H^{0}(T_{L,\tau \oplus \tau^{*}}, \operatorname{ad}^{0} \overline{W}_{\tau \oplus \tau^{*}}) \oplus \bigoplus_{\tau \in \Sigma'_{n}} H^{0}(T_{L,\tau}, \operatorname{ad} \overline{W}_{\tau}). \quad \Box$$

## **Bibliography**

- [BK90] Spencer Bloch and Kazuya Kato, *L-functions and Tamagawa numbers of motives*, The Grothendieck Festschrift, Vol. I, Progr. Math., vol. 86, Birkhäuser Boston, Boston, MA, 1990, pp. 333–400. MR 1086888
- [BT71] A. Borel and J. Tits, Éléments unipotents et sous-groupes paraboliques de groupes réductifs. I, Invent. Math. 12 (1971), 95–104. MR 0294349 (45 #3419)
- [Car85] Roger W. Carter, Finite groups of Lie type, Pure and Applied Mathematics (New York), John Wiley & Sons, Inc., New York, 1985, Conjugacy classes and complex characters, A Wiley-Interscience Publication. MR 794307 (87d:20060)
- [CGP15] Brian Conrad, Ofer Gabber, and Gopal Prasad, Pseudo-reductive groups, second ed., New Mathematical Monographs, vol. 26, Cambridge University Press, Cambridge, 2015. MR 3362817
- [CHT08] Laurent Clozel, Michael Harris, and Richard Taylor, Automorphy for some l-adic lifts of automorphic mod l Galois representations, Publ. Math. Inst. Hautes Études Sci. (2008), no. 108, 1–181, With Appendix A, summarizing unpublished work of Russ Mann, and Appendix B by Marie-France Vignéras. MR 2470687 (2010j:11082)
- [Con94] Brian Conrad, Filtered modules, galois representations, and big rings, for Wiles Seminar, 1994.
- [Con14] \_\_\_\_\_, Reductive group schemes, Autour des schémas en groupes. Vol. I, Panor. Synthèses, vol. 42/43, Soc. Math. France, Paris, 2014, pp. 93–444. MR 3362641
- [CPS75] Edward Cline, Brian Parshall, and Leonard Scott, Cohomology of finite groups of Lie type. I, Inst. Hautes Études Sci. Publ. Math. (1975), no. 45, 169–191. MR 0399283 (53 #3134)
- [SAGE] The Sage Developers, Sagemath, the Sage Mathematics Software System (Version 6.7), 2015, http://www.sagemath.org.
- [Eme11] Matthew Emerton, Local-global compatibility in the p-adic Langlands programme for  $GL_{2/\mathbb{Q}}$ , preprint, 2011.
- [FL82] Jean-Marc Fontaine and Guy Laffaille, Construction de représentations p-adiques, Ann. Sci. École Norm. Sup. (4) 15 (1982), no. 4, 547–608 (1983). MR 707328
- [Fon82] Jean-Marc Fontaine, Sur certains types de représentations p-adiques du groupe de Galois d'un corps local; construction d'un anneau de Barsotti-Tate, Ann. of Math. (2) 115 (1982), no. 3, 529–577. MR 657238
- [GHS15] T. Gee, F. Herzig, and D. Savitt, General Serre weight conjectures, ArXiv e-prints (2015).
- [Gro] Benedict Gross, Odd galois representations, available at http://www.math.harvard.edu/~gross/preprints/Galois\_Rep.pdf.
- [EGAIV<sub>3</sub>] A. Grothendieck, Éléments de géométrie algébrique. IV. Étude locale des schémas et des morphismes de schémas. III, Inst. Hautes Études Sci. Publ. Math. (1966), no. 28, 255. MR 0217086

- [GY03] Wee Teck Gan and Jiu-Kang Yu, Schémas en groupes et immeubles des groupes exceptionnels sur un corps local. I. Le groupe G<sub>2</sub>, Bull. Soc. Math. France **131** (2003), no. 3, 307–358. MR 2017142 (2004j:14049)
- [Ham08] Spencer Hamblen, Lifting n-dimensional Galois representations, Canad. J. Math.  $\bf 60$  (2008), no. 5, 1028-1049. MR 2442046 (2009j:11085)
- [Hat] Shin Hattori, Ramification of crystalline representations, available at http://www2.math.kyushu-u.ac.jp/~shin-h/RennesHodge/RennesHodge.pdf.
- [Hum06] James E. Humphreys, Modular representations of finite groups of Lie type, London Mathematical Society Lecture Note Series, vol. 326, Cambridge University Press, Cambridge, 2006. MR 2199819
- [Jan04] \_\_\_\_\_, Nilpotent orbits in representation theory, Lie theory, Progr. Math., vol. 228, Birkhäuser Boston, Boston, MA, 2004, pp. 1–211. MR 2042689 (2005c:14055)
- [Kis09] Mark Kisin, The Fontaine-Mazur conjecture for GL<sub>2</sub>, J. Amer. Math. Soc. **22** (2009), no. 3, 641–690. MR 2505297 (2010j:11084)
- [KW09a] Chandrashekhar Khare and Jean-Pierre Wintenberger, Serre's modularity conjecture. I, Invent. Math. 178 (2009), no. 3, 485–504. MR 2551763 (2010k:11087)
- [KW09b] \_\_\_\_\_, Serre's modularity conjecture. II, Invent. Math. 178 (2009), no. 3, 505–586. MR 2551764 (2010k:11088)
- [Liu07] Tong Liu, Torsion p-adic Galois representations and a conjecture of Fontaine, Ann. Sci. École Norm. Sup. (4) 40 (2007), no. 4, 633–674. MR 2191528 (2010h:11191)
- [Man09] Jayanta Manoharmayum, Lifting Galois representations of number fields, J. Number Theory 129 (2009), no. 5, 1178–1190. MR 2516981 (2010i:11077)
- [Mat89] Hideyuki Matsumura, Commutative ring theory, second ed., Cambridge Studies in Advanced Mathematics, vol. 8, Cambridge University Press, Cambridge, 1989, Translated from the Japanese by M. Reid. MR 1011461
- [Maz97] Barry Mazur, An introduction to the deformation theory of Galois representations, Modular forms and Fermat's last theorem (Boston, MA, 1995), Springer, New York, 1997, pp. 243–311. MR 1638481
- [McN08] George J. McNinch, The centralizer of a nilpotent section, Nagoya Math. J. 190 (2008), 129–181. MR 2423832 (2009d:20110)
- [MS03] George J. McNinch and Eric Sommers, Component groups of unipotent centralizers in good characteristic, J. Algebra **260** (2003), no. 1, 323–337, Special issue celebrating the 80th birthday of Robert Steinberg. MR 1976698 (2004d:20054)
- [MT11] Gunter Malle and Donna Testerman, Linear algebraic groups and finite groups of Lie type, Cambridge Studies in Advanced Mathematics, vol. 133, Cambridge University Press, Cambridge, 2011.
  MR 2850737
- [NSW08] Jürgen Neukirch, Alexander Schmidt, and Kay Wingberg, Cohomology of number fields, second ed., Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences], vol. 323, Springer-Verlag, Berlin, 2008. MR 2392026
- [Pat06] Stefan Patrikis, Lifting symplectic galois representations, Harvard Undergraduate Thesis, 2006.
- [Pat15] Stefan Patrikis, Deformations of galois representations and exceptional monodromy, Inventiones mathematicae (2015), 1–68.

- [Pre03] Alexander Premet, Nilpotent orbits in good characteristic and the Kempf-Rousseau theory, J. Algebra **260** (2003), no. 1, 338–366, Special issue celebrating the 80th birthday of Robert Steinberg. MR 1976699 (2004i:17014)
- [Ram93] Ravi Ramakrishna, On a variation of Mazur's deformation functor, Compositio Math. 87 (1993), no. 3, 269–286. MR 1227448 (94h:11054)
- [Ram99] \_\_\_\_\_, Lifting Galois representations, Invent. Math. 138 (1999), no. 3, 537–562. MR 1719819 (2000j:11167)
- [Ram02] \_\_\_\_\_, Deforming Galois representations and the conjectures of Serre and Fontaine-Mazur, Ann. of Math. (2) **156** (2002), no. 1, 115–154. MR 1935843 (2003k:11092)
- [Sch68] Michael Schlessinger, Functors of Artin rings, Trans. Amer. Math. Soc. 130 (1968), 208–222. MR 0217093
- [Sei00] Gary M. Seitz, *Unipotent elements, tilting modules, and saturation*, Invent. Math. **141** (2000), no. 3, 467–502. MR 1779618 (2001);20074)
- [Ser77] Jean-Pierre Serre, *Linear representations of finite groups*, Springer-Verlag, New York-Heidelberg, 1977, Translated from the second French edition by Leonard L. Scott, Graduate Texts in Mathematics, Vol. 42. MR 0450380 (56 #8675)
- [Ser94] \_\_\_\_\_, Sur la semi-simplicité des produits tensoriels de représentations de groupes, Invent. Math. 116 (1994), no. 1-3, 513-530. MR 1253203 (94m:20091)
- [SGA1] Revêtements étales et groupe fondamental (SGA 1), Documents Mathématiques (Paris) [Mathematical Documents (Paris)], 3, Société Mathématique de France, Paris, 2003, Séminaire de géométrie algébrique du Bois Marie 1960–61. [Algebraic Geometry Seminar of Bois Marie 1960-61], Directed by A. Grothendieck, With two papers by M. Raynaud, Updated and annotated reprint of the 1971 original [Lecture Notes in Math., 224, Springer, Berlin; MR0354651 (50 #7129)]. MR 2017446 (2004g:14017)
- [SGA3] Schémas en groupes. III: Structure des schémas en groupes réductifs, Séminaire de Géométrie Algébrique du Bois Marie 1962/64 (SGA 3). Dirigé par M. Demazure et A. Grothendieck. Lecture Notes in Mathematics, Vol. 153, Springer-Verlag, Berlin-New York, 1970. MR 0274460
- [SS70] T. A. Springer and R. Steinberg, Conjugacy classes, Seminar on Algebraic Groups and Related Finite Groups (The Institute for Advanced Study, Princeton, N.J., 1968/69), Lecture Notes in Mathematics, Vol. 131, Springer, Berlin, 1970, pp. 167–266. MR 0268192
- [Ste63] Robert Steinberg, Representations of algebraic groups, Nagoya Math. J. **22** (1963), 33–56. MR 0155937
- [Tay03] Richard Taylor, On icosahedral Artin representations. II, Amer. J. Math. 125 (2003), no. 3, 549–566. MR 1981033 (2004e:11057)
- [Tay08] \_\_\_\_\_, Automorphy for some l-adic lifts of automorphic mod l Galois representations. II, Publ. Math. Inst. Hautes Études Sci. (2008), no. 108, 183–239. MR 2470688 (2010j:11085)
- [Til96] Jacques Tilouine, Deformations of Galois representations and Hecke algebras, Published for The Mehta Research Institute of Mathematics and Mathematical Physics, Allahabad; by Narosa Publishing House, New Delhi, 1996. MR 1643682
- [Win84] Jean-Pierre Wintenberger, Un scindage de la filtration de Hodge pour certaines variétés algébriques sur les corps locaux, Ann. of Math. (2) 119 (1984), no. 3, 511–548. MR 744862
- [Yun14] Zhiwei Yun, Motives with exceptional Galois groups and the inverse Galois problem, Invent. Math. 196 (2014), no. 2, 267–337. MR 3193750